The Immune System Out of Shape? Shaping of adaptive immunity by persistent viral infections in young children

Diana van den Heuvel



ISBN: 978-94-91811-09-8

No parts of this thesis may be reproduced or transmitted in any form by any means, electronic or mechanical, including photocopying, recording or any information storage and retrieval system, without permission in writing from the author

The research for this thesis was performed within the framework of the Erasmus MC Postgraduate School Molecular Medicine

The studies described in this thesis were performed at the Department of Immunology, Erasmus MC, Rotterdam, the Netherlands

The studies were financially supported by an Erasmus MC Fellowship granted to dr. M.C. van Zelm

The printing of this thesis was financially supported by: Erasmus University Rotterdam

Cover design and thesis lay-out: Diana van den Heuvel Printing: Ridderprint BV, Ridderkerk

The Immune System Out Of Shape?

Shaping of adaptive immunity by persistent viral infections in young children

Het immuunsysteem uit vorm?

Het vormen van de verworven afweer door persisterende virale infecties in jonge kinderen

Proefschrift

ter verkrijging van de graad van doctor aan de Erasmus Universiteit Rotterdam op gezag van de rector magnificus

Prof.dr. H.A.P. Pols

en volgens besluit van het College voor Promoties

De openbare verdediging zal plaatsvinden op dinsdag 10 november om 13:30 uur

door

Diana van den Heuvel

geboren te Rotterdam

Frafing

Erasmus University Rotterdam

PROMOTIECOMMISSIE

Promotor:	Prof.dr. J.J.M. van Dongen, MD
Overige leden:	Prof.dr. H.A. Moll, MD Prof.dr. R.A.W. van Lier, MD Dr. A.M.C. van Rossum, MD

Copromotor: Dr. M.C. van Zelm

CONTENTS

CHAPTER 1 General introduction	page 7
CHAPTER 2 Effects of external determinants on age-related patterns of innate leukocyte and naive and memory B- and T-lymphocyte numbers in early childhood: The Generation R Study In preparation	37
CHAPTER 3 Decreased memory B cells and increased CD8 ⁺ memory T cells in blood of breastfed children: The Generation R study <i>PLoS One. 2015; 10(5): e0126019.</i>	67
CHAPTER 4 Can we explain ethnic differences in CMV, EBV and HSV-1 virus seroprevalences in childhood? The Generation R Study <i>Submitted</i>	87
CHAPTER 5 CMV- and EBV-induced T-cell expansions in young children do not impair naive T-cell populations or vaccination responses: The Generation R Study Journal of Infectious Diseases (2015), in press	113
CHAPTER 6 Transient reduction in IgA ⁺ and IgG ⁺ memory B-cell numbers in young children persistently infected with EBV: The Generation R Study In preparation	133
CHAPTER 7 Persistent subclinical immune defects in HIV-1-infected children treated with antiretroviral therapy <i>AIDS (2015), in press</i>	145
CHAPTER 8 General discussion	171
ADDENDA	187
Summary & Samenvatting	189
List of abbreviations	197
Curriculum Vitae & PhD portfolio & List of publications	201
Acknowledgements / Dankwoord	207

CHAPTER 1

GENERAL INTRODUCTION

PROTECTION OF THE HUMAN BODY: THE IMMUNE RESPONSE

The human body is continuously exposed to various pathogens. To protect itself against these potential threats, the human body possesses a combination of mechanisms to either inhibit the entrance of the pathogens, or to neutralize and quickly remove pathogens that do enter the human body. These defense mechanisms can be subdivided into three different layers of protection.

THE FIRST LAYER OF PROTECTION: PHYSICAL AND CHEMICAL BARRIERS

The first layer of protection, is composed of a physical and chemical barrier, which prevent the initial entrance of pathogens into the human system. The major physical barrier is the epithelial lining on the outside of our body, i.e. skin tissue and the mucosal membranes of our gut and lungs. The cells in this epithelial lining are tightly connected (e.g. via tight junctions, gap junctions or adherens junctions), thereby preventing pathogens to penetrate through the epithelium.^{1,2} In addition, air flow around the skin, peristaltic movement of the bowel, and movement of cilia on the apical side of epithelial cells mechanically remove pathogens from the human body. Thirdly, in addition to these physical barriers, chemical barriers are formed, which include mucus, sweat, saliva and tears. These fluids contain antimicrobial proteins, contain a high salt concentration and have a low pH, which together inhibit attachment and proliferation of pathogens.³ Finally, the human body is colonized with commensal microbiota. These bacteria do not only occupy potential niches and thereby prevent the attachment of harmful pathogens, but are also essential for epithelial cell proliferation and differentiation, and support the second and third layers of protection.⁴ (Figure 1).



Figure 1. First layer of protection: A physical and chemical barrier.

The physical barrier is composed of a tight epithelial lining near the exterior of the human body, closely connected via structures including tight junctions. The chemical barrier is composed of mucus, tears and sweat containing low pH, high salt concentration and containing antimicrobial proteins. The commensal microbiota and cilia on top of epithelial cells further prevent pathogen adherence.

THE SECOND LAYER OF PROTECTION: THE INNATE IMMUNE RESPONSE

In case a pathogen does enter the human body, an immune response is needed to eliminate it.⁵ Cells and molecules of the vertebrate immune system are able to recognize foreign particles and to initiate a cascade of responses for pathogen neutralization and removal. Important mediators of the second layer of protection are soluble proteins that form the complement system,⁶ and innate immune cells. These innate cells carry receptors that recognize conserved pathogenic structures, such as pathogen-associated, microbe-associated or damage-associated molecular

General Introduction

patterns on bacteria, fungi, parasites or damaged host cells.⁷⁻⁹ Various innate cell populations can be discriminated, each with a distinct role in the immune response (Figure 2).^{10,11} First of all, pathogenic threats can be fought directly by two types of innate cells, i.e natural killer (NK) cells and eosinophils. NK cells recognize and kill pathogen-infected cells, 12,13 whereas eosinophils locally release anti-pathogenic toxins upon infiltration of extracellular pathogens.¹⁴ Secondly, mast cells can release chemokines and pro-inflammatory cytokines that induce the dilation of blood vessels, thereby mediating the recruitment of additional innate immune cells from blood, including monocytes, neutrophils, basophils and dendritic cells (DCs).¹⁵ Basophils start to release their granule content, which stimulates further recruitment of innate cells.¹⁵ Monocytes migrate into tissue and differentiate into macrophages that, together with the newly recruited neutrophils, take up pathogens by phagocytosis and destroy the pathogens internally in endosomal structures.^{16,17} Finally, DCs take up the pathogen, and migrate through the lymphatic system into secondary lymphoid tissues. In these tissues, B and T lymphocytes are activated and form the third layer of protection.¹⁸ DCs thereby perform a crucial bridging function between the second and third layers of protection.

Innate leukocytes possess four basic mechanisms to fight pathogen infections: 1) direct killing of pathogen-infected cells by NK cells: or removal of free pathogen in the local tissue by eosinophils;

2) production of chemokines by mast cells and basophils;

3) phagocytosis and internal pathogen killing by macrophages and neutrophils; 4) phagocytosis and antigen presentation by dendritic cells (DCs) to initiate adaptive

THE THIRD LAYER OF PROTECTION: THE ADAPTIVE IMMUNE RESPONSE

B and T lymphocytes form the third line of defense: the adaptive immune response (Figure 3). These cells carry specialized receptors, i.e. the B-cell receptor (BCR) or T-cell receptor (TCR). In contrast to other proteins, these receptors differ between each newly generated B and T cell, and together these cells can recognize an enormous diversity of structures, i.e. epitopes. Mature B and T cells circulate through blood and lymphatic vessels between lymphoid tissues. B cells recognize soluble antigens dispersed into the lymph nodes via the lymphatic system. T cells, however, only recognize an antigen when it is presented on specialized protein complexes called the human leukocyte antigens (HLA's; also known as the major histocompatibility complex (MHC) proteins), by DCs that migrated from local tissue into the lymph nodes. Upon recognition of specific antigen, a B or T cell becomes activated, resulting in extensive clonal proliferation and differentiation into effector cells.

Two major types of T lymphocytes exist: cytotoxic and helper T cells. Cytotoxic T cells recognize antigens presented on HLA proteins A, B or C (HLA-class I), which



pathogen

are expressed on all cells of the body except for red blood cells. Presented in the groove of these HLA-class I molecules are intracellular peptides.¹⁹ By scanning the antigens presented in these HLA-class I molecules, cytotoxic T cells are able to scan the content of a cell. In case of a viral infection, intracellular viral peptides will also be presented in HLA-class I molecules on the cell membrane and can be recognized by cytotoxic T cells. Local recognition of virus-infected cells by activated cytotoxic T cells subsequently triggers the release of cytotoxic molecules resulting in the direct killing of the infected cells.²⁰ The TCR – HLA-class I interaction is stabilized by the co-receptor molecule cluster of differentiation 8 (CD8), expressed on the cytotoxic T cell, which is therefore also referred to as CD8⁺ T cell. Helper T cells recognize peptides presented on HLA proteins DP, DQ or DR (HLA-class II). Different from HLA-class I molecules, HLA-class II molecules present peptides taken up from the cell's exterior, such as bacterial antigens. As antigen uptake and presentation in the groove of HLA-class II proteins can only be performed by specialized cells called antigen-presenting cells (APCs), helper T-cell interactions are limited mainly to cells of the immune system.^{20,21} Activated helper T cells function predominantly in stimulating the activation of surrounding antigenspecific B and cytotoxic T lymphocytes.²² The TCR – HLA-class II interaction is stabilized by the co-receptor CD4, and helper T cells are therefore also referred to as CD4⁺ T cells.

Upon antigen recognition, and proper stimulation by helper T cells, B cells undergo extensive clonal expansion and can differentiate into plasma cells, which secrete large amounts of soluble variants of the BCR: immunoglobulins (Igs). These Igs can bind the pathogen and thereby prevent it from binding and infecting cells, thereby neutralizing the pathogenic threat. Furthermore, they can stimulate phagocytosis of the pathogen by phagocytic cells, such as neutrophils and macrophages, or they can induce antibody-dependent cytotoxicity, both processes resulting in the destruction of the pathogen.²⁰

The antigen-specific B- and T-cell responses are highly effective in controlling an infection. However, upon primary infection, the selective activation followed by extensive proliferation and maturation of lymphocytes takes several days. Therefore, the host depends largely on the immediate innate immune response in the first days after primary infection. Importantly, after primary infection with a pathogen and subsequent clearance by the immune system, a selected set of antigen-experience lymphocytes involved in this response remain present in the circulation as "memory" lymphocytes. In case of a subsequent encounter with the same pathogen, these memory lymphocytes are able to recognize the pathogen and to respond more quickly than naive B and T cells. The adaptive immune responses thereby provide immunological memory. Consequently, the pathogenic threat can be removed before the development of clinical symptoms and the host is functionally immune for this specific pathogen.

HEMATOPOIESIS

All immune cells, or leukocytes, are continuously generated throughout life from tissue-specific, pluripotent hematopoietic stem cell (HSC) in the bone marrow. This process is called hematopoiesis and involves a branched multi-step differentiation process in which the HSC proliferates and differentiates into each of the leukocyte lineages. First, the progenitor cells becomes restricted towards either the myeloid or lymphoid lineage. Myeloid progenitors can still differentiate into mast cells, monocytes, granulocyte subsets (i.e. neutrophilic, basophilic and eosinophilic



Figure 3. Third layer of protection: Adaptive immune responses.

A) Soluble antigen and antigenloaded DCs migrate to secondary lymphoid structures, including the lymph nodes. Lymphocytes enter the lymph node via the blood circulation. DCs activate T cells (Тн). helper Together they activate cytotoxic T cells (Tc) and B cells. Cytotoxic T cells migrate into the tissue, where they induce death of virusinfected cells. B cells differentiate immunoglobulin-producing in plasma Immunoglobulins cells. disperse into tissue where they neutralize antigens. B) Initiation of adaptive immune responses follows three phases: 1. Selection of antigen-specific B cells, TH cells and Tc cells. 2. Proliferation and differentiation of antigen-specific lymphocytes. TH cells stimulate B-cell and Tc-cell proliferation differentiation. В and cells differentiate into immunoglobulin producing plasma cells. Activated cytotoxic T cells release cytotoxic granules upon antigenic trigger. 3. Memory formation. A selection of lymphocytes remain after pathogen clearance to provide immunological memory.

granulocytes), certain DC subsets, erythrocytes (red blood cells) and thrombocytes (platelets). Mast cells originate directly from myeloid progenitors. Monocytes and granulocyte subsets develop via an intermediate myeloblast cell stage. Monocytes leave the bone marrow into the circulation, but upon their migration into tissue differentiate further into macrophages.

Lymphoid progenitors can differentiate into B, T and NK lymphocytes, and certain DC subsets (Figure 4). Commitment and differentiation into the B- or NK-cell lineages takes place in bone marrow, whereas early progenitors exit the bone marrow and migrate into the thymus where they develop into T cells (reviewed in 23).

ANTIGEN-INDEPENDENT B-CELL AND T-CELL ANTIGEN RECEPTOR FORMATION

In contrast to myeloid cell development, B- and T-cell differentiation is a lengthy process in which the cells undergo various stages of differentiation and proliferation.²⁴⁻²⁸ These stages are required to ensure regulated rearrangement of genetic elements in the genes encoding the BCRs and TCRs.^{27,29} The BCR is composed of four peptide chains; two identical Ig heavy chains (IgH) and two identical Ig light chains (IgL). The latter are either of the Lambda (Ig λ) or Kappa (Ig κ) isotype (Figure 5A).²⁹ The TCR is composed of only two TCR chains, being



Figure 4. Schematic representation of the hematopoietic differentiation pathways. Hematopoietic stem cells differentiate into a myeloid progenitor or lymphoid progenitor. Myeloid progenitors give rise to innate immune cells. They differentiate directly into mast cells, or via a myeloblast stage into neutrophils, basophils, eosinophils, monocytes and selected dendritic cell (DC) subsets (dashed line). Monocytes differentiate into macrophages upon migration into tissue. Lymphoid progenitors give rise to selected DC subsets (dashed line), NK cells and B cells in bone marrow, and T cells in the thymus. T-cell precursors differentiate further into TCRQ6⁺ T cells or TCRY0⁺ T cells. All lymphocytes undergo V(D)J recombination (V(D)J) to develop a functional antigen receptor. TCRX0⁺ T cells further differentiate into CD4-expressing helper T cells (TH) and CD8-expressing cytotxic T cells (Tc). For simplicity, erythrocytes and thrombocytes are excluded from the figure. Figure adapted from Rieger *et al.*²³

either composed of one TCRa and one TCR β chain (TCRa β^+ T cells; comprising ~95% of all T cells), or of one TCR γ paired with one TCR δ chain (TCR $\gamma\delta^+$ T cells; comprising ~5% of all T cells) (Figure 5B). Each BCR and TCR contains a constant domain (or backbone), providing stability to the receptor and anchoring it in the cell membrane, and a variable domain, which is unique in each B or T cell and is responsible for the specific recognition of antigens. An adult adaptive human immune system contains approximately 10¹² different lymphocytes. As it is not possible to code this large number of TCRs or BCRs in the human genome, the variable domains of BCRs and TCRs are formed in a process called V(D) J-recombination (Figure 5C).

V(D)J recombination is a highly structured and step-wise process in which the BCR and TCR are developed via a number of maturation steps.^{24-28,30,31} In the human genome, the gene *IGH* (coding for the B-cell IgH chain), and the genes *TCRB* and *TCRD* (coding for the TCR β and TCR δ proteins, respectively) are all composed of multiple non-identical copies of three different gene segments; a Variable (V) gene segment, a Diversity (D) gene segment and a Joining (J) gene segment.³²⁻³⁴ For each of the *IGH*, *TCRB* and *TCRD* loci, V(D)J recombination starts with the random selection of one D and one J gene segment, which are then brought in close

General Introduction

proximity. Next, the DNA in between the two segments is removed and excised in the form of a circular excision product and the D and J segments are rearranged into a DJ recombination product. Subsequently, this DJ segment is recombined to one V segment in a similar process, resulting in a VDJ recombination product. Together this VDJ segment forms a first exon of the IgH, TCRβ or TCRδ proteins. The resulting V(D)J exon is subsequently transcribed, spliced towards a downstream constant (C) region and translated into a functional protein (Figure 5C). A similar process occurs for the IGL, TCRA and TCRG gene loci encoding the IgL, TCRa and TCRy proteins respectively, though these loci contain only V and J gene segments and consequently require only V to J recombination.^{34,35} This largely random selection of V, (D) and J gene segments ensures high combinatorial diversity of BCRs and TCRs. Still, to even further increase the variability in BCRs and TCRs, random nucleotides are inserted or deleted at the sites of V(D) recombination. resulting in a high junctional diversity.³⁶⁻³⁸ Together, the combinatorial diversity and junctional diversity enable the formation of a repertoire of more than 10^{12} unique lymphocytes.²⁹



Figure 5. B-cell receptor and T-cell receptor formation.

A) The B-cell receptor (BCR) is composed of two identical Ig heavy (IgH) and two identical Ig light (IgL) peptide chains. Each IgH chain is composed of a Variable (V), Diversity (D), Joining (J) and Constant (C) region (the latter composed of four domains). Each IgL is composed of a V, J and C region. The BCR is anchored in the cell membrane by anchoring molecules CD79a and CD79b. **B)** The T-cell receptor (TCR) is composed of a combination of a TCRa and a TCR β peptide chain (left), or a combination of a TCR γ and TCR δ peptide chain (right). The TCR β and TCR δ chains are composed of V, D, J and C regions. The TCR α and TCR γ chains are composed of V, D, J and C regions. The TCR α and TCR δ chains are composed of V, D, J and C regions. The TCR α and TCR δ , CD3 δ , CD3 γ and CD3 ζ . **C)** V(D)J recombination starts with the rearrangement of a D to a J segment, followed by a V to DJ rearrangement, resulting in a V(D)J exon. Upon functional protein.

ANTIGEN-DEPENDENT B-CELL MATURATION

AFFINITY MATURATION AND CLASS-SWITCH RECOMBINATION

After V(D)J recombination, a functional BCR is produced and expressed on the cell membrane of the naive B cell, which exits the bone marrow and migrates to secondary lymphoid organs. Upon antigen encounter, the naive B cell becomes activated and, in case of a protein antigen, recruits antigen-specific CD4⁺ helper T cells through the presentation of antigenic peptides in HLA-class II proteins. Interactions between B cells and CD4⁺ helper T cells trigger clonal proliferation and formation of germinal centers. Germinal centers are specialized lymphoid tissue structures that support two important B-cell maturation processes: BCR affinity maturation and class switch recombination (CSR). BCR affinity maturation is a process to optimize the BCR variable domain for more optimal recognition of the antigen (Figure 6). During extensive proliferation, B cells induce point mutations, i.e. somatic hypermutations (SHM), particularly targeting the V(D)J exon of their IgH and IgL chains, under the control of activation-induced cytidine deaminase (AID).^{39,40} Subsequently, B cells expressing BCRs with higher affinity for the antigen are positively selected, resulting in increased overall affinity of the B-cell pool responding to a specific antigen.⁴¹⁻⁴⁴

CSR is a process to change the constant region of the BCR and thereby adapt its effector functions. In naive B cells the constant region of the IgH chain anchors the BCR in the B-cell membrane. Upon antigen encounter, however, the activated B cells differentiate into Ig-secreting plasma cells. The constant region of the Ig (also known as the Fc-tail), mediates its specific effector function. Nine different immunoglobulin isotypes exist in humans: IgM, IgD, IgG (IgG1, IgG2, IgG3, IgG4), IgA (IgA1, IgA2) and IgE (Figure 6).²⁹ In naive B cells the V(D)J exon is by default spliced towards the downstream constant region C μ or by alternative splicing toward the C δ , resulting in a BCR of the IgM or IgD isotype, respectively, both of which are expressed on naive B cells. During CSR, a switch region upstream of the C μ exons recombines with a switch region upstream of one of the C γ , Ca or C ϵ exons (Figure 6). This results in a new recombination product, replacing the IgM and IgD constant regions for either an IgG, IgA or IgE constant region (extensively reviewed in 45,46). Each of these Ig isotypes have their own predominant effector functions, as discussed in the following paragraph.

Immunoglobulin-producing plasma cells

Following proliferation, SHM and CSR, germinal center B cells differentiate either into plasma cells or memory B cells. Plasma cells are the effector B cells that, by producing Igs, directly fight the invading pathogen. As each Ig isotype has a distinct constant region; i.e. Fc-tail, each of them can be recognized by a distinct Fc receptor, subsequently inducing distinct downstream effector functions.²⁹ The nature of the pathogen and the combination of cytokines produced by the CD4⁺ helper T cells, together define the selection of a certain isotype.^{46,47} IgM is the first isotype to be expressed upon naive B-cell activation. Part of the IgM expressing cells does not undergo a germinal center response, but differentiates in extrafollicular niches, such as the marginal zone of the spleen. Consequently, IgM expressing cells can respond quickly to a pathogen trigger, but the IgM molecules contains low levels of SHM and consequently low affinity maturation. Although this low affinity can be (partially) overcome by forming large pentamer molecules, thereby

increasing binding efficiency, highly specific, long-lasting antibody responses will predominantly be of an Ig isotype different from IgM. Since B cells can undergo consecutive CSR steps (Figure 6), effector functions of the highly-specific Ig molecules can be adapted to specific needs.

IgG is the most predominant immunoglobulin in the circulation. It is mainly present as monomer and is easily transferred into local tissue, via binding to the neonatal Fc-Receptor, thereby providing protection throughout the human body.⁴⁸ In a similar process, IgG can also be transferred through the placenta and provide a fetus with maternal protection.⁴⁹ IgG coated pathogens can be recognized by Fcy receptors, which either activate the complement system, stimulate phagocytosis, or induces antibody-dependent cellular cytotoxicity by NK cells and other cytotoxic cells. These actions result in the direct destruction of the pathogen.⁵⁰ IgA is predominantly localized at mucosal tissues lining the outside of the human body. Via binding of polymeric IgA to the polymeric Ig receptor on epithelial cells, it can be transported through the epithelial layer into mucus layers or excretions.⁵¹ One of these IqA containing excretions is human breastmilk. IqA can thereby be passively transferred from a mother to her newborn's intestinal mucosa, providing early maternal protection to the child. In contrast to IgG, IgA is less capable of inducing antibody-dependent cytotoxicity. It, however, predominantly functions in blocking pathogen binding to host cells.⁵² IgE predominantly functions by activating mast cells through the cross-linking of Fcc receptors on these cells, resulting in the release of granule content, including histamines and other inflammatory proteins. IqE antibodies function in response to parasitic infections, but are also known for their function in allergic responses.



Figure 6. Class switch recombination and somatic hypermutation.

The human *IGH* locus contains 9 functional constant (C) regions. Apart from the $C\delta$ region, each of the constant regions is preceded by a switch (S) region. RNA splicing of the V(D)J exon to the Cµ results in a B-cell receptor (BCR) of the IgM isotype. During class switch recombination, a switch region (in this figure Sµ) recombines with a downstream switch region (e.g. Sγ1), which result in a new RNA splicing product and the production of a BCR of a different isotype (e.g. IgG1). During extensive proliferation, point mutations (somatic hypermutations; red dots) are introduced in the V(D)J exon to increase the affinity of the BCR. Consecutive switching to a downstream switch region (e.g. Sγ2), enables a further switching towards a more downstream BCR Ig isotype (e.g. IgG2).

MEMORY B CELLS

Memory B cells are able to respond fast upon subsequent encounter with the same antigen, providing long-lasting immunity to the host. Eight different memory B-cell populations can be defined (Figure 7). Five memory B-cell populations develop via classical T-cell dependent germinal center responses: IgM^+IgD^- (IgMonly), CD27⁻IgG⁺, CD27⁺IgE⁺, CD27⁺IgG⁺ and CD27⁺IgA⁺ memory B cells. All five have proliferated significantly and show high levels of SHM, indicative of affinity maturation. Still, CD27⁺IgG⁺ and CD27⁺IgA⁺ memory B cells underwent more proliferation and acquired more SHM than IqMonly, CD27⁻IqG⁺ and CD27⁺IqE⁺ memory B cells, suggesting that these cells underwent multiple cycles of affinity maturation.⁵³⁻⁵⁵ Besides these classical germinal center-dependent memory B-cell populations, three other populations have been identified: IaM⁺ memory B cells still co-expressing IqD (natural effector B cells) and CD27⁻IqA⁺ and CD27⁻IqE⁺ memory B cells. All three populations show low levels of proliferation and SHM, and can be detected in peripheral blood of immunodeficient patients who are unable to form germinal centers.⁵⁴⁻⁵⁷ Together, this indicates a germinal center-independent (T-cell independent) origin of these populations. Natural effector B cells have been suggested to develop in the marginal zone of the spleen and are also referred to as marginal-zone B cells.⁵⁷⁻⁵⁹ The CD27 IgA⁺ and CD27 IgE⁺ memory B-cell populations have been suggested to originate from mucosal tissues.^{54,55}

Altogether, plasma cell formation, antibody production and memory development result from maturation of a selected set of naive B cells, thereby restricting the BCR repertoire, but increasing pathogen-specific responsiveness.



Figure 7. Memory B-cell formation.

Naive mature B cells can undergo T cell-independent (TI) or T cell-dependent (TD) immune responses. TI responses occur locally in the mucosal tissue, inducing the formation of CD27·IgE⁺ and CD27·IgA⁺ cells, or systemically in the splenic marginal zone, inducing natural effector B cells. TI responses are characterized by low proliferation and B-cell receptor (BCR) maturation levels as compared with TD responses. TD responses occur in germinal center structures. Primary germinal center responses induce the formation of CD27⁺IgA⁺, CD27⁻IgG⁺ and CD27⁺IgE⁺ memory B cells with intermediate proliferation and BCR maturation. Secondary TD germinal center responses induce the formation. Figure adapted from Berkowska *et al.*⁵⁴

ANTIGEN-DEPENDENT T-CELL MATURATION

The majority of T cells produce a TCR using a TCRa and TCR β chain, thereby forming an TCRa β^+ T cell. Within these cells, around 65% comprises CD4⁺ helper T cells, whereas the remaining 35% are CD8⁺ cytotoxic T cells.^{22,60}

$CD4^+ \; TCRa\beta^+ \; T\text{-cell effector responses}$

When a naive CD4⁺ T cell recognizes an antigenic peptide presented in HLA-class II proteins by an APC, further co-stimulation via CD40 expressed on the cell membrane of the APC, and CD40Ligand on the CD4⁺ T cell, induces the activation and maturation of both cell types.⁶¹ This induces the upregulated expression of co-stimulatory molecules on the APC, and the production and secretion of signaling molecules, i.e. cytokines, to the surroundings by both the cell types. The upregulated co-stimulatory molecules, such as CD80, CD86 or CD70, can stimulate and stabilize a subsequent interaction between an APC and an antigen-specific CD8⁺ T cell.⁶²⁻⁶⁵ The cytokines, including interferon γ , interleukin (IL)-2, IL-12 and IL-15,^{22,63,66-68} furthermore stimulate proper activation of CD8⁺ T cells (Figure 8A).

In addition, B cells are able to present antigenic peptides in HLA-class II molecules, thereby enabling a cognate interaction with an antigen-specific activated CD4⁺ T cell in a germinal center. During this germinal center response (paragraph 'Antigen-dependent B-cell maturation'), the CD4⁺ T cell stimulates B-cell maturation and differentiation via CD40 – CD40L interaction and the production of cytokines including IL21 and IL4 (Figure 8B).⁴¹⁻⁴³ An activated CD4⁺ T cell can thereby provide help to both antigen-specific CD8⁺ T cells and B cells.



Figure 8. Helper T-cell functions.

A) An antigen-specific helper T (TH) cell initiates a cognate interaction with a dendritic cell (DC) presenting antigenic peptides in HLA-class II molecules. This initiates the upregulation of co-stimulatory molecules, which stabilize a subsequent cognate interaction between the DC, presenting antigen in HLA-class I molecules, and an antigen-specific cytotoxic T (Tc) cell. Furthermore, both the TH cell and DC produce cytokines to stimulate further Tc-cell differentiation. **B)** A B cell can present antigenic peptides in HLA-class II molecules to antigen-specific TH cells. Subsequent signaling via co-stimulatory molecules and the production of cytokines by the TH cell, stimulate the B cell towards plasma cell (pc) differentiation.

$CD8^+ \ TCRa\beta^+ \ T\text{-cell effector responses}$

When a naive CD8⁺ T cell recognizes an antigen presented in HLA-class I molecules, the additional co-stimulation provided by APCs, together with the cytokine production of both APCs and CD4⁺ T cells, results in its full activation (as described above). Activated CD8⁺ T cells then migrate towards the site of infection where they scan the cell membranes of cells in the surrounding tissue in search for cells that present their cognate antigen. Upon antigen encounter, the CD8⁺ T cell induces the death of its infected target-cell, predominantly via the release of toxic granules containing perforins and granzymes, or FAS – FASLigand dependent cell death.⁶⁹⁻⁷¹

$TCR\gamma\delta^+$ T-cell responses

TCR $\gamma\delta^+$ T cells are different from TCRa β^+ T cells, not only in the TCR chains they use, but also with respect to their development and responses. Though details of their development and function still remain largely unknown, two major subsets of TCR $\gamma\delta^+$ T cells can be defined: V $\delta1^+$ and V $\delta2^+$ T cells. V $\delta1^+$ T cells predominate in the intestine and have been described to recognize stress-antigens (e.g. MHCclass I–related molecules MICA and MICB),^{72,73} whereas V $\delta2^+$ T cells predominate in blood and recognize low molecular mass phospho-antigens.^{72,74} Antigen recognition might be independent of presentation in HLA-class I or II molecules,⁷³ and does not require clonal expansion. Besides cytotoxic functions, TCR $\gamma\delta^+$ T cells have also been described to produce cytokines or chemokines and to perform a function in antigen presentation. Altogether, their quick, diverse, and stressinduced response make them important in stress-surveillance.^{75,76}

MEMORY T CELLS

Upon pathogen clearance, the majority of effector T cells die by apoptosis to dampen the T-cell response and to prevent excessive effector T-cell accumulation after multiple pathogen encounters.^{77,78} Instead, subsets of T cells remain present after pathogen clearance, in the form of memory T cells. The total pool of memory T cells is highly diverse, with the development of each memory T cell depending on the specific strength of TCR-triggering and/or cytokine stimulation,⁷⁹⁻⁸¹ and each possessing its own migration capacities, requirements for optimal stimulation and potential effector functions (Figure 9).^{77,82-91}

One of the first proteins described to be differentially expressed in different T-cell subsets is the protein tyrosine phosphatase CD45. Naive T cells express CD45RA (an isotype of CD45), but upon antigen encounter and T-cell activation, the CD45RA switches isotype into CD45RO. Although the exact role of the CD45 isotypes is still largely unclear, this CD45-isotype-switch correlates with induced T-cell activation and proliferation.⁹² Memory T cells can express either CD45RO, or re-express the CD45RA isotype, the latter being considered as having more effector cell properties.^{93,94}

Similarly, depending on their function and predominant localization throughout the human body, T-cell subsets show differential expression of chemokine receptors (CCRs) and adhesion molecules.⁸⁶ Naive T cells express molecules important in lymph node homing, including CCR7 (also known as CD197) and L-selectin (also known as CD62L). Part of the CD45RO⁺ memory T cells express similar

General Introduction

proteins, and have been designated "central memory" T cells (Tcm).^{82,83} Due to the maintained expression of CCR7 and L-selectin, these Tcm cells circulate through the blood, lymphatics system and secondary lymphoid tissues. The remaining CD45RO⁺ memory T cells and the CD45RA⁺ memory T cells, however, lack the expression of both markers, but express other chemokine receptors, e.g. CCR5, CXCR1 or CX₃CR1, which enable their migration into local tissue, such as the skin or intestine.^{77,84-87,91,94} Both of these memory populations are referred to as effector memory T cells (Tem), either CD45RO⁺ (TemRO) or CD45RA⁺ (TemRA)



В

Α

		Thaive	Tcm	early	Tem RO interm	late	early	Tem RA interm	late
	CD45RA								
	CD45RO								
co-stimulation	CD27								
cell survival									
regulation of activation	CD127 (IL-7R)								
	CXCR4								
	CD62L								
cell migration and	CCR4								
aunesion	CCR5								
	CXCR1								
	CX3CR1								
	IL-2								
cytokine production	IFNγ								
and	TNFα								
cytotoxic function	Perforin/Granzyme B								
	CD95L (FASL)								

Figure 9. CD8⁺ T-cell responses.

A) Activated naive T cells differentiate into central memory T cells (Tcm) or effector memory T cells, the latter being either CD45RO⁺ (TemRO) or CD45RO⁻ (TemRA). Tcm cells circulate the lymphatics and blood circulation and lymphoid structures. TemRO and TemRA cells migrate into tissue and kill virus-infected cells. **B)** T-cell subsets expression levels of proteins involved in co-stimulation, cell survival and regulation of activation; cell migration and dhesion; and cytokine production and cytotoxic function. White, not expressed; light blue, intermediate expression; dark blue, strong expression.

The various memory T-cell subsets each possess distinct functional properties and different requirements for co-stimulation and survival. Naive T cells express high levels of survival factors, such as IL-7R (CD127), and different co-stimulatory molecules including CD27 and CD28, to enable their optimal activation upon antigen encounter.^{84,85,87,94} Similarly, the systemic and lymph node localization of Tcm cells, will enable them to quickly recognize a subsequent infection with the same antigen, and to perform a central function in the induction of an adaptive immune response. In line with this function, Tcm cells maintain the expression of IL-7R, CD27 and CD28. Effector memory T cells gradually lose these survival factors and co-stimulatory molecules upon further differentiation (from early towards late differentiated TemRO or TemRA cells), but (especially cytotoxic T cells) show increased effector molecule production, including granzymes and perforin,^{70,89} FASL (CD95L), and the cytokines IFNv and TNFa, increasing their cvtotoxic potential.^{84,88,89} Increasing differentiation is, however, also correlated with an increased expression of inhibitory molecules, including killer cell lectin like receptor G1 (KLRG1), Programmed death 1 (PD-1 or CD279) and CD57,95-97 and with increasing T-cell proliferative exhaustion.^{81,84,95-97}

The exact differentiation pathway through which the various memory T-cell populations are generated remains debated.⁹¹ Memory cells might develop from effector cells that escaped activation-induced cell death, or via a pathway independent from effector cell formation.⁹⁸ Furthermore, a consecutive differentiation pathway from naive T cells, via Tcm cells towards Tem cells has been described, in which Tcm cells might provide a renewing pool of cells continuously replenishing the Tem populations and in which effector memory T cells progressively lose co-stimulation, proliferative capacity and responsiveness.^{79,80,84,89,99} Others, however, described a parallel differentiation,¹⁰⁰ or even a two-directional interchanging phenotype in which effector memory cells can regain expression of lymph node homing potential, self-renewal and functional properties comparable to central memory T cells.¹⁰¹

Still, irrespective of the differentiation pathway, the extensive characterization of the distinct memory T-cell subsets by markers described above (Figure 9) – which are still a selection of the available expression profiles in literature – enables detailed functional discrimination of memory T-cell subsets.

MATURATION OF THE CHILDHOOD IMMUNE SYSTEM

During fetal development, a fetus resides in the protected environment of the uterus, shielded from the vast majority of external pathogens. Thus, the fetal immune system is largely immature and mainly composed of innate leukocytes and naive lymphocytes.¹⁰²⁻¹⁰⁴ Maturation of the immune system largely occurs in the first few years of life, in response to environmental pathogens and other antigens. Important partial immune protection for the first few months of life, will be provided by the mother. Already during pregnancy maternal IgG molecules, can be transferred to the child through the placenta and can provide protection to the child.⁴⁹ Also after birth the mother can pass on immune protection to her child via breastfeeding. Breastfeeding contains various particles, including maternal IgA and IgG molecules and cytokines, which passively provide the child with immune protection for the time of breastfeeding duration.¹⁰⁵

While receiving short-term passive protection via its mother, a child will have to develop its own protective immunity, mainly in the form of memory B and T cells. During childhood, the number of lymphocytes is relatively large, compared to adults.^{102,104} With each naive lymphocyte having a unique antigen receptor, the combined BCR and TCR repertoires are enormous, with a high potential to mount responses to all distinct types of pathogens. As each pathogen induces a distinct functional response, associated with the expansions of distinct lymphocyte populations and the generation of distinct memory lymphocyte pools, the combination of pathogens we encounter in the environment eventually shapes the composition and antigen-receptor repertoire of our immune system.

With increasing age, the production of naive lymphocytes declines, together with a reduction in the size and functionality of the thymus. Consequently, the repertoire of new naive lymphocytes will become smaller in adults.¹⁰⁶ Adults, therefore, largely depend on their memory lymphocyte pools, produced as a result of the many pathogen-encounters during the first few years of life, to protect them from the recurrence of clinical illness.¹⁰⁷⁻¹¹³ The formation of a stable memory lymphocyte pool during childhood, and the maintenance hereof in adulthood, is therefore essential for long-term immune protection. A disbalance in the immune system can result in clinical complications, ranging from lack of protection, to the development of excessive immune responses, and can even lead to auto-immune diseases.

PERSISTENT VIRAL INFECTIONS

Most of the encountered pathogens are eventually cleared by the immune system. Several viruses, however, escape from clearance by the immune system. Even though these can be seemingly well-controlled by the host, they can persist life-long, and cause complications in case of immunosuppression. Such persistent viruses include members of the family of herpesviruses and the human immunodeficiency virus (HIV).

HERPESVIRUSES

Herpesviruses are double-stranded DNA viruses, and many are highly prevalent in the human population.¹¹⁴ To date, 8 herpesviruses have been identified to infect humans. Four of these include the cytomegalovirus (CMV), Epstein Barr virus (EBV), herpes simplex virus (HSV) type 1 and varicella zoster virus (VZV). Seroprevalence for these viruses is high, varying between around 30 to >90% in adults.^{115,116} Infection predominantly occurs during childhood and is mostly mild or even asymptomatic. However, infection in neonates and immunocompromised individuals might result in severe clinical complications. HSV-1 and VZV are two neurotropic viruses, meaning that they reside in sensory neurons.^{117,118} HSV-1 can induce cold sores, recurring in situations of weakened immune surveillance. VZV induces chicken pox in children and can recur as shingles in most adults. CMV is able to infect a broad range of cell types including monocytes, epithelial cells, endothelial cell and to a lesser extend leukocytes.¹¹⁹⁻¹²¹ CMV infection in endothelial cells also induces the attraction of neutrophils. CMV is transferred to these neutrophils, after which they recirculate and promote further dissemination of the virus.¹²² EBV infects and persists in B cells.^{123,124} EBV targets B cells through interactions with complement receptor 2 (CR2; CD21), HLA-class II co-receptor surface molecules, or the Beta-1 integrin.¹²⁵⁻¹²⁷ Even though all B cells express these molecules, EBV persists preferentially in memory cells.^{123,128} Both CMV and EBV are able to induce infectious mononucleosis. CMV is, due to its tropism for endothelial cells, furthermore associated with vascular complications.^{129,130} EBV is associated with increased development of malignant complications, including Burkitt lymphomas.¹³¹

THE HUMAN IMMUNODEFICIENCY VIRUS

HIV is a retrovirus responsible for inducing the acquired immunodeficiency syndrome (AIDS). HIV infects cells through binding to the CD4 proteins expressed on the membrane of the target cell, and therefore the predominant target cells are the CD4⁺ T cells and monocytes/macrophages. HIV infection is characterized by CD4⁺ T-cell depletion. HIV-infected adults in whom the CD4⁺ T-cell numbers drop below 200 cells/µl of blood are eventually diagnosed with AIDS.

HIV uses co-receptors CCR5 and CXCR4, or integrin $a4\beta7$ for cell entry.¹³²⁻¹⁴⁰ The integrin $a4\beta7$ is more recently described to form a complex with the CD4 protein on the cell membrane and CD4⁺ $a4\beta7^+$ T cells have been observed to be enriched for metabolically active cells expressing high levels of CCR5.¹³⁸ By targeting $a4\beta7^+$ cells, HIV thereby seems to increase its chance for effective infection. Furthermore, HIV is able to interact with DC-specific C-type lectin (DC-SIGN) on local DCs and CD21 on B cells, two cell types that are themselves not permissive for HIV infection, but that facilitate HIV virus transfer towards secondary lymphoid structures and into close proximity of CD4⁺ T cells.^{141,142} Moreover, HIV seems to preferentially infect HIV-specific CD4⁺ T cells, most likely due to prolonged presence of both actively replicating HIV particles and HIV-specific CD4⁺ T cells in lymphoid structures.¹⁴³ The latter might even further impair HIV clearance.

IMMUNE EVASION BY PERSISTENT VIRAL INFECTIONS

Persistent viruses have developed various strategies to avoid their detection and destruction by the host's immune system. Although describing all the immunomodulatory effects of the viral proteins involved in immune evasion will be beyond the scope of this thesis (these have been extensively reviewed in 144-149), five major evasion strategies, used by herpesviruses and the HIV virus, will be addressed here (Table 1).

First, especially RNA viruses, including the HIV virus, have a high mutation rate of their viral genome, resulting in very high antigenic variability of the virus. Consequently, immune responses need to be continuously adapted to these mutating viruses. This provides the virus with a survival-advantage. DNA viruses, including the herpesviruses, seem to depend less on this evasion strategy as their DNA templates are more stable.

Second, CMV, EBV, HSV-1, VZV and HIV all inhibit antigen presentation by the infected host cell, masking them from the immune system. This is achieved through the low production of viral proteins during their state of latency. Furthermore, these viruses inhibit host HLA expression using mechanisms that can vary from decreasing HLA protein production, HLA stability, HLA transport to the cell membrane or HLA maintenance at the cell membrane.¹⁵⁰⁻¹⁵⁵ Finally, viruses might limit the generation of presentable peptides by inhibiting host cell protease activity

or the TAP protein (transporter associated with antigen processing).^{145,151,156-159}

Third, persistent viruses might directly inhibit the activation of immune cells; inhibiting cellular responses by avoiding NK cell and CD8⁺ T cells activation,¹⁶⁰⁻¹⁶³ or humoral responses by inhibiting antibodies from binding to their Fc-receptor and inducing antibody-dependent or complement-dependent cytotoxicity.¹⁴⁴

Fourth, persistent viruses possess many mechanisms to affect cytokine and chemokine production or their subsequent signaling function, thereby regulating the balance between immune-activating molecules and immune-inhibiting molecules (reviewed in 144,148).

The last, important mechanism that can contribute to immune evasion is the selective tropism in certain host cell types. HSV-1 and VZV, for example, induces latency mainly in neuronal cells, which are protected from immune targeting, because of their importance and incapability to renew. EBV and HIV directly target lymphocyte populations. Killing the virus-infected cells, either by the virus itself, or as a result of an anti-viral immune response, therefore directly leads to destruction of the body's own immune system, eventually weakening the anti-viral immune responses.

Combined, the various evasion strategies, of which the ones described in this thesis are only a limited selection, enable the persistent viruses to escape from virus clearance and to induce long-term latency in the host.

IMMUNOLOGICAL CONSEQUENCES OF PERSISTENT VIRAL INFECTION

As persistent viruses are continuously present, they also apply a constant pressure on the immune system. Consequently, important changes in the immune system have been correlated to the presence of a persistent viral infection. Some of these being common to persistent viruses, others being virus-specific (Figure 10).

	Persistent virus							
		CMV	EBV	HSV1	vzv	HIV		
1 High mut	ation rate and antigenic variability	±	±	±	±	+		
2 Inhibiting	virus peptide presentation							
reduc	ing protein expression during latency	+	+	+	+	+		
preve	nting HLA expression	+	+	+	+	+		
inhibi	ting proteases and TAP function	+	+	+	?	?		
3 Inhibiting	immune cell activation							
inhibi	ting CD8+ T/ NK cell killing	+	+	+	?	+		
inhibi	ting humoral immunity	+	+	+	+	+		
4 Reducing	cytokine/chemokine production	+	+	+	?	+		
5 Predomin	ant tropism	Broad range of cells	B cells	CD4 ⁺ T cells, macro- phages	Neu ce	ronal ells		

Table 1. Evasion strategies used by different persistent viruses

TAP, transporter associated with antigen processing; +, virus possesses the evasion strategy; \pm , virus can perform evasion strategy, but is less common; ?, evasion strategy not (yet) defined for the specific virus.

DIRECT DESTRUCTION OF THE IMMUNE SYSTEM

Some persistent viruses directly target the immune system and reside in immune cells. These viruses thereby directly affect the immune system and immune cell functioning (Figure 10a.). The two strongest examples hereof are the EBV and HIV viruses. EBV directly targets B cells, inducing persistence preferentially in memory B cells.^{123,128} It is thought that this is the result of EBV infection in naive B cells that are driven into an immune response, after which the virus persists in the long-lived memory B-cell compartment.^{123,164-166} Recent studies, however, indicate that both activated naive and memory B cells can be infected,¹⁶⁷ and that EBV can establish persistence in the absence of fully functional germinal center activity.^{128,168} The true effect of EBV on B cells, is therefore still a matter of debate. Furthermore, although primary EBV infection induces a strong expansion of memory B cells that normalizes within 1 week after the appearance of clinical complications, long-term consequences of EBV persistence in memory B cells are still unknown.¹⁶⁹

HIV directly targets CD4⁺ T cells, focusing on activated, CCR5-expressing T cells,¹³²⁻¹⁴⁰ and even HIV-specific CD4⁺ T cells.¹⁴³ While HIV thereby directly impairs the CD4⁺ T-cell functioning, it furthermore affects the function of CD8⁺ T cells and B cells, both of which need CD4⁺ T-cell help for optimal activation (as discussed earlier). B cells can furthermore be directly affected by HIV via direct binding and signaling of HIV to the B cell,¹⁴¹ or via the transfer of HIV proteins, such as Nef (negative factor) onto B cells, reducing B-cell class switching towards IgG and IgA.¹⁷⁰ (extensively reviewed in 171-173) Both CD4⁺ T-cell help-dependent and independent B-cell defects have been described, including hypergammaglobulinemia,^{174,175} a reduction of CD27⁺ class-switched memory B cells,^{170,175,176} and an expansion of an aberrant, anergic B-cell population expressing low level of CD21 (CD21^{low} B cells).^{177,178}

Current combined antiretroviral therapy (cART) can nowadays effectively reduce HIV virus loads in peripheral blood and recover CD4⁺ T-cell numbers to near-normal levels in both HIV-infected adults and children, suggesting that the treatment is effectively inhibiting viral replication and inducing immune recovery.¹⁷⁹⁻¹⁸² HIV thereby is changing from a lethal into a chronic disease. Still, full recovery of all immune defects is in adults not obtained, and long-term consequences of HIV infection and cART treatment in HIV-infected children remain largely unknown.

OVERCROWDING THE IMMUNE SYSTEM

The continuous presence of persistent viruses and the repeated cycles of activation and reactivation of virus production result in a continuous triggering of the immune system and a constant activation and stimulation of memory lymphocyte responses. Clinical consequences hereof have been best described for CMV and EBV infection. Memory lymphocyte pools directed against these viruses have been described to accumulate over time,^{84,99} resulting in very large populations in especially the elderly. In these elderly they seem to occupy such a large part of the immune system that they overcrowd memory lymphocytes against different, unrelated pathogens, resulting in the loss of immunity (Figure 10b.)¹⁸³⁻¹⁸⁷ Still, these effects are not consistently observed,¹⁸⁸⁻¹⁹¹ and despite high CMV and EBV seropositivity rates in elderly, only a minority develops clinical complications. Furthermore, childhood infection with these viruses is mostly asymptomatic and might even be protective against the development of allergic or autoimmune diseases.^{192,193} The reason for the different clinical outcome in children and adults, the discrepancy in literature on adult infection, as well as why only the minority of CMV- and/or EBV-infected elderly present clinical complications, remain largely unknown.

IMMUNE EXHAUSTION

Another important consequence of the continuous immune activation is immune exhaustion, also known as immunosenescence, a process that is also largely correlated to aging.^{81,194,195} Immune exhaustion is a state in which cells show high replication history, reduced proliferative potential and reduced functional responsiveness (Figure 10c.). Exhaustion has been predominantly described for CD8⁺ T cells and correlates with shortened telomere length, reduced expression of co-stimulatory molecules, e.g. CD27 and CD28, increased expression of inhibitory receptors, e.g. CD57, KLRG1 and PD-1, and reduced cytotoxic potential (see also Figure 9).^{84,85,95-97}

Immune exhaustion has also been described for a B-cell population expressing low levels of CD21. Increased numbers of these CD21^{low} B cells have been observed in various conditions of chronic immune activation, including immunodeficiencies, ¹⁹⁶ autoimmune diseases,¹⁹⁷ and persistent viral infection with HIV.¹⁹⁸ In HIV-infected individuals, these CD21^{low} B cells express increased levels of inhibitory receptors (including FcRL4), reduced replication potential, altered expression of homing receptors guiding them more towards inflamed tissue, and functional exhaustion.¹⁹⁸ They thereby share many characteristics with exhausted T cells. The presence of these CD21^{low} B cells correlates with the presence of detectable HIV-viremia, suggesting that continuous viral exposure also induces exhaustion of B cells.¹⁹⁸ The ability of HIV to directly interact with B cells through binding of the CD21 protein, has been suggested to increase the chance of close contact between HIV particles and B cells, and enhance the induction of B-cell exhaustion.¹⁴¹ This is even further underlined by the finding that CD21^{low} B cells are enriched for HIVspecific cells, as recognition of HIV particles by BCRs could prolong the interaction period and subsequently enhance immunomodulatory effects and the induction of immune exhaustion.198



Figure 10. Immunological consequences of persistent viral infection.

Persistent viral infections can affect the immune system via three different mechanism: a. the targeted depletion of immune cells; b. the induction of extensive antiviral immune responses that overcrowd the pool of non-related immune cells; and c. extensive stimulation of the immune system which induces functional exhaustion, characterized by reduced proliferative potential and effector functions, but increased susceptibility to apoptosis and increased expression of inhibitory receptors.

Chapter 1

AIMS OF THIS THESIS

The importance of a well-functioning immune system can be appreciated in cases of either insufficient immune protection, such as immunodeficiency or immunosenescence, or excessive immune responses leading to clinical complications, such as autoimmune or allergic diseases. An optimal collaboration between the various cell types of the immune system is crucial for providing long-term protection of the host without eventually losing control. Understanding the mechanism behind this delicately balanced immune system will largely contribute to the field of immunology and might have implications for situations varying from vaccine strategies, the treatment of autoimmune or allergic diseases and even cancer therapy.

Most of the priming of the immune system is achieved during childhood, when the majority of the antigens encountered are still seen by the immune system for the first time. The research presented in this thesis aimed to further unravel the development and shaping of the immune system during childhood, focusing predominantly on the adaptive immune compartment. This in order to understand the mechanism by which children become optimally prepared for adulthood. **Chapter 2** focuses on the homeostasis during healthy immune development. In this chapter, we describe the different dynamics of the innate and adaptive immune compartments during their simultaneous development from birth until 6 vears of age, and identify various environmental factors that associate to these dynamics. Chapter 3 describes the effects of breastfeeding on the immune maturation at the age of 6 months. Chapter 4 focuses on persistent herpesvirus infections, describing ethnic differences in herpesvirus seroprevalence. Chapter 5 describes the presence of CMV- and EBV-associated T-cell expansions in 6-yearold children and their effect on the total T-cell compartment. Chapter 6 focuses on the effect of EBV infection and persistence in memory B cells on the composition and functioning of the memory B-cell compartment in young children. In **Chapter** 7, we studied the effectiveness of current cART treatment in HIV-infected children and adolescents to enable B- and T-cell development.

REFERENCES

- 1. Kobielak A, Boddupally K. Junctions and inflammation in the skin. *Cell Commun Adhes*. 2014; 21(3): 141-147.
- 2. Kirschner N, Houdek P, Fromm M, Moll I, Brandner JM. Tight junctions form a barrier in human epidermis. *Eur J Cell Biol*. 2010; 89(11): 839-842.
- 3. Kim JJ, Khan WI. Goblet cells and mucins: role in innate defense in enteric infections. *Pathogens*. 2013; 2(1): 55-70.
- 4. Canny GO, McCormick BA. Bacteria in the intestine, helpful residents or enemies from within? *Infect Immun.* 2008; 76(8): 3360-3373.
- 5. Sperandio B, Fischer N, Sansonetti PJ. Mucosal physical and chemical innate barriers: Lessons from microbial evasion strategies. *Semin Immunol.* 2015.
- 6. Tomlinson S. Complement defense mechanisms. *Curr Opin Immunol*. 1993; 5(1): 83-89.
- 7. Akira S, Uematsu S, Takeuchi O. Pathogen recognition and innate immunity. *Cell*. 2006; 124(4): 783-801.
- 8. Kawai T, Akira S. The role of pattern-recognition receptors in innate immunity: update on Toll-like receptors. *Nat Immunol.* 2010; 11(5): 373-384.
- 9. Janeway CA, Jr., Medzhitov R. Innate immune recognition. *Annu Rev Immunol*. 2002; 20(197-216.
- 10. Gomez CR, Nomellini V, Faunce DE, Kovacs EJ. Innate immunity and aging. Exp Gerontol. 2008;

General Introduction

43(8): 718-728.

- 11. Mahbub S, Brubaker AL, Kovacs EJ. Aging of the Innate Immune System: An Update. *Curr Immunol Rev.* 2011; 7(1): 104-115.
- 12. Caligiuri MA. Human natural killer cells. Blood. 2008; 112(3): 461-469.
- Cichocki F, Sitnicka E, Bryceson YT. NK cell development and function--plasticity and redundancy unleashed. Semin Immunol. 2014; 26(2): 114-126.
- Stone KD, Prussin C, Metcalfe DD. IgE, mast cells, basophils, and eosinophils. J Allergy Clin Immunol. 2010; 125(2 Suppl 2): S73-80.
- 15. Voehringer D. Protective and pathological roles of mast cells and basophils. *Nat Rev Immunol*. 2013; 13(5): 362-375.
- 16. Chilvers ER, Cadwallader KA, Reed BJ, White JF, Condliffe AM. The function and fate of neutrophils at the inflamed site: prospects for therapeutic intervention. *J R Coll Physicians Lond*. 2000; 34(1): 68-74.
- 17. Borregaard N, Sorensen OE, Theilgaard-Monch K. Neutrophil granules: a library of innate immunity proteins. *Trends Immunol*. 2007; 28(8): 340-345.
- Banchereau J, Briere F, Caux C, Davoust J, Lebecque S, Liu YJ, et al. Immunobiology of dendritic cells. Annu Rev Immunol. 2000; 18(767-811.
- 19. Hewitt EW. The MHC class I antigen presentation pathway: strategies for viral immune evasion. *Immunology*. 2003; 110(2): 163-169.
- 20. Murphy K, Travers P, Walport M. Janeway's immunobiology (ed 7th): Garland Science Taylor & Francis Group; 2008.
- 21. Germain RN. MHC-dependent antigen processing and peptide presentation: providing ligands for T lymphocyte activation. *Cell*. 1994; 76(2): 287-299.
- 22. Zhang S, Zhang H, Zhao J. The role of CD4 T cell help for CD8 CTL activation. *Biochem Biophys Res Commun*. 2009; 384(4): 405-408.
- 23. Rieger MA, Schroeder T. Hematopoiesis. Cold Spring Harb Perspect Biol. 2012; 4(12).
- 24. Alt FW, Yancopoulos GD, Blackwell TK, Wood C, Thomas E, Boss M, et al. Ordered rearrangement of immunoglobulin heavy chain variable region segments. *Embo J*. 1984; 3(6): 1209-1219.
- Korsmeyer SJ, Hieter PA, Sharrow SO, Goldman CK, Leder P, Waldmann TA. Normal human B cells display ordered light chain gene rearrangements and deletions. J Exp Med. 1982; 156(4): 975-985.
- Ehlich A, Schaal S, Gu H, Kitamura D, Muller W, Rajewsky K. Immunoglobulin heavy and light chain genes rearrange independently at early stages of B cell development. *Cell*. 1993; 72(5): 695-704.
- 27. van Zelm MC, van der Burg M, de Ridder D, Barendregt BH, de Haas EF, Reinders MJ, et al. Ig gene rearrangement steps are initiated in early human precursor B cell subsets and correlate with specific transcription factor expression. *J Immunol*. 2005; 175(9): 5912-5922.
- Blom B, Verschuren MC, Heemskerk MH, Bakker AQ, van Gastel-Mol EJ, Wolvers-Tettero IL, et al. TCR gene rearrangements and expression of the pre-T cell receptor complex during human T-cell differentiation. *Blood*. 1999; 93(9): 3033-3043.
- 29. Schroeder HW, Jr., Cavacini L. Structure and function of immunoglobulins. J Allergy Clin Immunol; 125(2 Suppl 2): S41-52.
- Joachims ML, Chain JL, Hooker SW, Knott-Craig CJ, Thompson LF. Human alpha beta and gamma delta thymocyte development: TCR gene rearrangements, intracellular TCR beta expression, and gamma delta developmental potential--differences between men and mice. J Immunol. 2006; 176(3): 1543-1552.
- 31. Dik WA, Pike-Overzet K, Weerkamp F, de Ridder D, de Haas EF, Baert MR, et al. New insights on human T cell development by quantitative T cell receptor gene rearrangement studies and gene expression profiling. *J Exp Med.* 2005; 201(11): 1715-1723.
- 32. Honjo T, Nakai S, Nishida Y, Kataoka T, Yamawaki-Kataoka Y, Takahashi N, et al. Rearrangements of immunoglobulin genes during differentiation and evolution. *Immunol Rev.* 1981; 59(33-67.
- 33. Early P, Huang H, Davis M, Calame K, Hood L. An immunoglobulin heavy chain variable region gene is generated from three segments of DNA: VH, D and JH. *Cell*. 1980; 19(4): 981-992.
- 34. Krangel MS. Mechanics of T cell receptor gene rearrangement. *Curr Opin Immunol*. 2009; 21(2): 133-139.
- 35. Brack C, Hirama M, Lenhard-Schuller R, Tonegawa S. A complete immunoglobulin gene is created by somatic recombination. *Cell*. 1978; 15(1): 1-14.

Chapter 1

- 36. Kallenbach S, Doyen N, Fanton d'Andon M, Rougeon F. Three lymphoid-specific factors account for all junctional diversity characteristic of somatic assembly of T-cell receptor and immunoglobulin genes. *Proc Natl Acad Sci U S A*. 1992; 89(7): 2799-2803.
- Ma Y, Pannicke U, Schwarz K, Lieber MR. Hairpin opening and overhang processing by an Artemis/ DNA-dependent protein kinase complex in nonhomologous end joining and V(D)J recombination. *Cell*. 2002; 108(6): 781-794.
- Purugganan MM, Shah S, Kearney JF, Roth DB. Ku80 is required for addition of N nucleotides to V(D)J recombination junctions by terminal deoxynucleotidyl transferase. *Nucleic Acids Res.* 2001; 29(7): 1638-1646.
- 39. Diaz M, Casali P. Somatic immunoglobulin hypermutation. *Curr Opin Immunol*. 2002; 14(2): 235-240.
- Muramatsu M, Kinoshita K, Fagarasan S, Yamada S, Shinkai Y, Honjo T. Class switch recombination and hypermutation require activation-induced cytidine deaminase (AID), a potential RNA editing enzyme. *Cell*. 2000; 102(5): 553-563.
- 41. MacLennan IC. Germinal centers. Annu Rev Immunol. 1994; 12(117-139.
- 42. Vinuesa CG, Linterman MA, Goodnow CC, Randall KL. T cells and follicular dendritic cells in germinal center B-cell formation and selection. *Immunol Rev.* 2010; 237(1): 72-89.
- 43. Allen CD, Okada T, Cyster JG. Germinal-center organization and cellular dynamics. *Immunity*. 2007; 27(2): 190-202.
- 44. Neuberger MS. Antibody diversification by somatic mutation: from Burnet onwards. *Immunol Cell Biol.* 2008; 86(2): 124-132.
- 45. Matthews AJ, Zheng S, DiMenna LJ, Chaudhuri J. Regulation of immunoglobulin class-switch recombination: choreography of noncoding transcription, targeted DNA deamination, and long-range DNA repair. *Adv Immunol.* 2014; 122(1-57.
- 46. Stavnezer J, Guikema JE, Schrader CE. Mechanism and regulation of class switch recombination. *Annu Rev Immunol.* 2008; 26(261-292.
- 47. Stavnezer J, Radcliffe G, Lin YC, Nietupski J, Berggren L, Sitia R, et al. Immunoglobulin heavy-chain switching may be directed by prior induction of transcripts from constant-region genes. *Proc Natl Acad Sci U S A*. 1988; 85(20): 7704-7708.
- 48. Chen N, Wang W, Fauty S, Fang Y, Hamuro L, Hussain A, et al. The effect of the neonatal Fc receptor on human IgG biodistribution in mice. *MAbs*. 2014; 6(2): 502-508.
- 49. Virella G, Silveira Nunes MA, Tamagnini G. Placental transfer of human IgG subclasses. *Clin Exp Immunol*. 1972; 10(3): 475-478.
- 50. Clark MR. IgG effector mechanisms. *Chem Immunol*. 1997; 65(88-110.
- 51. Cerutti A, Rescigno M. The biology of intestinal immunoglobulin A responses. *Immunity*. 2008; 28(6): 740-750.
- 52. Cerutti A. The regulation of IgA class switching. Nat Rev Immunol. 2008; 8(6): 421-434.
- 53. Fecteau JF, Cote G, Neron S. A new memory CD27-IgG+ B cell population in peripheral blood expressing VH genes with low frequency of somatic mutation. *J Immunol*. 2006; 177(6): 3728-3736.
- 54. Berkowska MA, Driessen GJ, Bikos V, Grosserichter-Wagener C, Stamatopoulos K, Cerutti A, et al. Human memory B cells originate from three distinct germinal center-dependent and -independent maturation pathways. *Blood*. 2011; 118(8): 2150-2158.
- 55. Berkowska MA, Heeringa JJ, Hajdarbegovic E, van der Burg M, Thio HB, van Hagen PM, et al. Human IgE(+) B cells are derived from T cell-dependent and T cell-independent pathways. *J Allergy Clin Immunol.* 2014; 134(3): 688-697 e686.
- 56. Klein U, Rajewsky K, Kuppers R. Human immunoglobulin (Ig)M+IgD+ peripheral blood B cells expressing the CD27 cell surface antigen carry somatically mutated variable region genes: CD27 as a general marker for somatically mutated (memory) B cells. J Exp Med. 1998; 188(9): 1679-1689.
- 57. Weller S, Faili A, Garcia C, Braun MC, Le Deist FF, de Saint Basile GG, et al. CD40-CD40L independent Ig gene hypermutation suggests a second B cell diversification pathway in humans. *Proc Natl Acad Sci U S A*. 2001; 98(3): 1166-1170.
- 58. Weller S, Braun MC, Tan BK, Rosenwald A, Cordier C, Conley ME, et al. Human blood IgM "memory" B cells are circulating splenic marginal zone B cells harboring a prediversified immunoglobulin repertoire. *Blood*. 2004; 104(12): 3647-3654.
- 59. Weller S, Reynaud CA, Weill JC. Splenic marginal zone B cells in humans: where do they mutate

their Ig receptor? *Eur J Immunol*. 2005; 35(10): 2789-2792.

- 60. Rosendahl Huber S, van Beek J, de Jonge J, Luytjes W, van Baarle D. T cell responses to viral infections opportunities for Peptide vaccination. *Front Immunol*. 2014; 5(171.
- 61. Bennett SR, Carbone FR, Karamalis F, Flavell RA, Miller JF, Heath WR. Help for cytotoxic-T-cell responses is mediated by CD40 signalling. *Nature*. 1998; 393(6684): 478-480.
- 62. Prilliman KR, Lemmens EE, Palioungas G, Wolfe TG, Allison JP, Sharpe AH, et al. Cutting edge: a crucial role for B7-CD28 in transmitting T help from APC to CTL. *J Immunol*. 2002; 169(8): 4094-4097.
- 63. Cella M, Scheidegger D, Palmer-Lehmann K, Lane P, Lanzavecchia A, Alber G. Ligation of CD40 on dendritic cells triggers production of high levels of interleukin-12 and enhances T cell stimulatory capacity: T-T help via APC activation. *J Exp Med.* 1996; 184(2): 747-752.
- 64. Welten SP, Redeker A, Franken KL, Benedict CA, Yagita H, Wensveen FM, et al. CD27-CD70 costimulation controls T cell immunity during acute and persistent cytomegalovirus infection. *J Virol.* 2013; 87(12): 6851-6865.
- Keller AM, Schildknecht A, Xiao Y, van den Broek M, Borst J. Expression of costimulatory ligand CD70 on steady-state dendritic cells breaks CD8+ T cell tolerance and permits effective immunity. *Immunity*. 2008; 29(6): 934-946.
- 66. Kumaraguru U, Banerjee K, Rouse BT. In vivo rescue of defective memory CD8+ T cells by cognate helper T cells. *J Leukoc Biol*. 2005; 78(4): 879-887.
- 67. Smith CM, Wilson NS, Waithman J, Villadangos JA, Carbone FR, Heath WR, et al. Cognate CD4(+) T cell licensing of dendritic cells in CD8(+) T cell immunity. *Nat Immunol*. 2004; 5(11): 1143-1148.
- Curtsinger JM, Lins DC, Mescher MF. Signal 3 determines tolerance versus full activation of naive CD8 T cells: dissociating proliferation and development of effector function. J Exp Med. 2003; 197(9): 1141-1151.
- 69. Russell JH, Ley TJ. Lymphocyte-mediated cytotoxicity. Annu Rev Immunol. 2002; 20(323-370.
- Harari A, Bellutti Enders F, Cellerai C, Bart PA, Pantaleo G. Distinct profiles of cytotoxic granules in memory CD8 T cells correlate with function, differentiation stage, and antigen exposure. J Virol. 2009; 83(7): 2862-2871.
- 71. Peters PJ, Borst J, Oorschot V, Fukuda M, Krahenbuhl O, Tschopp J, et al. Cytotoxic T lymphocyte granules are secretory lysosomes, containing both perforin and granzymes. *J Exp Med.* 1991; 173(5): 1099-1109.
- 72. Pfeffer K, Schoel B, Gulle H, Kaufmann SH, Wagner H. Primary responses of human T cells to mycobacteria: a frequent set of gamma/delta T cells are stimulated by protease-resistant ligands. *Eur J Immunol*. 1990; 20(5): 1175-1179.
- 73. Groh V, Steinle A, Bauer S, Spies T. Recognition of stress-induced MHC molecules by intestinal epithelial gammadelta T cells. *Science*. 1998; 279(5357): 1737-1740.
- 74. Pauza CD, Poonia B, Li H, Cairo C, Chaudhry S. gammadelta T Cells in HIV Disease: Past, Present, and Future. *Front Immunol*. 2014; 5(687.
- 75. Hayday AC. [gamma][delta] cells: a right time and a right place for a conserved third way of protection. *Annu Rev Immunol*. 2000; 18(975-1026.
- 76. Hayday AC. Gammadelta T cells and the lymphoid stress-surveillance response. *Immunity*. 2009; 31(2): 184-196.
- Hertoghs KM, Moerland PD, van Stijn A, Remmerswaal EB, Yong SL, van de Berg PJ, et al. Molecular profiling of cytomegalovirus-induced human CD8+ T cell differentiation. J Clin Invest. 2010; 120(11): 4077-4090.
- 78. Strasser A, Pellegrini M. T-lymphocyte death during shutdown of an immune response. *Trends Immunol.* 2004; 25(11): 610-615.
- Geginat J, Lanzavecchia A, Sallusto F. Proliferation and differentiation potential of human CD8+ memory T-cell subsets in response to antigen or homeostatic cytokines. *Blood*. 2003; 101(11): 4260-4266.
- Geginat J, Sallusto F, Lanzavecchia A. Cytokine-driven proliferation and differentiation of human naive, central memory, and effector memory CD4(+) T cells. J Exp Med. 2001; 194(12): 1711-1719.
- 81. Gamadia LE, van Leeuwen EM, Remmerswaal EB, Yong SL, Surachno S, Wertheim-van Dillen PM, et al. The size and phenotype of virus-specific T cell populations is determined by repetitive antigenic stimulation and environmental cytokines. *J Immunol*. 2004; 172(10): 6107-6114.

- Chapter 1
- 82. Sallusto F, Geginat J, Lanzavecchia A. Central memory and effector memory T cell subsets: function, generation, and maintenance. *Annu Rev Immunol*. 2004; 22(745-763.
- 83. Sallusto F, Lenig D, Forster R, Lipp M, Lanzavecchia A. Two subsets of memory T lymphocytes with distinct homing potentials and effector functions. *Nature*. 1999; 401(6754): 708-712.
- Appay V, van Lier RA, Sallusto F, Roederer M. Phenotype and function of human T lymphocyte subsets: consensus and issues. *Cytometry A*. 2008; 73(11): 975-983.
- van Aalderen MC, Remmerswaal EB, ten Berge IJ, van Lier RA. Blood and beyond: properties of circulating and tissue-resident human virus-specific alphabeta CD8(+) T cells. *Eur J Immunol*. 2014; 44(4): 934-944.
- 86. Hess C, Means TK, Autissier P, Woodberry T, Altfeld M, Addo MM, et al. IL-8 responsiveness defines a subset of CD8 T cells poised to kill. *Blood*. 2004; 104(12): 3463-3471.
- Wills MR, Okecha G, Weekes MP, Gandhi MK, Sissons PJ, Carmichael AJ. Identification of naive or antigen-experienced human CD8(+) T cells by expression of costimulation and chemokine receptors: analysis of the human cytomegalovirus-specific CD8(+) T cell response. *J Immunol*. 2002; 168(11): 5455-5464.
- Hamann D, Baars PA, Rep MH, Hooibrink B, Kerkhof-Garde SR, Klein MR, et al. Phenotypic and functional separation of memory and effector human CD8+ T cells. J Exp Med. 1997; 186(9): 1407-1418.
- 89. Romero P, Zippelius A, Kurth I, Pittet MJ, Touvrey C, Iancu EM, et al. Four functionally distinct populations of human effector-memory CD8+ T lymphocytes. *J Immunol.* 2007; 178(7): 4112-4119.
- 90. Seder RA, Ahmed R. Similarities and differences in CD4+ and CD8+ effector and memory T cell generation. *Nat Immunol.* 2003; 4(9): 835-842.
- 91. Appay V, Rowland-Jones SL. Lessons from the study of T-cell differentiation in persistent human virus infection. *Semin Immunol*. 2004; 16(3): 205-212.
- 92. Johannisson A, Festin R. Phenotype transition of CD4+ T cells from CD45RA to CD45R0 is accompanied by cell activation and proliferation. *Cytometry*. 1995; 19(4): 343-352.
- Wills MR, Carmichael AJ, Weekes MP, Mynard K, Okecha G, Hicks R, et al. Human virus-specific CD8+ CTL clones revert from CD45ROhigh to CD45RAhigh in vivo: CD45RAhighCD8+ T cells comprise both naive and memory cells. J Immunol. 1999; 162(12): 7080-7087.
- 94. Henson SM, Riddell NE, Akbar AN. Properties of end-stage human T cells defined by CD45RA reexpression. *Curr Opin Immunol*. 2012; 24(4): 476-481.
- Sauce D, Almeida JR, Larsen M, Haro L, Autran B, Freeman GJ, et al. PD-1 expression on human CD8 T cells depends on both state of differentiation and activation status. *Aids.* 2007; 21(15): 2005-2013.
- Henson SM, Franzese O, Macaulay R, Libri V, Azevedo RI, Kiani-Alikhan S, et al. KLRG1 signaling induces defective Akt (ser473) phosphorylation and proliferative dysfunction of highly differentiated CD8+ T cells. *Blood*. 2009; 113(26): 6619-6628.
- Brenchley JM, Karandikar NJ, Betts MR, Ambrozak DR, Hill BJ, Crotty LE, et al. Expression of CD57 defines replicative senescence and antigen-induced apoptotic death of CD8+ T cells. *Blood*. 2003; 101(7): 2711-2720.
- Manjunath N, Shankar P, Wan J, Weninger W, Crowley MA, Hieshima K, et al. Effector differentiation is not prerequisite for generation of memory cytotoxic T lymphocytes. *J Clin Invest*. 2001; 108(6): 871-878.
- Appay V, Dunbar PR, Callan M, Klenerman P, Gillespie GM, Papagno L, et al. Memory CD8+ T cells vary in differentiation phenotype in different persistent virus infections. *Nat Med*. 2002; 8(4): 379-385.
- 100. Sallusto F, Lanzavecchia A. Exploring pathways for memory T cell generation. *J Clin Invest*. 2001; 108(6): 805-806.
- 101. Wherry EJ, Teichgraber V, Becker TC, Masopust D, Kaech SM, Antia R, et al. Lineage relationship and protective immunity of memory CD8 T cell subsets. *Nat Immunol*. 2003; 4(3): 225-234.
- 102. Comans-Bitter WM, de Groot R, van den Beemd R, Neijens HJ, Hop WC, Groeneveld K, et al. Immunophenotyping of blood lymphocytes in childhood. Reference values for lymphocyte subpopulations. *J Pediatr.* 1997; 130(3): 388-393.
- 103. Xanthou M. Leucocyte blood picture in healthy full-term and premature babies during neonatal period. *Arch Dis Child*. 1970; 45(240): 242-249.

- 104. Sagnia B, Ateba Ndongo F, Ndiang Moyo Tetang S, Ndongo Torimiro J, Cairo C, Domkam I, et al. Reference values of lymphocyte subsets in healthy, HIV-negative children in Cameroon. *Clin Vaccine Immunol*. 2011; 18(5): 790-795.
- 105. Hanson LA, Korotkova M, Lundin S, Haversen L, Silfverdal SA, Mattsby-Baltzer I, et al. The transfer of immunity from mother to child. *Ann N Y Acad Sci*. 2003; 987(199-206.
- 106. Qi Q, Zhang DW, Weyand CM, Goronzy JJ. Mechanisms shaping the naive T cell repertoire in the elderly thymic involution or peripheral homeostatic proliferation? *Exp Gerontol*. 2014; 54(71-74.
- Huck K, Feyen O, Ghosh S, Beltz K, Bellert S, Niehues T. Memory B-cells in healthy and antibodydeficient children. *Clin Immunol*. 2009; 131(1): 50-59.
- 108. van Gent R, van Tilburg CM, Nibbelke EE, Otto SA, Gaiser JF, Janssens-Korpela PL, et al. Refined characterization and reference values of the pediatric T- and B-cell compartments. *Clin Immunol*. 2009; 133(1): 95-107.
- 109. Morbach H, Eichhorn EM, Liese JG, Girschick HJ. Reference values for B cell subpopulations from infancy to adulthood. *Clin Exp Immunol*. 2010; 162(2): 271-279.
- 110. Piatosa B, Wolska-Kusnierz B, Pac M, Siewiera K, Galkowska E, Bernatowska E. B cell subsets in healthy children: reference values for evaluation of B cell maturation process in peripheral blood. *Cytometry B Clin Cytom*. 2010; 78(6): 372-381.
- 111. Driessen GJ, Dalm VA, van Hagen PM, Grashoff HA, Hartwig NG, van Rossum AM, et al. Common variable immunodeficiency and idiopathic primary hypogammaglobulinemia: two different conditions within the same disease spectrum. *Haematologica*. 2013; 98(10): 1617-1623.
- 112. Tosato F, Bucciol G, Pantano G, Putti MC, Sanzari MC, Basso G, et al. Lymphocytes subsets reference values in childhood. *Cytometry A*. 2015; 87(1): 81-85.
- 113. Duchamp M, Sterlin D, Diabate A, Uring-Lambert B, Guerin-El Khourouj V, Le Mauff B, et al. B-cell subpopulations in children: National reference values. *Immun Inflamm Dis.* 2014; 2(3): 131-140.
- 114. Brown JC, Newcomb WW. Herpesvirus capsid assembly: insights from structural analysis. *Curr Opin Virol*. 2011; 1(2): 142-149.
- 115. Staras SA, Dollard SC, Radford KW, Flanders WD, Pass RF, Cannon MJ. Seroprevalence of cytomegalovirus infection in the United States, 1988-1994. *Clin Infect Dis.* 2006; 43(9): 1143-1151.
- 116. Bradley H, Markowitz LE, Gibson T, McQuillan GM. Seroprevalence of herpes simplex virus types 1 and 2--United States, 1999-2010. *J Infect Dis.* 2014; 209(3): 325-333.
- 117. Heldwein EE, Krummenacher C. Entry of herpesviruses into mammalian cells. *Cell Mol Life Sci.* 2008; 65(11): 1653-1668.
- 118. Mettenleiter TC. Pathogenesis of neurotropic herpesviruses: role of viral glycoproteins in neuroinvasion and transneuronal spread. *Virus Res.* 2003; 92(2): 197-206.
- Taylor-Wiedeman J, Sissons JG, Borysiewicz LK, Sinclair JH. Monocytes are a major site of persistence of human cytomegalovirus in peripheral blood mononuclear cells. *J Gen Virol*. 1991; 72 (Pt 9)(2059-2064.
- 120. Gerna G, Percivalle E, Baldanti F, Sozzani S, Lanzarini P, Genini E, et al. Human cytomegalovirus replicates abortively in polymorphonuclear leukocytes after transfer from infected endothelial cells via transient microfusion events. *J Virol*. 2000; 74(12): 5629-5638.
- 121. Revello MG, Gerna G. Human cytomegalovirus tropism for endothelial/epithelial cells: scientific background and clinical implications. *Rev Med Virol*. 2010; 20(3): 136-155.
- 122. Grundy JE, Lawson KM, MacCormac LP, Fletcher JM, Yong KL. Cytomegalovirus-infected endothelial cells recruit neutrophils by the secretion of C-X-C chemokines and transmit virus by direct neutrophil-endothelial cell contact and during neutrophil transendothelial migration. *J Infect Dis.* 1998; 177(6): 1465-1474.
- 123. Babcock GJ, Decker LL, Volk M, Thorley-Lawson DA. EBV persistence in memory B cells in vivo. *Immunity*. 1998; 9(3): 395-404.
- 124. Niedobitek G, Agathanggelou A, Herbst H, Whitehead L, Wright DH, Young LS. Epstein-Barr virus (EBV) infection in infectious mononucleosis: virus latency, replication and phenotype of EBV-infected cells. *J Pathol.* 1997; 182(2): 151-159.
- 125. Fingeroth JD, Weis JJ, Tedder TF, Strominger JL, Biro PA, Fearon DT. Epstein-Barr virus receptor of human B lymphocytes is the C3d receptor CR2. *Proc Natl Acad Sci U S A*. 1984; 81(14): 4510-4514.
- 126. Li Q, Spriggs MK, Kovats S, Turk SM, Comeau MR, Nepom B, et al. Epstein-Barr virus uses HLA class II as a cofactor for infection of B lymphocytes. *J Virol*. 1997; 71(6): 4657-4662.

32

- Chapter 1
- Chapter 1
- 127. Dorner M, Zucol F, Alessi D, Haerle SK, Bossart W, Weber M, et al. beta1 integrin expression increases susceptibility of memory B cells to Epstein-Barr virus infection. J Virol. 2010; 84(13): 6667-6677.
- 128. Chaganti S, Heath EM, Bergler W, Kuo M, Buettner M, Niedobitek G, et al. Epstein-Barr virus colonization of tonsillar and peripheral blood B-cell subsets in primary infection and persistence. Blood. 2009; 113(25): 6372-6381.
- 129. van de Berg PJ, Yong SL, Remmerswaal EB, van Lier RA, ten Berge IJ. Cytomegalovirus-induced effector T cells cause endothelial cell damage. Clin Vaccine Immunol. 2012; 19(5): 772-779.
- 130. Bolovan-Fritts CA, Trout RN, Spector SA. High T-cell response to human cytomegalovirus induces chemokine-mediated endothelial cell damage. Blood. 2007; 110(6): 1857-1863.
- 131. Hsu JL, Glaser SL. Epstein-barr virus-associated malignancies: epidemiologic patterns and etiologic implications. Crit Rev Oncol Hematol. 2000; 34(1): 27-53.
- 132. Choe H, Farzan M, Sun Y, Sullivan N, Rollins B, Ponath PD, et al. The beta-chemokine receptors CCR3 and CCR5 facilitate infection by primary HIV-1 isolates. Cell. 1996; 85(7): 1135-1148.
- 133. Alkhatib G, Combadiere C, Broder CC, Feng Y, Kennedy PE, Murphy PM, et al. CC CKR5: a RANTES, MIP-1alpha, MIP-1beta receptor as a fusion cofactor for macrophage-tropic HIV-1. Science. 1996; 272(5270): 1955-1958.
- 134. Dragic T, Litwin V, Allaway GP, Martin SR, Huang Y, Nagashima KA, et al. HIV-1 entry into CD4+ cells is mediated by the chemokine receptor CC-CKR-5. Nature. 1996; 381(6584): 667-673.
- 135. Deng H, Liu R, Ellmeier W, Choe S, Unutmaz D, Burkhart M, et al. Identification of a major coreceptor for primary isolates of HIV-1. Nature. 1996; 381(6584): 661-666.
- 136. Feng Y, Broder CC, Kennedy PE, Berger EA. HIV-1 entry cofactor: functional cDNA cloning of a seven-transmembrane, G protein-coupled receptor. Science. 1996; 272(5263): 872-877.
- 137. Li H, Pauza CD. HIV envelope-mediated, CCR5/alpha4beta7-dependent killing of CD4-negative gammadelta T cells which are lost during progression to AIDS. Blood. 2011; 118(22): 5824-5831.
- 138. Cicala C, Martinelli E, McNally JP, Goode DJ, Gopaul R, Hiatt J, et al. The integrin alpha4beta7 forms a complex with cell-surface CD4 and defines a T-cell subset that is highly susceptible to infection by HIV-1. Proc Natl Acad Sci U S A. 2009; 106(49): 20877-20882.
- 139. Krzysiek R, Rudent A, Bouchet-Delbos L, Foussat A, Boutillon C, Portier A, et al. Preferential and persistent depletion of CCR5+ T-helper lymphocytes with nonlymphoid homing potential despite early treatment of primary HIV infection. Blood. 2001; 98(10): 3169-3171.
- 140. Arthos J, Cicala C, Martinelli E, Macleod K, Van Ryk D, Wei D, et al. HIV-1 envelope protein binds to and signals through integrin alpha4beta7, the gut mucosal homing receptor for peripheral T cells. Nat Immunol. 2008; 9(3): 301-309.
- 141. Moir S, Malaspina A, Li Y, Chun TW, Lowe T, Adelsberger J, et al. B cells of HIV-1-infected patients bind virions through CD21-complement interactions and transmit infectious virus to activated T cells. J Exp Med. 2000; 192(5): 637-646.
- 142. Geijtenbeek TB, Kwon DS, Torensma R, van Vliet SJ, van Duijnhoven GC, Middel J, et al. DC-SIGN, a dendritic cell-specific HIV-1-binding protein that enhances trans-infection of T cells. Cell. 2000; 100(5): 587-597.
- 143. Douek DC, Brenchley JM, Betts MR, Ambrozak DR, Hill BJ, Okamoto Y, et al. HIV preferentially infects HIV-specific CD4+ T cells. Nature. 2002; 417(6884): 95-98.
- 144. Tortorella D, Gewurz BE, Furman MH, Schust DJ, Ploegh HL. Viral subversion of the immune system. Annu Rev Immunol. 2000; 18(861-926.
- 145. Ressing ME, Luteijn RD, Horst D, Wiertz EJ. Viral interference with antigen presentation: trapping TAP. Mol Immunol. 2013; 55(2): 139-142.
- 146. Abendroth A, Kinchington PR, Slobedman B. Varicella zoster virus immune evasion strategies. Curr Top Microbiol Immunol. 2010; 342(155-171.
- 147. Suazo PA, Ibanez FJ, Retamal-Diaz AR, Paz-Fiblas MV, Bueno SM, Kalergis AM, et al. Evasion of early antiviral responses by herpes simplex viruses. Mediators Inflamm. 2015; 2015(593757.
- 148. Arens R. Rational design of vaccines: learning from immune evasion mechanisms of persistent viruses and tumors. Adv Immunol. 2012; 114(217-243.
- 149. Jackson SE, Mason GM, Wills MR. Human cytomegalovirus immunity and immune evasion. Virus Res. 2011; 157(2): 151-160.
- 150. Hill AB, Barnett BC, McMichael AJ, McGeoch DJ. HLA class I molecules are not transported to the cell surface in cells infected with herpes simplex virus types 1 and 2. J Immunol. 1994; 152(6):

2736-2741.

- 151. Hengel H, Flohr T, Hammerling GJ, Koszinowski UH, Momburg F. Human cytomegalovirus inhibits peptide translocation into the endoplasmic reticulum for MHC class I assembly. *J Gen Virol*. 1996; 77 (Pt 9)(2287-2296.
- 152. Cohen JI. Infection of cells with varicella-zoster virus down-regulates surface expression of class I major histocompatibility complex antigens. *J Infect Dis*. 1998; 177(5): 1390-1393.
- 153. Abendroth A, Lin I, Slobedman B, Ploegh H, Arvin AM. Varicella-zoster virus retains major histocompatibility complex class I proteins in the Golgi compartment of infected cells. *J Virol*. 2001; 75(10): 4878-4888.
- 154. Schwartz O, Marechal V, Le Gall S, Lemonnier F, Heard JM. Endocytosis of major histocompatibility complex class I molecules is induced by the HIV-1 Nef protein. *Nat Med.* 1996; 2(3): 338-342.
- 155. Kerkau T, Bacik I, Bennink JR, Yewdell JW, Hunig T, Schimpl A, et al. The human immunodeficiency virus type 1 (HIV-1) Vpu protein interferes with an early step in the biosynthesis of major histocompatibility complex (MHC) class I molecules. *J Exp Med.* 1997; 185(7): 1295-1305.
- 156. York IA, Roop C, Andrews DW, Riddell SR, Graham FL, Johnson DC. A cytosolic herpes simplex virus protein inhibits antigen presentation to CD8+ T lymphocytes. *Cell*. 1994; 77(4): 525-535.
- 157. Keating S, Prince S, Jones M, Rowe M. The lytic cycle of Epstein-Barr virus is associated with decreased expression of cell surface major histocompatibility complex class I and class II molecules. *J Virol.* 2002; 76(16): 8179-8188.
- 158. Hill A, Jugovic P, York I, Russ G, Bennink J, Yewdell J, et al. Herpes simplex virus turns off the TAP to evade host immunity. *Nature*. 1995; 375(6530): 411-415.
- 159. Ressing ME, Keating SE, van Leeuwen D, Koppers-Lalic D, Pappworth IY, Wiertz EJ, et al. Impaired transporter associated with antigen processing-dependent peptide transport during productive EBV infection. *J Immunol*. 2005; 174(11): 6829-6838.
- Reyburn HT, Mandelboim O, Vales-Gomez M, Davis DM, Pazmany L, Strominger JL. The class I MHC homologue of human cytomegalovirus inhibits attack by natural killer cells. *Nature*. 1997; 386(6624): 514-517.
- 161. Schepis D, D'Amato M, Studahl M, Bergstrom T, Karre K, Berg L. Herpes simplex virus infection downmodulates NKG2D ligand expression. *Scand J Immunol.* 2009; 69(5): 429-436.
- Cerboni C, Neri F, Casartelli N, Zingoni A, Cosman D, Rossi P, et al. Human immunodeficiency virus 1 Nef protein downmodulates the ligands of the activating receptor NKG2D and inhibits natural killer cell-mediated cytotoxicity. *J Gen Virol*. 2007; 88(Pt 1): 242-250.
- 163. Nachmani D, Stern-Ginossar N, Sarid R, Mandelboim O. Diverse herpesvirus microRNAs target the stress-induced immune ligand MICB to escape recognition by natural killer cells. *Cell Host Microbe*. 2009; 5(4): 376-385.
- Kurth J, Spieker T, Wustrow J, Strickler GJ, Hansmann LM, Rajewsky K, et al. EBV-infected B cells in infectious mononucleosis: viral strategies for spreading in the B cell compartment and establishing latency. *Immunity*. 2000; 13(4): 485-495.
- Kurth J, Hansmann ML, Rajewsky K, Kuppers R. Epstein-Barr virus-infected B cells expanding in germinal centers of infectious mononucleosis patients do not participate in the germinal center reaction. *Proc Natl Acad Sci U S A*. 2003; 100(8): 4730-4735.
- 166. Heath E, Begue-Pastor N, Chaganti S, Croom-Carter D, Shannon-Lowe C, Kube D, et al. Epstein-Barr virus infection of naive B cells in vitro frequently selects clones with mutated immunoglobulin genotypes: implications for virus biology. *PLoS Pathog*. 2012; 8(5): e1002697.
- Ehlin-Henriksson B, Gordon J, Klein G. B-lymphocyte subpopulations are equally susceptible to Epstein-Barr virus infection, irrespective of immunoglobulin isotype expression. *Immunology*. 2003; 108(4): 427-430.
- 168. Chaganti S, Ma CS, Bell AI, Croom-Carter D, Hislop AD, Tangye SG, et al. Epstein-Barr virus persistence in the absence of conventional memory B cells: IgM+IgD+CD27+ B cells harbor the virus in X-linked lymphoproliferative disease patients. *Blood*. 2008; 112(3): 672-679.
- 169. Souza TA, Stollar BD, Sullivan JL, Luzuriaga K, Thorley-Lawson DA. Influence of EBV on the peripheral blood memory B cell compartment. *J Immunol.* 2007; 179(5): 3153-3160.
- 170. Xu W, Santini PA, Sullivan JS, He B, Shan M, Ball SC, et al. HIV-1 evades virus-specific IgG2 and IgA responses by targeting systemic and intestinal B cells via long-range intercellular conduits. *Nat Immunol.* 2009; 10(9): 1008-1017.
- 171. Amu S, Ruffin N, Rethi B, Chiodi F. Impairment of B-cell functions during HIV-1 infection. Aids.

2013; 27(15): 2323-2334.

- 172. Moir S, Fauci AS. Pathogenic mechanisms of B-lymphocyte dysfunction in HIV disease. *J Allergy Clin Immunol.* 2008; 122(1): 12-19; quiz 20-11.
- 173. Moir S, Fauci AS. B cells in HIV infection and disease. Nat Rev Immunol. 2009; 9(4): 235-245.
- 174. Lane HC, Masur H, Edgar LC, Whalen G, Rook AH, Fauci AS. Abnormalities of B-cell activation and immunoregulation in patients with the acquired immunodeficiency syndrome. *N Engl J Med.* 1983; 309(8): 453-458.
- 175. De Milito A, Morch C, Sonnerborg A, Chiodi F. Loss of memory (CD27) B lymphocytes in HIV-1 infection. *Aids*. 2001; 15(8): 957-964.
- 176. D'Orsogna LJ, Krueger RG, McKinnon EJ, French MA. Circulating memory B-cell subpopulations are affected differently by HIV infection and antiretroviral therapy. *Aids*. 2007; 21(13): 1747-1752.
- 177. Scott ME, Landay AL, Lint TF, Spear GT. In vivo decrease in the expression of complement receptor 2 on B-cells in HIV infection. *Aids.* 1993; 7(1): 37-41.
- 178. Moir S, Malaspina A, Ogwaro KM, Donoghue ET, Hallahan CW, Ehler LA, et al. HIV-1 induces phenotypic and functional perturbations of B cells in chronically infected individuals. *Proc Natl Acad Sci U S A*. 2001; 98(18): 10362-10367.
- 179. Pakker NG, Notermans DW, de Boer RJ, Roos MT, de Wolf F, Hill A, et al. Biphasic kinetics of peripheral blood T cells after triple combination therapy in HIV-1 infection: a composite of redistribution and proliferation. *Nat Med.* 1998; 4(2): 208-214.
- 180. Cohen Stuart JW, Slieker WA, Rijkers GT, Noest A, Boucher CA, Suur MH, et al. Early recovery of CD4+ T lymphocytes in children on highly active antiretroviral therapy. Dutch study group for children with HIV infections. *Aids.* 1998; 12(16): 2155-2159.
- 181. Duong T, Judd A, Collins IJ, Doerholt K, Lyall H, Foster C, et al. Long-term virological outcome in children on antiretroviral therapy in the UK and Ireland. *Aids*. 2014; 28(16): 2395-2405.
- 182. van Rossum AM, Scherpbier HJ, van Lochem EG, Pakker NG, Slieker WA, Wolthers KC, et al. Therapeutic immune reconstitution in HIV-1-infected children is independent of their age and pretreatment immune status. *Aids*. 2001; 15(17): 2267-2275.
- 183. Savva GM, Pachnio A, Kaul B, Morgan K, Huppert FA, Brayne C, et al. Cytomegalovirus infection is associated with increased mortality in the older population. *Aging Cell*; 12(3): 381-387.
- Khan N, Hislop A, Gudgeon N, Cobbold M, Khanna R, Nayak L, et al. Herpesvirus-specific CD8 T cell immunity in old age: cytomegalovirus impairs the response to a coresident EBV infection. J Immunol. 2004; 173(12): 7481-7489.
- 185. Weinberger B, Lazuardi L, Weiskirchner I, Keller M, Neuner C, Fischer KH, et al. Healthy aging and latent infection with CMV lead to distinct changes in CD8+ and CD4+ T-cell subsets in the elderly. *Hum Immunol*. 2007; 68(2): 86-90.
- 186. Derhovanessian E, Maier AB, Hahnel K, McElhaney JE, Slagboom EP, Pawelec G. Latent infection with cytomegalovirus is associated with poor memory CD4 responses to influenza A core proteins in the elderly. *J Immunol.* 2014; 193(7): 3624-3631.
- Frasca D, Diaz A, Romero M, Landin AM, Blomberg BB. Cytomegalovirus (CMV) seropositivity decreases B cell responses to the influenza vaccine. *Vaccine*; 33(12): 1433-1439.
- den Elzen WP, Vossen AC, Cools HJ, Westendorp RG, Kroes AC, Gussekloo J. Cytomegalovirus infection and responsiveness to influenza vaccination in elderly residents of long-term care facilities. *Vaccine*. 2011; 29(29-30): 4869-4874.
- Odumade OA, Knight JA, Schmeling DO, Masopust D, Balfour HH, Jr., Hogquist KA. Primary Epstein-Barr virus infection does not erode preexisting CD8(+) T cell memory in humans. *J Exp Med.* 2012; 209(3): 471-478.
- Remmerswaal EB, Havenith SH, Idu MM, van Leeuwen EM, van Donselaar KA, Ten Brinke A, et al. Human virus-specific effector-type T cells accumulate in blood but not in lymph nodes. *Blood*. 2012; 119(7): 1702-1712.
- 191. van Leeuwen EM, Koning JJ, Remmerswaal EB, van Baarle D, van Lier RA, ten Berge IJ. Differential usage of cellular niches by cytomegalovirus versus EBV- and influenza virus-specific CD8+ T cells. *J Immunol.* 2006; 177(8): 4998-5005.
- 192. Nilsson C, Larsson Sigfrinius AK, Montgomery SM, Sverremark-Ekstrom E, Linde A, Lilja G, et al. Epstein-Barr virus and cytomegalovirus are differentially associated with numbers of cytokine-producing cells and early atopy. *Clin Exp Allergy*. 2009; 39(4): 509-517.
- 193. Saghafian-Hedengren S, Sverremark-Ekstrom E, Linde A, Lilja G, Nilsson C. Early-life EBV infection

protects against persistent IgE sensitization. J Allergy Clin Immunol. 2010; 125(2): 433-438.

- 194. Sansoni P, Vescovini R, Fagnoni FF, Akbar A, Arens R, Chiu YL, et al. New advances in CMV and immunosenescence. *Exp Gerontol*. 2014; 55(54-62.
- 195. Pawelec G. Immunosenenescence: role of cytomegalovirus. Exp Gerontol. 2014; 54(1-5.
- 196. Rakhmanov M, Keller B, Gutenberger S, Foerster C, Hoenig M, Driessen G, et al. Circulating CD21low B cells in common variable immunodeficiency resemble tissue homing, innate-like B cells. *Proc Natl Acad Sci U S A*. 2009; 106(32): 13451-13456.
- 197. Wehr C, Eibel H, Masilamani M, Illges H, Schlesier M, Peter HH, et al. A new CD21low B cell population in the peripheral blood of patients with SLE. *Clin Immunol.* 2004; 113(2): 161-171.
- Moir S, Ho J, Malaspina A, Wang W, DiPoto AC, O'Shea MA, et al. Evidence for HIV-associated B cell exhaustion in a dysfunctional memory B cell compartment in HIV-infected viremic individuals. *J Exp Med.* 2008; 205(8): 1797-1805.
CHAPTER 2

EFFECTS OF EXTERNAL DETERMINANTS ON AGE-RELATED PATTERNS OF INNATE LEUKOCYTE AND NAIVE AND MEMORY B- AND T-LYMPHOCYTE NUMBERS IN EARLY CHILDHOOD: THE GENERATION R STUDY

Running title: Determinants of leukocyte subset kinetics in children

D. van den Heuvel,¹ M.A.E. Jansen,^{2,3} K. Nasserinejad,⁴ W.A. Dik,¹ E.G. van Lochem,^{1*} L.E. Bakker-Jonges,^{1#} H. Bouallouch-Charif,¹ V.W.V. Jaddoe,^{2,3,5} H. Hooijkaas,¹ J.J.M. van Dongen,¹ H.A. Moll,³ M.C. van Zelm.^{1#}

¹ Dept. of Immunology, Erasmus MC, University Medical Center, Rotterdam, NL
 ² The Generation R Study Group, Erasmus MC, University Medical Center, Rotterdam, NL
 ³ Dept. of Pediatrics, Erasmus MC-Sophia, Rotterdam, NL
 ⁴ Dept. of Biostatistics Erasmus MC, University Medical Center, Rotterdam, NL
 ⁵ Dept. of Epidemiology Erasmus MC, University Medical Center, Rotterdam, NL
 * Present address: Dept. of Microbiology and Immunology, Rijnstate Hospital, Arnhem, NL
 # Present address: Dept. of Medical Laboratory, Reinier de Graaf Groep, Delft, NL
 * Present address: Dept. of Immunology, Central Clinical School, Monash University, Melbourne, Victoria, Australia

In preparation



ABSTRACT

BACKGROUND. Blood leukocyte subsets in young children are highly dynamic, resulting in wide reference ranges. This is likely related to variable exposure to external antigens, and the generation of immunological memory following primary antigen responses. However, it is unclear which external determinants affect the dynamics of innate leukocytes, and naive and memory lymphocyte subsets.

METHODS. Using 6-color flow cytometry and linear mixed effect modeling, we defined the dynamics of 62 leukocyte subsets from birth to 6 years of age in 1,182 children with one to five measurements per individual. Subsequently, we defined the impact of prenatal maternal lifestyle-related, prenatal maternal immune-mediated, birth characteristics and bacterial/viral exposure-related determinants on these leukocyte dynamics.

Results. Hierarchical clustering of leukocyte subset dynamics revealed four major age-related patterns that each contained functionally similar leukocyte populations. Innate leukocyte numbers were high at birth and were predominantly affected by maternal low education level. Naive lymphocytes peaked around 1 year, while most memory lymphocyte subsets more gradually increased during the first 1 to 4 years of life. Dynamics of CD4⁺ T cells were predominantly associated with gender, birth characteristics, and persistent infections with cytomegalovirus (CMV) or Epstein Barr virus (EBV). CD8⁺ T cells were predominantly associated with CMV and EBV infections. TCRy δ^+ T cells were associated with premature rupture of membranes and CMV infection. B-cell subsets were predominantly associated with gender, breastfeeding and Helicobacter Pylori carriership.

CONCLUSIONS. Our study identifies specific dynamic patterns of leukocyte subset numbers, as well as determinants that affect these patterns, thereby providing new insights into the shaping of the childhood immune system.

INTRODUCTION

The immune system in young children is highly dynamic in composition and cell numbers.¹⁻⁶ Innate cell numbers, such as neutrophils and NK cells, are higher in neonates than in children or adults,^{1,5,7} and already display dynamic changes within the first few days after birth.^{1,5,6} Furthermore, B and T cells are mostly naive in infants, while protective immunity is gradually built up in the form of increasing numbers of memory B and T cells during the first 5 years of life after which these numbers stabilize.^{2-4,8-11}

Age-associated changes in leukocyte numbers have been extensively studied in the past two decades, and findings were quite consistent.^{1-5,8-14} Both short-term follow-up of selected leukocyte subsets in infants in the first year of life,^{5,6,12,13,15} and cross-sectional studies in children and adults,^{3,4,6,10,11,14} have shown agerelated differences in B- and T-cell numbers, whereas innate leukocyte subset numbers are quite stable. Still, it is difficult to make correlations between various leukocyte lineages, due to either the restricted numbers of analyzed subsets or the short-term follow-up. Furthermore, the determinants that drive the kinetics in immune development remain less well studied. Likely, various environmental determinants affect the development of certain leukocyte subsets during the first years of life, such as prenatal maternal life style and immune-mediated diseases, birth characteristics, and bacterial and viral exposure.¹⁶⁻²⁶ Studying the impact of environmental determinants on the dynamics of childhood immune development requires large cohorts of healthy young children, with multiple measurements per individual.

We investigated which external determinants, related to the prenatal maternal lifestyle, prenatal maternal immune-mediated diseases, birth characteristics or bacterial/viral exposure-related characteristics, influence the dynamics of blood leukocyte populations from birth until 6 years of age. This concerned a total of 62 leukocyte populations, including innate leukocyte subsets, naive and memory B-cell and T-cell subsets and TCRy δ^+ T-cell subsets.

PATIENTS AND METHODS

STUDY SUBJECTS

This study was embedded in the Generation R Study, a prospective populationbased cohort study from fetal life until young adulthood.^{27,28} The current study focused on a subgroup of 1,182 two-generation Dutch children, born between August 2003 and August 2006. Peripheral blood was obtained at birth (cord blood; n=220), and median age of 6 months (5-9m; n=376), 14 months (13-17m; n=241), 25 months (22-30m; n=257) and 72 months (61-95m; n=916). Detailed immunophenotyping was performed at 1-5 time points per child, resulting in a total of 2,010 data points. Written informed consent was obtained from parents, according to the Medical Ethics Committee guidelines of Erasmus MC.

IMMUNOPHENOTYPING

Absolute numbers of leukocytes, NK cells, T cells and B cells were obtained with a diagnostic lyse-no-wash protocol, using a BD FACSCalibur. Six-color flow cytometry was performed to quantify 62 well-defined leukocyte populations

n= 62	n= 31	Name population	Phenotype definition
		Innate leukocytes	
1		Granulocytes	SSC ^{high} CD45 ⁺
2		CD15+ granulocytes	SSC ^{high} CD45 ⁺ CD15 ⁺
3	1′	Neutrophils	SSC ^{high} CD45 ⁺ CD16 ⁺
4	2′	Eosinophils	SSC ^{high} CD45 ⁺ CD16 ⁻
5		Monocytes	FSC ^{inter} SSC ^{inter} CD45 ⁺
6	3′	classical monocytes	FSC ^{inter} SSC ^{inter} CD45 ⁺ CD14 ⁺ CD16 ⁻
7	4′	intermediate monocytes	FSC ^{inter} SSC ^{inter} CD45 ⁺ CD14 ⁺ CD16 ⁺
8	5′	Non-classical monocytes	FSC ^{inter} SSC ^{inter} CD45 ⁺ CD14 ⁻ CD16 ⁺
9	6′	NK cells	SSClowCD45+CD3-CD16+or CD56+
		Lymphocytes	
10		T cells	SSC ^{low} CD45 ⁺ CD3 ⁺
11		TCRaβ+ T cells	SSC ^{low} CD3+TCRγδ-TCRaβ+
12		\Box CD4+ TCRa β + T cells	SSC ^{low} CD3+TCRγδ-TCRαβ+CD8-CD4+
13		\Box CD8+ TCRa β + T cells	SSC ^{Iow} CD3+TCRγδ-TCRαβ+CD8+CD4-
14		TCRγδ+ T cells	SSC ^{low} CD3+TCRγδ+TCRαβ ⁻
15		∟CD8+ TCRγδ+ T cells	SSC ^{low} CD3+TCRγδ+TCRaβ ⁻ CD8+CD4 ⁻
16		∟CD4+ TCRγδ+ T cells	SSC ^{low} CD3+TCRγδ+TCRaβ ⁻ CD8 ⁻ CD4+
17		\Box Vδ1-Vδ2- TCRaβ- T cells	SSC ^{low} CD3+TCRaβ-Vδ1-Vδ2-
18		$ LV\delta2+T$ cells	SSC ^{low} CD3 ⁺ TCRaβ ⁻ Vδ1 ⁻ Vδ2 ⁺
19	7′	∟Vδ1+ T cells	SSC ^{low} CD3+TCRaβ-Vδ1+Vδ2-
20		∟Vγ9+ T cells	FSC ^{low} SSC ^{low} CD3 ⁺ TCRaβ ⁻ Vγ9 ⁺
21	8′	∟Vδ2+Vγ9+ T cells	FSC ^{Iow} SSC ^{Iow} CD3 ⁺ TCRaβ ⁻ Vδ2 ⁺ Vγ9 ⁺
22		CD4+ T cells	SSClowCD45+CD3+CD4+CD8-
23	9′	∟CD4+ Tnaive	FSClowSSClowCD3+CD8-CD197+CD45R0-CD27+CD28+
24		∟CD4+ Tmem	FSC ^{low} SSC ^{low} CD3 ⁺ CD8 ⁻ CD197 ⁻ & CD197 ⁺ CD45RO ⁺
25	10'	∟CD4+ Tcm	FSC ^{low} SSC ^{low} CD3 ⁺ CD8 ⁻ CD197 ⁺ CD45R0 ⁺ CD27 ⁺ CD28 ⁺
26		∟CD4+ TemRO	FSC ^{low} SSC ^{low} CD3 ⁺ CD8 ⁻ CD197 ⁻ CD45R0 ⁺
27	11'	∟CD4+ early TemRO	FSC ^{low} SSC ^{low} CD3 ⁺ CD8 ⁻ CD197 ⁻ CD45R0 ⁺ CD27 ⁺ CD28 ⁺
28	12′	∟ CD4+ interm TemRO	FSC ^{low} SSC ^{low} CD3 ⁺ CD8 ⁻ CD197 ⁻ CD45R0 ⁺ CD27 ⁻ CD28 ⁺
29	13′	∟CD4+ late TemRO	FSClowSSClowCD3+CD8-CD197-CD45R0+CD27-CD28-
30		∟CD4+ TemRA	FSC ^{Iow} SSC ^{Iow} CD3 ⁺ CD8 ⁻ CD197 ⁻ CD45R0 ⁻
31	14'	CD4+ early TemRA	FSC ^{Iow} SSC ^{Iow} CD3 ⁺ CD8 ⁻ CD197 ⁻ CD45R0 ⁻ CD27 ⁺ CD28 ⁺
32	15'	∟ CD4+ interm TemRA	FSC ^I ^{ow} SSC ^I ^{ow} CD3 ⁺ CD8 ⁻ CD197 ⁻ CD45R0 ⁻ CD27 ⁻ CD28 ⁺
33	16′	∟CD4+ late TemRA	FSC/0wSSC/0wCD3+CD8-CD197-CD45R0-CD27-CD28-
34	4 7 (CD8+ I cells	SSC/0wCD45+CD3+CD4-CD8+
35	17	∟CD8+ Inaive	FSC/0wSSC/0wCD3+CD8+CD197+CD45R0-CD27+CD28+
36	4.07	LCD8+ Imem	FSC/0wSSC/0wCD3+CD8+ CD19/- & CD19/+CD45R0+
37	18'		FSC/0wSSC/0wCD3+CD8+CD197+CD45R0+CD27+CD28+
38		∟CD8+ TemRO	FSC ^{IOW} SSC ^{IOW} CD3 ⁺ CD8 ⁺ CD197 ⁻ CD45RO ⁺
39	19'	LCD8+ early lemRO	
40	20'	LCD8+ Interm IemRO	
41	21′	LCD8+ late lemRO	
42		∟CD8+ IemRA	FSC ¹⁰⁰ SSC ¹⁰⁰ CD3 ⁺ CD8 ⁺ CD197 ⁻ CD45R0 ⁻
43	22'	∟CD8+ early TemRA	FSC ^{10W} SSC ^{10W} CD3 ⁺ CD8 ⁺ CD197 ⁻ CD45R0 ⁻ CD27 ⁺ CD28 ⁺
44	23'	∟CD8+ interm TemRA	FSC ^{10W} SSC ^{10W} CD3 ⁺ CD8 ⁺ CD197 ⁻ CD45R0 ⁻ CD27 ⁺ CD28 ⁻
45	24′	∟CD8+ late TemRA	FSCIOWSSCIOWCD3+CD8+CD197-CD45R0-CD27-CD28-

Table 1. Definition of leukocyte subsets

Table 1	(cont	inued)	
n= 62	n= 31	Name population	Phenotype definition
46		B cells	SSC ^{low} CD45 ⁺ CD19 ⁺
47	25′	Bnaive	SSC ^{low} CD19 ⁺ CD27 ⁻ IgD ⁺
48		Bmem	SSC ^{low} CD19 ⁺ IgD ⁻ & CD27 ⁺ IgD ⁺
49		∟CD27- Bmem	SSC ^{low} CD19 ⁺ CD27 ⁻ IgD ⁻
50		∟CD27+ Bmem	SSC ^{low} CD19 ⁺ CD27 ⁺ IgD ⁻
51		∟IgM+ Bmem	SSC ^{low} CD19 ⁺ CD27 ⁺ IgM ⁺
52	26′	∟Natural effector	SSC ^{low} CD19 ⁺ CD27 ⁺ IgM ⁺ IgD ⁺
53	27′	∟IgMonly	SSC ^{low} CD19 ⁺ CD27 ⁺ IgM ⁺ IgD ⁻
54		∟IgA+ Bmem	SSC ^{low} CD19 ⁺ IgA ⁺
55	28′	∟CD27-IgA+	SSC ^{low} CD19 ⁺ CD27 ⁻ IgA ⁺
56	29′	∟CD27+IgA+	SSC ^{low} CD19 ⁺ CD27 ⁺ IgA ⁺
57		∟IgG+ Bmem	SSC ^{low} CD19 ⁺ IgG ⁺
58	30′	∟CD27-IgG+	SSC ^{low} CD19 ⁺ CD27 ⁻ IgG ⁺
59	31′	∟CD27+IgG+	SSC ^{low} CD19 ⁺ CD27 ⁺ IgG ⁺
60		CD21low B cells	SSC ^{low} CD19 ⁺ CD38 ^{low} CD21 ^{low}
61		Igλ+ B cells	SSC ^{low} CD19 ⁺ Igκ ⁻ Igλ ⁺
62		Igκ+ B cells	SSC ^{low} CD19 ⁺ Igκ ⁺ Igλ ⁻

(Table 1), which included NK-cell, monocyte, granulocyte, naive and memory B-cell, naive and memory TCRa β^+ T-cell and TCR $\gamma\delta^+$ T-cell subsets (Supplemental Table 1). Flowcytometric analyses were performed on an LSRII (BD Biosciences) using standardized measurement settings.²⁹

DETERMINANTS

Information on 26 dichotomized determinants was obtained (Table 2).^{27,28} Six of these were related to prenatal maternal life style, and were evaluated using questionnaires in the first, second and third trimester of pregnancy:²⁸ maternal age (above 35y defined as a group with increased risks during pregnancy), education, net household income, smoking or alcohol use during pregnancy, and maternal body mass index before pregnancy (>25 defined as above healthy range). Three determinants were related to prenatal maternal immune-mediated diseases: maternal atopy defined by guestionnaire data,²⁸ and serum anti-thyroid peroxidase IgG (anti-TPO) and anti-tissue transglutaminase IgA antibody (TG2A) levels.³⁰ Three birth-related determinants included gender, gestational age (preterm birth <37w) and birth weight (low birth weight <2,500g). Fourteen determinants related to bacterial or viral exposure included three perinatal determinants, i.e. premature rupture of membranes, birth season and mode of delivery; and eleven postnatal determinants, i.e. breastfeeding and breastfeeding duration, having siblings, antibiotics usage and presence of upper or lower respiratory tract infection in the first year of life (obtained using questionnaires at the child's age of 2, 3, 6 and 12 months);²⁸ Helicobacter pylori (H. pylori) carriership, and seropositivity (IgG) for cytomegalovirus (CMV), Epstein Barr virus (EBV), herpes simplex virus 1 (HSV-1) and varicella zoster virus (VZV) were defined at the age of 6 years.^{26,31,32} We did not include the postnatal determinants of housing conditions, pets, day care, home environment or other potential sources for differences in antigenic or microbial pressure, because of incomplete data sets.

STATISTICAL MODELING

To model leukocyte dynamics between birth and the age of 6 years, linear mixed effect analyses were performed on the relationship between the age of the children and the size of individual leukocyte populations. By including random-effects in the model, this approach enabled modeling of cross-sectional data, with further improvement of the accuracy by incorporating longitudinal follow-up data from individual children. To capture the trend in the data more precisely, we included a natural spline with different knots (0-3 knots) into the models. Basically, the number of knots is inversely related to the smoothness of the curve. Positions of the knots in the 1-knot model was defined as the 50th percentile (25.5 months), the 2-knots model at 33rd and 66th percentiles (14.1 and 70 months); the knots in the 3-knot model were defined manually at 6, 14 and 24 months, focusing around the time points of data inclusion. Model selection was performed by likelihood ratio test.

Next, for each leukocyte population the effect of all 26 determinants on the models was assessed by first adding each determinant univariately into the model and analyzing fixed effect estimates. To correct for potential multiple testing errors, a correction for the four groups of determinants was performed, and consequently only determinants with an effect of p<0.0125 were defined as significant. Subsequently, for each leukocyte population, the determinants with a significant effect were combined in a multivariable model to correct for possible confounding effects. The p-value for significance for the multivariable analyses was p<0.05. Up to 3 determinants were added per model. Statistical analyses were performed in R (version R-3.2.1).³³

CLUSTERING ANALYSES

To cluster patterns of leukocyte subsets kinetics, the modelled data of each population was normalized into zero mean and unit standard deviation (z-score), using the following calculation:

$$z$$
-score= $\frac{(\text{population size (at month x) - average population size (at months 0-72))}{(\text{standard deviation of population size (at months 0-72))}}$

The normalized leukocyte models were subsequently clustered using agglomerative ('bottom-up') Ward's hierarchical clustering, at each step clustering two clusters with minimum between-cluster distance, using the Euclidean distance measure.

RESULTS

Linear mixed effect modeling of leukocyte-subset cell numbers versus the child's age

To study the dynamics of different leukocyte subsets in young children, 2,010 measurements were performed in 1,182 children between birth and 72 months (6y) of age to quantify 11 leukocyte subsets: four major innate leukocyte subsets (monocytes, neutrophils, eosinophils and NK cells), naive B cells (Bnaive) and memory B cells (Bmem), naive and memory CD4⁺ T cells (CD4⁺ Tnaive; CD4⁺ Tmem), naive and memory CD8⁺ T cells (CD8⁺ Tnaive; CD8⁺ Tmem), and TCR $\gamma\delta^+$ T cells. All four innate leukocyte subsets followed a similar profile, starting with

high numbers at birth, which quickly declined within the first 6 months of age and subsequently remained stable (Figure 1A). In contrast, naive B-cell and T-cell numbers strongly increased after birth, peaked around 6-14 months of age and subsequently decreased to stable levels around the age of 2-6 years (Figure 1B). Memory B-cell and T-cell numbers slowly increased within the first 6-14 months of life, after which numbers declined marginally and stabilized from ~3 years onwards (Figure 1B). Total TCR $\gamma\delta^+$ T-cell numbers increased until 6 months and remained quite stable at these levels (Figure 1C).



Figure 1. Dynamics of innate leukocyte and naive and memory lymphocyte populations in children between birth and 6 years of age.

A) Dynamics of monocytes, neutrophils, eosinophils and NK cells. **B)** Dynamics of naive and memory B-cell and T-cell subsets **C)** Dynamics of TCRy δ^+ T cells. Flow cytometry plots depict population definitions in one representative 6-year-old individual. Graphs depict absolute numbers of cells in blood of 1,182 children with in total 2,010 measurements (gray dots). Linear mixed effect models were generated for each population (solid black line) and represented with the 90% confidence interval (CI) of the model (dashed black lines). For clarity, only direct follow-up time points within one individual were connected with gray lines; i.e. 0-6m, 6-14m, 14-25m or 25-72m.

Thus, innate leukocytes, naive lymphocytes and memory lymphocytes seem to display three distinct patterns in cell numbers versus the child's age, but patterns seemed common to subsets within each of these lineages.

DISTINCT DYNAMICS OF INNATE LEUKOCYTES, AND NAIVE AND MEMORY ${\sf B}$ and ${\sf T}$ cells

To evaluate whether functionally similar immune populations would follow similar dynamics with increasing age, we extended our analysis to 62 leukocyte subsets, including 31 phenotypically non-overlapping subsets, using a more detailed classification of innate cells, and naive and memory B and T cells (Table 1), in the same children. Linear mixed effect models were generated as described above and presented with the raw data points in Supplemental Figures 1-5.

Hierarchical clustering of the total 62 leukocyte populations resulted in 4 distinct clusters of leukocyte dynamics, in which functionally related populations, as well as populations with phenotypic overlapping definitions clustered together (Figure 2A). To test the effect of including populations with overlapping population definitions on the clustering, an additional clustering was performed on a selection of 31 non-overlapping populations (Table 1 and Supplemental Figure 6). The resulting clusters showed similar dynamics (Figure 2B and Supplemental Figure 6B) with only 3 populations being assigned to a different cluster, indicating that the overlapping populations did not overtly skew our clustering approach.

Cell numbers of all populations in the first cluster were high at birth, followed by a sharp decrease within the first 6 months of life, after which they stabilized (Figure 2B). This cluster exclusively contained innate leukocyte subsets, including the initially defined NK cells, monocytes, neutrophils and eosinophils (Figure 1A). All defined innate leukocyte subsets clustered within cluster 1, except for the intermediate monocytes, and except for the non-classical monocytes in the clustering of 31 non-overlapping populations (Supplemental Figure 6).

The populations in the second cluster strongly increased shortly after birth, peaked before 14 months of age, followed by a long-term gradual decrease. Populations in cluster 2 included the three major naive lymphocyte subsets defined in Figure 1B: Bnaive, CD4⁺ Tnaive and CD8⁺ Tnaive. Also the total B and T cells, the intermediate monocytes and the V δ 1⁺ T cells clustered in cluster 2, as well as the non-classical monocytes in the clustering in Supplemental Figure 6. Furthermore, 5 memory populations also showed this pattern: CD4⁺ central memory T cells (Tcm) and CD8⁺ Tcm cells, early differentiated CD8⁺ CD45RO⁻ effector memory T cells (CD8⁺ early TemRA), CD27⁻IgA⁺ memory B cells and natural effector memory B cells.

The populations in cluster 3 and 4 gradually increased in cell number and peaked either at 14 months (cluster 3) or after ~4 years (cluster 4). Clusters 3 and 4 contained only memory B-cell and T-cell subsets, and the TCR $\gamma\delta^+$ T-cell subsets. Cluster 3 contained the three major memory populations described in Figure 1B: Bmem, CD4⁺ Tmem and CD8⁺ Tmem. In particular, these memory populations included the early and intermediate CD8⁺ TemRO subpopulations and the IgMonly, CD27⁺IgA⁺ and CD27⁻IgG⁺ memory B-cell subsets. Cluster 4 contained total TCR $\gamma\delta^+$ T cells and the V $\gamma9^+$ T-cell subsets, as well as the intermediate CD8⁺ TemRO and TemRA cells, most of the CD4⁺ effector memory T-cell subsets and the CD27⁺IgG⁺ memory B cells. CD8⁺ late TemRO and TemRA cells clustered in cluster 3 in the analysis of all 62 populations (Figure 2) and in cluster 4 when only the 31 non-overlapping populations were included (Supplemental Figure 6).



Figure 2. Hierarchical clustering of leukocyte subset dynamics in early childhood. A) Absolute numbers of the 62 populations were calculated between 0-72 months of age using the linear mixed models. For each population, the data was converted to zero mean and unit standard deviation (z score). The resulting patterns of all 62 subsets were subjected to Ward's hierarchical clustering. (*Figure 2 continues on next page*)

Associations between external determinants and leukocyte subset dynamics in the four clusters

To study which external determinants were associated with the four distinct age-related patterns of leukocyte numbers (Figure 2), we analyzed 6 prenatal maternal life style-related determinants, 3 prenatal maternal immune-mediated determinants, 3 birth characteristics, and 14 bacterial/viral exposure-related determinants (Table 2). We observed 1 prenatal maternal life style-related determinant, no prenatal maternal immune-mediated determinants, gender, and 9 bacterial/viral exposure-related determinants that showed a significant association with the dynamics of one or more of the 62 leukocyte populations after multivariable correction (Table 2). All of these 11 determinants also showed a significant association with the 31 non-overlapping leukocyte subsets.

Within cluster 1, 5 determinants were associated to changes in the kinetics of one or more of the leukocyte subsets. The strongest association was observed with maternal education level, both for the analysis of the total 62 populations (37.5%, Figure 3A,C) and the 31 non-overlapping populations (40%, Figure



Figure 2. Hierarchical clustering of leukocyte subset dynamics in early childhood. (*continued*) **B)** Average patterns +/- 1 standard deviation of the subsets in each of the 4 major clusters. A comparable hierarchical cluster analysis on 31 non-overlapping subsets is presented in Supplementary Figure 6.

3B,D). Specifically, low maternal education was associated with a reduction in the patterns of eosinophils and classical monocytes (Figure 4). Female gender was associated with a significant increase in the pattern of neutrophils, and the phenotypically overlapping CD15⁺ granulocyte population. Having more than 1 sibling was associated with a reduction, and CMV infection with an increase, in NK cells. Antibiotics usage in the first year of life was associated with an increase in non-classical monocytes. Prenatal maternal immune-mediated determinants, and the bacterial/viral exposure-related determinants breastfeeding, respiratory tract infection or bacterial/ persistent viral infections did not affect any of the populations within cluster 1.

Bacterial/ viral exposure-related determinants were more frequently found to affect cell numbers in clusters 2, 3 and 4 than in cluster 1 (Figure 3A-B). None of the determinants included in our study affected the patterns of more than

Table 2. Characteristics of external determinants					
	negative individuals (%)	positive individuals (%)	MISSING (%)	sign errect on cluster ª	sign errect conrounder- corrected ^b
Prenatal maternal life style					
Maternal age >35yr	951 (80.5)	231 (19.5)	0 (0)	I	T
Low maternal educational level	400 (33.8)	760 (64.3)	22 (1.9)	1,2,3,4	1,2,3,4
Net household income per month $> \in 2,200$	242 (20.5)	847 (71.7)	93 (7.9)	ı	
Smoking during pregnancy	815 (69)	256 (21.7)	111 (9.4)	I	ı
Alcohol use continued during pregnancy	299 (25.3)	750 (63.5)	133 (11.3)	ı	·
Body Mass Index before pregnancy >25	761 (64.4)	240 (20.3)	181 (15.3)	I	ı
Prenatal maternal immune-mediated diseases					
Maternal atopy (eczema, allergy HDM, hay-fever)	683 (57.8)	383 (32.4)	116 (9.8)		
Anti-TPO >60 IU/ml (before 18 weeks of pregnancy)	849 (71.8)	55 (4.7)	278 (23.5)	·	
Maternal TG2A during pregnancy >6 IU/ml	969 (82)	5 (0.4)	208 (17.6)	4	ı
Birth characteristics					
Girl Gender	600 (50.8)	582 (49.2)	0 (0)	1,2,3,4	1,2,3,4
Preterm birth <37 weeks	76 (6.4)	1,106 (93.6)	0 (0)	ı	
Birth weight <2,500g	1,119 (94.7)	63 (5.3)	0 (0)	I	
Bacterial/viral exposure-related characteristics					
Premature rupture of membranes	1,089 (92.1)	53 (4.5)	40 (3.4)	4	4
Caesarian section versus vaginal/ forceps/vacuum assisted	903 (76.4)	154 (13)	125 (10.6)	3,4	3,4
Birth season; born in fall/winter	682 (57.7)	500 (42.3)	0 (0)	ı	ı
Breastfeeding ever	109 (9.2)	969 (82)	104 (8.8)	I	ı
Breastfeeding at 6 months of age	754 (63.8)	299 (25.3)	129 (10.9)	2,4	2,4
Siblings >1	687 (58.1)	480 (40.6)	15 (1.3)	1	1
Antibiotics/Penicillin use in $1^{st} \gamma$	636 (53.8)	342 (28.9)	204 (17.3)	1,3	1,3
Upper respiratory tract infections in 1^{st} y	547 (46.3)	475 (40.2)	160 (13.5)	I	ı
Lower respiratory tract infections in 1^{st} y (doctor attended)	819 (69.3)	131 (11.1)	232 (19.6)	ı	ı
Carrier of Helicobacter pylori within 6yrs	887 (75)	49 (4.1)	246 (20.8)	2,3	2,3
Seropositivity for Cyomegalovirus (CMV) at 6y	665 (56.3)	269 (22.8)	248 (21)	1,2,3,4	1,2,3,4
Seropositivity for Epstein Barr virus (EBV) at 6y	534 (45.2)	400 (33.8)	248 (21)	2,3,4	2,3,4
Seropositivity for Herpes simplex virus-1 (HSV-1) at 6y	810 (68.5)	124 (10.5)	248 (21)	2	2
Seropositivity for Varicella zoster virus (VZV) at 6y	73 (6.2)	861 (72.8)	248 (21)	ı	I
 Effect in single variable analyses: each determinant was added to the beffect after multivariable correction: per leutecyte population, all det up to a single correction: per leutecyte population, all det 	e linear mixed effect m erminants with a signi	odel univariately to te ficant individual effect	st for a significant were added in a n	effect (Significance wa nultivariable model to c	s defined as p<0.0125) correct for confounding effects.

Determinants of leukocyte subset kinetics in children

Chapter 2

25% of the populations in cluster 2 (Figure 3C-D). However, female gender was associated with an increase in the patterns of CD4⁺ naive and Tcm cells, and consequently in the phenotypically overlapping CD4⁺TCRa β^+ and total CD4⁺ T cells. Whereas breastfeeding for more than 6 months was associated with a reduction in the pattern of total IgA⁺ and CD27⁻IgA⁺ memory B cells, H. pylori carriership was associated with an increase in these populations. Furthermore, H. pylori carriership was associated with an increase in total Igk⁺ B cells and TCRa β^+ T cells. CMV and EBV were both associated with an increase in the strongly related total and TCRa β^+ CD8⁺ T cells, and CMV with an increase in V δ 1⁺ T cells. HSV-1 seropositivity was associated with a decrease in the large population of naive B cells, and consequently with total B cells and Igk⁺ B-cell subsets.

In cluster 3, CMV and EBV seropositivity were most strongly associated with changes in lymphocyte dynamics (Figure 3C). Both viruses were associated with an increase in CD8⁺ intermediate and late TemRO cells, and the phenotypically-related total CD8⁺ TemRO and total CD8⁺ Tmem populations. In addition, CMV was associated with an increase in CD8⁺TCR $\gamma\delta^+$ T cells, and CD8⁺ late TemRA cells, and the phenotypically-related total CD8⁺ temRA population. EBV was associated with an increase in CD8⁺ early TemRO cells, and a decrease in total Bmem, and the CD27⁻ and CD27⁺ Bmem subsets. The effects on CD8⁺ late TemRO and TemRA cells were still present in the analysis of 31 non-overlapping population, even though these were now included in cluster 4 (Figure 3D).

In cluster 4, gender, premature rupture of membranes, CMV and EBV were most strongly associated with changes in lymphocyte dynamics (Figure 3C-D). Female gender was associated with an increase in the patterns of CD27⁺IgG⁺ memory B cells, and CD4⁺ early TemRO cells, and with a reduction in total TCR $\gamma\delta^+$ T cells and CD4⁺ late TemRA populations. Premature rupture of membranes was associated with an increase in CD4⁺ late TemRO and V δ^2 ⁺ and V γ 9⁺ T-cell subsets. CMV and EBV seropositivity were both associated with an increase in total CD4⁺ and CD4⁺ late TemRO and TemRA cells, as well as CD8⁺ intermediate TemRA cells. In addition, CMV was associated with increased CD4⁺ intermediate TemRO cells, and EBV with an increase in CD4⁺ early TemRO cells.

Altogether, we identified multiple determinants that had a significant effect on the dynamics of leukocyte subset numbers within one or more of the four distinct agerelated patterns. Still, considerable variation of effects could be observed within individual clusters.

EFFECTS OF EXTERNAL DETERMINANTS ON LEUKOCYTE DYNAMICS WITHIN DISTINCT LEUKOCYTE LINEAGES

As clusters 2, 3 and 4 each contained various B-cell and T-cell populations, each with distinct humoral or cellular functions, we next studied the effects of the 26 determinants on the patterns of individual subsets within the innate leukocytes, TCR $\gamma\delta^+$ T-cell, naive and memory B-cell, naive and memory CD4⁺ T-cell and naive and memory CD8⁺ T-cell and lineages (Figure 4).

Within the B-cell lineage, low maternal educational level was the only prenatal maternal life style-related factor that was associated with a change in B-cell dynamics, and was associated with a reduction in the patterns of total memory B cells. None of the prenatal maternal immune-mediated diseases were associated with leukocyte dynamics in any of the cell lineages. Of the birth characteristics,





Twenty-six determinants were studied for having an independent effect on the linear mixed effect models of each of the 62 leukocyte populations (A and C) and each of the 31 non-overlapping leukocyte subsets (B and D). Subsequently, all determinants with a significant individual effect were added to a multivariable linear mixed effect analysis to correct for confounding effects. **A and B** Relative distribution of significant effects by determinants within the groups: prenatal maternal life style, prenatal maternal immune-mediated diseases, birth characteristics or bacterial/ exposure-related determinants. Analyses were independently performed for all 62 defined subsets (**A**) and the subgroup of 31 non-overlapping subsets (**B**). **C and D**) Frequencies of populations within a cluster that were affected by individual determinants. In **C.** all 62 leukocyte populations and all 26 studied determinants are presented. In **D**., the independent analysis on the 31 non-overlapping populations is depicted containing only the determinants that affected one or more cell types. Abbreviations: PROMM, premature rupture of membranes; RTI, respiratory tract infection. Numbers in each bar refer to the number of each population as indicated in Figure 2.

female gender was associated with an increase in the patterns of CD27⁺IgA⁺ and CD27⁺IgG⁺ memory B cells, though these effects were not reflected in the total memory B-cell populations. Breastfeeding for more than 6 months was associated with a selective reduction in the pattern of CD27⁻IgA⁺ memory B cells, antibiotics usage in the first year of life was associated with a selective increase in the pattern of CD27⁻IgA⁺ memory B cells, antibiotics of CD27⁻IgG⁺ memory B cells, and H. pylori carriership was associated with an increase in the patterns of both CD27⁻IgA⁺ and CD27⁺IgA⁺ memory B cells. EBV infection was associated with a reduction in the pattern of total memory B cells, though not with a specific memory B-cell subset, HSV-1 infection was associated with a selective reduction in the pattern of naive B cells.

Within the CD4⁺ and CD8⁺ T-cell lineage, female gender was associated with an increase in the patterns of CD4⁺ naive, Tcm and early TemRO cells, although these effects were not reflected in the total CD4⁺ memory T-cell populations. In contrast, female gender was associated with a decrease in the patterns of CD4⁺ and CD8⁺ late TemRA cells. Whereas premature rupture of membranes was associated with an increase in CD4⁺ late TemRO cells, child birth via caesarian section was associated with a reduction in the pattern of CD4⁺ early TemRO cells. Breastfeeding, antibiotics usage and H. pylori carriership were not associated with CD4⁺ or CD8⁺ T-cell dynamics. Persistent viral infection with CMV and EBV strongly associated with CD4⁺ and CD8⁺ T-cell dynamics. CMV infection associated coL4⁺ and CD8⁺ TemRO and TemRA cells. Associations with EBV were slightly more variable, also including early differentiated CD4⁺ and CD8⁺ TemRO cells. In contrast to both CMV and EBV, HSV-1 and VZV did not associate with CD4⁺ and CD8⁺ T-cell dynamics.

Within the TCR $\gamma\delta^+$ T-cell lineage, CMV infection associated with an increase in the pattern of V $\delta1^+$ T cells; premature rupture of membranes was associated with an increase in V $\delta2^+$ V $\gamma9^+$ T cells.

DISCUSSION

We here modelled the kinetics of 62 leukocyte subsets, including 31 nonoverlapping subsets, and identified distinct patterns of cell numbers in the first 6 years of life for innate leukocytes, naive B and T cells, and memory B and T cells. Furthermore, we identified multiple determinants related to prenatal maternal life style, prenatal maternal immune-mediated diseases, birth characteristics or bacterial/viral exposure, that affected the dynamics of each of these leukocyte subsets between birth and 6 years of age.

Unsupervised clustering of all 62 leukocyte subsets, as well as of the set of 31 phenotypically non-overlapping populations, resulted in 4 clusters with the first cluster containing exclusively innate leukocyte subsets. These subsets followed highly similar patterns of kinetics with age: they were abundant at birth and quickly decreased before stabilizing at 6-14 months of age. Consistent with their early predominance, innate leukocyte kinetics was affected by maternal education level, which might already influence the fetus prenatally. Maternal education level is strongly related to life style with low education being an indicator for a less healthy life style, including differences in smoking and alcohol consumption, BMI, stress-levels, higher infection load and increased crowding.³⁴ These observations might suggests that maternal life style is especially important for shaping of innate



population definitions, and the total memory B-cell, CD4* memory T-cell or CD8* memory T-cell subsets (indicated as "mem" for each lineage). Significant associations with a subset are indicated with blue boxes representing a reduction, or red boxes representing an increase in the pattern of the indicated population. Details of each determinant are provided in Table 2. Additional abbreviations: PROMM, premature rupture of membranes; RTI, respiratory tract infection; neu, neutrophils; eos, eosinophils; class, classical monocytes; inter, intermediate monocytes; n-class, non-classical monocytes. leukocyte populations, though the exact mediator of the effect will still have to be determined.

The non-classical and intermediate monocytes did not consistently cluster with the innate leukocyte subsets due to a later peak in numbers or a slower decline. This altered kinetics could be the result of these being derived from classical monocytes, and not directly from precursors in bone marrow.³⁵ Furthermore, the various subsets of monocytes show distinct expression levels of proteins involved in HLA-class II-dependent antigen presentation and CD40-CD40L co-stimulation,³⁵ and display distinct parasite pattern recognition abilities.³⁵⁻³⁸ Thus, our data support the need to discriminate intermediate and non-classical monocytes from the dominant population of classical monocytes in kinetics studies.

All naive B-cell and T-cell populations clustered together in cluster 2 and their numbers increased in the first 14 months followed by a more gradual decline. After the first 1-2 years of life, B-lymphocyte production in bone marrow decreases,^{39,40} and the thymus starts to involute.⁴¹ These processes are likely causes of the decline we observed. The nature for decreased bone marrow output remains unclear. Recent studies, however, indicate a role for circulating long-lived memory B cells in this process.⁴²

Memory B-cell subsets were found in several clusters. Natural effector and CD27⁻ IgA⁺ memory B cells clustered together with naive B cells. Both memory B-cell populations are suggested to derive, at least in part, from germinal center-independent responses without T-cell help in the splenic marginal zone and the intestinal lamina propria.⁴³⁻⁴⁵ The fast rise in cell numbers within the first 6 months of life contrasts the kinetics of the other, T-cell dependent, memory B-cell subsets in clusters 3 and 4. This supports the concept of these cells being generated in first and fast responses, in the absence of the extensive proliferation and selection processes of the germinal center.⁴⁵

IgMonly, CD27⁻IgG⁺ and CD27⁺IgA⁺ memory B cells clustered in cluster 3, representing gradual memory formation. Only the CD27⁺IgG⁺ population clustered in cluster 4, and developed slightly slower than the other memory B-cell populations. In adults, this subset shows a higher level of affinity maturation, more extensive replication history, and more frequent development via consecutive class-switching, than CD27⁻IgG⁺ memory B cells.^{45,46} It is perceivable that many of these CD27⁺IgG⁺ cells are generated from CD27⁻IgG⁺ memory B cells in secondary responses. This could explain the gradual decline in CD27⁻IgG⁺ B cell numbers after 14 months of age.

Several external determinants had a significant effect on the B-cell kinetics. First, we confirmed our previous observations regarding lower numbers of memory B cells in breastfed children.^{21,23} However, in the current analysis we could not reproduce the previously reported association between breastfeeding duration and germinal center-dependent memory B-cell numbers at 6 months of age.⁴⁵ The difference is likely due to our current analysis in which duration of breastfeeding was not included and the overall pattern between birth and 6 years of age was studied.

IgA⁺ memory B cells were significantly increased in H. Pylori positive children. The colonization of the gastric mucus by H. Pylori has been found to correlate with an increase in total blood B-cell counts,⁴⁷ on top of the local expansion of antibody-secreting IgA⁺ cells,⁴⁸ which are important for the protection against H. Pylori

infection.⁴⁹ Our observed expansion of circulating IgA⁺ memory B cells included both the CD27⁺IgA⁺ and mucosa-derived CD27⁻IgA⁺ subsets. This would suggest that the presence of H. pylori does not only result in a local expansion of plasma cells, but also a systemic expansion of memory B cells in otherwise asymptomatic carriers. It remains to be determined whether this expansion is beneficial for the host as the bacterial protein CagA inhibits B-cell apoptosis and thereby increases the risk for mucosa-associated B-cell malignancies.^{50,51}

Memory T-cell subsets were found in clusters 2, 3 and 4. CD4⁺ and CD8⁺ Tcm (cluster 2) express lymph node homing markers and are the presumed precursors of effector memory T cells.^{25,52,53} The early peak in Tcm numbers prior to those of effector memory T cells in young children would fit with this function. Similarly, CD8⁺ early TemRA cells might be precursors for further differentiated TemRA subsets. V δ 1⁺ T-cell numbers (cluster 2) peaked prior to the V δ 2⁺ and V γ 9⁺ subsets (cluster 4), thereby confirming the previously observed early shift from V δ 1⁺ to V δ 2⁺V γ 9⁺ predominance in children,^{54,55} Of note, V δ 1⁺ T cells are abundant in adult intestine; their decline in blood might, at least in part, result from altered distribution towards mucosal tissue.

Both vaginal delivery and premature rupture of membranes were associated with an increase in the pattern of one of the CD4⁺ TemRO subsets, in line with previous literature.⁵⁶ Stress upon child birth via vaginal delivery has been correlated with an increase in CD4⁺ regulatory T cells.⁵⁷ As a large fraction of regulatory T cells has a memory phenotype, this could explain the observed increase of CD4⁺ TemRO cells in children born by vaginal delivery. Since reduced numbers of regulatory T cells in cord blood are associated with the development of atopic dermatitis in the first year of life,⁵⁸ it will be of interest to study whether this effect is modified by the mode of delivery.

Both CD4⁺ and CD8⁺ memory T-cell dynamics were predominantly affected by CMV and/ or EBV seropositivity. This is consistent with previous observations in both children and elderly,^{26,59-63} and in agreement with their function in cellular immunity. Moreover, these viruses displayed distinct effects with CMV being associated to increased late effector memory T-cell numbers and EBV with a slightly more variable effect, including also early effector memory T-cell subsets.^{25,64} Furthermore, we confirmed that HSV-1 did not affect CD4⁺ and CD8⁺ T-cell populations.^{26,63} In contrast, HSV-1 seropositivity was associated with a loss of naive B cells. This association has, to our knowledge, not been described before. EBV infection was associated with a reduction in memory B cells, likely due to the selective EBV persistence in these cells.^{65,66} Our large-scale analysis allowed us to separate the herpesvirus-associated effects, with CMV and EBV being associated with memory T-cell expansions, and HSV-1 and EBV, with a decrease in naive or memory B-cell numbers, respectively.

Gender had a widespread effect on 13 subsets within multiple leukocyte lineages. Interestingly, female gender was associated with an increase in neutrophils, T-cell dependent IgA⁺ and IgG⁺ memory B-cell populations and CD4⁺ naive, Tcm and early differentiated TemRO cells, while numbers of late differentiated CD4⁺ and CD8⁺ TemRA cells and TCR $\gamma\delta^+$ T cells were reduced. These combined effects result in a skewing of humoral and early differentiated CD4⁺ T-cell responses over cellular cytotoxic responses in girls. These effects might be associated with differences in sex-hormone levels (testosterone, estradiol) in females versus males that are already detectable during early infancy, as well as with genetic

differences between females and males.⁶⁷ In adults, sex-hormones have been found to affect immune responses in both quantitative and qualitative aspects.^{68,69} These insight can especially be important for dissection of auto-immune diseases, which are much more prevalent in females than in males, though predisposition for allergic diseases seems to be opposite in infancy.^{67,70}

METHODOLOGICAL CONSIDERATIONS

The strength of this study is its prospective longitudinal population-based design with >1,000 children and the possibility to study 26 external determinants with adjustments for major confounders. The linear mixed model approach enabled modeling of cross-sectional data, with further improvement of the accuracy by incorporating additionally available longitudinal follow-up data. Furthermore, we included only children with a two-generation Dutch ancestry, which prevented interference of our analyses by strong ethnic and cultural influences. However, extrapolation of our findings to different ethnic and cultural populations might be limited, and would require additional analysis of ethnically-different population cohorts.

The inclusion of 62 leukocyte populations allowed for the large-scale analysis of the effect of external determinants on both total cell lineages as well as on small subsets defined by extensive and detailed phenotypic definition. The overlap in some populations could have skewed the hierarchical clustering. However, our selection of 31 phenotypically non-overlapping populations resulted in clusters with similar patterns, indicating the robustness of the 4 major patterns of leukocyte dynamics and the observed effects of external determinants.

Our study was primarily explorative with a focus on the identification of determinants related to prenatal maternal life style, maternal immune-mediated diseases, birth characteristics or bacterial/viral exposure, that affected leukocyte dynamics between birth and 6 years of age. Although we defined whether the determinants had a positive or negative association with leukocyte numbers, we were unable to identify the exact nature of the effect, i.e. exactly when these effects presented and whether these effects will be transient, persistent or potentially even increasing over time. Consequently, more research into individual determinants will be needed to extend our observations by specifying these effects.

Finally, special consideration needs to be taken for serology of infectious agents that were measured at the age of 6 years. We cannot determine the exact timing of primary infections, and this can consequently be in the whole period preceding the age of 6 years. Still, most H. Pylori infections already occur in early childhood, and IgG seropositivity to herpesviruses only appears several weeks to 3 months after infection. Thus, it is likely that these determinants were present during or prior to the 6th year of life.

CONCLUSIONS

With our unbiased approach, we were, for the first time, able to classify leukocyte populations according to their dynamics between birth and 6 years of age, and to assess determinants that are associated with the dynamics of cell lineages or specific leukocyte subsets. We confirmed previously described effects of persistent CMV and EBV infections on memory T cells and identified new effects, including

the effects of HSV-1 on naive B cells. These newly identified determinants can provide new targets for studies on the molecular processes that regulate leukocyte development and immune responses, and that together underlie formation of longlasting immunity without inducing destructive, excessive or insufficient immune responses.

ACKNOWLEDGEMENTS

This work was supported by an Erasmus MC Fellowship to M.C.v.Z, and was performed within the framework of the Erasmus MC Postgraduate School Molecular Medicine. The Generation R Study is conducted by the Erasmus MC, Erasmus University Rotterdam in close collaboration with the School of Law and Faculty of Social Sciences of the Erasmus University Rotterdam, the Municipal Health Service Rotterdam Metropolitan Area, the Rotterdam Homecare Foundation, and the Stichting Trombosedienst & Artsenlaboratorium Rijnmond. We gratefully acknowledge the contributions of children and parents, general practitioners, hospitals, midwives and pharmacies in Rotterdam. We thank D. Zhao, K.A.M. van Kester, M.A.W. Smits-te Nijenhuis, M.J Kolijn-Couwenberg and N.M.A. Nagtzaam for technical support.

CONFLICT OF INTEREST

All authors declare that no competing interests exist.

J.J.M.v.D., E.G.v.L., L.E.B-J, H.H., W.A.D., and M.C.v.Z. designed and supervised the flowcytometric experiments; D.v.d.H., M.A.E.J. and H.B.-C. performed and analyzed most of the experiments and contributed to data analyses; V.W.V.J. designed the Generation R Study; H.A.M. designed the study, the collection and analyses of determinant data; K.N. contributed to the statistical modeling; D.v.d.H., M.C.v.Z., and M.A.E.J. wrote the manuscript; and all authors commented on the manuscript.

REFERENCES

- 1. Comans-Bitter WM, de Groot R, van den Beemd R, Neijens HJ, Hop WC, Groeneveld K, et al. Immunophenotyping of blood lymphocytes in childhood. Reference values for lymphocyte subpopulations. *J Pediatr.* 1997; 130(3): 388-393.
- van Gent R, van Tilburg CM, Nibbelke EE, Otto SA, Gaiser JF, Janssens-Korpela PL, et al. Refined characterization and reference values of the pediatric T- and B-cell compartments. *Clin Immunol*. 2009; 133(1): 95-107.
- 3. Tosato F, Bucciol G, Pantano G, Putti MC, Sanzari MC, Basso G, et al. Lymphocytes subsets reference values in childhood. *Cytometry A*. 2015; 87(1): 81-85.
- Duchamp M, Sterlin D, Diabate A, Uring-Lambert B, Guerin-El Khourouj V, Le Mauff B, et al. B-cell subpopulations in children: National reference values. *Immun Inflamm Dis.* 2014; 2(3): 131-140.
- 5. Xanthou M. Leucocyte blood picture in healthy full-term and premature babies during neonatal period. *Arch Dis Child*. 1970; 45(240): 242-249.
- Manroe BL, Weinberg AG, Rosenfeld CR, Browne R. The neonatal blood count in health and disease. I. Reference values for neutrophilic cells. *J Pediatr*. 1979; 95(1): 89-98.
- Sagnia B, Ateba Ndongo F, Ndiang Moyo Tetang S, Ndongo Torimiro J, Cairo C, Domkam I, et al. Reference values of lymphocyte subsets in healthy, HIV-negative children in Cameroon. *Clin Vaccine Immunol.* 2011; 18(5): 790-795.
- 8. Huck K, Feyen O, Ghosh S, Beltz K, Bellert S, Niehues T. Memory B-cells in healthy and antibody-

deficient children. Clin Immunol. 2009; 131(1): 50-59.

- 9. Morbach H, Eichhorn EM, Liese JG, Girschick HJ. Reference values for B cell subpopulations from infancy to adulthood. *Clin Exp Immunol*. 2010; 162(2): 271-279.
- Piatosa B, Wolska-Kusnierz B, Pac M, Siewiera K, Galkowska E, Bernatowska E. B cell subsets in healthy children: reference values for evaluation of B cell maturation process in peripheral blood. *Cytometry B Clin Cytom.* 2010; 78(6): 372-381.
- 11. Driessen GJ, Dalm VA, van Hagen PM, Grashoff HA, Hartwig NG, van Rossum AM, et al. Common variable immunodeficiency and idiopathic primary hypogammaglobulinemia: two different conditions within the same disease spectrum. *Haematologica*. 2013; 98(10): 1617-1623.
- de Vries E, de Bruin-Versteeg S, Comans-Bitter WM, de Groot R, Hop WC, Boerma GJ, et al. Longitudinal survey of lymphocyte subpopulations in the first year of life. *Pediatr Res.* 2000; 47(4 Pt 1): 528-537.
- 13. Tsao PN, Chiang BL, Yang YH, Tsai MJ, Lu FL, Chou HC, et al. Longitudinal follow-up of lymphocyte subsets during the first year of life. *Asian Pac J Allergy Immunol*. 2002; 20(3): 147-153.
- 14. Schatorje EJ, Gemen EF, Driessen GJ, Leuvenink J, van Hout RW, van der Burg M, et al. Agematched reference values for B-lymphocyte subpopulations and CVID classifications in children. *Scand J Immunol.* 2011; 74(5): 502-510.
- 15. Schmutz N, Henry E, Jopling J, Christensen RD. Expected ranges for blood neutrophil concentrations of neonates: the Manroe and Mouzinho charts revisited. *J Perinatol*. 2008; 28(4): 275-281.
- Pabst HF, Spady DW, Pilarski LM, Carson MM, Beeler JA, Krezolek MP. Differential modulation of the immune response by breast- or formula-feeding of infants. *Acta Paediatr*. 1997; 86(12): 1291-1297.
- 17. Winkler B, Aulenbach J, Meyer T, Wiegering A, Eyrich M, Schlegel PG, et al. Formula-feeding is associated with shift towards Th1 cytokines. *Eur J Nutr*. 2015; 54(1): 129-138.
- Hanson L, Silfverdal SA, Stromback L, Erling V, Zaman S, Olcen P, et al. The immunological role of breast feeding. *Pediatr Allergy Immunol.* 2001; 12 Suppl 14(15-19.
- 19. Hanson LA, Korotkova M, Lundin S, Haversen L, Silfverdal SA, Mattsby-Baltzer I, et al. The transfer of immunity from mother to child. *Ann N Y Acad Sci*. 2003; 987(199-206.
- Duijts L, Ramadhani MK, Moll HA. Breastfeeding protects against infectious diseases during infancy in industrialized countries. A systematic review. *Matern Child Nutr.* 2009; 5(3): 199-210.
- 21. Hawkes JS, Neumann MA, Gibson RA. The effect of breast feeding on lymphocyte subpopulations in healthy term infants at 6 months of age. *Pediatr Res.* 1999; 45(5 Pt 1): 648-651.
- M'Rabet L, Vos AP, Boehm G, Garssen J. Breast-feeding and its role in early development of the immune system in infants: consequences for health later in life. J Nutr. 2008; 138(9): 1782S-1790S.
- Jansen MA, van den Heuvel D, van Zelm MC, Jaddoe VW, Hofman A, de Jongste JC, et al. Decreased memory B cells and increased CD8 memory T cells in blood of breastfed children: the generation R study. *PLoS One.* 2015; 10(5): e0126019.
- Bekker V, Scherpbier H, Pajkrt D, Jurriaans S, Zaaijer H, Kuijpers TW. Persistent humoral immune defect in highly active antiretroviral therapy-treated children with HIV-1 infection: loss of specific antibodies against attenuated vaccine strains and natural viral infection. *Pediatrics*. 2006; 118(2): e315-322.
- 25. Appay V, van Lier RA, Sallusto F, Roederer M. Phenotype and function of human T lymphocyte subsets: consensus and issues. *Cytometry A*. 2008; 73(11): 975-983.
- 26. Heuvel Dvd, Jansen MAE, Dik WA, Bouallouch-Charif H, Zhao D, Kester KAMv, et al. CMV- and EBVinduced T-cell expansions in young children do not impair naive T-cell populations or vaccination responses. The Generation R Study *Journal of Infectious Diseases* 2015.
- Jaddoe VW, van Duijn CM, Franco OH, van der Heijden AJ, van Iizendoorn MH, de Jongste JC, et al. The Generation R Study: design and cohort update 2012. *Eur J Epidemiol*. 2012; 27(9): 739-756.
- 28. Kruithof CJ, Kooijman MN, van Duijn CM, Franco OH, de Jongste JC, Klaver CC, et al. The Generation R Study: Biobank update 2015. *Eur J Epidemiol*. 2014; 29(12): 911-927.
- 29. Kalina T, Flores-Montero J, van der Velden VH, Martin-Ayuso M, Bottcher S, Ritgen M, et al. EuroFlow standardization of flow cytometer instrument settings and immunophenotyping protocols. *Leukemia*. 2012; 26(9): 1986-2010.
- 30. Medici M, de Rijke YB, Peeters RP, Visser W, de Muinck Keizer-Schrama SM, Jaddoe VV, et al. Maternal early pregnancy and newborn thyroid hormone parameters: the Generation R study. *J Clin Endocrinol Metab*. 2012; 97(2): 646-652.

- 31. Jansen MA, Tromp, II, Kiefte-de Jong JC, Jaddoe VW, Hofman A, Escher JC, et al. Infant feeding and anti-tissue transglutaminase antibody concentrations in the Generation R Study. *Am J Clin Nutr*. 2014; 100(4): 1095-1101.
- den Hollander WJ, Holster IL, van Gilst B, van Vuuren AJ, Jaddoe VW, Hofman A, et al. Intergenerational reduction in Helicobacter pylori prevalence is similar between different ethnic groups living in a Western city. *Gut.* 2014.
- 33. R Core Team; R Foundation for Statistical Computing V, Austria. R: A language and environment for statistical computing. . 2015.
- 34. Bouthoorn SH, Silva LM, Murray SE, Steegers EA, Jaddoe VW, Moll H, et al. Low-educated women have an increased risk of gestational diabetes mellitus: the Generation R Study. *Acta Diabetol*. 2015; 52(3): 445-452.
- Wong KL, Tai JJ, Wong WC, Han H, Sem X, Yeap WH, et al. Gene expression profiling reveals the defining features of the classical, intermediate, and nonclassical human monocyte subsets. *Blood*. 2011; 118(5): e16-31.
- 36. Turner JD, Bourke CD, Meurs L, Mbow M, Dieye TN, Mboup S, et al. Circulating CD14brightCD16+ 'intermediate' monocytes exhibit enhanced parasite pattern recognition in human helminth infection. *PLoS Negl Trop Dis.* 2014; 8(4): e2817.
- 37. Hijdra D, Vorselaars AD, Grutters JC, Claessen AM, Rijkers GT. Phenotypic characterization of human intermediate monocytes. *Front Immunol*. 2013; 4(339.
- 38. Ziegler-Heitbrock L, Hofer TP. Toward a refined definition of monocyte subsets. *Front Immunol*. 2013; 4(23.
- Kogut I, Scholz JL, Cancro MP, Cambier JC. B cell maintenance and function in aging. Semin Immunol. 2012; 24(5): 342-349.
- 40. Jensen K, Schaffer L, Olstad OK, Bechensteen AG, Hellebostad M, Tjonnfjord GE, et al. Striking decrease in the total precursor B-cell compartment during early childhood as evidenced by flow cytometry and gene expression changes. *Pediatr Hematol Oncol*. 2010; 27(1): 31-45.
- 41. Flores KG, Li J, Sempowski GD, Haynes BF, Hale LP. Analysis of the human thymic perivascular space during aging. *J Clin Invest*. 1999; 104(8): 1031-1039.
- 42. Keren Z, Naor S, Nussbaum S, Golan K, Itkin T, Sasaki Y, et al. B-cell depletion reactivates B lymphopoiesis in the BM and rejuvenates the B lineage in aging. *Blood*. 2011; 117(11): 3104-3112.
- 43. Weller S, Braun MC, Tan BK, Rosenwald A, Cordier C, Conley ME, et al. Human blood IgM "memory" B cells are circulating splenic marginal zone B cells harboring a prediversified immunoglobulin repertoire. *Blood*. 2004; 104(12): 3647-3654.
- 44. Weller S, Faili A, Garcia C, Braun MC, Le Deist FF, de Saint Basile GG, et al. CD40-CD40L independent Ig gene hypermutation suggests a second B cell diversification pathway in humans. *Proc Natl Acad Sci U S A*. 2001; 98(3): 1166-1170.
- 45. Berkowska MA, Driessen GJ, Bikos V, Grosserichter-Wagener C, Stamatopoulos K, Cerutti A, et al. Human memory B cells originate from three distinct germinal center-dependent and -independent maturation pathways. *Blood*. 2011; 118(8): 2150-2158.
- 46. Jackson KJ, Wang Y, Collins AM. Human immunoglobulin classes and subclasses show variability in VDJ gene mutation levels. *Immunol Cell Biol*. 2014; 92(8): 729-733.
- 47. Helmin-Basa A, Michalkiewicz J, Gackowska L, Kubiszewska I, Eljaszewicz A, Mierzwa G, et al. Pediatric Helicobacter pylori infection and circulating T-lymphocyte activation and differentiation. *Helicobacter*. 2011; 16(1): 27-35.
- 48. Mattsson A, Quiding-Jarbrink M, Lonroth H, Hamlet A, Ahlstedt I, Svennerholm A. Antibodysecreting cells in the stomachs of symptomatic and asymptomatic Helicobacter pylori-infected subjects. *Infect Immun*. 1998; 66(6): 2705-2712.
- 49. Lee CK, Weltzin R, Thomas WD, Jr., Kleanthous H, Ermak TH, Soman G, et al. Oral immunization with recombinant Helicobacter pylori urease induces secretory IgA antibodies and protects mice from challenge with Helicobacter felis. *J Infect Dis.* 1995; 172(1): 161-172.
- 50. Umehara S, Higashi H, Ohnishi N, Asaka M, Hatakeyama M. Effects of Helicobacter pylori CagA protein on the growth and survival of B lymphocytes, the origin of MALT lymphoma. *Oncogene*. 2003; 22(51): 8337-8342.
- 51. Lin WC, Tsai HF, Kuo SH, Wu MS, Lin CW, Hsu PI, et al. Translocation of Helicobacter pylori CagA into Human B lymphocytes, the origin of mucosa-associated lymphoid tissue lymphoma. *Cancer Res.* 2010; 70(14): 5740-5748.

58

Chapter 2

- 52. Sallusto F, Geginat J, Lanzavecchia A. Central memory and effector memory T cell subsets: function, generation, and maintenance. *Annu Rev Immunol*. 2004; 22(745-763.
- 53. Sallusto F, Lenig D, Forster R, Lipp M, Lanzavecchia A. Two subsets of memory T lymphocytes with distinct homing potentials and effector functions. *Nature*. 1999; 401(6754): 708-712.
- 54. Sandberg Y, Almeida J, Gonzalez M, Lima M, Barcena P, Szczepanski T, et al. TCRgammadelta+ large granular lymphocyte leukemias reflect the spectrum of normal antigen-selected TCRgammadelta+ T-cells. *Leukemia*. 2006; 20(3): 505-513.
- Breit TM, Wolvers-Tettero IL, van Dongen JJ. Unique selection determinant in polyclonal V delta 2-J delta 1 junctional regions of human peripheral gamma delta T lymphocytes. *J Immunol*. 1994; 152(6): 2860-2864.
- 56. Duijts L, Bakker-Jonges LE, Labout JA, Jaddoe VW, Hofman A, Steegers EA, et al. Perinatal stress influences lymphocyte subset counts in neonates. The generation R study. *Pediatr Res.* 2008; 63(3): 292-298.
- 57. Yildiran A, Yurdakul E, Guloglu D, Dogu F, Arsan S, Arikan M, et al. The effect of mode of delivery on T regulatory (Treg) cells of cord blood. *Indian J Pediatr*. 2011; 78(10): 1234-1238.
- Hinz D, Bauer M, Roder S, Olek S, Huehn J, Sack U, et al. Cord blood Tregs with stable FOXP3 expression are influenced by prenatal environment and associated with atopic dermatitis at the age of one year. *Allergy*. 2012; 67(3): 380-389.
- 59. Bekker V, Bronke C, Scherpbier HJ, Weel JF, Jurriaans S, Wertheim-van Dillen PM, et al. Cytomegalovirus rather than HIV triggers the outgrowth of effector CD8+CD45RA+CD27- T cells in HIV-1-infected children. *Aids*. 2005; 19(10): 1025-1034.
- 60. Weinberger B, Lazuardi L, Weiskirchner I, Keller M, Neuner C, Fischer KH, et al. Healthy aging and latent infection with CMV lead to distinct changes in CD8+ and CD4+ T-cell subsets in the elderly. *Hum Immunol.* 2007; 68(2): 86-90.
- 61. van Leeuwen EM, Remmerswaal EB, Vossen MT, Rowshani AT, Wertheim-van Dillen PM, van Lier RA, et al. Emergence of a CD4+CD28- granzyme B+, cytomegalovirus-specific T cell subset after recovery of primary cytomegalovirus infection. *J Immunol*. 2004; 173(3): 1834-1841.
- 62. Chidrawar S, Khan N, Wei W, McLarnon A, Smith N, Nayak L, et al. Cytomegalovirus-seropositivity has a profound influence on the magnitude of major lymphoid subsets within healthy individuals. *Clin Exp Immunol.* 2009; 155(3): 423-432.
- 63. Derhovanessian E, Maier AB, Hahnel K, Beck R, de Craen AJ, Slagboom EP, et al. Infection with cytomegalovirus but not herpes simplex virus induces the accumulation of late-differentiated CD4+ and CD8+ T-cells in humans. *J Gen Virol*. 2011; 92(Pt 12): 2746-2756.
- Appay V, Dunbar PR, Callan M, Klenerman P, Gillespie GM, Papagno L, et al. Memory CD8+ T cells vary in differentiation phenotype in different persistent virus infections. *Nat Med*. 2002; 8(4): 379-385.
- 65. Babcock GJ, Decker LL, Volk M, Thorley-Lawson DA. EBV persistence in memory B cells in vivo. *Immunity*. 1998; 9(3): 395-404.
- 66. Chaganti S, Heath EM, Bergler W, Kuo M, Buettner M, Niedobitek G, et al. Epstein-Barr virus colonization of tonsillar and peripheral blood B-cell subsets in primary infection and persistence. *Blood*. 2009; 113(25): 6372-6381.
- 67. Ober C, Loisel DA, Gilad Y. Sex-specific genetic architecture of human disease. *Nat Rev Genet*. 2008; 9(12): 911-922.
- 68. Melzer S, Zachariae S, Bocsi J, Engel C, Loffler M, Tarnok A. Reference intervals for leukocyte subsets in adults: Results from a population-based study using 10-color flow cytometry. *Cytometry B Clin Cytom*. 2015.
- 69. Oertelt-Prigione S. The influence of sex and gender on the immune response. *Autoimmun Rev.* 2012; 11(6-7): A479-485.
- 70. Whitacre CC, Reingold SC, O'Looney PA. A gender gap in autoimmunity. *Science*. 1999; 283(5406): 1277-1278.

Supp	lemental Table 1. A	ntibody details					
Tube				Fluorochrome	-		
		FITC	PE	PerCP-Cy5.5	PE-Cy7	APC	APC-Cy7
	Antibody	CD3	CD16+CD56	CD45	CD4	CD19	CD8
H	Clone	SK7	B73.1 + C5.9	2D1	SK3	SJ25C1	SK1
	Manufacturer	BD Biosciences	BD Biosciences+Dako	BD Biosciences	BD Biosciences	BD Biosciences	BD Biosciences
	Antibody	CD15	1	CD45	CD16		CD14
2	Clone	ММА	1	2D1	3G8		MO-P9
	Manufacturer	BD Biosciences	I	BD Biosciences	BD Biosciences		BD Biosciences
	Antibody	Igk	Igλ	CD19	CD21		
m	Clone	polyclonal	polyclonal	SJ25C1	B-ly-4	ı	,
	Manufacturer	Dako	SBA	BD Biosciences	BD Biosciences	1	,
	Antibody	CD38	I	CD19	CD21		8
4	Clone	HB7	I	SJ25C1	B-ly-4	ı	,
	Manufacturer	BD Biosciences	1	BD Biosciences	BD Biosciences	1	,
	Antibody	IgD	I	CD19		MgI	CD27
ы	Clone	polyclonal	I	SJ25C1	I	polyclonal	L128
	Manufacturer	Southern Biotech	1	BD Biosciences	1	BD Biosciences	BD Biosciences
	Antibody	IgA	IgG	CD19	I	MgI	CD27
9	Clone	polyclonal	polyclonal	SJ25C1	ı	polyclonal	L128
	Manufacturer	Kallestad	Southern Biotech	BD Biosciences	1	BD Biosciences	BD Biosciences
	Antibody	TCRaβ	ΤΟΒΥδ	CD3	CD4	CD8	1
~	Clone	WT31	11F2	SK7	SK3	SK1	,
	Manufacturer	BD Biosciences	BD Biosciences	BD Biosciences	BD Biosciences	BD Biosciences	
	Antibody	V 0 2	Vð1 *	CD3	CD4	TCRaß	CD8
8	Clone	B6.1	R9.12	SK7	SK3	IP26	SK1
	Manufacturer	BD Biosciences	Beckman Coulter	BD Biosciences	BD Biosciences	eBiosciences	BD Biosciences
	Antibody	CD28	CD197	CD3	CD8	CD45RO	CD27
6	Clone	CD28.2	3D13	SK7	SK1	UCHL-1	L128
	Manufacturer	BD Biosciences	eBiosciences	BD Biosciences	BD Biosciences	BD Biosciences	BD Biosciences
	Antibody	Vð2	νγ9	CD3	CD4	τςκαβ	CD8
10	Clone	B6.1	B3.1	SK7	SK3	IP26	SK1
	Manufacturer	BD Biosciences	BD Biosciences	BD Biosciences	BD Biosciences	eBiosciences	BD Biosciences
*, un	conjugated antibody,	detected with Goat an	ti-Mouse IgG PE (polyclon	al; Invitrogen)			



Chapter 2



Supplemental Figure 1. Dynamics of innate leukocyte subsets in children between birth and 6 years of age.

Absolute numbers and modelled dynamics of granulocytes, monocytes and NK cells (A), CD15⁺, neutrophilic and eosinophilic granulocytes (B), and classical, intermediate and non-classical monocyte subsets (C), from birth until 6 years of age. Linear mixed effect models were generated for each population (solid black line) and represented with the 90% confidence interval (CI) of the model (dashed black lines). For clarity of the graphs, gray lines connect only consecutive time points within one individual; i.e. 0-6m, 6-14m, 14-25m or 25-72m.



Supplemental Figure 2. Dynamics of TCRa β^+ T-cell and TCRy δ^+ T-cell subsets in children between birth and 6 years of age.

Absolute numbers and modelled dynamics of T cells (**A**), TCRa β^+ T-cell populations (**B**), and TCR $\gamma\delta^+$ T-cell populations (**C**), from birth until 6 years of age. Linear mixed effect models were generated for each population (solid black line) and represented with the 90% confidence interval (CI) of the model (dashed black lines). For clarity of the graphs, gray lines connect only consecutive time points within one individual; i.e. 0-6m, 6-14m, 14-25m or 25-72m.



Supplemental Figure 3. Dynamics of CD4⁺ T-cell subsets in children between birth and 6 years of age. Absolute numbers and modelled dynamics of total, naive and memory CD4⁺ T cells (**A**); CD4⁺ central memory T cells (Tcm), CD45RO⁺ effector memory (TemRO) and CD45RO⁻ effector memory (TemRA) T-cell subsets (**B**); early, intermediate or late differentiated TemRO subsets (**C**); or early, intermediate or late differentiated TemRA subsets (**D**), between birth and 6 years of age. Linear mixed effect models were generated for each population (solid black line) and represented with the 90% confidence interval (CI) of the model (dashed black lines). For clarity of the graphs, gray lines connect only consecutive time points within one individual; i.e. 0-6m, 6-14m, 14-25m or 25-72m.



Supplemental Figure 4. Dynamics of CD8⁺ T-cell subsets in children between birth and 6 years of age. Absolute numbers and modelled dynamics of total, naive and memory CD8⁺ T cells (A); CD8⁺ central memory T cells (Tcm), CD45R0⁺ effector memory (TemRO) and CD45R0⁻ effector memory (TemRA) T cell subsets (B); early, intermediate or late differentiated TemRO subsets (C); or early, intermediate or late differentiated TemRA subsets (D), between birth and 6 years of age. Linear mixed effect models were generated for each population (solid black line) and represented with the 90% confidence interval (CI) of the model (dashed black lines). For clarity of the graphs, gray lines connect only consecutive time points within one individual; i.e. 0-6m, 6-14m, 14-25m or 25-72m.



Supplemental Figure 5. Dynamics of B-cell subsets in children between birth and 6 years of age. Absolute numbers and modelled dynamics of total B cells (A); naive and total memory B cells (B); total Igk⁺ and IgA⁺ B-cells populations (C); and CD21^{tw} B cells (D); CD27⁻ and CD27⁺ memory B cells (E); total IgM⁺, IgA⁺ or IgG⁺ memory B cells (F); IgM⁺, IgA⁺ or IgG⁺ memory B-cell subsets (G) between birth and 6 years of age. Linear mixed effect models were generated for each population (solid black line) and represented with the 90% confidence interval (CI) of the model (dashed black lines). For clarity of the graphs, gray lines connect only consecutive time points within one individual; i.e. 0-6m, 6-14m, 14-25m or 25-72m.



 \sim

Supplemental Figure 6. Hierarchical clustering of the dynamics of 31 non-overlapping leukocyte subsets in early childhood.

A) Ward's hierarchical clustering was performed as in Figure 2, including now only the 31 non-overlapping leukocyte subsets. Indicated in gray squares in front of the heat map is the cluster each population was assigned to in the analysis of the total 62 leukocyte subsets in Figure 2. **B)** Average patterns +/- 1 standard deviation of the subsets in each of the 4 major clusters. Indicated in black squares are 3 subsets that were assigned to different clusters than in Figure 2. (Figure on next page)

CHAPTER 3

DECREASED MEMORY B CELLS AND INCREASED CD8 MEMORY T CELLS IN BLOOD OF BREASTFED CHILDREN: THE GENERATION R STUDY

Running title: Impact of breastfeeding on adaptive immunity

M.A.E. Jansen,^{1,2,3} D. van den Heuvel,³ M.C. van Zelm,³ V.W.V. Jaddoe,^{1,2,4} A. Hofman,⁴ J.C. de Jongste,² H. Hooijkaas,³ H.A. Moll.²

¹ The Generation R Study Group, Erasmus MC, University Medical Center, Rotterdam, NL ² Dept. of Pediatrics, Erasmus MC-Sophia, Rotterdam, NL ³ Dept. of Immunology, Erasmus MC, University Medical Center, Rotterdam, NL ⁴ Dept. of Epidemiology, Erasmus MC, University Medical Center, Rotterdam, NL

PLoS One. 2015; 10(5): e0126019



ABSTRACT

BACKGROUND. Breastfeeding provides a protective effect against infectious diseases in infancy. Still, immunological evidence for enhanced adaptive immunity in breastfed children remains inconclusive.

OBJECTIVE. To determine whether breastfeeding affects B- and T- cell memory in the first years of life.

METHODS. We performed immunophenotypic analysis on blood samples within a population-based prospective cohort study. Participants included children at 6 months (n=258), 14 months (n=166), 25 months (n=112) and 6 years of age (n=332) with both data on breastfeeding and blood lymphocytes. Total B-cell and T-cell numbers and their memory subsets were determined with 6-color flow cytometry. Mothers completed questionnaires on breastfeeding when their children were aged 2, 6, and 12 months. Multiple linear regression models with adjustments for potential confounders were performed.

Results. Per month continuation of breastfeeding, a 3% (95% CI -6,-1) decrease in CD27⁺IgM⁺, a 2% (95 CI % -5,-1) decrease in CD27⁺IgA⁺ and a 2% (95% CI -4,-1) decrease in CD27⁻IgG⁺ memory B-cell numbers were observed at 6 months of age. CD8⁺ T-cell numbers at 6 months of age were 20% (95% CI 3,37) higher in breastfed than in non-breastfed infants. This was mainly found for central memory CD8⁺ T cells and associated with exposure to breast milk, rather than duration. The same trend was observed at 14 months, but associations disappeared at older ages.

CONCLUSIONS. Longer breastfeeding is associated with increased CD8⁺ T-cell memory, but not B-cell memory numbers in the first 6 months of life. This transient skewing towards T-cell memory might contribute to the protective effect against infectious diseases in infancy.

INTRODUCTION

Breast milk contains factors that enhance nutrient absorption, stimulate growth and enhance the defense against pathogens.¹ Consequently, breastfeeding provides protection against infectious diseases during infancy.²⁻⁴ The protective effect persists during childhood,^{5,6} and modulates vaccination responses.⁷⁻⁹ Thus, it is likely that breastfeeding not only provides passive immunization, but also enhances adaptive immunity.¹⁰

B and T lymphocytes comprise the cellular components of adaptive immunity, and are generated throughout life. B cells contribute to humoral immunity through the production of immunoglobulins (Ig), whereas CD8⁺ cytotoxic T cells provide cellular immune responses, CD4⁺ helper T cells support both humoral and cellular immune responses. Each B and T cell generates a unique antigen receptor during precursor differentiation in bone marrow or thymus, respectively. Only those cells that specifically recognize antigen with their receptor will undergo clonal proliferation and are involved in the antigen response. Cells generated from the clonal expansion will remain present in the body as long-lived memory cells and will initiate a fast and quantitatively stronger response upon secondary antigen encounter. In addition to CD27⁻ naive B cells, six memory B-cell subsets can be identified.¹¹ Four of these express CD27 and are either positive for IgM, IgM and IqD, IqA or IqG. In addition, CD27 IqA+ and CD27 IqG+ memory B cells can be identified. Within both the CD4⁺ and CD8⁺ T-cell lineages, central memory (CD45RO⁺CCR7⁺), CD45RO⁺CCR7⁻ effector memory (TemRO) and CD45RO⁻ CCR7⁻ (TemRA) can be distinguished from naive T cells (CD45RO⁻CCR7⁺).¹² Central memory T cells are most efficient in generating a new immune response by proliferating extensively in response to an antigen upon secondary antigen encounter.¹³⁻¹⁵ The diversity and composition of the B-cell and T-cell compartments are highly dynamic in the first years of life; blood cell counts are especially high up to 2 years of age, following which they slowly decline to reach adult levels between 6 and 10 years.¹⁶⁻²⁰ At birth, nearly all B and T cells are naive, and memory cells are gradually built up in the first 6 years of life.¹⁹

Several studies have addressed the effects of breastfeeding on adaptive immunity. Breastfeeding was found to be associated with decreased frequencies of blood CD4⁺ T cells.²¹ This was mostly due to lower frequencies of naïve (CD45RA⁺) T cells in breastfed children. Still, these observations were not consistently reproduced with some studies showing increased, and some decreased numbers of CD4⁺ T cells.²¹⁻²⁴ Furthermore, long term breastfeeding was found to be associated with increasing numbers of CD4⁺ and CD8⁺ T cells.²⁴ Thus, although previous studies have addressed the influence of breastfeeding on blood lymphocyte populations,²¹⁻²⁴ the results remained inconclusive, mainly due to small samples sizes and limitations in the detection of memory cells.²¹⁻²⁴

In the present study we used 6-color flow cytometric analysis of lymphocyte subsets in a population-based prospective cohort study to assess the impact of breastfeeding on build-up of memory B and T cells in infants and young children.

MATERIALS AND METHODS

DESIGN AND STUDY POPULATION

This study was embedded in the Generation R Study, a prospective populationbased cohort study that follows pregnant women and their children from fetal life onwards in the Netherlands.²⁵ The study has been approved by the Medical Ethics Committee of the Erasmus MC, University Medical Centre Rotterdam. Written informed consent was obtained from all parents of participants. We included 1,079 Dutch pregnant women and their children participating in a detailed subgroup study.²⁵ All children were born between February 2002 and August 2006. We excluded twins (n=27) in the present analysis to prevent bias due to correlation. Of these, data on both breastfeeding and immunophenotyping were available from 258 children at 6 months, 166 at 14 months, 112 at 25 months and 332 at 6 years of age. The main reasons for missing samples were due to non-consent of the parents (approximately 55% per visit) and technical or logistical failure (approximately 10% per visit).

BREASTFEEDING

Information regarding breastfeeding was obtained in postnatal questionnaires at the ages of 2, 6 and 12 months.²⁵ Mothers were asked whether they had ever breastfed their child (yes or no) and, if yes, at what age (months) they had stopped.^{26,27} Breastfeeding duration was then categorized into four groups: never, \leq 3 months, between 3 and 6 months and \geq 6 months. An approximation of exclusive breastfeeding was performed according to whether the child received breastfeeding without any other bottle feeding, milk or solids.²⁷ Partial breastfeeding indicates infants receiving both breast-feeding, bottle feeding and/or solids in this period. Subsequently, the information on exclusiveness of breastfeeding was combined and categorized into the following breastfeeding categories: never; partial until 4 months and exclusive until 4 months.

IMMUNOPHENOTYPING OF LYMPHOCYTE SUBSETS

Flow cytometry was performed within 24 hours following sampling on fresh whole blood at the ages of 6 months (median 6.2; range 5.2; 8.2), 14 months (median 14.4; range 13.1-17.4), 25 months (median 25.2; range 23.3-29.8) and 6 years (median 5.9; range 5.1-7.2). Absolute counts of blood CD3⁺ T cells, CD16/56⁺ NK cells, and CD19⁺ B cells were obtained with a diagnostic lyse-no-wash protocol. Lymphocytes were gated on the basis of CD45, FSC and SSC characteristics. Gates were set based on cells that are known to lack expression of the indicated marker. Additionally, 6-color flow cytometry was performed on an LSRII (BD Biosciences) to distinguish naive and memory B-cell and T-cell subsets as defined previously (Supplemental Table 3).^{11,12} All flow cytometry acquisition was performed on whole blood after red blood cell lysis with ammonium chloride.

COVARIATES

The covariates that were assessed in this study were obtained from midwife and hospital registries at birth (birth weight, gestational age and gender) or through measurements at the research center (child anthropometrics). Additionally,

information on smoking and alcohol use during pregnancy and socio-economic status was obtained by prenatal questionnaires sent during the first, second and third trimesters of pregnancy.^{28,29} Information on day-care attendance was obtained from parent-reported questionnaires at the ages of 6 and 12 months.

STATISTICAL METHODS

Because the distribution of lymphocyte numbers in different age groups was skewed, these values were normalized by transformation to a natural log-scale. Differences in maternal and infant characteristics between breastfed versus never breastfed children were tested using independent samples t-tests and Chi-Square tests. Differences in baseline characteristics among the groups with different duration of breastfeeding were assessed using ANOVA tests and Chi-Square tests. Additionally, the associations of breastfeeding, breastfeeding duration (measured in groups, and measured continuously per month continuation of breastfeeding) and breastfeeding exclusivity with the change in lymphocyte numbers were assessed using multiple linear regression models with adjustment for potential confounders. For all ages, the category with no breastfeeding was the reference. First, associations between breastfeeding and total B, T, CD4 and CD8 counts were assessed. Subsequently, associations for B, CD4⁺ and CD8⁺ T-cell subpopulations were studied to assess the effect of breastfeeding on memory cells specifically. Multivariable regression models were created with stepwise adjustment for potential confounders, which were selected based on previous literature. Potential confounders included: maternal age, socio-economic status (SES), marital status, maternal BMI, maternal smoking and alcohol consumption during pregnancy, maternal reported autoimmune disease (including thyroid disease, multiple sclerosis, systemic lupus erythematosus, diabetes mellitus and arthritis), elevated anti-tTG level during pregnancy, maternal fever in the last trimester of pregnancy, family history of asthma or atopy (hay-fever, allergy, eczema), multiple parities, mode of delivery (caesarean section), gender, birth weight, gestational age, preterm birth, APGAR score, birth season, weight and age at focus visit, fever in the first 6 months (yes/no), frequency of upper and lower respiratory tract infections, and day-care attendance in the first year of life. Covariates were kept in the final multivariate model if the covariate resulted in an alteration in effect estimate of > 10%,³⁰ or if the variables were associated with breastfeeding (determinant) and lymphocyte numbers (outcome) in our study. Because of the small numbers in the never breastfed group, final adjustment for potential confounders was restricted to those who attained the strongest alteration (%) in effect estimates. Because of the strong correlation between our outcomes (e.g. Pearson's correlation between total CD8⁺ T cells and naive CD8⁺ T cells r =0.82, and between total B cells and IqA r=0.57), and unweighted calculations only hold if the tests are independent,^{31,32} we did not perform adjustments for multiple testing. All measures of associations are presented with their 95% confidence interval. All statistical analyses were performed using the Statistical Package for the Social Sciences version 20.0 for Windows (SPSS Inc, Chicago, IL, USA). P values <0.05 were considered to be statistically significant.

RESULTS

POPULATION CHARACTERISTICS

No major differences in characteristics between children included at 6, 14, 25 months and 6 years of age were observed (Table 1, Supplemental Table 1) Overall, more than 86% of mothers started breastfeeding. Mother's educational level was significantly associated with the start of breastfeeding at 6 and 25 months, and 6

		Not	Breastfed
		breastfed (n=35)	(n=223)
Maternal characteristics	s (n=258)		
Age (Mean <u>+</u> SD; years)		32 (3.6)	32 (3.8)
Educational loval (n: %)	Lower	24 (69%)	71 (32%)
	Higher	11 (31%)	152 (68%)*
Net household income per	< € 2400	2 (7%)	23 (11%)
month (n; %)	<u>≥</u> € 2400	29 (93%)	178 (89%)
Smoking continued during p	pregnancy (n; %)	4 (14%)	21 (13%)
Alcohol use continued durin	ng pregnancy (n; %)	8 (29%)	55 (35%)
Body Mass Index before pre	egnancy (Mean <u>+</u> SD; kg/m²)	23 (3)	23 (4)
Fever in third trimester of p	regnancy (n; %)	3 (9)	14 (6%)
Maternal atopy (eczema, all	ergy HDM, hay-fever)(n; %)	8 (25%)	72 (35%)
Paternal atopy (eczema, all	ergy HDM, hay-fever) (n; %)	8 (28%)	55 (28%)
Family history of asthma / a	atopy (n; %)	13 (37%)	107 (49%)
Any reported autoimmune of SLE, arthritis, MS, thyroid d	lisease (diabetes mellitus, lisorder, or celiac disease) (n; %)	0 (0%)	5 (0.02%)
	Vaginal	21 (64%)	129 (61%)
Mode of delivery (n; %)	Forceps or vacuum assisted	5 (15%)	44 (20%)
	Caesarian section	7 (21%)	40 (19%)
Premature rupture of membranes (n; %)		3 (9%)	6 (3%)
Infant characteristics (n	=258)		
Males (n; %)		19 (54%)	113 (51%)
Gestational age (Mean \pm SD; weeks)		39.4 (2.4)	40.0 (1.7)
Preterm birth (<37 weeks)	(n; %)	2 (6%)	11 (5%)
Birth weight (Mean \pm SD;	grams)	3439 (632)	3504 (524)
Apgar score at 5 min <7 (n	; %)	0 (0%)	3 (1%)
	Winter (dec-jan-feb)	5 (14%)	36 (16%)
Birth coscon (n: %)	Spring (mar-apr-may)	10 (29%)	81 (36%)
Dirtil Season (II, 70)	Summer (jun-jul-aug)	11 (31%)	62 (28%)
	Autumn (sept-oct-nov)	9 (26%)	44 (20%)
Siblings <u>≥</u> 1 (n; %)		4 (11%)	26 (12%)
Day-care >16 hours /week	(n; %)	5 (33%)	77 (46%)
Fever in first 6 months (n;	%)	12 (60%)	121 (62%)
Age at focus visit (Median <u>+</u>	range; months)	6.6 (6.2-8.2)	6.2 (5.2-7.9)*

Table 1 Maternal a	nd infort chore at a	whether of the study	a manulation at 6 months
Table 1. Maternal a	nu infant characte	eristics of the study	v population at o months

Values are means (SD), absolute numbers (percentages) or #medians (90% range). *Significantly different between groups.

Data were missing on: Household income (n=26) Smoking during pregnancy (n=72), Alcohol during pregnancy (n=72) BMI before pregnancy (n=37), Fever in third trimester (n=9) Mode of delivery (n=12) Maternal atopy (n=22) Paternal atopy (n=29), Family history of asthma or atopy (n=3) Mode of delivery (n=12), maternal reported autoimmune disease (n=29), maternal reported any other chronic condition (n=31), premature rupture of membranes (n=5), Apgar (n=4), Day-care (n=42), weight at focus visit (n=1), fever in first 6 months (n=42). Any reported autoimmune disease included thyroid disease (n=4) and elevated anti-tTG level during pregnancy (n=1).
years of age. Moreover, mother's educational level was significantly associated with both the duration of breastfeeding and with B-cell memory subsets at 6 months of age [data not shown]. In addition, maternal alcohol use was related to both breastfeeding duration and total T, B, CD4⁺ and CD8⁺ cell numbers at 6 months [data not shown]. Both maternal educational level and alcohol use influenced the regression coefficients by more than 10%. Therefore, all subsequent analysis on breastfeeding duration and cell numbers were adjusted for both maternal education and maternal alcohol use during pregnancy.



Figure 1. Impact of breastfeeding duration on B lymphocyte subsets at 6 months of age. A) Gating strategy for dissection of CD19⁺ B cells into 1 naive and 5 memory B-cell subsets by flow cytometry. The reference plots depict density plots of total lymphocytes and were used to set the gates accordingly. B) Backtransformed B-cell counts (cells/ μ L) at 6 months of age according to different breastfeeding duration categories (reference category is never). Categories of breastfed children contain both partial and exclusively breasted children. C) The estimated backtransformed regression line reflects B-cell counts (cells/ μ L) at 6 months of age, per month increase in breastfeeding duration.*, P<0.05.

DECREASE IN B-CELL MEMORY

Breastfeeding exposure (Figure 1B) and duration (Figure 1C) were not associated with total B-cell numbers at 6, 14, 25 months and 6 years of age (Table 2, Supplemental Table 2). Furthermore, no associations were observed between

breastfeeding exposure and duration and naive B cells, which constitute the majority of total B-cell numbers (Table 3, Figure 1B and 1C) However, a longer duration of breastfeeding, was associated with changes in the memory B-cell compartment. Per month longer breastfeeding, a 3% decrease in absolute numbers of CD27⁺IgM⁺, and a 2% decrease in both CD27⁺IgA⁺ and CD27⁻IgG⁺ memory B cells at 6 months of age were observed (Table 3, Figure 1C). CD27⁺IgG⁺ and CD27⁻IgA⁺ memory B cells at 6 months of age did not change with differences in breastfeeding duration. Stronger negative trends were observed for the associations between breastfeeding duration and frequencies of CD27⁺IgM⁺,





A) Gating strategy for T-cell subset delineation. CD4⁺ and CD8⁺ T-cell subsets were defined within total CD3⁺ T cells, and subsequently dissected using CD45RO and CCR7 into naive, central memory, CD45RO⁺ effector memory (TemRO) and CD45RO⁻ effector memory (TemRA) T cells. The reference depict density plots of total lymphocytes and were used to set the gates accordingly. **B)** Backtransformed CD4 and CD8 cell counts (cells/µL) at 6 months of age according to different breastfeeding duration categories (reference category is never). Categories of breastfed children contain both partial and exclusively breasted children. **C)** The estimated backtransformed regression line reflects CD4 and CD8 cell counts (cells/µL) at 6 months of age, per month increase in breastfeeding duration. *, P<0.05.

Table 2. Adjusted as	sociatio	ns betw	een breas	stfeeding a	nd cell nu	imbers at	6 and 14	months						
				Reg	Jression c	oefficient	s for logtr	ansform	ed cell nu	umbers (*	10e9/L)			
				6 month	ıs (n=258					14 mo	onths (n=	:166)		
		c	8	NK	F	CD4	CD8	5	8	NK	F	8	54	CD8
DURATION														
Breastfeeding increase	e/month	223 -1	(-3,1)	0 (-2,2) -	-1 (-2,1)	1 (-2,1)	1 (-1,2)	150 -	2 (-4,0)	0 (-2,2)	0 (-2,1	(-) 0	2,1) ((-2,2)
EXPOSURE														
Never		35	REF	REF	REF	REF	REF	16	REF	REF	REF	RE	L.	REF
< 3 months		86 9 ((-8,25) 4	(-16,24) 7	7 (-5,19)	6 (-7,19)	19 (3,35)	* 55 1	5 (10,40)	21 (-5,47)	16 (-3,3	5) 13 (-(6,32) 19	(-5,44)
\geq 3 < 6 months		64 6 (-12,23) 0	(-21,22) 8	3 (-5,21)	7 (-7,22)	20 (3,37)	* 40 2	6 (0,52)	-2 (-28,25) 20 (0,39	9) 18 (-5	2,38) 2	5 (0,50)
\geq 6 months		73 0 (-18,18) 9	(-13,31) 2	(-11,16)	0 (-14,15)	16 (-2,33) 55 2	(-22,27)	9 (-16,34)	13 (-5,3	1) 13 (-(6,31) 13	(-11,36)
Values are log transfor	med reg	ression (coefficients	(95% conf	fidence into	erval of ce	ll numbers braactfaadi	(*10 ^{e9} /L)) derived	from lines	ar regress	ion mode	is and re	eflect the
breastfed children. *P.	exposed exposed -value <0	to brea 0.05.Adju	stfeeding r steeding r	relative to th aternal edu	he reference ication and	maternal	never breas alcohol use	tfeeding during p). Breastfe regnancy.	ed groups i	nclude boi	th partiall	ly and ex	clusively
Table 3. Adjusted as:	sociatio	ns betw	een breas	stfeeding a	nd memo	ry B-cell	and T-cell	subsets	s at 6 moi	nths				
				Regress	sion coeff	icients for	r logtransf	ormed o	cell numb	ers (*10e	(1/6			
			B (n=258)				r cD4+	(n=258)) +8dD .	n=258)	
	Naive B cells	CD27+ IgM+	- CD27+ IgA+	+ CD27- IgA+	CD27+ IgG+	CD27- IgG+	CD4+ T naive	CD4+ T cm	CD4+ TemRO	CD4+ TemRA	CD8+ T naive	CD8+ T cm	CD8+ TemRO	CD8+ TemRA
DURATION														
Breastfeeding increase/ month (n=223)	-1 (-3,1)	-3 (-6,-1)	-2 * (-5,-1)	-2 * (-5,1)	1 (-3,2)	-2 (-4,-1)*	-2 (-3,0)	-1 (-2,1)	-1 (-3,2)	0 (-3,2)	-2 (-4,1)	0 (-2,3)	0 (-3,3)	2 (-1,5)
EXPOSURE														
Never (n= 35)	REF	REF	REF	REF	REF	REF	REF	REF	REF	REF	REF	REF	REF	REF
< 3 months	12	6	Ч	15	6-	-	7	ς.	17	14	23	30	15	20
(n= 86)	(-6,31)	(-19,36) (-28,30) (-14,43)	(-34,16)	(-22,22)	(-10,24)	(-20,14)	(-4,38)	(-8,37)	(0,46)	(4,57)* ((-18,47)	(-8,47)
\geq 3 < 6 months	18	18		14	0	1	4	 5	6	8	23	17	4 4	6
(n= 64)	(-1,38)	(-11,48) (-30,32	2) (-17,44)	(-26,27)	(-22,25)	(-14,22)	(-23,13)	(-14,31)	(-16,32)	(-2,448) (-11,45) ((-38,31)	(-20,39)
> 6 months	6	4	-11	-2	2- 2-	-13	-5		13	10	10	29	-5	21
(n= 73)	(-13,26)	(-34,25) (-40,19) (-31,28)	(-33,19)	(-36,10)	(-23,14)	(-21,15)	(-10,35)	(-14,34)	(-15,34) (1,57)* ((-39,30)	(-9,51)
Values are log transfor crease or decrease (%)	med regr in loa tr	ession c	oemcients ed cell num	лыегs ner mi	dence intei onth contir	val of cell	numbers (preastfeedir	* 10°°/L)) na.(durat	derived fr	the increas	regressior e or decre	i models ; ase (%) i	and rerie in lod tra	ct the In- nsformed
cell numbers of groups	exposed	to breas	stfeeding r	elative to th	ne referen	ce group (r	never breas	stfeeding). Breastf	ed groups	include bo	th partial	lly and e	clusively
preastred children. *P-	vaiue <0	ulpa.cu.	sted ror m	aternal equo	cation and	maternal a	alconol use	auring p	regnancy.					

Impact of breastfeeding on adaptive immunity

Chapter 3

CD27⁺IgA⁺ and CD27⁻IgG⁺ memory B cells within total B cells at 6 months of age, resp. -4% (95% CI -7, -1), -4% (95% CI -7, -1), and -3% (95% CI -5, -1) (data not shown). At 14 months of age, similar trends for breastfeeding duration and total B-cell numbers were observed, although not significant (Table 2). At older ages the effects disappeared. (Supplemental Table 2). Thus, a longer breastfeeding duration seems to negatively impact B-cell numbers in infants at least until the age of 6 months, at older ages these effects disappeared.

Table 4. Unadjusted associations between breastfeeding exclusivity and CD8 $^{\scriptscriptstyle +}$ T cell numbers at 6 months

Regression coefficients for logt	ransformed	cell numbe	ers (*10e9/L); 6 months	(n=258)
	CD8 Total	CD8	CD8	CD8	CD8
		Tnaive	Tcm	TemRA	TemRO
EXCLUSIVENESS					
Never (n= 35)	REF	REF	REF	REF	REF
Partial until 4 months (n=157)	13 (-3,29)	9 (-11,29)	17 (-7,41)	-5 (-30,21)	3 (-27,32)
Exclusive until 4 months (n=80)	14 (-3,31)	3 (-19,25)	33 (7,59)*	12 (-16,40)	28 (-3,59)
Unadjusted values are log transforme (*10 ^{e9} /L)) derived from linear regre transformed cell numbers of partial (never breastfeeding). Partial breas feeding and/or solids in this period.*F	d regression ession modele and exclusiv tfeeding indi 2-value <0.05	coefficients (s and reflec e breastfed cates infant: 5.	95% confiden t the increase groups relativ s receiving bo	ce interval of e or decrease ve to the refe oth breast-fee	cell numbers (%) in log rence group eding, bottle

INCREASE IN T-CELL MEMORY

Breastfeeding exposure and duration were not associated with total T-cell and CD4⁺ T-cell numbers at 6, 14, 25 months and 6 years of age (Table 2, Figures 2B and 2C, and Supplemental Table 2). However, the exposure to breastfeeding was associated with CD8⁺ T cell numbers. CD8⁺ T-cell numbers were 19% (95% CI 3, 35) higher in 6-month old infants who were breastfed for less than 3 months and remained 20% (95% CI 3, 37) higher in children who were breastfed until 6 months (Table 2, Figure 2B), than in children who were never breastfed. Comparable effect sizes were observed for exclusiveness of breastfeeding in relation to CD8⁺ T cells (Table 4 and 5). Partial breastfeeding until 4 months was associated with a 20% increase in total CD8 T cells, and remained 21% higher in infants who were exclusively breastfed (Table 5). At 14 and 25 months of age, similar tendencies for breastfeeding exposure and CD8⁺ T cell numbers were observed, however the associations were not significant. (Table 2, Supplemental Table 2).

Within the CD8⁺ T-cell compartment at 6 months of age, central memory cell numbers (CD45RO⁺CCR7⁺) were 30% (95% CI 4, 57) higher in children who were breastfed until 3 months than in non-breastfed children (Table 3, Figure 2B). This effect was irrespective of duration; central memory cell numbers were still 29% (95% CI 1, 57) higher in infants who were breastfed for 6 months or longer than in infants who were never breastfed (Table 3, Figure 2B). In line with this, no significant associations were found between the duration of breastfeeding and CD8⁺ central memory numbers (Table 3, Figure 2C).

Thus, CD8⁺ central memory T cells were associated with breastfeeding exposure, but not with the duration of breastfeeding.

Table 5. Adjusted associations between breastfeeding exclusivity and CD8 $^{\scriptscriptstyle +}$ T cell numbers at 6 months

Regression coefficients for lo	gtransforme	d cell numbe	ers (*10e9/L	.); 6 months	(n=258)
	CD8 Total	CD8	CD8	CD8	CD8
		Tnaive	Tcm	TemRA	TemRO
EXCLUSIVENESS					
Never (n= 35)	REF	REF	REF	REF	REF
Partial until 4 months (n=143)	20 (3,38)*	20 (-5,44)	12 (-17,40)	18 (-11,47)	4 (-31,39)
Exclusive until 4 months (n=80)	21 (2,40)*	14 (-14,41)	16 (-15,48)	27 (-5,59)	12 (-28,51)
Values are log transformed regres derived from linear regression mor numbers of partial and exclusive br Partial breastfeeding indicates infa period.*P-value <0.05. Adjusted for	sion coefficient dels and reflec reastfed groups nts receiving b or maternal edu	ts (95% confi t the increase s relative to th ooth breast-fe ucation and m	dence interva e or decrease ne reference g eding, bottle i aternal alcoho	l of cell numb (%) in log tra roup (never b feeding and/o ol use during p	pers (*10 ^{e9} /L)) ansformed cell preastfeeding). r solids in this pregnancy.

DISCUSSION

Overall we found that breastfeeding was associated with a decrease in B-cell memory and an increase in CD8 T-cell memory at 6 months of age. The same trend was observed at 14 months of age, and disappeared at older ages. The decrease in B-cell memory was associated with the duration of breastfeeding, while the increase in total CD8⁺ T cells and central memory CD8⁺ T cells did not depend on the duration of breastfeeding. This suggests that breastfeeding enhances T-cell maturation, but not B-cell maturation.

Total B-cell numbers nor frequencies were significantly related with breastfeeding. This is in line with previous studies.^{21,23} Only one study reported higher frequencies of total B cells in breastfed children at 6 months of age, however sample sizes were small (n=7 breastfed infants).²² We found that the duration of breastfeeding was associated with decreased numbers of CD27⁺IgA⁺, CD27⁺IgM⁺ and CD27⁻ IgG⁺ memory B cells, which are mostly derived from systemic T-cell dependent responses.¹¹ This suggests that continuous breastfeeding inhibits memory B-cell development. One of the candidate factors in breast milk that is likely to mediate this, is secretory IqA (sIqA).^{33,34} Continuous breastfeeding will provide a constant supply of maternal sIqA onto the epithelial surface of the infant's gastrointestinal tract. This sIgA might prevent exposure of microorganisms to the infant's humoral immune system, and translocation of gut bacteria.^{34,35} Thus, fewer naive B cells might be activated to differentiate into memory B cells (Supplemental Figure 1). Indeed, in suckling mice it has been suggested that maternal sIqA blocks mucosal B cell responses in the offspring.^{36,37} Alternatively, the unchanged numbers of T-cell independent CD27⁻IqA⁺ B cells suggest that breastfeeding does not affect local IqA responses. It is therefore possible that maternal IqA helps to block translocation of intestinal bacteria, thereby only preventing systemic T-cell dependent B-cell memory formation. Other growth factors in breast milk that can reduce exposure of microbiota to B cells are epidermal growth factor, IGF-1, TGF- β , leptin and prolactin. These factors enhance maturation of the epithelial barrier,^{33,38} and decrease uptake of foreign protein antigens. Factors such as lactoferrin, oligosaccharides and lipids may directly prevent attachment of the bacterial outer membrane to the mucosal surface.³³ Finally, the passive transfer of functional Iq-secreting plasma cells in breast milk may prevent bacterial or viral transmission.39,40

Total T-cell numbers nor CD4⁺ T-cell numbers were associated with breastfeeding exposure or breastfeeding duration. However, infants who were breastfed until 3 months had higher CD8⁺ T-cell numbers than infants who were never breastfed, and this change persisted when breastfeeding was prolonged until 6 months, suggesting an ongoing activation of CD8⁺ T cells by exposure to breast milk, but no accumulation over time. These results extend previous observations in small studies (n<40) of increased frequencies of CD8⁺ T cells or decreased CD4/ CD8 ratios.^{21,22,24} In addition, an increase in CD8⁺ T cells from 8 to 10 months of age was observed before,²⁴ suggesting a longer-lasting effect of breastfeeding. However, the children received breastfeeding until 8 months of age.²⁴ In addition, the number of children in the breastfeeding group was small (n=35), and no detailed analysis of CD8⁺ subsets was performed. Therefore, future studies will be needed to validate these findings.

Within total CD8⁺ T cells, the central memory subset was most significantly increased in breastfed children. This expansion will be the result of stimulation of mature T cells. Candidate immune stimulatory factors in breast milk are lactoferrin and exosomes. Lactoferrin has direct microbicidal properties, including binding of the bacterial cell wall and the concomitant release of lipopolysaccharide (LPS, endotoxin). Moreover, lactoferrin is known to enhance T-cell proliferation.^{21,33} Exosomes are carrier vesicles of 50-100 nm that can bud from the membrane of eukaryotic cells.⁴¹ Exosomes formed by B cells or dendritic cells contain MHC class I and class II, and have the potential to stimulate T cells.^{42,43} Human breast milk has been found to contain immune modulatory exosomes that express MHC molecules, IL-2, IFNy and TNFa. In line with our findings that breastfeeding did not affect memory CD4⁺ T cells, these vesicles did not appear to stimulate CD4⁺ T cells.⁴⁴ It is, however, likely that immunostimulatory compounds such as exosomes, do have CD8⁺ T-cell stimulating capacities and contribute to the increase in central memory T cells in breastfed infants (Supplemental Figure 1). In addition, it has been described that the infants take up live and functional maternal immune cells from the breast milk.⁴⁵⁻⁴⁷ This might result in stimulation of CD8⁺ T cells. Thus, it remains unclear whether the increase in central memory CD8⁺ T cells can be attributed to immunostimulatory factors in breast milk, such as lactoferrin and exosomes, or reflects a role for maternal immune cells. Nevertheless, central memory T cells might mediate reactive immunity,⁴⁸ because they circulate between the spleen, blood and lymph nodes and proliferate extensively in response to a second encounter of an antigen.¹³⁻¹⁵ Therefore, it could be hypothesized that an increase in central memory T cells is associated with increased reactive immunity.

Beneficial effects of breastfeeding on adaptive immune responses have previously been demonstrated in vaccination studies.⁷⁻⁹ Breasted children showed increased interferon-γ production, and increased frequencies of CD8⁺ T cells after vaccination with measles, mumps and rubella.⁹ Furthermore, breastfeeding had beneficial effects on virus-specific immune responses to poliovirus, diphtheria toxoid and tetanus toxoid,⁴⁹ whereas the responses to rotavirus are not clearly enhanced.^{50,51} These studies included relatively few children, and confounding factors could not be taken into account. Thus, our results extend these previous observations from vaccination studies. Unfortunately, we were not able to study virus-specific memory cells and studies determining associations between central memory cell numbers and functional immunity are lacking. Thus, interpretations regarding cell-mediated immunity should be made with caution.

We did not observe associations between breastfeeding and naive T cells or effector memory T cells. In contrast, a previous study reported lower frequencies of naive CD4⁺ T cells,²¹ but no differences in memory T-cell frequencies. Therefore, it was suggested that the adaptive immune system develops slower in breastfed infants.²¹ However, the study reported on a relatively small number of children in the breastfeeding group (n=34), and lacked data on absolute B-cell and T-cell counts.²¹

Our study was conducted in a large population-based prospective birth cohort. Previous studies that addressed the effects of breastfeeding on adaptive immunity, had smaller sample size and/or lacked follow-up.²¹⁻²⁴ Most of these retrospective studies were based on recall of infant feeding after several years, making recall bias of feeding habits an important concern.⁵² Because of the prospective design of our study, detailed information on the duration of breastfeeding was collected at multiple time points shortly after breastfeeding was finished, thereby limiting potential recall bias.⁵² Still, data on breastfeeding was collected retrospectively. and due to the use of questionnaires, misclassification may occur.⁵² Because any misclassification would be independent of laboratory determined T-cell and B-cell numbers, it is unlikely that these affected the outcome. In addition, our study design provided information on a large number of potential confounders. We used an unbiased approach, investigating a broad panel of determinants on lymphocyte numbers and frequencies. Moreover, we performed measurements of lymphocytes at different ages, enabling us to study the effect of breastfeeding on adaptive immunity over a longer period of time. Finally, using detailed 6-color flow cytometry, we were able to discriminate multiple, functionally distinct B-cell and T-cell subsets.^{11,12,53} Thus, we could evaluate the effects of breastfeeding on specific aspects of adaptive immunity.

At 14 months or age, we observed the same trends of decreasing B-cell numbers with longer breastfeeding duration, and higher CD8 T-cell numbers with breastfeeding exposure, as we did for 6 months, although effects were not significant. The sample size at 14 months was smaller than the sample size at 6 months. Hence, we cannot exclude that non-significant effects at 14 months are due to loss of statistical power, and some effects of breastfeeding remain at this age.

Our measurements were limited to peripheral blood. Thus, it is not possible to deduce whether B-cell and T-cell numbers in lymphoid tissue were affected. Theoretically, preferential homing to or away from tissue could result in changes in blood lymphocyte counts.⁵⁴ If breastfeeding affects this preferential homing, the effects will be lost once breastfeeding is discontinued. Nearly all children were no longer receiving breast milk and still showed lower memory B-cell numbers and higher CD8⁺ T-cell numbers. Thus, we conclude that the effects of breastfeeding on preferential tissue homing was limited.

In conclusion, this prospective population-based cohort study among a large number of healthy children showed that CD27⁺IgA⁺, CD27⁺IgM⁺ and CD27⁻IgG⁺ memory B-cell numbers decreased with a longer breastfeeding duration. CD8⁺ T cells, and especially CD8⁺ central memory T-cell numbers, were higher in breastfed children up to 6 months of age. This suggests that breastfeeding enhances T-cell maturation in the first 6 months of life. On top of the protective effects of maternal IgA in breast milk, this might contribute to the protective effect against infectious diseases in infancy.

ACKNOWLEDGEMENTS

The Generation R Study is conducted by the Erasmus MC, Erasmus University Rotterdam in close collaboration with the School of Law and Faculty of Social Sciences of the Erasmus University Rotterdam, the Municipal Health Service Rotterdam Metropolitan Area, the Rotterdam Homecare Foundation, and the Stichting Trombosedienst & Artsenlaboratorium Rijnmond. We gratefully acknowledge the contributions of children and parents, general practitioners, hospitals and midwives and pharmacies in Rotterdam. We would like to thank Ms. M. Smits-te Nijenhuis, Ms. M. Kolijn-Couwenberg, Ms. H. Charif and Mr. K. van Kester for technical support, and prof. dr. O.H. Franco for critically reading of the manuscript.

REFERENCES

- 1. Lonnerdal B. Bioactive proteins in breast milk. J Paediatr Child Health. 2013; 49 Suppl 1(1-7.
- 2. Duijts L, Ramadhani MK, Moll HA. Breastfeeding protects against infectious diseases during infancy in industrialized countries. A systematic review. *Matern Child Nutr*. 2009; 5(3): 199-210.
- 3. Walker A. Breast milk as the gold standard for protective nutrients. *J Pediatr*. 2010; 156(2 Suppl): S3-7.
- 4. Zinkernagel RM. Maternal antibodies, childhood infections, and autoimmune diseases. *N Engl J Med*. 2001; 345(18): 1331-1335.
- Wilson AC, Forsyth JS, Greene SA, Irvine L, Hau C, Howie PW. Relation of infant diet to childhood health: seven year follow up of cohort of children in Dundee infant feeding study. *BMJ*. 1998; 316(7124): 21-25.
- Silfverdal SA, Bodin L, Hugosson S, Garpenholt O, Werner B, Esbjorner E, et al. Protective effect of breastfeeding on invasive Haemophilus influenzae infection: a case-control study in Swedish preschool children. *Int J Epidemiol*. 1997; 26(2): 443-450.
- Pabst HF, Godel J, Grace M, Cho H, Spady DW. Effect of breast-feeding on immune response to BCG vaccination. *Lancet*. 1989; 1(8633): 295-297.
- Pabst HF, Spady DW. Effect of breast-feeding on antibody response to conjugate vaccine. Lancet. 1990; 336(8710): 269-270.
- Pabst HF, Spady DW, Pilarski LM, Carson MM, Beeler JA, Krezolek MP. Differential modulation of the immune response by breast- or formula-feeding of infants. Acta Paediatr. 1997; 86(12): 1291-1297.
- Hanson LA, Korotkova M, Lundin S, Haversen L, Silfverdal SA, Mattsby-Baltzer I, et al. The transfer of immunity from mother to child. Ann N Y Acad Sci. 2003; 987(199-206.
- 11. Berkowska MA, Driessen GJ, Bikos V, Grosserichter-Wagener C, Stamatopoulos K, Cerutti A, et al. Human memory B cells originate from three distinct germinal center-dependent and -independent maturation pathways. *Blood*. 2011.
- 12. Appay V, van Lier RA, Sallusto F, Roederer M. Phenotype and function of human T lymphocyte subsets: consensus and issues. *Cytometry A*. 2008; 73(11): 975-983.
- 13. Mueller SN, Gebhardt T, Carbone FR, Heath WR. Memory T cell subsets, migration patterns, and tissue residence. *Annu Rev Immunol*. 2013; 31(137-161.
- 14. Mackay CR. Dual personality of memory T cells. Nature. 1999; 401(6754): 659-660.
- 15. Pepper M, Jenkins MK. Origins of CD4(+) effector and central memory T cells. *Nat Immunol*. 2011; 12(6): 467-471.
- Comans-Bitter WM, de Groot R, van den Beemd R, Neijens HJ, Hop WC, Groeneveld K, et al. Immunophenotyping of blood lymphocytes in childhood. Reference values for lymphocyte subpopulations. J Pediatr. 1997; 130(3): 388-393.
- 17. Driessen GJ, Dalm VA, van Hagen PM, Grashoff HA, Hartwig NG, van Rossum AM, et al. Common variable immunodeficiency and idiopathic primary hypogammaglobulinemia: two different conditions within the same disease spectrum. *Haematologica*. 2013; 98(10): 1617-1623.
- Shearer WT, Rosenblatt HM, Gelman RS, Oyomopito R, Plaeger S, Stiehm ER, et al. Lymphocyte subsets in healthy children from birth through 18 years of age: the Pediatric AIDS Clinical Trials Group

P1009 study. J Allergy Clin Immunol. 2003; 112(5): 973-980.

- 19. van Gent R, van Tilburg CM, Nibbelke EE, Otto SA, Gaiser JF, Janssens-Korpela PL, et al. Refined characterization and reference values of the pediatric T- and B-cell compartments. *Clin Immunol*. 2009; 133(1): 95-107.
- 20. Tosato F, Bucciol G, Pantano G, Putti MC, Sanzari MC, Basso G, et al. Lymphocytes subsets reference values in childhood. *Cytometry A*. 2014.
- Andersson Y, Hammarstrom ML, Lonnerdal B, Graverholt G, Falt H, Hernell O. Formula feeding skews immune cell composition toward adaptive immunity compared to breastfeeding. *J Immunol.* 2009; 183(7): 4322-4328.
- 22. Carver JD, Pimentel B, Wiener DA, Lowell NE, Barness LA. Infant feeding effects on flow cytometric analysis of blood. *J Clin Lab Anal*. 1991; 5(1): 54-56.
- 23. Hawkes JS, Neumann MA, Gibson RA. The effect of breast feeding on lymphocyte subpopulations in healthy term infants at 6 months of age. *Pediatr Res.* 1999; 45(5 Pt 1): 648-651.
- Jeppesen DL, Hasselbalch H, Lisse IM, Ersboll AK, Engelmann MD. T-lymphocyte subsets, thymic size and breastfeeding in infancy. *Pediatr Allergy Immunol*. 2004; 15(2): 127-132.
- 25. Jaddoe VW, van Duijn CM, Franco OH, van der Heijden AJ, van Iizendoorn MH, de Jongste JC, et al. The Generation R Study: design and cohort update 2012. *Eur J Epidemiol*. 2012; 27(9): 739-756.
- van Rossem L, Oenema A, Steegers EA, Moll HA, Jaddoe VW, Hofman A, et al. Are starting and continuing breastfeeding related to educational background? The generation R study. *Pediatrics*. 2009; 123(6): e1017-1027.
- Kiefte-de Jong JC, Escher JC, Arends LR, Jaddoe VW, Hofman A, Raat H, et al. Infant nutritional factors and functional constipation in childhood: the Generation R study. *Am J Gastroenterol*. 2010; 105(4): 940-945.
- Statistics N. Dutch Standard Classification of Education 2006. Voorburg/Heerlen, Statistics Netherlands 2008
- 29. Statistics Netherlands. Welfare in the Netherlands. Income, welfare and spending of households and individuals. *Den Haag/Heerlen: Statistics Netherlands, 2012*.
- 30. Mickey RM, Greenland S. The impact of confounder selection criteria on effect estimation. *Am J Epidemiol.* 1989; 129(1): 125-137.
- 31. Bender R, Lange S. Adjusting for multiple testing--when and how? *J Clin Epidemiol*. 2001; 54(4): 343-349.
- 32. Rothman KJ. No adjustments are needed for multiple comparisons. Epidemiology. 1990; 1(1): 43-46.
- 33. M'Rabet L, Vos AP, Boehm G, Garssen J. Breast-feeding and its role in early development of the immune system in infants: consequences for health later in life. J Nutr. 2008; 138(9): 1782S-1790S.
- 34. Pabst O. New concepts in the generation and functions of IgA. *Nat Rev Immunol*. 2012; 12(12): 821-832.
- 35. Wold AE, Adlerberth I. Does breastfeeding affect the infant's immune responsiveness? *Acta Paediatr*. 1998; 87(1): 19-22.
- 36. Kramer DR, Cebra JJ. Role of maternal antibody in the induction of virus specific and bystander IgA responses in Peyer's patches of suckling mice. *Int Immunol.* 1995; 7(6): 911-918.
- Kramer DR, Cebra JJ. Early appearance of "natural" mucosal IgA responses and germinal centers in suckling mice developing in the absence of maternal antibodies. *J Immunol*. 1995; 154(5): 2051-2062.
- Lawrence RM, Pane CA. Human breast milk: current concepts of immunology and infectious diseases. Curr Probl Pediatr Adolesc Health Care. 2007; 37(1): 7-36.
- 39. Hanson LA, Silfverdal SA, Korotkova M, Erling V, Strombeck L, Olcen P, et al. Immune system modulation by human milk. *Adv Exp Med Biol*. 2002; 503(99-106.
- Tuaillon E, Valea D, Becquart P, Al Tabaa Y, Meda N, Bollore K, et al. Human milk-derived B cells: a highly activated switched memory cell population primed to secrete antibodies. *J Immunol.* 2009; 182(11): 7155-7162.
- Thery C, Ostrowski M, Segura E. Membrane vesicles as conveyors of immune responses. Nat Rev Immunol. 2009; 9(8): 581-593.
- 42. Raposo G, Nijman HW, Stoorvogel W, Liejendekker R, Harding CV, Melief CJ, et al. B lymphocytes secrete antigen-presenting vesicles. J Exp Med. 1996; 183(3): 1161-1172.
- 43. Zitvogel L, Regnault A, Lozier A, Wolfers J, Flament C, Tenza D, et al. Eradication of established

murine tumors using a novel cell-free vaccine: dendritic cell-derived exosomes. *Nat Med.* 1998; 4(5): 594-600.

- 44. Admyre C, Bohle B, Johansson SM, Focke-Tejkl M, Valenta R, Scheynius A, et al. B cell-derived exosomes can present allergen peptides and activate allergen-specific T cells to proliferate and produce TH2-like cytokines. *J Allergy Clin Immunol*. 2007; 120(6): 1418-1424.
- 45. Jain L, Vidyasagar D, Xanthou M, Ghai V, Shimada S, Blend M. In vivo distribution of human milk leucocytes after ingestion by newborn baboons. *Arch Dis Child*. 1989; 64(7 Spec No): 930-933.
- Schlesinger JJ, Covelli HD. Evidence for transmission of lymphocyte responses to tuberculin by breast-feeding. *Lancet*. 1977; 2(8037): 529-532.
- 47. Mohr JA. Lumphocyte sensitisation passed to the child from the mother. Lancet. 1972; 1(7752): 688.
- 48. Sallusto F, Geginat J, Lanzavecchia A. Central memory and effector memory T cell subsets: function, generation, and maintenance. *Annu Rev Immunol*. 2004; 22(745-763.
- 49. Hahn-Zoric M, Fulconis F, Minoli I, Moro G, Carlsson B, Bottiger M, et al. Antibody responses to parenteral and oral vaccines are impaired by conventional and low protein formulas as compared to breast-feeding. *Acta Paediatr Scand*. 1990; 79(12): 1137-1142.
- 50. Groome MJ, Moon SS, Velasquez D, Jones S, Koen A, van Niekerk N, et al. Effect of breastfeeding on immunogenicity of oral live-attenuated human rotavirus vaccine: a randomized trial in HIV-uninfected infants in Soweto, South Africa. *Bull World Health Organ*. 2014; 92(4): 238-245.
- 51. Rennels MB. Influence of breast-feeding and oral poliovirus vaccine on the immunogenicity and efficacy of rotavirus vaccines. *J Infect Dis.* 1996; 174 Suppl 1(S107-111.
- 52. Li R, Scanlon KS, Serdula MK. The validity and reliability of maternal recall of breastfeeding practice. *Nutr Rev.* 2005; 63(4): 103-110.
- 53. Freud AG, Caligiuri MA. Human natural killer cell development. Immunol Rev. 2006; 214(56-72.
- 54. Bunders MJ, van der Loos CM, Klarenbeek PL, van Hamme JL, Boer K, Wilde JC, et al. Memory CD4(+) CCR5(+) T cells are abundantly present in the gut of newborn infants to facilitate mother-to-child transmission of HIV-1. *Blood*. 2012; 120(22): 4383-4390.

14 months (n=166) 25 months (n=112) 6 years (n=332) Not Not Not breastfed Breastfed breastfed Breastfed breastfed Breastfed (n=16) (n=150)(n=16) (n=96) (n=37)(n=295) Maternal characteristics Age (Mean + SD; years) 31 (4.2) 32 (3.9) 31 (4.1) 32 (3.8) 31 (3.6) 32 (3.6) Educational level (n; %) Lower 7 (44%) 59 (40%) 12 (75%) 31 (33%) 22 (59%) 96 (33%) 4 (25%) 64 (67%)* Higher 9 (56%) 90 (60%) 15 (41%) 196 (67%)* Net household income/month (n: %) 0 (0%) < € 2400 21 (15%) 1 (8%) 13 (15%) 4 (12%) 33 (13%) > € 2400 13 (100%) 112 (84%) 12 (92%) 73 (85%) 30 (88%) 227 (87%) Smoking during pregnancy (n; %) 2 (14%) 16 (15%) 1 (7%) 8 (11%) 6 (20%) 32 (15%) Alcohol use during pregnancy (n; %) 6 (43%) 41 (38%) 4 (29%) 29 (39%) 11 (37%) 83 (37%) Body Mass Index before pregnancy (Mean, SD) 25 (5) 24 (5) 25 (5) 23 (4) 24 (5) 24 (4) Fever in third trimester of pregnancy (n; %) 1(6%) 9 (6%) 2 (13%) 5 (6%) 2 (6%) 19 (7%) atopy (eczema, allergy HDM, hay-fever) (n; %) Maternal 3 (21%) 51 (36%) 6 (40%) 41 (46%) 9 (27%) 109 (40%) Paternal 3 (23%) 41 (29%) 5 (39%) 24 (28%) 13 (39%) 77 (29%) 56 (59%) Family history of asthma / atopy (n; %) 5 (31%) 77 (52%) 9 (56%) 20 (54%) 156 (54%) Any reported autoimmune disease (diabetes mellitus, SLE, arthritis, MS, thyroid disorder, or 0 (0%) 3 (2%) 0 (0%) 2 (2%) 2 (5%) 8 (3%) celiac disease) (n; %) Mode of delivery (n; %) 9 (56%) 7 (47%) Vaginal 19 (64%) 48 (55%) 20 (59%) 182 (65%) Forceps or vacuum assisted 2 (13%) 30 (21%) 2 (13%) 21 (24%) 5 (15%) 59 (21%) 6 (40%) 19 (22%) 40 (14%) Caesarian section 5 (31%) 21 (15%) 9 (26%) Premature rupture of membranes (n; %) 1 (6%) 2 (1%) 1 (6%) 2 (2%) 3 (8%) 4 (2%) Infant characteristics (n=258) Males (n; %) 9 (56%) 77 (51%) 7 (44%) 47 (49%) 20 (54%) 144 (49%) Gestational age (Mean + SD; weeks) 39.2 (2.2) 40.1 (1.9) 39.7 (1.9) 39.9(1.5) 39.7 (1.7) 40.1 (1.6) Preterm birth (<37 weeks) (n; %) 1 (6%) 3 (2%) 1 (6%) 4 (4%) 2 (5%) 8 (3%) Birth weight (Mean \pm SD; grams) 3,455 (738) 3,505 (513) 3,516 (614) 3,542 (480) 3,543 (554) 3,531 (508) Appar score at 5 min <7 (n; %) 0 (0%) 3 (2%) 0 (0%) 0 (0%) 0 (0%) 3 (1%) Birth season (n; %) Winter 2 (13%) 38 (25%) 3 (19%) 28 (29%) 6 (16%) 73 (25%) Spring 5 (31%) 41 (27%) 3 (19%) 26 (27%) 6 (16%) 93 (31%) Summer 3 (18%) 32 (21%) 4 (25%) 21 (22%) 11 (30%) 65 (22%) Autumn 6 (38%) 39 (26) 6 (37%) 21 (22%) 14 (38%) 64 (22%) 2 (12%) 17 (11%) 1 (6%) 8 (8%) 2 (5%) 26 (9%) Siblings ≥ 1 (n; %) 51 (47%) Day-care >16 hours /week (n; %) 3 (33%) 4 (25%) 33 (34%) 5 (33%) 99 (47%) 7 (78%) 80 (63%) 3 (60%) 51 (62%) 13 (57%) 175 (67%) Fever in first 6 months (n; %) 14.4 14.4 25.3 25.2 6.0 5.8 Age at focus visit (Median + range; months/years) (13.2-16.3) (13.1-17.4) (23.4-27.5) (23.3-29.8) (5.8-6.6)(5.1-7.2)Values are means (SD), absolute numbers (percentages) or #medians (90% range). * Significantly different

Supplemental Table 1. Maternal and infant characteristics of the study population at 14, 25 months and 6 years

between breastfeeding and no breastfeeding groups

•	
5	_
(υ
÷	2
	2
(0
÷	Ę
C)

				Logtran	sformed cell nu	umbers (*10e9	(T)			
			25 mc	onths (n=112	0			6 yea	rs (n=332)	
	5	B	NK	F	CD4	CD8	5	B	NK	F
DURATION										
Breastfeeding increase/month	96	-1 (-4,1)	0 (-3,3)	0 (-2,2)	1 (-2,3)	-2 (-4,1)	295	-1(-2,1)	0(-2,1)	1(-1,2)
EXPOSURE										
Never	16	REF	REF	REF	REF	REF	37	REF	REF	REF
< 3 months	29	14 (-10,38)	27 (-3, 58)	14 (-7,35)	11 (-10,32)	18 (-9,45)	112	-6(-19,8)	11(-5,26)	1(-10,12)
\geq 3 < 6 months	30	0 (-23,23)	7 (-23, 37)	-2 (-22,18)	-4 (-24,17)	1 (-25,27)	76	-5(-19,10)	7(-10,24)	2(-10,14)
<u>></u> 6 months	37	11 (-14,35)	11 (-20,42)	12 (-9,33)	18 (-3,30)	-3 (-30, 24)	107	-12(-26,3)	-1(-17,16)	3(-8,15)
Never	15	REF	REF	REF	REF	REF	52	REF	REF	REF
<u><</u> 6 months	50	-17 (-35, 2)	3 (-22,27)	-7 (-23,10)	-11 (-28,6)	3 (-18,25)	150	3(-8,13)	6(-6,18)	1(-8,10)
6-9 months	18	-4 (-29,21)	1 (-33,34)	-2 (-24,21)	0 (-23,23)	-2 (-31,28)	50	5(-10,20)	2(-15,19)	1(-11,13)
\geq 9 months	15	3 (-26, 32)	-17 (-55, 21)	9 (-17,35)	10 (-17,36)	8 (-26,41)	34	-3(-20,14	-6(-25,14)	1(-13,15)
Values are log transformed re	egressic	on coefficients ('95% confidenc	te interval of o	cell numbers (*:	10e9/L)) derived	from li	near regressi	on models ar	id reflect the
increase or decrease (%) in It	og tran	sformed cell nu	mbers per mon	iths continuation	on of breastfeed	ing (duration), a	and the	increase or	decrease (%)	in log trans-
formed cell numbers of group:	s expos	sed to breastfee	ding relative to	the reference	e group (never b	reastfeeding). N	dissing	in in categori	ies 0-6-9 mon	ths (n=14 at
25 months, n=46 at 6 years)	Breast	fed groups inclu	ide both partia	illy and exclusi	vely breastfed c	hildren. Adjuste	d for m	iaternal educa	ation and mat	ernal alcohol
use during pregnancy. *P-valu	Je <0.0	5.								

Supplemental Table 2. Adjusted association between breastfeeding and cell numbers at 25 months and 6 years

Labeling		Conju	igated monoclo	nal antibodi	es (clone)	
	FITC	PE	PerCP-Cy5.5	PE-Cy7	APC	APC-Cy7
1	CD3	CD16.56	CD45	CD4	CD19	CD8
1	(SK7)	(B73.1C5.9)	(2D1)	(SK3)	(SJ25C1)	(SK1)
2	IgD	CD23	CD19	CD21	IgM	CD27
2	(poly)	(EBVCS5)	(SJ25C1)	(B-ly-4)	(G20-127)	(M-T271)
3	IgA	IgG	CD19	CD40	IgM	CD27
	(poly)	(poly)	(SJ25C1)	(5C3)	(G20-127)	(M-T271)
4	CD28	CD197	CD3	CD8	CD45RO	CD27
4	(CD28.2)	(3D13)	(SK7)	(SK1)	(UCHL-1)	(M-T271)

Supplemental Table 3. Antibody pane	I used for 6-color flow cytometry
-------------------------------------	-----------------------------------

FITC= fluorescein isothiocyanate, PE=phycoerythrin, PerCPCy5.5= peridin chlorophyll protein, PE-Cy7= phycoerythrin-cyanin dye, APC=allophycocyanin and APC-Cy7= allophycocyanin-cyanin dye, poly = polyclonal antibody



Supplemental Figure 1. Summarizing mechanism of how breastfeeding might affect adaptive memory. In absence of breast milk, the infant's B and T cells respond to microorganisms in the intestine and generate long-lived memory cells and IgA (blue) that circulate through the body (left). Breast milk contains immune modulating components (right). Of these, maternal SIGA (green) is able to catch microorganisms and prevent recognition of these by B-cells. This might inhibit B-cell responses and B-cell memory formation. Other immunostimulatory components, such as exosomes, might stimulate naive T cells and increase T-cell memory formation. Abbreviations: Bn, naïve B cell; Bm, memory B cell; DC, dendritic cell; pc,plasma cell; Tn, naive T cell; Tm, memory T cell.

CHAPTER 4

CAN WE EXPLAIN ETHNIC DIFFERENCES IN CMV, EBV AND HSV-1 VIRUS SEROPREVALENCES IN CHILDHOOD? THE GENERATION R STUDY

Running title: Determinants of herpesviridae

M.A.E. Jansen,^{1,2} D. van den Heuvel,³ S.H. Bouthoorn,^{1,4} V.W.V. Jaddoe,^{1,2,6} H. Hooijkaas,³ H. Raat,⁴ P.L.A. Fraaij,^{2,5} M.C. van Zelm,³ H.A. Moll.²

¹ The Generation R Study Group, Erasmus MC, University Medical Center, Rotterdam, NL
 ² Dept. of Pediatrics, Erasmus MC, University Medical Center, Rotterdam, NL
 ³ Dept. of Immunology, Erasmus MC, University Medical Center, Rotterdam, NL
 ⁴ Dept. of Public Health, Erasmus MC, University Medical Center, Rotterdam, NL
 ⁵ Dept. of Viroscience, Erasmus MC, University Medical Center, Rotterdam, NL
 ⁶ Dept. of Epidemiology, Erasmus MC, University Medical Center, Rotterdam, NL

Submitted



ABSTRACT

OBJECTIVE. To identify whether there are ethnic differences in CMV, EBV and HSV-1 seroprevalence rates in children at 6 years of age, and when present, to evaluate how these differences can be explained by socio-demographic and environmental factors.

METHODS. This study was embedded within a multi-ethnic population-based prospective cohort study. Serum IgG levels against CMV, EBV and HSV-1 were measured by ELISA in 4,464 children (median age 6.0 years). Information on demographics and characteristics were assessed by questionnaires. Herpesvirus seroprevalences between Surinamese-Creole, Surinamese-Hindustani, Turkish, Moroccan, Cape Verdean and Antillean and Native Dutch children were compared.

Results. Non-western ethnicity was an independent risk factor for CMV (aOR, 2.16; 95% CI 1.81;2.57), EBV (aOR 1.76; 95% CI 1.48;2.09) and HSV-1 seropositivity (1.52; 1.39;1.66). Among the ethnic groups, CMV seroprevalences ranged between 29-65%, EBV between 43-69% and HSV-1 between 13-39%. Socio-economic position, crowding and lifestyle factors explained up to 48% of the ethnic differences in HSV-1 seroprevalences, and up to 39% of the ethnic differences in EBV seroprevalences. These factors did not explain ethnic differences in CMV seroprevalences.

CONCLUSION. Socio-economic position and factors related to lifestyle only explain a part of the large ethnic differences in EBV and HSV-1 seroprevalences, whereas they do not explain ethnic differences in CMV seroprevalences in childhood.

INTRODUCTION

Cytomegalovirus (CMV), Epstein Barr virus (EBV), and herpes simplex virus 1 (HSV-1) are ubiquitous herpesviruses, and often acquired in childhood. They are usually transmitted by contact with infected saliva (all three viruses), breastfeeding, placental transfer (CMV), {Longnecker, 2007 #3882;Boppana, 2007 #3880}or through contact with infectious skin lesions or secretions (HSV-1).^{1,2} The viruses establish a lifelong latency after primary infection, and can periodically reactivate with shedding of the virus. Most primary infections are mild or asymptomatic in childhood, but they can cause serious complications in fetuses, immunocompromised individuals, or in elderly.³⁻⁹ The incidence and severity of these herpesvirus associated diseases vary geographically, and many studies link age at primary infection to the pathogenesis of herpesvirus associated diseases.¹⁰⁻¹⁴

CMV seroprevalences vary between 40-60% in children aged 4-12 years,^{8,15,16} with exclusion of African countries where almost all children have already been seroconverted by 3 years of age. Approximately 50% of children in the US and Europe are seropositive for EBV in the first years of life, but seroprevalences vary from 20-80% depending on age, race/ethnicity, geographic location and socio-economic development of the country.^{9,12,17,18} HSV-1 is widespread during the first few years of life in developing countries, whereas in some European countries primary infection occurs relatively at older age.¹⁹⁻²² Besides geographical differences, also ethnic differences have been described in population subgroups.²³ Ethnic and geographical differences in CMV, EBV and HSV-1 seroprevalences might be explained by differences in behavior that facilitate transmission, such as breastfeeding, daycare, family size and socio-economic position.^{17,22,24,25} Because socio-economic position and ethnicity are strongly related, it remains unclear to what extent socio-economic factors explain ethnic differences in herpesvirus seroprevalences.²⁶ In addition, knowledge on other lifestyle related determinants is mainly based on prevalence studies from the late 90's,27-30 but social conditions and lifestyle have changed over time.^{31,32} Furthermore, the relative importance of each risk factor needs to be assessed,²⁶ and many studies lack the assessment of multiple risk factors. Previously, it has been suggested that early life exposures, in particular breastfeeding, are more important for CMV infection than adulthood exposures.³³ On the other hand, the ALADDIN birth cohort study found no strong influence of these early life risk factors.³⁴ Understanding transmission dynamics of herpesvirus infections within populations, and understanding how socioeconomic and ethnic groups are differentially exposed to these infections, can improve existing preventive policies and interventions, including on the long run, vaccination programs.35

Therefore, the first aim is to assess whether there are ethnic differences in CMV, EBV and HSV-1 infection in children at 6 years of age. The second aim is to evaluate how ethnic differences in CMV, EBV and HSV-1 infections, when present, can be explained by socio-demographic and environmental factors.

METHODS

STUDY DESIGN

This study was embedded within the Generation R Study, a prospective population based cohort study from fetal life onwards.^{36,37} All children were born between 2002 and 2006 in Rotterdam, the second largest city in the Netherlands. The study area covers more than half of the cities inhabitants, reflecting Rotterdam's multi-ethnic population.³⁶ The largest ethnic groups are of Dutch, Surinamese, Turkish, Moroccan, Dutch Antilles and Cape Verdean origin.³⁶ In total, n=8,305 children participated in the postnatal phase of the study, of whom 81% (n=6,690) visited the research centre (median age 6.0 years).³⁷ During this visit, blood samples were collected from 4,593 children. In total, n=4,464 children provided serological data on EBV, CMV, HSV-1 status (Supplemental Figure 1). The study was approved by the Medical Ethical Committee of Erasmus MC, University Medical Centre Rotterdam. Parents of the children gave written informed consent.

HERPESVIRUS SEROLOGY

Venous blood plasma samples were analysed using enzyme immunoassays for IgG antibodies against CMV, EBV (capsid antigen), HSV-1 (glycoprotein C1; all from EUROIMMUN®, Lübeck, Germany). Results were evaluated semiquantitatively relative to a manufacturer-provided reference threshold sample. Per manufacturer's instructions, an optical density in the patient sample >10% above the provided threshold sample was defined as positive.

ETHNIC BACKGROUND

Ethnicity of the child was determined by country of birth of the parents. The child was of non-native Dutch origin if one of the parents was born in another country than the Netherlands.³⁸ If both parents were born abroad, the country of birth of the mother prevailed. To study whether ethnic background was related to herpesvirus infections, we constructed a dichotomous variable 'Western/nonwestern' ethnicity. Next, to study to what extent socio-economic and lifestyle factors explained associations between ethnicity and herpesvirus infections, and to study the effect of cultural background of the mother (most often primary caregivers),³⁹ a distinction was made between mothers of Dutch, Turkish, Moroccan, Cape Verdean, Antillean, Surinamese Creole and Surinamese Hindustani ethnic background. Maternal ethnic background was based on country of birth of mother's parents and was assessed by questionnaires at enrollment. Mothers were considered non-native Dutch if one of her parents was born abroad.⁴⁰ If both parents were born abroad, country of birth of the mother's mother prevailed. Generational status of non-native Dutch mothers was based on their own country of birth. Foreign-born mothers were classified as 'first generation' and mothers born in the Netherlands were classified as 'second generation'.³⁹

OTHER DETERMINANTS

Other potential determinants of herpesvirus infections were selected based on existing literature.^{2,12,24,32,33,41,42} These are: socio-economic position, family size/ parity, maternal lifestyle, breastfeeding, daycare and gender. Data on gender,

gestational age, birth weight, parity and mode of delivery were obtained from obstetric records from hospitals and mid-wife practices. Data on socio-demographic and lifestyle factors were obtained by a combination of pre- and postnatal questionnaires completed by both parents. Prenatal questionnaires included information on maternal lifestyle and socio-economic position. Information on socio-economic position was also determined by questionnaires obtained at 6 years of age. Information on breastfeeding and daycare was obtained by postnatal questionnaires. All questionnaires were available in three languages (Dutch, English and Turkish) and further support for verbal translation of questionnaires was available in Arabic, Portuguese and French.

EXPLANATORY VARIABLES

Most effects of ethnicity on herpes virus infections are probably indirect ones, acting through more proximal determinants of CMV, EBV, and HSV-1. We considered the following factors to be potential explanatory variables: socio-economic position, breastfeeding, parity as indicator for siblings/family size, and day-care attendance.^{2,12,24,32,33,41,42} Information on household income⁴³ and educational level,⁴⁴ both indicators of socio-economic position, were obtained using prenatal questionnaires. The Dutch Standard Classification of Education was used to categorize 4 subsequent levels of education: 1=high, 2= mid-high, 3=mid-low, and 4=low.⁴⁴ For additional analyses, income was subdivided into three categories: 1. Below modal <2,000 euro's per month, 2. Between 2,000 and 3,200 euro's per month, 3. > 3,200 euro's per month.^{45,46} Child's age and gender were treated as potential confounders in the association between ethnicity and herpesvirus seroprevalences.

STATISTICAL ANALYSIS

First, independent student's T-tests, and χ^2 tests were used to test whether characteristics were different between children who were seropositive for CMV, EBV and HSV-1, and who were not. To investigate which determinants were associated with CMV, EBV and HSV-1 seropositivity (y/n), logistic regression analyses were performed with herpesvirus seropositivity at the age of 6 years as dependent variables (CMV y/n, EBV y/n, and HSV-1 y/n). To study to which extent associations between ethnicity and CMV, EBV, and HSV-1 seropositivity were explained by factors associated with socio-economic position, crowding, breastfeeding behaviour and maternal lifestyle, we used Baron and Kenny's causal step approach.⁴⁷ Only those factors that were unequally distributed across the ethnicity groups (determinant) (Table 6; online) and were significant (p<0.05) associated with CMV, EBV and HSV-1 seropositivity (outcomes) (independent of ethnicity, gender and age); (Table 7; online) were added separately to the model.⁴⁷ To assess their mediating effects, the corresponding percentages of attenuation of effect estimates were calculated by comparing differences of model 1 with the adjusted ones (100 x (OR model 1 - OR model 2 with explanatory factor)/ (OR model 1)). Finally, a full model containing ethnicity and all the explanatory factors assessed the joint effects of explanatory factors. Interaction terms between ethnic background and the explanatory variables were tested for significance. If the test was significant (p < 0.05), we also stratified the analysis by the variables. To reduce attrition bias, multiple imputation of the covariables was performed

Table 1. Maternal and infant characteristics of tl	he study popula	ation (n=4,464)				
	Ш	BV	Ū	۸v	SH	V-1
	EBV - (n=2,185)	EBV + (n=2,279)	CMV - (n=2,772)	CMV + (n=1,692)	HSV-1 - (n=3,592)	HSV-1 + (n=877)
Maternal characteristics				-		
Age (Mean ± SD; years)	31.4 (4.9)	30.1 (5.3)***	31.1 (5.0)	30.1 (5.3)***	31.0 (5.0)	29.5 (5.4)***
Educational level (n; %)						
Low	333 (16%)	535 (27%)	495 (19%)	373 (25%)	613 (19%)	255 (35%)
Mid Iow	604 (30%)	630 (31%)	794 (31%)	440 (29)	966 (30%)	238 (32%)
Mid high	504 (25%)	390 (19%)	582 (23%)	312 (21%)	768 (23%)	126 (17%)
Higher	595 (29%)	462 (23%)	689 (27%)	368 (25%)	936 (28%)	121 (16%)
Net household income per month (n; %)						
< € 2,200	594 (35%)	799 (48%) ***	802 (28%)	591 (48%)***	1061 (38%)	332 (57%)***
≥ € 2,200	1102 (65%)	864 (52%)	1320 (63%)	650 (52%)	1718 (62%)	252 (43%)
Smoking during pregnancy (n; %)	235 (14%)	301 (17%)**	349 (16%)	187 (14%)	419 (15%)	117 (18%)
Alcohol use during pregnancy (n; %)	782 (45%)	695 (39%)***	918 (44%)	519 (39%)**	1262 (44%)	215 (32%)***
Body Mass Index before pregnancy (Mean, SD)	23.3 (4.0)	23.9 (4.3)***	23.7 (4.3)	23.5 (4.0)	23.4 (1.6)	24.3 (4.9)**
Fever in 3rd trimester of pregnancy (n; %)	22 (7%)	17 (7%)	28 (7%)	11 (6%)	33 (7%)	6 (8%)
Maternal atopy (n; %)	670 (35%)	628 (33%)	845 (35%)	453 (32%)*	1069 (34%)	229 (32%)
Family history of asthma atopy (n; %)	1018 (50%)	937 (46%)*	1263 (49%)	692 (46%)*	1611 (49%)	344 (45%)
Parity						
0	1198 (55%)	1172 (54%)***	495 (19%)	373 (25%)*	1947 (57%)	423 (51%)*
1	661 (32%)	677 (31%)	794 (31%)	440 (30%)	1072 (31%)	266 (32%)
2	192 (9%)	257 (12%)	582 (23%)	312 (21%)	345 (10%)	104 (12%)
23	36 (2%)	76 (4%)	689 (27%)	368 (25%)	95 (2%)	51 (6%)
Mode of delivery (n; %)						
Vaginal	1342 (71%)	1468 (75%)	1746 (74%)	1066 (72%)	2240 (72%)	572 (76%)
Forceps or vacuum assisted	292 (16%)	240 12%)	305 (13%)	228 (15%)	449 (15%)	84 (11%)
Caesarian section	249 (13%)	255 (13%)	315 (13%)	190 (13%)	408 (13%)	97 (13%)
Premature rupture of membranes (n; %)	96 (5%)	85 (4%)	100 (4%)	81 (5%)	148 (4%)	33 (4%)
Table 1. continues on next page						

teristics of the study population (n=4.464) 1.1 7 te M

Chapter 4

Table 1. continued						
	ш	BV	υ	MV	SH	SV-1
	EBV - (n=2,185)	EBV + (n=2,279)	CMV - (n=2,772)	CMV + (n=1,692)	HSV-1 - (n=3,592)	HSV-1 + (n=877)
Infant characteristics (n=258)				-		
Males (n; %)	1121 (51%)	1183 (52%)	1469 (53%)	836 (49%)*	1855 (52%)	450 (51%)
Western ethnicity (n;%)	1598 (74%)	1266 (58%)***	1991 (73%)	877 (54%)***	2426 (70%)	407 (49%)***
Gestational age (Mean \pm SD; weeks)	39.7 (1.9)	39.8 (1.8)	39.8 (1.8)	39.8 (1.8)	39.8 (1.8)	39.7 (1.8)*
Preterm birth (<37 weeks) (n; %)	159 (7%)	140 (6%)	181 (7%)	119 (7%)	246 (7%)	54 (6%)
Birth weight (Mean \pm SD; grams)	3,427 (584)	3,401 (553)	3427 (576)	3391 (555)*	3,424 (523)	3,369(549)*
Day-care in first year (n; %)	940 (88%)	773 (88%)	1069 (86%)	646 (92%)***	1493 (89%)	222 (84%)*
Breastfeeding (n; %)						
Never	140 (9%)	124 (9%)	211 (11%)	53 (5%)	215 (9%)	49 (10%)
< 3 months	518 (34%)	472 (34%)	650 (35%)	341 (33%)	812 (34%)	179 (36%)
3-6 months	305 (20%)	313 (23%)	395 (21%)	224 (22%)	524 (22%)	95 (19%)
>6 months	565 (37%)	479 (35%)	626 (33%)	421 (40%)	870 (36%)	177 (35%)
Infections first year (n; %)						
RTI	877 (58%)	808 (61%)	1054 (58%)	631 (63%)**	1404 (60%)	281 (61%)
LRTI	190 (13%)	182 (15%)	241 (14%)	131 (14%)	314 (14%)	58 (14%)
URTI	810 (52%)	758 (54%)	975 (51%)	593 (56%)**	1296 (52%)	272 (55%)
GI	530 (57%)	473 (61%)	623 (56%)	380 (64%)**	833 (58%)	170 (61%)
Atopy first year (n; %)¶¶						
Eczema	440 (29%)	351 (26%)	486 (26%)	305 (30%)*	678 (28%)	113 (24%)
Wheezing (<u>></u> 1 episode)	545 (44%)	520 (48%)*	680 (45%)	385 (48%)	884 (45%)	181 (48%)
Asthma diagnosis at age 6 (n; %)	133 (7%)	118 (7%)	168 (8%)	83 (7%)	209 (7%)	42 (7%)
Age (Median <u>+</u> range; years)	6.1 (5.0-9.0)	6.2 (4.9-9.1)***	6.2 (5.0-9.0)	6.2 (4.9-9.1)***	6.2 (4.9-9.0)	6.2 (5.0-9.1)***
BMI for age (mean; sd; SDS)	0.21 (0.87)	0.34 (0.94)***	0.25 (0.90)	0.32 (0.92)**	0.25 (0.89)	0.41 (0.97)***
Values are means (SD), absolute numbers (percentages)	or *medians (rang	je). Significantly diffe	rent from not infe	ected *(p<0.05), ** (p<0.01) *** (p<	0.001)
Table is based on observed dataset.						
Western ethnicity includes children from Dutch, Europe	an, American West	cern, and Asian Weste	rn origin. Non-W	estern ethnicity includ	led children from	Turkish,

Determinants of herpesviridae

Moroccan, Cape Verdean, Antillean, Surinamese-Creole and Surinamese-Hindustani origin.

Chapter 4

 $(n=5 \text{ imputations}).^{48}$ Regression coefficients were pooled by taking the average of the coefficients of the 5 imputed datasets. The pooled results of the 5 imputed datasets were reported in this paper as Odds Ratios (ORs) and 95% confidence intervals (CI's). A 95% CI was calculated around the percentage attenuation using a bootstrap method with 1,000 resamplings per imputed dataset in the statistical program R.⁴⁹ All other statistical analyses were performed in SPSS 20.0 for Windows (SPSS Inc., Chicago, II).

RESULTS

Maternal and child characteristics related to CMV, EBV and HSV-1 are shown in (Table 1). Within the total group of 4464 children (median age 6.0 years), n=1,692 (38%) were infected with CMV, n=2,279 (51%) were infected with EBV, and n=877 (20%) were infected with HSV-1. Thirty percent (n=1,319) were seronegative for CMV, EBV and HSV-1, 39% (n=1,735) were infected with one of these viruses, and 32% (n=1,410) were infected with two or more viruses. Of children with available data on herpesvirus serology, a total of 2,368 children were Native Dutch, 181 were Cape Verdean, 247 Moroccan, 109 Dutch-Antillean, 131 Suriname-Creole, and 136 children were of Surinamese-Hindustani origin (Figure 1).

DETERMINANTS OF CMV, EBV AND HSV-1 SEROPOSITIVITY

CMV seropositive children were more likely of non-Western ethnicity (aOR 2.16; 95% CI 1.81-2.57), to be girls (aOR 1.18, 95% CI 1.02, 1.36), and to be breastfed for 6 months or longer (aOR 1.96; 95% CI 1.25-3.07) than CMV seronegative children. Multiparity was inversely associated with CMV seropositivity (aOR 0.76; 95% CI 0.65-0.90).



Figure 1. Seroprevalence of herpesvirus infections in childhood according to maternal ethnic background. Values are seroprevalences of CMV, EBV, and HSV-1 infections (%) per ethnicity category. CMV, EBV, and HSV-1 seroprevalences of Cape Verdean, Moroccan, Dutch Antillean, Turkish, Surinamese-Creole, and Surinamese-Hindustani children are compared to Dutch children. Footnotes: \P p-value <0.05, #P-value <0.01, *P-value <0.001 based on χ^2 test.

Similar to CMV, non-Western ethnicity was an independent risk factor for EBV (aOR 1.76; 95% CI 1.48-2.09) and HSV-1 (aOR 1.52; 95% CI 1.39-1.66). In contrast, EBV and HSV-1 seropositive children more often had 2 or more siblings (aOR EBV parity ≥ 2 1.35; 95% CI 1.05-1.740, aOR HSV-1 parity ≥ 2 1.31; 95% CI 1.14-1.50), than EBV and HSV-1 seronegative children (in contrast with the observations for CMV). Daycare and socio-economic position were only associated with HSV-1 seropositive children were lower educated (aOR 1.55, 95% CI 1.35, 1.79) and their family household income was lower (aOR 1.11, 95% CI 1.01-1.23), than mothers of HSV-1 seroprevalence of herpes virus infections and net household income more in detail strengthened the inverse associations between socio-economic position and herpesvirus seroprevalences (Table 8; online).

ETHNIC BACKGROUND AND CMV SEROPOSITIVITY

Children of Cape Verdean, Moroccan, Turkish, Surinamese-Creole and Surinamese Hindustani mothers were more often seropositive for CMV, than children of native Dutch mothers. CMV seroprevalences in the non-native Dutch groups ranged between 44-65%, compared to 29% in the native Dutch group (Figure 1). To explain the observed ethnic differences in CMV seroprevalences, the following explanatory variables were selected: maternal educational level, parity, daycare and breastfeeding duration,⁴⁷ (Table 6 and 7; online). However, after including all the explanatory factors in the model, differences in CMV seroprevalences compared to Dutch children increased with 55% (95% CI 24 to 124) for Surinamese-Hindustani children, 36% (95% CI 19 to 60) for Moroccan children, 34% (95% CI 14 to 66) for Surinamese-Creole children, 33% (95% CI 13 to 59) for Turkish children, and 23% (95% CI 5 to 47) for Cape Verdean children (Table 2).

ETHNIC BACKGROUND AND EBV SEROPOSITIVITY

Children in non-native Dutch groups were more often EBV seropositive than native Dutch children. EBV seroprevalences in the non-native Dutch minority groups ranged between 59-69%, compared to 43% in the native Dutch group (Figure 1). Maternal educational level, net household income and parity were selected to explain ethnic differences in EBV seroprevalences,⁴⁷ (Table 6 and 7; online). Socioeconomic position was the most important contributor to the association between ethnicity and EBV seroprevalence. For example, net household income explained 22% (95% CI -41 to -6) of the differences in EBV seroprevalences between Cape Verdean and native Dutch children, followed by maternal educational level, which explained 18% (95% CI -31 to -7) (Table 3). Overall, 39% (95% CI -66 to -19) of the difference in EBV seroprevalences prevalence between Cape Verdean and Dutch children was explained by differences in socio-economic position and family size. These factors explained 36% (95% CI -55 to -19) of the differences in EBV seroprevalences between Moroccan and Dutch children, 33% (95% CI -48 to -17) of the differences in EBV seroprevalences between Turkish and Dutch children, and 25% (95% CI -60 to -3) of the differences in EBV seroprevalences between Antillean and Dutch children (Table 3).

	1	_	
	0	D	
	ŧ	5	
	2	ב	
	(σ	
	2	=	
ľ	٢	٦.	

4

	Surinam	ese-	Surinan	nese-	Turkis	sh °	Moroco	an ^d	Cap	a	Antille	ean f
	Creole	s ^a Att.	Hindusi OR	ani ° Att.	OR	Att.	OR	Att.	Verde	an ° Att.	OR	Att.
CMV	_	-										
Model 1 (includes child sex and current age)	2.29 (1.61,3.28)		1.92 (1.35,2.72)		4.41 (3.47,5.62)		3.64 (2.77,4.79)		3.17 (2.33,4.32)		n.s.	NA
Socioeconomic status												
Model 1+ education	2.72 (1.87,3.88)	+33% (17, 60) ***	2.26 (1.56,3.19)	+37% (19, 80) ***	5.36 (4.01,6.76)	+28% (14,44) ***	4.23 (3.17,5.61)	+22% (11,36) ***	3.70 (2.69,5.12)	+24% (12,40) ***		
Crowding												
Model 1 + parity	2.29 (1.60,3.28)	0% (-10,10)	1.89 (1.33,2.71)	-3% (-20,14)	4.49 (3.51,5.74)	+2% (-4,9)	3.94 (2.97,5.24)	+11% (0.5,24) *	3.14 (2.29,4.32)	-1% (-12,11)		
Model 1 + daycare	2.37 (1.62,3.32)	+6% (1,14) **	2.14 (1.49,3.07)	+25% (8,58) **	5.08 (3.84,6.42)	+20% (7,36) **	3.90 (3.00,5.40)	+10% (3,19) ***	3.15 (2.28,4.24)	-1% (-3,2)		
Model 1 + breastfeeding	2.35 (1.63,3.33)	+5% (0.4,12) *	1.99 (1.37,2.79)	+8% (2,9) **	4.24 (3.33,5.43)	-5% (-8,-2) **	3.62 (2.70,4.69)	-1% (-4,1)	3.33 (2.45,4.58)	+8% (2,15) **		
Fully adjusted model	_	-				-				_		
	2.73 (1.81,3.83)	+34% (14,66) ***	2.42 (1.62,3.42)	+55% (24,124) ***	5.53 (3.98,6.97)	+33% (13,59) **	4.60 (3.36,6.31)	+36% (19,60) ***	3.68, (2.6-,5.07)	+23% (5,47) **		
Abbreviations: OR, Odds Rati ^{a -f} The OR's represent the atte the attenuations of effect esti	o; CI, confiden enuated effect imates and the	ice interva estimate ir 95% CI	al. and their 95 ⁶ is for the diff	% CIs rela erent ethn	tive to model icities relativ	1 after in e to mode	idividual adju I 1 (includes	istment fc confound	r explanatory ers) after adi	r factors. A ustment fc	vtt. repres or explana	sent atorv

variables (100x (OR model 1. OR model 1. OR model 1. OR model 1. (OR m

.

.

. . .

. . .

	Suriname Creoles	ese-	Surinam Hindust	ese- ani ^b	Turki	sh °	Moroco	an ^d	Cap Verde	e an °	Antille	ean f
	OR	Att.	OR	Att.	OR	Att.	OR	Att.	OR	Att.	OR	Att.
EBV												
Model 1 (includes child sex and current	1.77 (1.24,2.54)		2.85 (1.96,4.13)		2.69 (2.11,3.44)		2.40 (1.82,3.17)		2.18 (1.59,2.99)		1.98 (1.33,2.96)	
socioeconomic stat	SN.											
Model 1+ education	1.66 (1.15,2.40) (-	-15% -43,0.2)	2.71 (1.85,3.95)	-7% (-17,3)	2.37 (1.84,3.08)	-19% (-31,-7) **	2.19 (1.63,2.91)	-15% (-26,-6) **	1.97 (1.41,2.70)	-18% (-31,-7) **	1.86 (1.24,2.80)	-12% (-32,-0.6) *
Model 1+ income	1.63 (1.13,2.36) (·	-19% -50,-5) *	2.60 (1.80,3.86)	-13% (-24,-4) *	2.36 (1.86,3.16)	-19% (-33,-5) *	2.11 (1.60,2.91)	-21% (-36,-6) *	1.92 (1.41,2.74)	-22% (-41,-6) *	1.77 (1.19,2.73)	-22% (-50,-6) *
Crowding												
Model 1 + parity	1.80 (1.25,2.59) (+4% (-9,24)	2.86 (1.96,4.19)	1% (-10,16)	2.58 (2.02,3.31)	-6% (-13,0.7)	2.21 (1.66,2.94)	-14% (-25,-4)*	2.03 (1.47,2.81)	-12% (-28,1)	1.97 (1.32,2.96)	-1% (-12,14)
Fully adjusted mod	e											
	1.61 (1.12,2.37) (-21% (-56,3)	2.59 (1.77,3.87)	-14% (-30,5)	2.14 (1.67,2.88)	-33% (-48,-17) ***	1.89 (1.42,2.61)	-36% (-55,-19) ***	1.72 (1.23,2.44)	-39% (-66,-19) ***	1.74 (1.17,2.69)	-25% (-60,-3) *
Abbreviations: OR, O ^{a-f} The OR's represent the attenuations of ef variables (100x (OR	dds Ratio; CI, (the attenuatec fect estimates OR	confiden J effect ∈ and the	ce interval. estimate and ir 95% CIs fo	their 95% r the diffe	 CIs relative erent ethniciti of the relative 	to model : es relative ference ar	L after individ to model 1 ()	ual adjustr includes co	ment for expl infounders) a	anatory fa Ifter adjust < 0.001)	ictors. Att. re tment for exp	oresent Ilanatory

seropositivity for different ethnicities after individual adjustment for the explanatory Table 3. Attenuation of the OR of Epstein Barr virus

ETHNIC BACKGROUND AND HSV-1 SEROPOSITIVITY

All ethnic minority groups, except for Surinamese-Creole, were more often HSV-1 seropositive than native Dutch children. HSV-1 seroprevalences in the nonnative Dutch groups ranged between 22-39%, compared to 13% in the native Dutch group (Figure 1). To explain ethnic differences in HSV-1 seroprevalences, maternal educational level, net household income and breastfeeding duration were selected as explanatory variables,⁴⁷ (Table 6 and 7; online). Maternal educational level and net household income were the most important factors to explain ethnic differences in HSV-1 seroprevalence. For example, 27% (95% CI -38 to -16) of this differences in HSV-1 seroprevalences between Turkish and Dutch children was explained by maternal educational level (Table 4). Socio-economic position and breastfeeding together explained 24% (95% CI -40 to -7) of the difference in HSV-1 seroprevalence between Turkish and Dutch children; they explained 48% (95% CI -150 to -18) of the differences in HSV-1 seroprevalences between Cape Verdean and Dutch children; 30% (95% CI -68 to -10) of the differences in HSV-1 prevalence between Surinamese-Hindustani and Dutch children and 22% (95% CI -36 to -5) of the differences in HSV-1 seroprevalence between Moroccan and Dutch children. Maternal educational level was the most important explanatory factor in the association between ethnicity and HSV-1 seroprevalence (Table 4).

Next, we observed that children who were seropositive for a single virus, were more likely to be infected with multiple herpesviruses. For example, the odds for CMV seropositivity was 1.69 times greater, and the odds for HSV-1 seropositivity was 2.17 times greater in EBV infected children than in EBV seronegative children. Also, the odds for both CMV and HSV-1 seropositivity was almost 3 times greater in EBV seropositive children than in EBV seronegative children.

Also, CMV, EBV and HSV-1 seropositive children tended to have more frequently 'first' generation mothers, than seronegative children, although differences were not significant (Figure 2). Differences in characteristics between seronegative children and children with multiple herpes virus infections are described in Table 10; online. Mothers of seronegative children were more often higher educated, had a higher net household income, than mothers of children who had 1 or 2 herpes virus infections. Seronegative children were more often of Western origin, and had fewer siblings, than children who had 1 or 2 herpes virus infections (Table 10; online).

DISCUSSION

In this large and multi-ethnic population-based cohort, we found 1) large ethnic differences in seroprevalences of EBV (range 43-69%), CMV (range 29-65%) and HSV-1 (range 13-39%) among 6 year old children 2) the highest seroprevalence in non-Dutch children. Socio-economic factors and factors related to crowding early in life only partly explained ethnic differences in EBV and HSV-1 seroprevalence in childhood, but did not explain ethnic differences in CMV seroprevalence.

Our study design provided a unique opportunity to study determinants of these infections, to investigate variation in seroprevalence by ethnic group, and to study to which extent socio-economic and lifestyle factors explained these differences. Our results confirm and extend previous studies by observing that socio-economic position partly explains ethnic variation in EBV and HSV-1 seroprevalences, but not in CMV seroprevalences.^{12,35,42} Although these three viruses are all members of the

Table 4. Attenuation of t variables	the OR of H	erpes s	simplex virus 1	seroprevalence	for diff	erent ethnic	ities afteı	· individual a	ıdjustmen	t for the exp	lanatory
	Surinames Creoles ^a		Surinamese- Hindustani ^b	Turkish	U	Morocc	an ^d	Cap Verde	e an °	Antille	an ^f
	OR Att		OR Att.	OR	Att.	OR	Att.	OR	Att.	OR	Att.
HSV-1											
Model 1 (includes child sex and current age)	ns NA	(1.40	2.10 3, 3.16)	4.08 (3.18,5.24)		3.73 (2.80,4.98)		1.75 (1.20,2.54)		1.70 (1.06,2.73)	
Socioeconomic status											
Model 1+ education		1 (1.1)	L.82 -26% 9,2.73) (-6,-10) **	3.24 - (2.49,4.26) (- 3	27% 88,-16) ***	3.15 (2.28,4.17) (-21% (-30,-12) ***	1.45 (0.95,2.07) (-39% -119,-19) **	1.47 (0.91,2.36)	-33% (-160,-7) *
Model 1+ income		1 (1.2 [,]	l.92 - 17% 4,2.87) (-4,-0.1 *	3.58 - (2.66,4.68) (-	16% 30,-1) *	3.29 (2.35,4.42) (-16% -30,-0.5) *	1.54 (1.01,2.24)	-28% (-92,1)	1.52 (0.91,2.43)	-26% (-115,11)
Crowding											
Model 1 + breastfeeding		2 (1.4	2.04 -6% 0,3.18) (-14,-1) *	4.29 (3.29,5.50) (+7% 2,14) **	3.80 (2.89,5.16)	+2% (0,6)	1.67 (1.16,2.47)	-10% (-32,-2) *	1.65 (1.02,2.65)	-8% (-43,2)
Fully adjusted model											
		1 (1.1	1.78 -30% 7,2.75) (-68,-10	3.33 - 3.33 - (2.43,4.43) (-	24% 40,-7) **	3.14 (2.21,4.26)	-22% (-36,-5) *	1.39 (0.90,2.02) (-48% -150,-18) *	1.42 (0.85,2.30)	-40% (-213,9)
Abbreviations: OR, Odds R: ^{a-f} The OR's represent the ar the attenuations of effect e	atio; CI, con ttenuated eff stimates and	fidence fect esti 1 their 9	interval. imate and their 9. 35% CIs for the d	5% CIs relative to ifferent ethnicitie	o model s relativ	1 after indivi e to model 1	dual adjust (includes c	ment for expl onfounders) a	anatory fac	ctors. Att. rep ment for expl	resent anatory
variables (100x (OR model 1 - No significant difference in estimates for the Suriname	OR model 1 with ex HSV-1 preva se-Creole gr	alence w oup are	ctor) / (OR model 1)). vas observed for t not presented. *	Dutch is the refe the Surinamese-C p<0.05, ** p<0	erence g Creole gr 0.01, **:	oup. oup compare * p<0.001	d with the	Dutch group,	therefore a	attenuations i	ו effect

Determinants of herpesviridae



Figure 2. Percentages of first generation mothers according to CMV, EBV or HSV-1 seropositivity in childhood.



herpesvirus family, and thus share significant biological properties, transmission dynamics between these viruses seem to be different. Our results suggest that ethnic differences in CMV seroprevalences in childhood are mainly determined by factors that facilitate vertical transmission (e.g. pregnant women that transmit CMV to their child), which is in line with the idea that the most important source for CMV infection in childhood is breastmilk.⁵⁰⁻⁵² In contrast, ethnic differences in EBV and HSV-1 seroprevalences are partly determined by factors that facilitate horizontal transmission, such as family size and daycare, which are strongly related to socio-economic position. In fact, we were able to explain up to 39% of the ethnic differences in EBV seroprevalences, of which socio-economic position was the most important contributor, which is in line with a recent study.⁵³ However, in contrast with our study, the authors suggested that ethnic differences could not be explained by crowding. However, it could be speculated that socio-economic position reflects more proximal determinants (e.g. factors that facilitate horizontal transmission, such as family size, bed sharing and childcare arrangements). Indeed, the results from our multivariable model confirm that family size is important in estimating the risk of EBV infection.¹⁸ In fact, we observed a doseresponse relationship between parity and the risk of EBV and HSV-1 seropositivity.

In contrast with the results on EBV and HSV-1, ethnic differences in CMV seroprevalences could not be explained by socio-economic position, nor by other 'explanatory variables' such as breastfeeding, daycare and parity. In fact, these factors rather strengthened than weakened the association between ethnicity and CMV. We expected breastfeeding to partly mediate this association, because breastfeeding can transmit CMV, and breastfeeding behavior is influenced by ethnic background.⁵⁴ Nevertheless, vertical transmission from mother to child is probably more important in explaining ethnic differences in CMV seroprevalences than factors that facilitate horizontal transmission. Given the economic burden of CMV associated diseases, it has been suggested that vaccination in the general population would be the most straightforward way to control for all CMV

associated diseases.³ Some recommend to include 12-year old children, whereas others suggested to include vaccination of toddlers.³³ ⁵⁵ However, thus far, safe and effective vaccines have not been developed. Nevertheless, on the basis of our results, ethnic and socio-economic differences should be taken into account when vaccination is contemplated.

HSV-1 prevalence was highest in Turkish and Moroccan children, which is in line with a previous Dutch study in adolescents.⁴² Our results confirm that differences in HSV-1 seroprevalences between European countries could be partly explained by socio-economic factors.²² It has been shown that lower socio-economic position indeed increases the risk for HSV-1.⁴² All factors together, including socio-economic position, breastfeeding and crowding, explained up to 48% of ethnic differences in HSV-1 seroprevalence, of which socio-economic position was the most important contributor, suggesting an important role for factors that facilitate horizontal transmission.

METHODOLOGICAL CONSIDERATIONS

An important strength of this study is that we had a large multi-ethnic cohort study population drawn from the general population of Rotterdam. Because immigration is common in Western countries, our findings may be more broadly applicable. In addition, our study design enabled us to assess multiple risk factors, and to study the relative importance of each using multivariate models. Moreover, an explanation of ethnic differences in herpesvirus infections using a mediator approach has never been performed before. Besides socio-economic position, numerous other factors vary between countries, and adjustment for these factors might decrease ethnic/racial or geographical variation.¹² Because our study was performed within a population based study of a region within a single country, we were able to adjust for all of these variables. A limitation of this study is that we did not have data on maternal CMV, EBV, HSV-1 seroprevalences, preventing analysis of mother-to-child transmission. A second limitation was lack of data on EBV, CMV and HSV-1 seroprevalences in fathers and siblings, precluding examination of their roles in horizontal transmission.

CONCLUSION

In this multiethnic population-based cohort, we found large ethnic differences in seroprevalences of CMV, EBV and HSV-1 among 6 year old children, with higher seroprevalences in non-Dutch children. Socio-economic factors and factors related to crowding early in life partly explained ethnic differences in EBV and HSV-1 seroprevalence, but did not explain ethnic differences in CMV seroprevalence. When immunization against these viruses is considered, ethnic and socio-economic differences should be taken into account.

CONTRIBUTORS' STATEMENT

Michelle A.E. Jansen contributed to the design of the study, completed the background literature research, data collection, performed the analyses, wrote the first draft of the manuscript, and collated comments from other authors. Diana van den Heuvel contributed to the design of the study, data collection,

immunological data analyses, data interpretation and writing of the manuscript. Selma H. Bouthoorn contributed to the data collection, statistical analyses, data interpretation and writing of the manuscript. Vincent W.V. Jaddoe conceptualized and designed the study and contributed to the writing of the manuscript. Herbert Hooijkaas contributed to the design of the study, interpretation of data analysis, and writing of the manuscript. Hein Raat contributed to the design of the study, interpretation of data analysis, and writing of the manuscript. Pieter L.A. Fraaij contributed to the design of the study, interpretation of data, and writing of the manuscript Menno C. van Zelm contributed to the design of the study, immunological data analyses, data interpretation, and writing of the manuscript. Henriette A. Moll obtained funds, designed the study, supervised the project, contributed to data interpretation and writing of the manuscript. All authors have access to all of the data, take responsibility for the integrity of the data, the accuracy of the data analysis and approved the final version of the manuscript, MAEJ and HAM are guarantors of the study and accept full responsibility for the work and the conduct of the study. All authors confirm that neither this manuscript nor any part of it has been published or is being considered for publication elsewhere.

REFERENCES

- 1. Crough T, Khanna R. Immunobiology of human cytomegalovirus: from bench to bedside. *Clin Microbiol Rev.* 2009; 22(1): 76-98, Table of Contents.
- 2. Boppana SB, Fowler KB. Persistence in the population: epidemiology and transmisson. 2007.
- 3. Griffiths PD. Burden of disease associated with human cytomegalovirus and prospects for elimination by universal immunisation. *Lancet Infect Dis.* 2012; 12(10): 790-798.
- 4. Griffiths P, Baraniak I, Reeves M. The pathogenesis of human cytomegalovirus. J Pathol. 2014.
- Simanek AM, Dowd JB, Pawelec G, Melzer D, Dutta A, Aiello AE. Seropositivity to cytomegalovirus, inflammation, all-cause and cardiovascular disease-related mortality in the United States. *PLoS One*. 2011; 6(2): e16103.
- 6. Dreyfus DH. Herpesviruses and the microbiome. J Allergy Clin Immunol. 2013; 132(6): 1278-1286.
- 7. Longnecker R, Neipel F. Introduction to the human gamma-herpesviruses. 2007.
- Sidorchuk A, Wickman M, Pershagen G, Lagarde F, Linde A. Cytomegalovirus infection and development of allergic diseases in early childhood: interaction with EBV infection? J Allergy Clin Immunol. 2004; 114(6): 1434-1440.
- 9. Hjalgrim H, Friborg J, Melbye M. The epidemiology of EBV and its association with malignant disease. 2007.
- Ogwang MD, Bhatia K, Biggar RJ, Mbulaiteye SM. Incidence and geographic distribution of endemic Burkitt lymphoma in northern Uganda revisited. *Int J Cancer*. 2008; 123(11): 2658-2663.
- 11. Stefan DC, Lutchman R. Burkitt lymphoma: epidemiological features and survival in a South African centre. *Infect Agent Cancer*. 2014; 9(19.
- Flavell KJ, Biddulph JP, Powell JE, Parkes SE, Redfern D, Weinreb M, et al. South Asian ethnicity and material deprivation increase the risk of Epstein-Barr virus infection in childhood Hodgkin's disease. *Br J Cancer*. 2001; 85(3): 350-356.
- 13. Bagni R, Whitby D. Age of infection and risk of virally associated cancers: new clues to an old puzzle. *J Infect Dis.* 2012; 205(6): 873-874.
- 14. Hsu JL, Glaser SL. Epstein-barr virus-associated malignancies: epidemiologic patterns and etiologic implications. *Crit Rev Oncol Hematol*. 2000; 34(1): 27-53.
- Grillner L, Strangert K. A prospective molecular epidemiological study of cytomegalovirus infections in two day care centers in Sweden: no evidence for horizontal transmission within the centers. J Infect Dis. 1988; 157(5): 1080-1083.
- 16. Aarnisalo J, Ilonen J, Vainionpaa R, Volanen I, Kaitosaari T, Simell O. Development of antibodies against cytomegalovirus, varicella-zoster virus and herpes simplex virus in Finland during the first eight years of life: a prospective study. *Scand J Infect Dis.* 2003; 35(10): 750-753.

- 17. Balfour HH, Jr., Sifakis F, Sliman JA, Knight JA, Schmeling DO, Thomas W. Age-specific prevalence of Epstein-Barr virus infection among individuals aged 6-19 years in the United States and factors affecting its acquisition. *J Infect Dis.* 2013; 208(8): 1286-1293.
- Dowd JB, Palermo T, Brite J, McDade TW, Aiello A. Seroprevalence of Epstein-Barr virus infection in U.S. children ages 6-19, 2003-2010. *PLoS One*. 2013; 8(5): e64921.
- 19. Mertz GJ, Rosenthal SL, Stanberry LR. Is herpes simplex virus type 1 (HSV-1) now more common than HSV-2 in first episodes of genital herpes? *Sex Transm Dis*. 2003; 30(10): 801-802.
- 20. Smith JS, Robinson NJ. Age-specific prevalence of infection with herpes simplex virus types 2 and 1: a global review. J Infect Dis. 2002; 186 Suppl 1(S3-28.
- 21. Whitley RJ, Kimberlin DW, Roizman B. Herpes simplex viruses. *Clin Infect Dis.* 1998; 26(3): 541-553; quiz 554-545.
- 22. Pebody RG, Andrews N, Brown D, Gopal R, De Melker H, Francois G, et al. The seroepidemiology of herpes simplex virus type 1 and 2 in Europe. *Sex Transm Infect*. 2004; 80(3): 185-191.
- 23. Wald A, Corey L. Persistence in the population: epidemiology, transmission. 2007.
- 24. Joseph SA, Beliveau C, Muecke CJ, Rahme E, Soto JC, Flowerdew G, et al. Cytomegalovirus as an occupational risk in daycare educators. *Paediatr Child Health*. 2006; 11(7): 401-407.
- Staras SA, Flanders WD, Dollard SC, Pass RF, McGowan JE, Jr., Cannon MJ. Cytomegalovirus seroprevalence and childhood sources of infection: A population-based study among pre-adolescents in the United States. J Clin Virol. 2008; 43(3): 266-271.
- Cannon MJ, Schmid DS, Hyde TB. Review of cytomegalovirus seroprevalence and demographic characteristics associated with infection. *Rev Med Virol*. 2010; 20(4): 202-213.
- 27. Schuster V, Kreth HW. Epstein-Barr virus infection and associated diseases in children. I. Pathogenesis, epidemiology and clinical aspects. *Eur J Pediatr*. 1992; 151(10): 718-725.
- 28. Kangro HO, Osman HK, Lau YL, Heath RB, Yeung CY, Ng MH. Seroprevalence of antibodies to human herpesviruses in England and Hong Kong. *J Med Virol*. 1994; 43(1): 91-96.
- 29. Leogrande G, Jirillo E. Studies on the epidemiology of child infections in the Bari area (south Italy). VII. Epidemiology of Epstein-Barr virus infections. *Eur J Epidemiol*. 1993; 9(4): 368-372.
- 30. Wang PS, Evans AS. Prevalence of antibodies to Epstein-Barr virus and cytomegalovirus in sera from a group of children in the People's Republic of China. *J Infect Dis.* 1986; 153(1): 150-152.
- Svahn A, Berggren J, Parke A, Storsaeter J, Thorstensson R, Linde A. Changes in seroprevalence to four herpesviruses over 30 years in Swedish children aged 9-12 years. *J Clin Virol*. 2006; 37(2): 118-123.
- 32. Stagno S, Cloud GA. Working parents: the impact of day care and breast-feeding on cytomegalovirus infections in offspring. *Proc Natl Acad Sci U S A*. 1994; 91(7): 2384-2389.
- Pembrey L, Raynor P, Griffiths P, Chaytor S, Wright J, Hall AJ. Seroprevalence of cytomegalovirus, Epstein Barr virus and varicella zoster virus among pregnant women in Bradford: a cohort study. *PLoS One*. 2013; 8(11): e81881.
- 34. Hesla HM, Gutzeit C, Stenius F, Scheynius A, Dahl H, Linde A, et al. Herpesvirus infections and allergic sensitization in children of families with anthroposophic and non-anthroposophic lifestyle the ALADDIN birth cohort. *Pediatr Allergy Immunol.* 2013; 24(1): 61-65.
- 35. Dowd JB, Aiello AE, Alley DE. Socioeconomic disparities in the seroprevalence of cytomegalovirus infection in the US population: NHANES III. *Epidemiol Infect*. 2009; 137(1): 58-65.
- 36. Jaddoe VW, van Duijn CM, Franco OH, van der Heijden AJ, van Iizendoorn MH, de Jongste JC, et al. The Generation R Study: design and cohort update 2012. *Eur J Epidemiol*. 2012; 27(9): 739-756.
- 37. Kruithof CJ, Kooijman MN, van Duijn CM, Franco OH, de Jongste JC, Klaver CC, et al. The Generation R Study: Biobank update 2015. *Eur J Epidemiol*. 2014; 29(12): 911-927.
- 38. Swertz O, Duimelaar P, Thijssen J. Migrants in the Netherlands 2004. Voorburg/Heerlen: Statistics Netherlands, 2004.
- 39. Wijtzes AI, Jansen W, Jaddoe VW, Moll HA, Tiemeier H, Verhulst FC, et al. Ethnic background and television viewing time among 4-year-old preschool children: the generation R study. *J Dev Behav Pediatr.* 2013; 34(2): 63-71.
- 40. Statistics Netherlands. Jaarrapport Integratie 2010; Den Haag/Heerlen, the Netherlands: Statistics Netherlands. 2010.
- 41. Tookey PA, Ades AE, Peckham CS. Cytomegalovirus prevalence in pregnant women: the influence of parity. *Arch Dis Child*. 1992; 67(7 Spec No): 779-783.

- Kramer MA, Uitenbroek DG, Ujcic-Voortman JK, Pfrommer C, Spaargaren J, Coutinho RA, et al. Ethnic differences in HSV1 and HSV2 seroprevalence in Amsterdam, the Netherlands. *Euro Surveill*. 2008; 13(24).
- 43. Statistics Netherlands. Welfare in the Netherlands. Income, welfare and spending of households and individuals. *Den Haag/Heerlen: Statistics Netherlands, 2012*.
- 44. Statistics Netherlands. Dutch Standard Classification of Education 2003. Voorburg/Heerlen: Statistics Netherlands, 2004.
- 45. Wijtzes AI, Jansen W, Bouthoorn SH, Pot N, Hofman A, Jaddoe VW, et al. Social inequalities in young children's sports participation and outdoor play. *Int J Behav Nutr Phys Act*. 2014; 11(155.
- 46. Netherlands Bureau for Economic Policy Analysis. http://cpb.nl.
- 47. Baron RM, Kenny DA. The moderator-mediator variable distinction in social psychological research: conceptual, strategic, and statistical considerations. *J Pers Soc Psychol.* 1986; 51(6): 1173-1182.
- 48. Sterne JA, White IR, Carlin JB, Spratt M, Royston P, Kenward MG, et al. Multiple imputation for missing data in epidemiological and clinical research: potential and pitfalls. *BMJ*. 2009; 338(b2393.
- 49. Venables WN SD. An Introduction to R. Notes on R: A Programming Environment for Data Analysis and Graphics Version 2.12.1 (2012-12-16). http://www.r-project.org
- 50. Forsgren M. Cytomegalovirus in breast milk: reassessment of pasteurization and freeze-thawing. *Pediatr Res.* 2004; 56(4): 526-528.
- 51. Hayes K, Danks DM, Gibas H, Jack I. Cytomegalovirus in human milk. N Engl J Med. 1972; 287(4): 177-178.
- 52. Ahlfors K, Ivarsson SA, Harris S. Report on a long-term study of maternal and congenital cytomegalovirus infection in Sweden. Review of prospective studies available in the literature. *Scand J Infect Dis.* 1999; 31(5): 443-457.
- 53. Condon LM, Cederberg LE, Rabinovitch MD, Liebo RV, Go JC, Delaney AS, et al. Age-specific prevalence of Epstein-Barr virus infection among Minnesota children: effects of race/ethnicity and family environment. *Clin Infect Dis.* 2014; 59(4): 501-508.
- 54. Kelly YJ, Watt RG, Nazroo JY. Racial/ethnic differences in breastfeeding initiation and continuation in the United kingdom and comparison with findings in the United States. *Pediatrics*. 2006; 118(5): e1428-1435.
- 55. Plotkin SA. Vaccination against cytomegalovirus, the changeling demon. *Pediatr Infect Dis J.* 1999; 18(4): 313-325; quiz 326.

Determinants Gender Male	(aOR 95% CI)	(aOR 95% CI)	(aOR 95% CI)
Gender Male	1.0	1.0	
Male	1.0	1.0	
	1 10 (1 02 1 26)*	1.0	1.0
Female	1.10 (1.02, 1.30)*	0.99 (0.87, 1.15)	1.05 (0.97, 1.13)
Ethnicity			
Western	1.0	1.0	1.0
Non western	2.16 (1.81, 2.57)***	1.76 (1.48, 2.09)***	1.52 (1.39, 1.66)*
Income			
Low (<2200)	1.18 (0.98, 1.42)	1.09 (0.91, 1.30)	1.11 (1.01, 1.23)*
High (<u>></u> 2200)	1.0	1.0	1.0
Maternal educational level			
Low	0.95 (0.73, 1.23)	1.09 (0.85, 1.41)	1.55 (1.35, 1.79)***
Mid-low	0.82 (0.66, 1.01)	0.93 (0.76, 1.15)	1.19 (1.05, 1.35)**
Mid-high	0.96 (0.77, 1.19)	0.94 (0.77, 1.15)	1.16 (1.03, 1.31)*
High	1.0	1.0	1.0
Daycare	1.20 (1.00, 1.43)	1.06 (0.86, 1.31)	0.89 (0.81, 0.97)*
Breastfeeding duration			
Never	1.0	1.0	1.0
<3 months	1.53 (1.04, 2.23)*	0.98 (0.73, 1.33)	1.02 (0.89, 1.16)
3-6 months	1.77 (1.21, 2.60)**	1.20 (0.85, 1.69)	1.08 (0.93, 1.26)
>6 months	1.96 (1.25, 3.07)**	1.08 (0.78, 1.85)	1.06 (0.92, 1.22)
Parity			
0	1.0	1.0	1.0
1	0.76 (0.65, 0.90)**	1.11 (0.95, 1.30)	1.18 (1.08, 1.29)***
2	0.90 (0.69, 1.18)	1.35 (1.05, 1.74)*	1.31 (1.14, 1.50)***
<u>></u> 3	0.85 (0.54, 1.33)	1.58 (1.01, 2.48)*	1.60 (1.29, 1.99)***

Abbreviations: aOR, adjusted Odds Ratio; CI, confidence interval. Values reflect aOR's (95% CI). Table is based on imputed data set. *P<0.05, ** P<0.01, *** P<0.001. Models were adjusted for age, gender, socio-economic position (ethnicity, income, maternal educational level), crowding (daycare, breastfeeding, parity) and maternal lifestyle (maternal age, smoking and alcohol use during pregnancy, maternal BMI).

Ethnic background	Dutch	Surinamese-	Surinamese-	Turkish	Moroccan	Cape Verdean	Antillean	P-value ^a
	(n=2386)	creates (n=131)	(n=136)	(n=338)	(n=247)	(n=181)	(n=109)	
Potential confounders ^b								
Age (years)	6.1 (0.4)	6.4 (0.7)	6.3 (0.6)	6.3 (0.6)	6.4 (0.7)	6.3 (0.7)	6.6 (0.9)	<0.001
Parity (%)								
Nulliparous	60.3	54.3	53.8	43.7	37.9	47.7	51.9	<0.001
Potential mediators								
Maternal educational level (%)								
Low	12.7	32.8	31.5	53.3	45.1	50.7	36.0	<0.001
Mid-low	25.7	50.4	49.6	33.0	37.9	39.5	49.0	
Mid-high	26.7	14.3	15.0	10.0	14.6	8.6	0.6	
High	34.9	2.5	3.9	3.7	2.4	1.3	6.0	
Net household income (%)								
≤ 2200 euro	25.1	70.5	70.2	87.1	88.2	91.6	83.3	<0.001
> 2200 euro	74.9	29.5	29.8	12.9	11.8	8.4	16.7	
Daycare (%yes)	89.5	83.3	70.4	57.1	65.4	92.1	82.6	<0.001
Breastfeeding duration (%)								
never	10.1	8.3	4.2	1.2	3.7	11.7	13.0	<0.001
<3 months	31.8	41.7	56.9	35.5	41.1	51.9	52.2	
3-6 months	22.7	18.1	20.8	18.7	18.7	18.2	8.7	
>6 months	35.4	31.9	18.1	44.6	36.4	18.2	26.1	
Maternal smoking (% continued)	14.0	27.2	15.2	32.3	4.9	17.8	20.4	<0.001
Maternal alcohol use (% continued)	53.4	40.4	18.6	6.8	3.0	28.5	36.3	<0.001
BMI child (kg/m²)	15.9 (1.4)	16.9 (2.7)	15.8 (2.1)	17.2 (2.3)	16.9 (2.0)	17.0 (2.3)	17.2 (2.5)	<0.001
Maternal BMI (kg/m²)	23.2 (3.9)	24.9 (4.9)	23.8 (4.5)	24.9 (4.8)	25.2 (4.3)	23.7 (3.6)	25.5 (5.6)	<0.001
Values are percentages or means (SL	D).							
^b Data was missing for maternal educa	-square test ior ational leval /0	Categorical var	Tables and ANUV	A IOF CONUMUO	us variables.	otfaading gunstin	em (%2757 u	tarnal emolino
(13.3%), maternal alcohol use (21.0%)	%). BMI child (.2 %) and mate	ernal BMI (26.19	r.o /u/, uaycare %)		סוובכטוווש טמומנוס		

	Cytomegalovi	rus	Epstein-Barr v	irus	Herpes Simplex	virus 1
	OR	P-	OR	P-	OR	P-
	(95% CI)	value	(95% CI)	value	(95% CI)	value
Potential mediators						
Parity	0.89 (0.82,0.98)	0.012	1.08 (0.99,1.17)	0.096	1.07 (0.96,1.18)	0.22
Maternal educational le	vel					
Low	0.67 (0.53,0.85)	0.001	1.34 (1.07,1.69)	0.011	1.88 (1.41,2.51)	< 0.001
Mid-low	0.66 (0.54,0.82)	<0.001	0.99 (0.81,1.21)	0.93	1.37 (1.03,1.83)	0.03
Mid-high	0.89 (0.72,1.11)	0.31	0.92 (0.75,1.13)	0.43	1.14 (0.80,1.63)	0.46
High	Ref		Ref		Ref	
Net household income						
≤ 2200 euro	0.87 (0.72, 1.06)	0.16	1.20 (1.00, 1.44)	0.057	1.29 (1.02,1.64)	0.036
> 2200 euro	Ref		Ref		Ref	
Daycare						
No	Ref		Ref		Ref	
Yes	1.38 (1.09, 1.76)	0.009	0.97 (0.76,1.23)	0.77	0.86 (0.68, 1.09)	0.20
Breastfeeding duration						
never	Ref		Ref		Ref	
<3 months	1.34 (0.94,1.90)	0.10	0.89 (0.66,1.20)	0.44	0.74 (0.51,1.08)	0.12
3-6 months	1.48 (1.07,2.06)	0.019	1.01 (0.72,1.42)	0.95	0.67 (0.43,1.03)	0.067
>6 months	1.63 (1.13,2.36)	0.012	0.83 (0.60,1.14)	0.23	0.67 (0.44,1.02)	0.061

Table 7 (online)	. Associations bet	ween potential	mediators and	herpes virus	s seropositivity
------------------	--------------------	----------------	---------------	--------------	------------------

OR=odds ratio, CI=confidence interval. Values are OR's (95% CI)

Table is based on imputed dataset (potential confounders and mediators are multiple imputed, determinant observed).

Values are derived from logistic regression models and represent effect estimates (95% confidence intervals), adjusted for gender, age and ethnic background.

Table 8 (online). Odds Ratios	s for Infec	ctions and Coi	nfections	s According to	o Socio E	conomic Positi	uo				
	CMV		EBV		HSV-1		CMV and HSV	/-1	EBV and H	SV-1	EBV and	CMV
	Odds Ratio (95% CI)	<i>P</i> Value	Odds Ratio (95% CI)	<i>P</i> Value	Odds Ratio (95% CI)	<i>P</i> Value	Odds Ratio (95% CI) P	Value	Odds Ratio (95% CI)	<i>P</i> Value	Odds Ratio (95% CI)	<i>P</i> Value
Income												
<2000	1.70 (1.45, 2.01)	<0.0001	1.92 (1.63, 2.27)	< 0.0001	2.56 (2.09, 3.12)	<0.0001	3.43 (2.60, 4.54) ^{<(}	0.0001	3.10 (2.44, 3.93)	<0.0001	2.19 (1.81, 2.64)	< 0.0001
2000-3200	0.93 (0.79, 1.10)	0.41	1.32 (1.13, 1.55)	0.001	1.52 (1.23, 1.88)	<0.0001	1.66 (1.21, 2.28)	0.002	1.58 (1.21, 2.06)	0.001	1.14 (0.93, 1.40)	0.22
> 3200	1.00 (ref)		1.00 (ref)		1.00 (ref)		1.00 (ref)		1.00 (ref)		1.00 (ref)	
Education												
Low	1.36 (1.08, 1.72)	0.011	1.92 (1.59, 2.33)	< 0.0001	2.63 (2.11, 3.27)	<0.0001	3.21 (2.37, 4.35) ^{<(}	0.0001	2.93 (2.21, 3.87)	<0.0001	2.01 (1.50, 2.68)	< 0.0001
Mid Low	0.97 (0.80, 1.19)	0.79	1.23 (1.04, 1.46)	0.02	1.47 (1.17, 1.86)	0.001	1.40 (1.02, 1.94)	0.04	1.62 (1.22, 2.15)	0.001	1.21 (0.95, 1.54)	0.13
Mid High	0.95 (0.79, 1.14)	0.58	0.94 (0.77, 1.14)	0.52	1.07 (0.77, 1.94)	0.67	1.02 (0.69, 1.51)	0.93	1.05 (0.71; 1.57)	0.79	1.12 (0.88, 1.42)	0.37
High	1.00 (ref)		1.00 (ref)		1.00 (ref)		1.00 (ref)		1.00 (ref)		1.00 (ref)	
Abbreviations: C Data on income n=1202	II, confidence in < 2000 n=870,	terval; CM 2000-320(V, cytomegalov 0 n=915, >320	irus, EBV, 0 n=1827	Epstein-Barr v ', Data on mate	irus; HSV ernal edu	/-1, herpes simp cational level low	lex virus v n=106	s type-1. ,4, mid low n=	=1258, m	nid high n=940), high
Table 9 (online). Odds Ratios	, for Coinf	iections Accor	dina to F	:BV Seronosit	ivity Sta	ş					
		VMC			HSV-1		CMV an	HSV-1				
	ΡO	ds Ratio		DhD	s Ratio		Odds Ratio					

	CMV		HSV-1		CMV and HSV	-1
	Odds Ratio		Odds Ratio		Odds Ratio	
	(95% CI)	<i>P</i> Value	(95% CI)	P Value	(95% CI)	<i>P</i> Value
EBV Positive	1.69 (1.50; 1.91)	<0.0001	2.17 (1.86; 2.53)	<0.0001	2.95 (2.02; 3.11)	<0.0001
EBV Negative	1.00 (reference)		1.00 (reference)		1.00 (reference)	
Abbreviations: CI,	confidence interval; CMV, o	ytomegalov	irus, EBV, Epstein-Barr	- virus; HSV	'-1, herpes somplex vi	rus type-1.

Chapter 4
	EBV, CMV or HSV-1		
	0 infections (n=1,319)	1 Infection (n=1,735)	≥2 infections (n=1,410)
Maternal characteristics			
Age (Mean \pm SD; years)	31.7 (4.7)	30.8 (5.0)***	29.7 (5.4)***
Educational level (n; %)			
Low	180 (14.4%)	322 (20.3%)***	366 (30.1%)***
Mid-Low	376 (30.0%)	476 (30.0%)	382 (31.5%)
Mid-High High	329 (26.3%)	336 (21.2%) 452 (28.5%)	229 (18.9%)
Net household income per month (n: %)	500 (25.570)	452 (20.570)	237 (15.570)
< € 2,000	177 (16%)	325 (23%)	368 (35%)***
€ 2,000-3,200	291 (22%)	370 (26%)	254 (24%)
> € 3,200	668 (59%)	716 (51%)	443 (42%)
Smoking continued during pregnancy (n;%)	138 (13%)	232 (17%)**	166 (15%)
Alcohol use continued during pregnancy (n;%)	487 (46%)	595 (43.4)	395 (35.7%)***
BMI before pregnancy (Mean \pm SD; kg/m ²)	23.4 (3.9)	23.6 (1.7)	23.9 (4.3)**
Maternal atopy (n; %)	396 (33.8%)	548 (36.9%)	354 (30.2%)
Family history of asthma / atopy (n; %)	605 (48.7%)	809 (51.1%)	541 (43.7%)*
Parity	. ,		
0	717 (57%)	917 (55%)*	736 (54%)***
1	404 (32%)	533 (32%)	401 (29%)
2	118 (9%)	162 (10%)	169 (12%)
≥ 3	23 (2%)	61 (3%)	62 (5%)
Mode of delivery (n; %)	825 (73 406)	1071 (72%)	014 (74 6%)
Forceps or vacuum assisted	166 (14.8%)	203 (13.6%)	163 (13.3%)
Caesarian section	133 (11.8%)	214 (14.4%)	148 (12.1%)
Premature rupture of membranes (n; %)	51 (4.1%)	72 (4,4%)	58 (4.3%)
Infant characteristics (n=258)	, , , , , , , , , , , , , , , , , , ,		
	683 (51.8%)	922 (53.1%)	699 (49.6%)
Western ethnicity (n:%)	1047 (80%)	1162 (69%)***	655 (49%)***
Gestational age (Mean + SD: weeks)	39.8 (1.9)	39.8 (1.7)	39.8 (1.8)
Preterm hirth (<37 weeks) (n; %)	76 (6 0%)	104 (6.0%)	79 (5.6%)
Birth weight (Moon \pm SD: grams)	3442 (599)	3413 (557)	3386 (552)**
Low hight (real \pm 3D, grains)	3442(399)	09 (F 70)	67 (4.90/)
Low birth weight ($<2500 \text{ gr}$)	75 (5.7%)	90 (5.7%)	07 (4.0%)
Day-care in first year (n; %)	567 (85%)	699 (89%)*	447 (88%)
Never	98 (10.2%)	114 (10.0%)	52 (6 3%)*
< 3 months	340 (35.5%)	364 (32.0%)	286 (34.8%)
3-6 months	195 (20.4%)	237 (20.8%)	186 (22.7%)
>6 months	324 (33.9%)	423 (37.2%)	297 (36.2%)
Infections first year (n; %)			
RTI	538 (57.2%)	652 (59.4%)	495 (63.3%)*
	125 (13.8%)	140 (13.5%) 607 (52.6%)	107 (15.0%) 471 (56 1%)**
GI	329 (55.7%)	378 (58.0%)	296 (64.1%)**
Atopy first year			
Eczema	264 (27.4%)	312 (28.2%)	215 (27.2%)
Wheezing (\geq 1 epsiode)	338 (42.8%)	412 (45.5%)	315 (49.5%)*
Nasopharyngeal bacterial carriership 6 years (n;	%)		
No pathogen	606 (47.3%)	794 (46.8%)	606 (44.2%)
S Aureus	338 (26.4%)	459 (27.0%)	408 (29.8%)
5. Fileumoniae H Influenza	200 (19.9%) 125 (9.8%)	290 (17.4%) 197 (11.6%)	240 (17.9%) 152 (11.1%)
M. Catarrhalis	117 (9.1%)	156 (9.2%)	156 (11.4%)

Table 10 (online). Maternal and infant characteristics of the study population (n=4,464)

Table 10.	(continued)
-----------	-------------

	EBV, CMV or HSV-1		
	0 infections	1 Infection	≥2 infections
	(n=1,319)	(n=1,735)	(n=1,410)
Antibiotic use (n; %)			
year 1	386 (47.6%)	504 (54.1%)**	340 (52.7%)
year 2	413 (41.6%)	520 (44.9%)	374 (47.3%)*
year 3	313 (32.4%)	396 (36.4%)	281 (37.5%)*
year 4	260 (27.6%)	331 (30.1%)	247 (32.2%)*
year 6	227 (18.9%)	339 (23.0%)*	266 (23.9%)**
ever (at age 6 years)	830 (85.0%)	1059 (89.4%)**	774 (90.0%)**
Age at focus visit		. ,	
(Median <u>+</u> range; years)	6.0 (5.0-8.8)	6.0 (5.0-9.0)***	6.1 (4.9-9.1)***
BMI for age (mean; sd; SDS)	0.20 (0.85)	0.25 (0.92)	0.39 (0.95)***
Values are means (SD), absolute number	ers (percentages) or #mee	dians (90% range). E	Based on observed
detect Circuificently, different from 0 in		0 01) *** (0 00	1) Matawal

dataset. Significantly different from 0 infections (p<0.05), (p<0.01), (p<0.001). Maternal atopy: eczema, allergy HDM, hay-fever. Any reported autoimmune disease: diabetes mellitus, SLE, arthritis, MS, thyroid disorder, or celiac disease

CHAPTER 5

CMV- AND EBV-INDUCED T-CELL EXPANSIONS IN YOUNG CHILDREN DO NOT IMPAIR NAIVE T-CELL POPULATIONS OR VACCINATION RESPONSES: THE GENERATION R STUDY

Running title: CMV and EBV T-cell expansion in children

D. van den Heuvel,¹ M.A.E. Jansen,^{2,3} W.A. Dik,¹ H. Bouallouch-Charif,¹
D. Zhao,¹ K.A.M. van Kester,¹ M.A.W. Smits-te Nijenhuis,¹
M.J. Kolijn-Couwenberg,¹ V.W.V. Jaddoe,^{2,3,4} R. Arens,⁵
J.J.M. van Dongen,¹ H.A. Moll,³ M.C. van Zelm.¹

¹ Dept. of Immunology, Erasmus MC, University Medical Center, Rotterdam, NL
² The Generation R Study Group, Erasmus MC, University Medical Center, Rotterdam, NL
³ Dept. of Pediatrics, Erasmus MC-Sophia, Rotterdam, NL
⁴ Dept. of Epidemiology, Erasmus MC, University Medical Center, Rotterdam, NL
⁵ Dept. of Immunohematology and Blood Transfusion, Leiden University Medical Center, Leiden, NL.

Journal of Infectious Diseases (2015), in press



ABSTRACT

BACKGROUND. Cytomegalovirus (CMV) and Epstein Barr virus (EBV) induce effector memory T-cell expansion, which are variable and potentially depend on the age at primary exposure and co-infections. We evaluated the T-cell compartment and herpesvirus infections in 6-year-old children.

METHODS. T-cell subsets and IgG seropositivity for CMV, EBV, herpes simplex virus-1 (HSV-1) and varicella zoster virus (VZV) were studied in 1,079 6-year-old children. A random subgroup of 225 children was evaluated for CMV and EBV seropositivity before 2y of age and vaccination responses against measles and tetanus.

Results. CMV and EBV infections were associated with significant expansions of CD27⁻ and CD27⁺ effector memory T cells, respectively. These expansions were enhanced in CMV+EBV+ children and were independent of VZV or HSV-1 co-infection. Naive and central memory T-cell numbers were not affected, nor were anti-tetanus and anti-measles IgG levels. Children infected before 2y of age showed smaller effector memory T-cell expansions than children infected between 2-6y.

CONCLUSIONS. CMV- and EBV-related T-cell expansions do not impair naive T-cell numbers nor maintenance of protective responses against non-related pathogens. Duration of infection was not directly related to larger expansions of effector memory T cells in children, suggesting that other mechanisms affect these expansions at later age.

INTRODUCTION

Cytomegalovirus (CMV) and Epstein Barr Virus (EBV) are ubiquitous in the human population and persist with presumed viral latency.^{1,2} Infection occurs mainly in childhood, reaching ~50% seropositivity around the age of 6y and 80-90% in adults within the Western world.^{3,4} CMV and EBV are associated with changes in immunological memory. Infected adults display persistent expansions of virus-specific effector memory T cells in both the CD8⁺ and the CD4⁺ lineages.^{5,6} CMV-specific CD8⁺ T cells are predominantly CCR7⁻CD45R0⁺CD27^{+/-} in young adults, while more clonal CCR7⁻CD45RA⁺CD27⁻ populations are found in elderly.^{5,7-14} EBV-specific T cells are mainly CCR7⁻CD45R0⁺CD27⁺.^{5,8,15,16} The majorities of these populations have an extended lifespan, but a poor response to T-cell mitogens.^{7-9,16-18}

Co-infection with CMV and EBV can have both synergistic and antagonistic effects: CMV co-infection can restore defective vaccination responses in EBV-infected children, while EBV co-infection can boost CMV-induced NK-cell differentiation, and together they affect the chance for developing allergic complications.¹⁹⁻²¹ In contrast, infections with herpes simplex virus 1 (HSV-1) do not result in persistent effector T-cell expansions.¹⁵ Less is known for co-infection with varicella zoster virus (VZV), because this already reaches 90-100% seropositivity in young adulthood in the Western world.

Accumulations of CMV- and EBV-induced effector memory T-cells have been suggested to overcrowd T cells with other specificities,^{5,22} and to negatively impact immune responses to other infections and/or vaccinations.^{9,17,23-25} Moreover, CMV and EBV persistence are associated with cardiovascular disease, infectious complications and with increased mortality rates, especially in immunosuppressed individuals or the very elderly.^{23,26-31} Still, these effects are not consistently observed,³²⁻³⁵ and despite high CMV and EBV seropositivity rates in elderly, only a minority develops clinical complications. An explanation could be the variation in T-cell responses: virus-specific T-cell numbers vary from barely detectable to >30% of total memory T-cells in peripheral blood. This diversity might be affected by the infectious dose,³⁶ and with long-term infection, and thus especially develop in individuals who had been infected early in life. However, in contrast to infection after puberty, primary infections with CMV or EBV in early childhood are mostly asymptomatic and might even be protective against the development of Celiac disease and allergies.^{37,38}

To study whether the developing immune system in childhood provides more effective control of persistent viruses,³⁹ we studied 1,079 6-year old children in the Generation R cohort. The combined analysis of CMV, EBV, HSV-1 and VZV infection in this large cohort allowed us to study virus-specific and modifying effects on the T-cell compartment and their relation with vaccination responses to tetanus and measles.

MATERIALS AND METHODS

STUDY SUBJECTS

This study was conducted in the context of the Generation R Study, a prospective population-based cohort study from fetal life until young adulthood.⁴⁰ We included 1,079 6-year-old children (range 5.0-7.9y) from whom CMV, EBV, HSV-1 and VZV IgG serology, and detailed immunophenotyping of blood T cells was performed. From a random selection of 225 of these children, additional virus serology and immunophenotyping was performed on blood samples obtained in the second year of life (age range 13.1-29.9 months). Written informed consent was obtained from all parents of participating children. Ethical approval for the study was obtained from the Medical Ethical Committee of the Erasmus MC.

SEROLOGY

Blood plasma samples were subjected to enzyme-linked immunoassays for IgG antibodies against CMV, EBV capsid antigen (EBV-VCA), HSV-1 glycoprotein C1, VZV, Tetanus toxoid or Measles antigen (EUROIMMUN®, Lübeck, Germany). Results were evaluated relative to a manufacturer-provided reference threshold sample. Seropositivity, and assumed virus persistence, was defined by a sample/ threshold ratio above 0.6 (CMV), 0.8 (EBV-VCA) 1.6 (HSV-1) and 1.0 (VZV). Antimeasles and anti-tetanus IgG levels were analyzed in international units/ml (IU/ ml) as a continuous variable and with plasma levels above 275 IU/ml (measles) or 0.5 IU/ml (tetanus) being defined as protective vaccination responses.

IMMUNOPHENOTYPING

Absolute numbers of CD3⁺ T cells were obtained with a routine diagnostic lyseno-wash protocol. Detailed analysis of T-cell subsets was performed with 6-color flowcytometry (Supplemental Table 1 and Supplemental Figure 1).^{22,41}

Antigen-specific T-cells were detected in thawed post-Ficoll PBMC of 15 CMV+ and 14 EBV+ children who carried the HLA-A*0201 allele, defined by SNPtags rs2844821(G) and rs762324(C) in previously generated SNP arrays.^{40,42} Virusspecific CD8⁺ T cells were detected with HLA-A*0201 tetramers loaded with CMV peptides NLVPMVATV (from pp65; APC-labeled) or VLEETSVML (from IE-1; PElabeled) proteins and EBV peptides GLCTLVAML (from BMLF-1; APC-labeled) or YVLDHLIVV (BRLF-1; PE-labeled). Flowcytometric data were acquired on a FACSCalibur or LSRII (BD Biosciences) and analyzed using FACSDiva software (BD Biosciences; version 6.2) and Infinicyte (Cytognos; version 1.7) analysis software.⁴³

STATISTICAL ANALYSES

Differences in infection prevalence were assessed using Chi-square tests. Differences in lymphocyte numbers (or frequencies) between uninfected controls and virus-infected groups were assessed with a Kruskal-Wallis test, followed by post-hoc Dunn's tests. Differences in longitudinal frequencies of lymphocyte populations were assessed using paired t-tests. P-values <0.05 were considered statistically significant.

RESULTS

HERPESVIRUS SEROPOSITIVITY

Among the 1,079 children included in the study, 36.2% was IgG-seropositive for CMV, 47.1% for EBV, 14.0% for HSV-1 and 92.1% for VZV. Because mainly CMV and EBV are reported to result in chronic T-cell expansions, we divided the 1,079 children into 4 groups: CMV-EBV- (uninfected; n=399); CMV+EBV- (CMV+; n=172); CMV-EBV+ (EBV+; n=289); CMV+EBV+ (n=219) (Table 1).

Co-infection with VZV was >90% in all groups. HSV-1 infection was significantly more frequent in EBV+ (p=0.02) or CMV+EBV+ (p<0.001) children (Table 1). Therefore, especially HSV-1 infection needs to be considered for possible confounding effects in our analyses on EBV.

CMV- AND EBV-SPECIFIC EFFECTOR MEMORY T-CELLS

To analyze whether 6-year-old children carried virus-specific T-cell expansions, we phenotyped virus-specific CD8⁺ T cells in 15 CMV+ and 14 EBV+ children using CD27 and CD45RA and HLA-A*0201 tetramers loaded with CMV-specific peptides of pp65 and IE-1 and EBV-specific peptides of BMLF-1 and BRLF-1 (Figure 1). Virus-specific CD8⁺ T cells were detected in blood at frequencies of 0.01-0.52% (pp65-NLV) and 0.01-2.32% (IE-1-VLE) for CMV peptides and 0.02-0.64% (BMLF-1-GLC) and 0.02-2.6% (BRLF-1-YVL) for EBV peptides. The CMV- and EBV-specific CD8⁺ T-cells were phenotypically diverse and predominantly consisted of CD45RA⁻CD27^{+/-} and CD45RA⁺CD27⁻ memory T-cells (Figure 1). Thus, 6-year old children already display expansions of effector memory T cells directed against CMV and EBV antigens.

CMV AND EBV ASSOCIATED MEMORY T-CELL EXPANSIONS

To study the effects of CMV and EBV on the T-cell compartment, we immunophenotyped CD8⁺ and CD8⁻ (CD4⁺) T cells in 18-20 randomly-selected children being either uninfected, CMV+, EBV+ or CMV+EBV+. T-cell data from each group of children were merged and subjected to automatic population separation (APS) with Infinicyte software,⁴³ based on the expression of CD3, CD8, CCR7, CD45RO, CD27 and CD28. Within both CD8⁺ and CD4⁺ T cells, four populations were distinguished in 2D plots of principle component (PC)1 vs PC2 (Figure 2). CMV was associated with a relative increase of two CD8⁺ populations: population 2 (CCR7⁻CD45RO⁻; 9.2% uninfected vs 15.1% CMV+) and population

Table 1. HSV-1 or VZV co-infection in correlation to CMV- and EBV-IgG seropositivity at (6 years
of age.	

			HSV+	VZV+	HSV+VZV+
	n	%	n (% of group)	n (% of group)	n (% of group)
Uninfected	399	37.0	39 (9.8%)ref ^a	367 (92%)ref ^a	35 (8.8%)ref ^a
CMV+	172	15.9	18 (10.5%)	165 (95.9%)	18 (10.5%)
EBV+	289	26.8	46 (15.9%)*	260 (90%)	42 (14.5%)*
CMV+EBV+	219	20.3	48 (21.9%)****	202 (92.2%)	40 (18.3%)***
total	1,079	100	151 (14%)	994 (92.1%)	135 (12.5%)
^a Significance of co-infection in the CMV+, EBV+ or CMV+EBV+ group is tested relative to the CMV-EBV- uninfected controls, using the Chi-square test; *, $p < 0.05$; ***, $p < 0.001$; ****, $p < 0.0001$.					



Figure 1. Frequencies and phenotypes of CMV- and EBV-specific CD8⁺ **T cells in 6-year-old children.** CD8⁺ T cells recognizing CMV- (**A-B**) or EBV-specific peptides (**C-B**) were detected with multicolor flowcytometry and further analyzed for CD27 and CD45RA expression. Flowcytometry plots consist of merged data files of 14 representative HLA-A*0201+ CMV+ (A-B) or EBV+ (C-D) children. The bottom graphs depict for each individual the relative distributions of tetramer-positive cells as fold increase compared to the relative distribution of total CD8⁺ T-cells.

3 (CCR7⁻CD45RO⁺; 2.8% uninfected vs 5.9% CMV+). EBV-infected children showed an increase in CCR7⁻CD45RO⁺ populations, both CD27⁻ (population 3; 2.8% uninfected vs 4.1% EBV+) and CD27⁺ (population 4; and 10.5% uninfected vs 14.1% EBV+, respectively). CMV- and EBV-infections were also associated with CD4⁺ T-cell memory expansions: CCR7⁻CD45RO⁺ (population 3; 14.2% uninfected; 20.6% CMV+; 16.6% EBV+) and CCR7⁻CD45RO⁻ (population 4; 6.7% uninfected; 10.9% CMV+; 8.6% EBV+) (Figure 2B/D). Thus, CMV and EBV infection in young children are associated with relative expansions of memory T cells.

Normal numbers of naive and central memory ${\bf T}$ cells in ${\bf CMV}\text{-}$ or ${\bf EBV}\text{-}$ infected children

To quantify whether the relative memory T cell expansions in CMV- and/or EBVcarriers also affected absolute T-cell numbers, we next evaluated the CD4⁺ and CD8⁺ T-cell lineages in 1,079 6-year-old children. CCR7⁺CD45R0⁻CD27⁺CD28⁺ naive T cells were distinguished from CCR7⁺CD45R0⁺CD27⁺CD28⁺ central memory T cells (Tcm), CCR7⁻CD45R0⁺ effector memory (TemRO) and CCR7⁻ CD45R0⁻ effector memory (TemRA) (Supplemental Figure 1A).^{22,41} TemRO and



Figure 2. Relative change in the T-cell phenotype in CMV- or EBV-infected children.

Unbiased flowcytometric analyses of CD8⁺ T cells (**A**) and CD8⁺ T cells (**B**) in uninfected (uninf), CMV+, EBV+ and CMV+EBV+ children. Data from 18-20 children in each group were merged into one file, and subjected to automatic population separation (APS) based on the expression of CD3, CD8, CCR7, CD45R0, CD27 and CD28. 2D projections of principle component (PC)1 versus PC2 revealed four populations. The relative contributions of CCR7, CD45R0, CD27 and CD28 to PC1 or PC2 are indicated. **C and D**) CD45R0 vs CCR7 expression of the populations defined by APS. Populations 3 and 4 in panel C were both CCR7-CD45R0+, but differed in CD27 and CD28 expression (small inset). TemRA CD8⁺ T cells were significantly increased in children with CMV or EBV infection compared to uninfected controls (Figure 3B). For CMV, these specifically concerned a 4.4 fold increase of CD27 CD28 late differentiated TemRA cells and 1.3-3.3 fold increase of CD27^{-/+}CD28⁻ intermediate to late-differentiated TemRO cells over uninfected controls (Figure 3C-D). EBV-infected children showed a 1.2-1.3 fold increase in CD27^{-/+}CD28⁻ intermediate to late-differentiated TemRA cells over uninfected controls and a 1.4-1.8 fold increase in all TemRO subsets (Figure 3C-D). Naive CD8⁺ T cells and CD8⁺ Tcm cells were present in normal numbers. Thus, the expansions of effector memory cells resulted in a significant 1.1 fold increase in the total number of CD8⁺ T cells (Figure 3A). Combined CMV and EBV infections resulted in a further increase in total CD8⁺ T-cell numbers (Figure 3A), with seemingly additive effects of the two viruses in double-infected children versus the single-infected children (Figure 3). Thus, in these 6-year-old children, infections with CMV and EBV were both associated with T-cell memory expansions. The phenotypes of the expanded populations differed for each of the viruses, did not affect each other in CMV+EBV+ double infection, and did not result in loss of naive CD8⁺ T cells.

Total CD4⁺ T-cell numbers were not affected by CMV and/or EBV infection (Supplemental Figure 1B). Still, EBV-infected children had 1.5 fold more CD27⁻ CD28⁻ late-differentiated TemRO cells than uninfected controls. Furthermore, CMV-infected children showed a 1.3-2 fold increase in CD27⁻ intermediate and late-differentiated TemRO cells, and a 1.9 fold increase in CD27⁻CD28⁻ late-differentiated TemRA cells over uninfected controls (Supplemental Figure 1B). Similar to CD8⁺ T cells, CMV and EBV infection independently resulted in effector memory CD4⁺ T-cell expansions, which did not affect naive T-cell numbers.



Figure 3. Absolute numbers of CD8⁺ T-cell subsets in CMV- or EBV-infected children.

A) Absolute numbers of total CD8⁺ T-cells in children uninfected with CMV or EBV (uninf), infected with only CMV, only EBV or CMV and EBV. **B)** Similar as in A, for 4 main CD8⁺ T-cell subsets: CCR7⁺CD45R0⁻CD27⁺CD28⁺ Central memory (Tcm), CCR7⁻CD45R0⁻ Effector memory (TemRO) and CCR7⁻CD45R0⁻ Effector memory (TemRA) cells ⁴¹. **C-D)** Similar as in A, for CD27⁺CD28⁺ early, CD27⁺CD28⁻ intermediate and CD27⁻CD28⁻ late TemRO (C) and TemRA (D) populations ²². Bars depict stacked median values per T-cell population. The number of individuals per category is indicated underneath each plot. Significance was tested first by a Kruskal-Wallis test per T-cell population relative to the uninfected controls, and in case of significance (p<0.05) followed by a Dunn's test of individual patient groups. Significance for the Dunn's test is indicated in the plots: *, p<0.00; **, p<0.001; ****, p<0.0001.

EFFECT OF HSV-1 AND VZV ON T-CELL SUBSET NUMBERS

Through combined analysis of children with single CMV, or single EBV, and double CMV/EBV infection, we could distinguish distinct effects of these viruses on the T-cell compartment in young children. Still, the observed effects could be influenced by infection with other viruses, such as HSV-1 or VZV, especially considering the increased HSV-1 positivity in EBV+ children (Table 1). Therefore, we separated the groups of uninfected controls, CMV+, EBV+ and CMV+EBV+ children further into VZV-HSV-1-, VZV+HSV-1 and VZV+HSV-1+ subgroups. Because VZV prevalence was >90% in our cohort, the effect of HSV-1, without VZV, could not be determined. Though the TemRO and TemRA populations were significantly different between all virus-infected groups (Kruskal Wallis; p<0.0001), these effects were only caused by CMV- or EBV-associated expansions. The presence of VZV and HSV-1 within the CMV+ or CMV+EBV+ group was only associated with a slight, but not significant increase of TemRA cells (Figure 4). Thus, co-infection with VZV and HSV-1 did not significantly affect CMV- or EBV-associated effector memory T-cell expansion.



Figure 4. Effect of HSV-1 and VZV on blood CD8⁺ TemRO and TemRA cell numbers. Uninfected, CMV+, EBV+ and CMV+EBV+ children were further subdivided in children without HSV-1 and VZV (white), co-infected with only VZV (light gray) or with both HSV-1 and VZV (dark gray). Absolute numbers of CD8⁺ TemRO (**A**) and TemRA (**B**) cells in all population groups depicted as 10-90% box-whisker-plots. The number of individuals per category is indicated underneath each plot. Significance was tested first by a Kruskal-Wallis test per T-cell population, and in case of significance (p<0.05) followed by a Dunn's test of individual patient groups, comparing HSV-1 and/or VZV subgroups within each category of uninfected, CMV, EBV or CMV+EBV+ children.

NORMAL VACCINATION RESPONSES IN CMV- OR EBV-INFECTED CHILDREN

The effector memory T-cell expansions in CMV- or EBV-infected children did not result in a reduction of naive and Tcm numbers (Figures 2 and 3). To study if immunological memory to other pathogens was normally present in these 6-year-old children, we tested their responses to previous vaccinations with tetanus at 2, 3, 4, 11 months and 4 years, and measles at 14 months according to the Dutch national vaccination protocol.⁴⁴ We defined vaccination responses in a randomly selected subgroup of 225 6-year-old children that showed similar seroprevalence for CMV and EBV (Supplemental Table 2), and similar effector memory T-cell

expansions (data not shown) as the total cohort of 1,079 children. Furthermore, the seroprevalence of co-infection with HSV-1 and VZV was not significantly different between the CMV+ or EBV+ groups in this selected cohort (Supplemental Table 3). In total, 81.2% of the children had IgG antibodies against measles, and 63.2% against tetanus. These frequencies were not significantly different in children with CMV and/or EBV infection (Figure 5). Furthermore, median titers of anti-measles and anti-tetanus IgG did not differ between the four groups. Thus, EBV and CMV seropositivity did not impair immunity to measles and tetanus vaccination in 6-year-old children.



Figure 5. Vaccination responses in 6-year-old CMV or EBV infected children.

Anti-measles (A) and anti-tetanus (B) IgG levels in uninfected, CMV+, EBV+ or CMV+EBV+ children. Specific IgG levels were not significantly different between the virus positive groups as compared with uninfected controls as determined with the Kruskal-Wallis test; p-values are indicated at the top right of the plot. Percentages at the top represent the number of children with a protective anti-measles IgG response >275 international units/ml (IU/ml) (A) or anti-tetanus IgG response >0.5 IU/ml (B). Percentages were not significantly different as tested by Chi-square test.

No association between T-cell expansions and duration of $\ensuremath{\mathsf{CMV}}$ or $\ensuremath{\mathsf{EBV}}$ infection

Our results indicate that the EBV- and CMV-specific T-cell expansions do not affect naive T-cell numbers or vaccination responses to tetanus and measles in 6-year old children. To study whether infection early in childhood has different immunological consequences than infection later in childhood, we determined CMV and EBV seropositivity around 2 years of age in a subgroup of 225 children. We observed that of the CMV+EBV+ children, 80% already carried anti-CMV IgG before 2 years of age, and 46.8% anti-EBV IgG.

We subdivided the EBV+ and CMV+ children into early-infected (before 2y) and late-infected subgroups (2-6y), and compared their T-cell compartments at 6 years of age. Early nor late infection affected naive and Tcm cell numbers (not shown), whereas CD4⁺ and CD8⁺ effector memory T cells were significantly increased (Figure 6 and Supplemental Figure 1C). CMV+EBV+ early-infected children had significantly smaller CD8⁺ TemRO and TemRA T-cell populations than late-infected children, with numbers in early-infected children being similar to uninfected controls (Figure 6A). Moreover, retrospective longitudinal analysis of early-infected children indicated that CD8⁺ effector memory T-cell numbers

were stable between the age of 2y and 6y (Figure 6B). CD4⁺ effector memory T-cell numbers did not significantly differ between early-infected and late-infected children. Still, some subsets in late-infected children were significantly higher than in uninfected controls, whereas these were not increased in early-infected children (Supplemental Figure 1C). Furthermore, CD4⁺ effector memory T-cell numbers increased between the ages of 2y to 6y in early-infected children (Supplemental Figure 1D). Together, these results suggest that CD8⁺ effector memory T-cell numbers are not directly related to the time post infection. Rather, the early infections were associated with smaller expansions of effector memory T cells at 6 years.



Figure 6. Limited effector memory T-cell expansions in children infected with EBV and CMV <2y. 6-Year old CMV+EBV+ children were subgrouped based on infection <2y of age (early; light gray; n=11) and >2y of age (late; dark gray; n=10), and compared to CMV-EBV- uninfected controls (white; n=397). A-B) Absolute number of CD8⁺ TemRO (A) and TemRA (B) cell subsets. Plots depict 10-90% box-whisker plots. Significance was tested first by a Kruskal-Wallis test per T-cell population, and in case of significance (p<0.05) followed by a Dunn's test of individual patient groups. Significance for the Dunn's test is indicated in the plots: *, p<0.05; **, p<0.01; ***, p<0.001; ****, p<0.001; ****, p<0.001; C-D) Longitudinal follow-up of frequencies of CD8⁺ TemRO (C) and TemRA (D) subsets at the age of 2y and 6y in early infected children. No significant differences between 2y and 6y could be observed by paired T-test (p>0.12).

DISCUSSION

We studied the effects of single or combined infection with CMV, EBV, HSV-1 and/ or VZV on naive and memory T cells in a large cohort of more than 1,000 6-yearold children. CMV and EBV infections each resulted in distinct effector memory T-cell expansions, which were additive in case of co-infection. HSV-1 and VZV infections did not significantly affect the T-cell compartment, but might slightly enhance CMV- and EBV-associated T-cell expansions. In contrast to adults, CD8⁺ Tem-cell expansions (either TemRO or TemRA) in CMV- or EBV-infected children did neither result in overcrowding of naive and Tcm compartments, nor in loss of vaccination responses to tetanus or measles. Notably, children infected <2y showed fewer Tem cells at 6-years than children infected between 2-6y. Thus, depending on the age at the time of infection, CMV and EBV infections seem to be controlled differently.

We, here, detected sizeable CMV- and EBV-specific CD8⁺ memory T-cell expansions in 6-year-old children. Unfortunately, the real extent of virus-specific T-cell expansions is difficult to assess due to the large number of viral epitopes and the diverse HLA backgrounds of the children in our study. However, since 1) the phenotype of the antigen-specific T cells is largely similar to the phenotype of the expanded effector memory T-cell populations, both relatively and absolute, within the total CD8⁺ T-cell pool in our cohort of children, and 2) the antigen-specific phenotypes are in accordance with previous literature on CMV and EBV infections in adults,⁴⁵ we are convinced that phenotyping the total CD8⁺ T-cell compartment generates an reliable representation of virus-associated changes of the immune compartment.

Our observation that CMV and EBV were associated with an increase of distinct CD8⁺ Tem subsets is consistent with previous literature in adults and children.^{5,7-12,15,46,47} Importantly, our large cohort allowed us to further analyze the possible antagonistic, synergistic or independent impact of CMV and EBV on the immune compartment. Importantly, in the group of CMV and EBV co-infected children, an additive effect was seen for the Tem expansions compared to that seen in single-infected patients. Furthermore, the size of the CMV-associated TemRA late population was not (negatively) affected by the size of the EBV-associated TemRO populations in individual patients (data not shown). Thus, the distinct effects of CMV and EBV on CD8⁺ Tem expansions, appear to be independent of each other.

In addition to CD8⁺ Tem expansions, we found that CD4⁺ Tem-cell numbers were increased in both CMV and EBV infections, which is similar to observations in adults.⁶ Although these expansions were smaller and did not result in a significant increase in total CD4⁺ T-cell numbers, they were distinct between both viruses. Furthermore, the phenotypes of the CD4⁺ Tem expansions were remarkably similar to their CD8⁺ counterparts and were additive in CMV and EBV double positive children. Despite these seemingly minor expansions, CD4⁺ T cells have an important role in controlling primary CMV infection.⁴⁸ Apparently, the formation of sizeable Tem populations is necessary to successfully suppress the virus. Importantly, these expansions did not result in overcrowding of more immature T-cell subsets, something that for CMV infection in elderly has been associated with poor CD4⁺ memory T-cell responses to influenza proteins.²⁴

Our large cohort allowed us to study the effects of HSV-1 and VZV in EBV and CMV double negative children, as well as any modifying effects on CMV and/or EBV infections. Because virtually all adults are infected with VZV, little is known on the effects on T-cell memory¹⁵ and we could for the first time conclude that VZV does not modify T-cell memory alone or in combination with CMV and/or EBV. Co-infection with HSV-1 was more frequent in EBV+ and CMV+EBV+ children and might therefore be an important contributor to the immune modulation that is currently assumed to be EBV-associated. Still, HSV-1 did not affect Tem in EBV+

children, suggesting that these expansions are EBV-specific and not due to HSV-1. The difference in T-cell modulation between CMV and EBV on the one hand and VZV and HSV-1 on the other hand is most likely due to tropism and anatomical localization of the viruses. In fact, it is well-possible that HSV-1 and VZV induce T-cell expansions, but these cells are thought to reside mostly in the human skin or locally around virus-infected cells.⁴⁹

Despite the expansions of CD8⁺, and CD4⁺, Tem in children infected with CMV and/ or EBV, these did not result in decreased naive or Tcm cell numbers. In fact, infected children had significantly more total CD8⁺ T cells. Especially in immunosuppressed individuals, but also in CMV-infected elderly, effector memory T-cell expansions with a subsequent loss of naive T cells and loss of vaccination responses have been described to be hallmarks of CMV-associated immunosenescence.^{9,17,24,25} However, literature is inconsistent,³²⁻³⁵ and the loss of vaccination responsiveness in children, is not consistently observed.¹⁹ In our cohort, IgG titers to previous measles and tetanus vaccinations were similar between uninfected and infected children. Combined, our data suggest that 6-year-old CMV- and/or EBV-infected children do not lose their immune responsiveness due to virus-related effects.

An important explanation for the discrepancy in the literature on the loss of naive T cells could be differences in analysis strategies. In fact, due to the Tem expansions, the relative proportions of naive and Tcm are decreased in infected children in our study. Moreover, the expansion of total CD8⁺ T-cell numbers causes a shift in the CD4:CD8 T-cell ratio. These relative shifts are, however, the results of data analysis, rather than real defects, because the absolute naive CD8⁺ and total CD4⁺ T-cell numbers were normal. This is why we have for long time advocated for consistent analysis of absolute cell numbers with age-matched controls.⁵⁰

Although 90% of elderly are infected with CMV and EBV, only a minority of these individuals shows loss of naive T-cell numbers, impaired vaccination responses and clinical complications. The reason why only a minority develops complications is still unclear. Our study at the age of 6y indicates that early infection before the age of 2y results in smaller Tem expansions than infection between 2-6y. Since the size of the expanded virus-specific Tem pool has been directly correlated to the extent of CMV-induced endothelial damage, the children that have been infected before 2y might have a reduced risk for developing cardiovascular complications.³¹ CMV-related clinical complications might occur only in individuals infected later in life and who show the strongest CMV-associated T-cell expansions. An important factor influencing the variability in the response might be the infectious dose.³⁶ It would therefore be interesting to correlate the age at infection and the subsequent size of the Tem-cell expansions with the infectious dose and the development of clinical complications in children and adults, in a separate cohort or in longitudinal follow-up of the uninfected children in our cohort.

In conclusion, we here provide evidence that in young children, CMV and EBV infections lead to sizeable Tem expansions, but not to the associated immunosenescence. In fact, the Tem expansion result in increased total T-cell numbers, while naive and Tcm remain normally present, as are responses to previous vaccinations. Moreover, 6-year old children who were infected prior to 2 year of age maintained a stable control of these persistent viruses with only limited Tem expansions. These new insights into the immuno-modulatory effects of herpesviruses in young children are important to understand herpesvirus-associated immunosenescence in elderly.

ACKNOWLEDGMENTS

This work was supported by an Erasmus MC Fellowship to M.C.v.Z. The Generation R Study is conducted by the Erasmus MC, Erasmus University Rotterdam in close collaboration with the School of Law and Faculty of Social Sciences of the Erasmus University Rotterdam, the Municipal Health Service Rotterdam Metropolitan Area, the Rotterdam Homecare Foundation, and the Stichting Trombosedienst & Artsenlaboratorium Rijnmond. We gratefully acknowledge the contributions of children and parents, general practitioners, hospitals, midwives and pharmacies in Rotterdam. We thank R. Bouzid, N.M.A. Nagtzaam and B.C.M. Dufour-van den Goorbergh for technical support.

CONFLICT OF INTEREST

J.J.M.v.D. discloses financial support by BD Biosciences for Educational Services in the field of flowcytometric diagnosis and classification of leukemias and lymphomas. In addition, because of the J.J.M.v.D.'s chairmanship of the scientific EuroFlow consortium, the department of Immunology at Erasmus MC receives the Infinicyt software free-of-charge. M.C.v.Z. reports grants from Erasmus MC during the conduct of the study; grants from Sophia Children's Hospital, outside the submitted work. All other authors declare that no competing interests exist.

D.v.d.H., W.A.D., J.J.M.v.D. and M.C.v.Z designed the experiments; D.v.d.H., M.A.E.J., H.B.-C., D.Z., K.A.M.v.K., M.A.W.S-t.N. and M.J.K.-C. performed and analyzed most of the experiments and contributed to data analyses; V.W.V.J. and H.A.M. designed the Generation R study and coordinated material collection. R.A. contributed to phenotyping virus-specific T-cell populations. D.v.d.H. and M.C.v.Z. wrote the manuscript; and all authors commented on the manuscript.

REFERENCES

- 1. Sinclair J. Human cytomegalovirus: Latency and reactivation in the myeloid lineage. *J Clin Virol.* 2008; 41(3): 180-185.
- Wada T, Toga A, Sakakibara Y, Toma T, Hasegawa M, Takehara K, et al. Clonal expansion of Epstein-Barr virus (EBV)-infected gammadelta T cells in patients with chronic active EBV disease and hydroa vacciniforme-like eruptions. *Int J Hematol.* 2012; 96(4): 443-449.
- 3. Bate SL, Dollard SC, Cannon MJ. Cytomegalovirus seroprevalence in the United States: the national health and nutrition examination surveys, 1988-2004. *Clin Infect Dis*. 2010; 50(11): 1439-1447.
- Dowd JB, Palermo T, Brite J, McDade TW, Aiello A. Seroprevalence of Epstein-Barr virus infection in U.S. children ages 6-19, 2003-2010. *PLoS One*. 2013; 8(5): e64921.
- 5. Appay V, Dunbar PR, Callan M, Klenerman P, Gillespie GM, Papagno L, et al. Memory CD8+ T cells vary in differentiation phenotype in different persistent virus infections. *Nat Med*. 2002; 8(4): 379-385.
- van Leeuwen EM, Remmerswaal EB, Vossen MT, Rowshani AT, Wertheim-van Dillen PM, van Lier RA, et al. Emergence of a CD4+CD28- granzyme B+, cytomegalovirus-specific T cell subset after recovery of primary cytomegalovirus infection. *J Immunol*. 2004; 173(3): 1834-1841.
- Griffiths SJ, Riddell NE, Masters J, Libri V, Henson SM, Wertheimer A, et al. Age-associated increase of low-avidity cytomegalovirus-specific CD8+ T cells that re-express CD45RA. *J Immunol*; 190(11): 5363-5372.
- 8. Brunner S, Herndler-Brandstetter D, Weinberger B, Grubeck-Loebenstein B. Persistent viral infections and immune aging. *Ageing Res Rev.* 2011; 10(3): 362-369.
- 9. Weinberger B, Lazuardi L, Weiskirchner I, Keller M, Neuner C, Fischer KH, et al. Healthy aging and latent infection with CMV lead to distinct changes in CD8+ and CD4+ T-cell subsets in the elderly. *Hum Immunol.* 2007; 68(2): 86-90.

- Almanzar G, Schwaiger S, Jenewein B, Keller M, Herndler-Brandstetter D, Wurzner R, et al. Longterm cytomegalovirus infection leads to significant changes in the composition of the CD8+ T-cell repertoire, which may be the basis for an imbalance in the cytokine production profile in elderly persons. J Virol. 2005; 79(6): 3675-3683.
- 11. Sylwester AW, Mitchell BL, Edgar JB, Taormina C, Pelte C, Ruchti F, et al. Broadly targeted human cytomegalovirus-specific CD4+ and CD8+ T cells dominate the memory compartments of exposed subjects. *J Exp Med.* 2005; 202(5): 673-685.
- Chidrawar S, Khan N, Wei W, McLarnon A, Smith N, Nayak L, et al. Cytomegalovirus-seropositivity has a profound influence on the magnitude of major lymphoid subsets within healthy individuals. *Clin Exp Immunol.* 2009; 155(3): 423-432.
- 13. Henson SM, Riddell NE, Akbar AN. Properties of end-stage human T cells defined by CD45RA reexpression. *Curr Opin Immunol.* 2012; 24(4): 476-481.
- Komatsu H, Inui A, Sogo T, Fujisawa T, Nagasaka H, Nonoyama S, et al. Large scale analysis of pediatric antiviral CD8+ T cell populations reveals sustained, functional and mature responses. *Immun Ageing*. 2006; 3(11.
- 15. Derhovanessian E, Maier AB, Hahnel K, Beck R, de Craen AJ, Slagboom EP, et al. Infection with cytomegalovirus but not herpes simplex virus induces the accumulation of late-differentiated CD4+ and CD8+ T-cells in humans. *J Gen Virol*. 2011; 92(Pt 12): 2746-2756.
- Ouyang Q, Wagner WM, Walter S, Muller CA, Wikby A, Aubert G, et al. An age-related increase in the number of CD8+ T cells carrying receptors for an immunodominant Epstein-Barr virus (EBV) epitope is counteracted by a decreased frequency of their antigen-specific responsiveness. *Mech Ageing Dev*. 2003; 124(4): 477-485.
- 17. Khan N, Hislop A, Gudgeon N, Cobbold M, Khanna R, Nayak L, et al. Herpesvirus-specific CD8 T cell immunity in old age: cytomegalovirus impairs the response to a coresident EBV infection. *J Immunol*. 2004; 173(12): 7481-7489.
- Khan N, Shariff N, Cobbold M, Bruton R, Ainsworth JA, Sinclair AJ, et al. Cytomegalovirus seropositivity drives the CD8 T cell repertoire toward greater clonality in healthy elderly individuals. *J Immunol*. 2002; 169(4): 1984-1992.
- 19. Holder B, Miles DJ, Kaye S, Crozier S, Mohammed NI, Duah NO, et al. Epstein-Barr virus but not cytomegalovirus is associated with reduced vaccine antibody responses in Gambian infants. *PLoS One*. 2010; 5(11): e14013.
- Sidorchuk A, Wickman M, Pershagen G, Lagarde F, Linde A. Cytomegalovirus infection and development of allergic diseases in early childhood: interaction with EBV infection? *J Allergy Clin Immunol*. 2004; 114(6): 1434-1440.
- Saghafian-Hedengren S, Sohlberg E, Theorell J, Carvalho-Queiroz C, Nagy N, Persson JO, et al. Epstein-Barr virus coinfection in children boosts cytomegalovirus-induced differentiation of natural killer cells. J Virol. 2013; 87(24): 13446-13455.
- 22. Appay V, van Lier RA, Sallusto F, Roederer M. Phenotype and function of human T lymphocyte subsets: consensus and issues. *Cytometry A*. 2008; 73(11): 975-983.
- 23. Savva GM, Pachnio A, Kaul B, Morgan K, Huppert FA, Brayne C, et al. Cytomegalovirus infection is associated with increased mortality in the older population. *Aging Cell*; 12(3): 381-387.
- 24. Derhovanessian E, Maier AB, Hahnel K, McElhaney JE, Slagboom EP, Pawelec G. Latent infection with cytomegalovirus is associated with poor memory CD4 responses to influenza A core proteins in the elderly. *J Immunol.* 2014; 193(7): 3624-3631.
- 25. Frasca D, Diaz A, Romero M, Landin AM, Blomberg BB. Cytomegalovirus (CMV) seropositivity decreases B cell responses to the influenza vaccine. *Vaccine*; 33(12): 1433-1439.
- 26. Jonasson L, Tompa A, Wikby A. Expansion of peripheral CD8+ T cells in patients with coronary artery disease: relation to cytomegalovirus infection. *J Intern Med.* 2003; 254(5): 472-478.
- 27. Sansoni P, Vescovini R, Fagnoni FF, Akbar A, Arens R, Chiu YL, et al. New advances in CMV and immunosenescence. *Exp Gerontol*. 2014; 55(54-62.
- 28. Pawelec G. Immunosenenescence: role of cytomegalovirus. Exp Gerontol. 2014; 54(1-5.
- 29. White DW, Suzanne Beard R, Barton ES. Immune modulation during latent herpesvirus infection. *Immunol Rev.* 2012; 245(1): 189-208.
- 30. van de Berg PJ, Yong SL, Remmerswaal EB, van Lier RA, ten Berge IJ. Cytomegalovirus-induced effector T cells cause endothelial cell damage. *Clin Vaccine Immunol.* 2012; 19(5): 772-779.
- 31. Bolovan-Fritts CA, Trout RN, Spector SA. High T-cell response to human cytomegalovirus induces

chemokine-mediated endothelial cell damage. Blood. 2007; 110(6): 1857-1863.

- den Elzen WP, Vossen AC, Cools HJ, Westendorp RG, Kroes AC, Gussekloo J. Cytomegalovirus infection and responsiveness to influenza vaccination in elderly residents of long-term care facilities. *Vaccine*. 2011; 29(29-30): 4869-4874.
- Odumade OA, Knight JA, Schmeling DO, Masopust D, Balfour HH, Jr., Hogquist KA. Primary Epstein-Barr virus infection does not erode preexisting CD8(+) T cell memory in humans. *J Exp Med.* 2012; 209(3): 471-478.
- 34. Remmerswaal EB, Havenith SH, Idu MM, van Leeuwen EM, van Donselaar KA, Ten Brinke A, et al. Human virus-specific effector-type T cells accumulate in blood but not in lymph nodes. *Blood*. 2012; 119(7): 1702-1712.
- van Leeuwen EM, Koning JJ, Remmerswaal EB, van Baarle D, van Lier RA, ten Berge IJ. Differential usage of cellular niches by cytomegalovirus versus EBV- and influenza virus-specific CD8+ T cells. J Immunol. 2006; 177(8): 4998-5005.
- 36. Redeker A, Welten SP, Arens R. Viral inoculum dose impacts memory T-cell inflation. *Eur J Immunol*; 44(4): 1046-1057.
- Nilsson C, Larsson Sigfrinius AK, Montgomery SM, Sverremark-Ekstrom E, Linde A, Lilja G, et al. Epstein-Barr virus and cytomegalovirus are differentially associated with numbers of cytokineproducing cells and early atopy. *Clin Exp Allergy*. 2009; 39(4): 509-517.
- 38. Saghafian-Hedengren S, Sverremark-Ekstrom E, Linde A, Lilja G, Nilsson C. Early-life EBV infection protects against persistent IgE sensitization. *J Allergy Clin Immunol*. 2010; 125(2): 433-438.
- 39. Prendergast AJ, Klenerman P, Goulder PJ. The impact of differential antiviral immunity in children and adults. *Nat Rev Immunol*. 2012; 12(9): 636-648.
- 40. Kruithof CJ, Kooijman MN, van Duijn CM, Franco OH, de Jongste JC, Klaver CC, et al. The Generation R Study: Biobank update 2015. *Eur J Epidemiol*. 2014; 29(12): 911-927.
- 41. Sallusto F, Lenig D, Forster R, Lipp M, Lanzavecchia A. Two subsets of memory T lymphocytes with distinct homing potentials and effector functions. *Nature*. 1999; 401(6754): 708-712.
- 42. de Bakker PI, McVean G, Sabeti PC, Miretti MM, Green T, Marchini J, et al. A high-resolution HLA and SNP haplotype map for disease association studies in the extended human MHC. *Nat Genet*. 2006; 38(10): 1166-1172.
- Pedreira CE, Costa ES, Lecrevisse Q, van Dongen JJ, Orfao A, EuroFlow C. Overview of clinical flow cytometry data analysis: recent advances and future challenges. *Trends Biotechnol*. 2013; 31(7): 415-425.
- 44. F. Abbink, H.G.A.M. van de Avoort, W.A.M. Berbers, R.S. van Binnendijk, H.J. Boot, Y.T.H.P. van Duynhoven, et al. The National Immunisation Programme in the Netherlands_Developments in 2006 *Centre for Infectious Diseases Control, RIVM.* 2006.
- 45. Gamadia LE, van Leeuwen EM, Remmerswaal EB, Yong SL, Surachno S, Wertheim-van Dillen PM, et al. The size and phenotype of virus-specific T cell populations is determined by repetitive antigenic stimulation and environmental cytokines. *J Immunol*. 2004; 172(10): 6107-6114.
- 46. Huygens A, Dauby N, Vermijlen D, Marchant A. Immunity to cytomegalovirus in early life. *Front Immunol*. 2014; 5(552.
- Kuijpers TW, Vossen MT, Gent MR, Davin JC, Roos MT, Wertheim-van Dillen PM, et al. Frequencies of circulating cytolytic, CD45RA+CD27-, CD8+ T lymphocytes depend on infection with CMV. *J Immunol*. 2003; 170(8): 4342-4348.
- Gamadia LE, Remmerswaal EB, Weel JF, Bemelman F, van Lier RA, Ten Berge IJ. Primary immune responses to human CMV: a critical role for IFN-gamma-producing CD4+ T cells in protection against CMV disease. *Blood*. 2003; 101(7): 2686-2692.
- 49. Farber DL, Yudanin NA, Restifo NP. Human memory T cells: generation, compartmentalization and homeostasis. *Nat Rev Immunol*. 2014; 14(1): 24-35.
- 50. Comans-Bitter WM, de Groot R, van den Beemd R, Neijens HJ, Hop WC, Groeneveld K, et al. Immunophenotyping of blood lymphocytes in childhood. Reference values for lymphocyte subpopulations. *J Pediatr*. 1997; 130(3): 388-393.

Supplemental Table 1. Antibody details			
Antibody	clone	manufacturer	
CD45-PerCP	2D1	BD Biosciences	
CD3-FITC	SK7	BD Biosciences	
CD3-PerCP	SK7	BD Biosciences	
CD4-PE-Cy7	SK3	BD Biosciences	
CD8-APC-H7	SK1	BD Biosciences	
CD8-PE-Cy7	SK1	BD Biosciences	
CD45RO-APC	UCHL-1	BD Biosciences	
CD28-FITC	CD28.2	BD Biosciences	
CD27-APC-H7	M-T271	BD Biosciences	
CD197(CCR7)-PE	3D13	e-Bioscience	

tel Telelo 4. Autile du deteil

Supplemental Table 2. CMV- and EBV- IgG seropositivities in the total and selected cohorts

	Total cohort	Selected cohort		
	n(%)	n(%)		
Uninfected	399 (37.0)	90 (40.0)		
CMV+	172 (15.9)	41 (18.2)		
EBV+	289 (26.8)	52 (23.1)		
CMV+EBV+	219 (20.3)	42 (18.7)		
total	1,079	225		
No significant	differences were	observed in IgG		
seropositivity between the two cohorts using the				
Chi-square test.				

Supplemental Table 3. HSV-1 or VZV co-infection in relation to CMV- and EBV-carriership.

	n	%	HSV+ n(% of group)	VZV+ n(% of group)	HSV+VZV+ n(% of group)
Uninfected	90	40.0	9 (10%)ref ^a	83 (92.2%)ref ^a	8 (8.9%)ref ^a
CMV+	41	18.2	5 (12.2%)	41 (100%)	5 (12.2%)
EBV+	52	23.1	10 (19.2%)	48 (92.3%)	9 (17.3%)
CMV+EBV+	42	18.7	6 (14.3%)	39 (92.9%)	6 (14.3%)
total	225	100.0	30 (13.3%)	211 (93.8%)	28 (12.4%)
^a No significant differences were observed of co-infection in the CMV+, EBV+ or CMV+EBV+ groups					
relative to the CMV-EBV- uninfected controls using the Chi-square test.					



(Figure on previous page)

Supplemental Figure 1. CD4⁺ T cell subset analyses

A) Gating strategy for delineating T-cell subsets. Total CD3⁺ T cells were subdivided into CD8⁺ (blue) and CD8⁺ (CD4⁺; green) T cells. Both CD4⁺ and CD8⁺ T cells were further subdivided into CCR7⁺CD45R0⁻ Naive, CCR7⁺CD45R0⁺ central memory (Tcm), CCR7⁻CD45R0⁺ effector memory (TemR0) and CCR7⁺CD45R0⁻ effector memory (TemRA) cells [41]. TemR0 and TemRA subsets were further subdivided into CD27⁺CD28⁺ certy. (CD27⁺CD28⁺ (CD4⁺) intermediate and CD27⁻CD28⁺ late populations [22]. **B)** Absolute numbers of total CD4⁺ T-cells and CD4⁺ T-cell subsets in children uninfected with CMV or EBV (uninf), infected with only CMV (CMV), only EBV (EBV) or both CMV and EBV (CMV EBV). Bars depict stacked median values per T-cell population. The number of individuals per category is indicated underneath each plot. **C)** Absolute numbers of CD8⁺ TemR0 (left) and TemRA (right) cell subsets in 6-year-old children who were either CMV-EBV-, became CMV+EBV+ before the age of 2y (early) or CMV+EBV+ between 2 and 6y of age (late). Plots depict 10-90% box-whisker plots. **B-C**) Significance was tested first by a Kruskal-Wallis test per lymphocyte population, and in case of significance (p<0.05) followed by a Dunn's test of individual patient groups. Significance for the Dunn's test is indicated in the plots: *, p<0.05; **, p<0.01; ***, p<0.001. **D)** Longitudinal follow-up of frequencies of CD8⁺ (CD4⁺) TemR0 (left) and TemRA (right) subsets at the age of 2y and 6y in 9 early infected children. Significance between 2y and 6y was tested by paired T-tests: *, p<0.05.

CHAPTER 6

TRANSIENT REDUCTION IN IGA⁺ AND IGG⁺ MEMORY B-CELL NUMBERS IN YOUNG CHILDREN PERSISTENTLY INFECTED WITH EBV: THE GENERATION R STUDY

Running title: EBV-associated depletion of memory B cells

D. van den Heuvel,¹ M.A.E. Jansen,^{2,3} A. Bell,⁴ A.B. Rickinson,⁴ V.W.V. Jaddoe,^{2,3,5} J.J.M. van Dongen,¹ H.A. Moll,³ M.C. van Zelm.^{1‡}

¹ Dept. of Immunology, Erasmus MC, University Medical Center, Rotterdam, NL
² The Generation R Study Group, Erasmus MC, University Medical Center, Rotterdam, NL
³ Dept. of Pediatrics, Erasmus MC-Sophia, Rotterdam, NL
⁴ School of Cancer Sciences, College of Medical and Dental Sciences, University of Birmingham, Edgbaston Birmingham, UK
⁵ Dept. of Epidemiology, Erasmus MC, University Medical Center, Rotterdam, NL
[‡] Present address: Dept. of Immunology, Central Clinical School, Monash University, Melbourne, Victoria, Australia

In preparation



ABSTRACT

The Epstein Barr virus (EBV) can persists both in IgM⁺ and in Ig class-switched memory B cells. However, it remains unclear whether EBV infection results in depletion of these target cells. We here studied the presence of EBV genomes in 6 memory B-cells subset in adults, and we investigated how EBV infection during the first year of life affects the numbers of these memory B cells in young children at the ages of 14 months and 6 years. EBV genomes were equally numerous in CD27⁺IqG⁺, CD27⁺IqA⁺ and CD27⁻IqA⁺ memory B cells, and to a lesser extent in IqM-only, natural effector and CD27⁻IqG⁺ B cells. The blood counts of all memory B-cell subsets were highly dynamic in children in the first two years of life with peak levels around 14 months of age. IgM-only, CD27⁺IgG⁺, CD27⁺IgA⁺ and CD27⁻IgA⁺ memory B-cell counts at 14 months of age were significantly lower in EBV seropositive children than in uninfected controls. However, at 6 years, these counts were normalized, as were plasma IgG levels to previous primary measles and booster tetanus vaccinations. Thus, EBV persists mainly in Iq classswitched memory B-cells, even when derived from T-cell independent responses (CD27⁻IqA⁺), and results in transient depletion of these cells in young children. Combined, our studies demonstrate the impact of EBV infection on immunological memory in young children, as well as the plasticity of the immune system to overcome this.

INTRODUCTION

The Epstein Barr virus (EBV) is a gamma-herpesvirus that is ubiquitous in the human population. Primary infection occurs predominantly in childhood, after which EBV establishes life-long persistence by inducing viral latency in B cells.^{1,2} EBV targets B cells through interactions with complement receptor 2 (CR2; CD21), MHC-II co-receptor surface molecules, or the Beta-1 integrin.³⁻⁵ Even though all B cells express these molecules, EBV persists preferentially in memory B cells.^{1,6} It is thought that this is the result of EBV infection in activated B cells that undergo an immune response after which the virus persists in the long-lived memory B-cell compartment.^{1,7-9} Recent studies, however, indicate that both activated naive and memory B cells can be infected,¹⁰ and that EBV can establish persistence in the absence of fully functional germinal center activity.¹¹

Primary EBV infection induces a strong expansion of memory B cells that normalizes within 1 week after the appearance of clinical complications.¹² It is likely that infected cells are killed by the virus or anti-viral immune responses. However, there is limited information on how EBV infection and persistence affect the memory B-cell compartment.^{9,12} Importantly, about 40% of EBV infections occur in the first 5 years of life,^{13,14} during which the memory B-cell compartment is gradually build up.¹⁵ Human memory B cells are phenotypically diverse. The majority originates from T-cell dependent germinal center responses: CD27⁺IgD⁻IgM⁺ 'IgM-only', CD27⁻IgG⁺, CD27⁺IgG⁺ and CD27⁺IgA⁺ memory B cells.¹⁶ In addition, T-cell independent responses in the splenic marginal zone and the intestinal lamina propria can generate CD27⁺IgD⁺ IgM⁺ 'natural effector' and CD27⁻IgA⁺ memory B cells, respectively.¹⁶⁻¹⁸

We here studied EBV persistence in all 6 memory B-cell subsets and investigated how EBV infection during the first year of life affects the memory B-cell counts in young children at the ages of 14 months and 6 years.

METHODS

DETECTION OF EBV GENOMES IN ISOLATED MEMORY B-CELL SUBSETS IN EBV+ ADULTS

Buffy coat material of 5 EBV-positive adults was obtained from Sanquin (Amsterdam, NL) and mononuclear cells were isolated by Ficoll-density gradient separation. Genomic DNA of 3.0×10^6 mononuclear cells was isolated (Sigma-Aldrich) to determine the presence of EBV genomes with a quantitative PCR directed against specific for BALF5 (EBV DNA polymerase).^{6,19}

The remaining post-Ficoll mononuclear cells were presorted using human CD19⁺ MicroBeads (Miltenyi Biotech) on an autoMACS system and stored in liquid nitrogen. From EBV-positive donors, the CD19⁺ B cells were thawed and B-cell subsets were purified after labeling with CD38-PE-Cy7 (clone HB7), CD24-APC-H7 (ML5), CD19-PerCP-Cy5.5 (SJ25C1), CD27-APC (L128), IgD-PO (IA6-2), IgM-PB (G20-127), IgG-PE (G18-145; all from BD Biosciences), and IgA-FITC (IS11-8E10; Miltenyi Biotech). Cell sorting was performed on a FACS Aria I (BD Biosciences) using standardized measurement settings.²⁰ Genomic DNA was isolated, and EBV genome copies per 1x10⁶ isolated cells were determined as described above.

THE GENERATION R POPULATION COHORT

Analysis of EBV infection in young children was conducted in the context of the Generation R study, which is a prospective population-based cohort study from fetal life until young adulthood.^{21,22} In this study, a subgroup of 1,182 Dutch children were included, who were all born between August 2003 and August 2006. Ethical approval for the study was obtained from the Medical Ethical Committee of the Erasmus MC, University Medical Center Rotterdam. Written informed consent was obtained from all parents of participating children.

FLOWCYTOMETRIC IMMUNOPHENOTYPING

Peripheral blood was obtained at the mean ages of 0 (cord blood; n=220), 6 (n=376), 14 (n=241), 25 (n=257) or 72 (n=916) months for detailed immunophenotyping. With 1-5 measurements per child this resulted in a total of 2,010 data points. Absolute numbers of CD19⁺ B cells were obtained with a routine diagnostic lyseno-wash protocol and measured on a FACSCalibur (BD Biosciences). Six memory B-cell populations were defined,¹⁶ using the following antibodies: CD19-PerCP (clone SJ25C1), CD27-APC-H7 (clone M-T27/L128), IgM-APC (polyclonal; all from BD Biosciences), IgD-FITC (polyclonal), IgG-PE (polyclonal; both Southern Biotech), and IgA-FITC (polyclonal; Kallestad). Flowcytometry was performed on a BD LSRII (BD Biosciences) using standardized measurement settings.²⁰

EBV AND MEASLES SEROLOGY

IgG antibody levels against the EBV viral capsid antigen (EBV-VCA) were determined in plasma samples from age 14 months and age 6 years of a subset of 219 children with enzyme-linked immunoassays (EUROIMMUN®). EBV positivity was determined based on a ratio of 0.8 of the sample over a manufacturer-provided reference threshold sample. In the same 219 children, anti-measles and anti-tetanus IgG levels were determined at 6 years of age (EUROIMMUN®). The presence of more than 275 international units per ml (IU/ml) for measles or 0.5IU/ml for tetanus were defined as protective vaccination responses.

STATISTICAL ANALYSIS

Statistical comparison of EBV- and EBV+ groups were performed using the Mann-Whitney U test, paired T-test or Fisher's exact test as indicated in figure legends. P-values <0.05 were considered statistically significant.

To model the memory B-cell dynamics between birth and the age of 6 years, linear mixed effect analyses were performed on the relationship between the age of the children and the size of individual memory B-cell populations. By including random-effects in the model, this approach enabled modeling of cross-sectional data, with further improvement of the accuracy by incorporating longitudinal follow-up data from individual children. To capture the trend in the data more precisely, we included a natural spline with different knots (0-3 knots) into the models. Basically, the number of knots is inversely related to the smoothness of the curve. Positions of the knots in the 1-knot model was defined as the 50th percentile (25.5 months), the 2-knots model at 33rd and 66th percentiles (14.1 and 70 months); the knots in the 3-knot model were defined manually at 6, 14

and 24 months, focusing around the time points of data inclusion. Model selection was performed by likelihood ratio test. Statistical analyses were performed in R (version R-3.2.1).²³

RESULTS

EBV persists in IgA^+ and IgG^+ class-switched memory B cells in adults

To test whether EBV preferentially persists in certain B-cell subsets, we purified 8 previously defined populations from human blood.¹⁶ Naive B cells were separated into CD38⁺CD24⁺ transitional and CD38^{dim}CD24^{dim}CD27⁻IgD⁺IgM⁺ naive mature B cells. In addition 6 CD38^{dim}CD24^{dim} memory B-cell subsets were studied: CD27⁺IgM⁺IgD⁺ 'natural effector', CD27⁺IgM⁺IgD⁻ 'IgM-only', CD27⁺IgA⁺, CD27⁻IgA⁺, CD27⁻IgA⁺, CD27⁻IgG⁺ and CD27⁻IgG⁺ memory B cells (Figure 1A). The numbers of EBV genomes per million cells were determined in all subsets from 5 EBV+ adults and represented relative to the number of EBV genome copies in CD27⁺IgG⁺ memory B cells (Figure 1B). EBV genomes were readily detectable in CD27⁺Ig-class switched memory B cells, but hardly detected in transitional, naive mature and IgM-only B cells (containing <8% of the number of copies detected in CD27⁺IgG⁺ memory B cells).^{1,6,11} While low numbers of CD27⁺IgM⁺IgD⁺ and CD27⁻IgG⁺ memory B cells, carried EBV (25% and 39% compared to CD27⁺IgG⁺ memory B cells,



Figure 1. EBV persistence in isolated memory B-cell subsets in adults.

A) Definition of transitional B cells, naive mature B cells and IgM⁺, IgG⁺ or IgA⁺ memory B-cell subsets. B) Relative EBV load in the populations defined as in A in peripheral blood of 5 EBV+ adults. Data is depicted as mean (+ SEM) EBV load relative to EBV copies/106 cells CD27+IgG+ in memory B cells. Number of donors tested in each individual populations is indicated under the plot. Significance was tested between all subsets and CD27+IgG+ memory B cells with paired T-tests. *, p<0.05; **, p<0.01; ****, p<0.0001.

respectively), EBV was as frequent in CD27⁻IgA⁺ memory B cells as in the CD27⁺ Ig-class switched memory B cells. Thus, EBV can persist in all 6 memory B-cell populations, but is most abundant in the T-cell dependent CD27⁺IgG⁺, CD27⁺IgA⁺ subsets and the T-cell independent CD27⁻IgA⁺ subset.

Dynamics of memory ${\bf B}$ cells in the first two years of life

In previous transversal studies, it has been found that memory B cells are gradually build-up in the first years of life.^{15,24-27} To study the longitudinal dynamics of memory B cells in the first years of life, we performed repeated flowcytometric immunophenotyping measurements between birth and 6 years of age in 1,182 children in the context of the Generation R study. We performed linear mixed effect modeling of memory B-cell populations to define the immune dynamics while taking into account the information from longitudinal measurements within individuals (Figure 2). All memory B-cell populations showed the largest dynamics during the first two years of life. Blood B-cell counts strongly increased within the first 14 months of age, followed by a gradual decline and finally a stabilization of cell counts between 4-6 years. Thus, the build-up of the various memory B-cell subsets occurs most strongly in the first year of life.



Figure 2. Dynamics of memory B-cell populations from birth until 6 years of age.

Included were 1,182 individuals with 1-5 measurements per individual, resulting in a total of 2,010 data points. Individual data is depicted in gray dots; directly consequetive timepoints within one individual are connected by gray lines (0-6m, 6-14m, 14-25m, 25-72m). Longitudinal linear mixed effect models describe the overall dynamics between birth and 6 years of age. Solid black lines represent the modelled data, dashed lines are 90% confidence intervals.

TRANSIENT DECLINE IN MEMORY **B** CELLS OF **EBV** POSITIVE CHILDREN

To study whether EBV infection in the first year of life affects the dynamic development of memory B cells, we studied the naive and memory B-cell counts in blood of 149 14-month-old children, consisting of 13 EBV carriers and 136 EBV-negative children, as assessed by anti-EBV-VCA IgG serology. While total, as well as naive, natural effector and CD27⁻IgG⁺ B cells were similar between EBV+ and EBV- children, four memory B-cell populations were significantly reduced in EBV-infected children: CD27⁺IgD⁻IgM⁺ 'IgM-only', CD27⁻IgA⁺, CD27⁺IgA⁺ and CD27⁺IgG⁺ (40.6-60.6% reduction in median values of EBV+ vs EBV- children) (Figure 3A).

The long-term effects of this decline were studied by follow-up analysis of the blood B cells of the same children at 6 years of age. At this age, 81 children were still EBV negative and 53 became seropositive between 14 months and 6 years of age. Regardless of early (<14 months) or late (>14 months) EBV infection, nearly all memory B-cell counts were comparable to those in uninfected controls (Figure 3B). Only the numbers of IgM-only memory B cells were significantly lower in late EBV+ than in EBV- children (31% decrease) (Figure 3B). Thus, the decline in memory B cells at 14 months seems to be transient and the numbers are normalized at 6 years of age.

EFFECTS OF **B**-CELL DEPLETION ON VACCINATION RESPONSES

To study whether the low memory B-cell counts at 14 months of age were associated with abnormal antibody responses, we measured IgG levels in plasma samples of 6-year-old children to previous primary vaccination against measles and booster vaccination against tetanus. Both vaccinations are performed in the context of the Dutch national vaccination protocol.²⁸ Primary measles vaccination is performed at the age of 14 months when EBV+ children show a decline in memory B cells. Tetanus vaccination is performed at the ages of 2, 3, 4 and 11 months during the





A) Absolute numbers of total B cells and naive, IgM⁺ memory, IgA⁺ memory or IgG⁺ memory B-cell subsets in EBV- (n=131) and EBV+ (n=13) children at the age of 14 months. **B)** Absolute numbers of total B cells and naive, IgM⁺ memory, IgA⁺ memory or IgG⁺ memory B-cell subsets at the age of 6 years in EBV- children (n=81), children infected with EBV between 1-6 years of age (late EBV+; n=52) and children infected with EBV within the first 14 months of age (early EBV+; n=13). Bars depict median and interquartile range. Significance was tested by Mann-Whitney U test relative to EBV- individuals. *, p<0.05; **, p<0.01; ***, p<0.001.

first year of life, followed by a booster at 4 years (Figure 4A). Thus, in plasma of children at 6 years of age, we could evaluate build-up and/or maintenance of protective primary responses against measles, as well as memory responses against tetanus. Both the anti-measles (Figure 4B) and the anti-tetanus (Figure 4C) IgG levels at 6 years of age were similar between early EBV+, late EBV+ and EBV- children. Furthermore, the percentage of children with a protective response was similar between early EBV+, late EBV+ and EBV- children, with >80% (measles) and >67% (tetanus) of children having a protective titer at 6 years. Thus, both primary responses against measles and memory responses against tetanus were not affected by EBV infection and/or memory B-cell depletion in the first year of life.



Figure 4. Vaccination responses in EBVinfected children.

A) Schematic representation of the measles and tetanus vaccination protocol in the Netherlands ²⁸. Tetanus vaccinations were performed at the age of 2,3,4,11 months and 4 years, and measles vaccination at 14 months of age. EBV+ children carried low numbers of memory B cells at 14 months of age, and IgG titers to previous vaccinations were determined at 6 years of age. **B,C)** Anti-measles IgG titers (B) and anti-tetanus IgG titers (C) in international units per ml (IU/ml) at 6-years of age in EBV-(n=73), late EBV+ (n=47) or early EBV+ children (n=12). The threshold for a protective vaccination response (anti-measles, 275 IU/ ml; anti-tetanus, 0.5 IU/ml) is indicated by the dashed line. The frequencies of individuals with vaccination responses above this threshold are indicated at the top of the plots. Significance was tested by Mann-Whitney U test or Fisher's exact test relative to EBV- individuals. No significant differences were observed; all p values >0.46.

DISCUSSION

We here studied the persistence of EBV in memory B-cell populations and its effect on blood memory B-cell counts. EBV was able to persist in all class-switched memory B-cell populations. Memory B-cell populations were highly dynamic in size in the first 2 years of life, peaking around 14 months of age. Early EBV infection associated with a transient decline of memory B-cell numbers at 14 months of age, which normalized before 6 years and which did not affect vaccination responses against measles and tetanus.

In line with previous observations,^{1,6,11} we hardly detected any EBV genomes in naive B cells, and only to a limited extent in IgM-only memory B cells and natural effector B cells.^{1,6,11} Interestingly, EBV genome copies were equally numerous in CD27⁻IgA⁺ memory B cells as in the classical CD27⁺ IgA⁺ and IgG⁺ subsets. CD27⁻IgA⁺ memory B cells appear to originate through germinal centerindependent maturation pathways in local tissue, as they have been observed in patients with a lack of germinal centers due to a mutation in the CD40L gene and show strong similarities with IgA⁺ memory B cells in the intestinal lamina propria.¹⁶ The observation that these mucosa-derived CD27⁻IgA⁺ memory B cells contained the highest number of EBV copies, therefore, fits with the oral pathway of EBV infection. Previously, EBV was also detected in natural effector B cells that originate from T-cell independent responses in the splenic marginal zone.^{11,18,29,30} Our observations therefore strengthen the concept that EBV is able to infect cells outside of the germinal center, not only in the spleen, but also in mucosal tissue.

We here observed decreased numbers of memory B cells in blood of 14-month-old EBV-infected children. These children were unlikely to be in the phase of acute infection, as they had a positive IqG serology against the viral capsid antigen (VCA) of EBV. Since these antibodies typically appear a week after acute infection, and remain positive for life,³¹ these children most likely were infected during their first year of life. Even though the impact of EBV on T-cell populations has been extensively described,^{13,32} effects of EBV on the B-cell compartment are less well described. Primary EBV infection was shown to result in a large relative expansion of EBV+ B cells, which subsequently contracted resulting in low frequencies of virus-carrying cells 1 week after appearance of clinical symptoms.¹² This depletion of virus-expressing cells is most likely the result of virus-induced or immunemediated cell death. Our observations extend these observation with long-term effects on memory-B-cell numbers in young children. Importantly, we did not observe effects on memory B cell numbers in older children who were infected >14 month of age. During the first 5 years of life, memory B cells show dynamic changes in cell number with peaks around 14 month of age. It is therefore conceivable that EBV infection has the greatest effect at this age.

Despite a reduction in memory B-cell numbers at 14 months of age, early EBVinfected children had normal memory B-cells counts at 6 years of age, and carried normal IgG titers to previous measles and tetanus vaccinations. Since antigen-specific memory B-cell numbers correlate well with anti-measles IgG titers in the first years after vaccination,^{33,34} it appears that these were normally formed in our EBV+ children at 14 months. Moreover, anti-tetanus IgG titers were normal in 6-year-old EBV-infected children. These children had received a booster vaccination at 4 years after initial vaccinations during the first year of life. This booster is expected to trigger long-lived memory B cells. Thus, despite a depletion in memory B cell, the remaining cells were sufficient to mount an antigen-specific response with protective IgG levels. Thus, the B-cell response shows large plasticity in immunological memory in these young children. Still, the outcomes we measured are of low-resolution. We cannot exclude that the antigenspecific memory B cells have a more restricted Ig gene repertoire following EBVassociated memory B cell depletion.

It is unlikely that natural exposure to measles or tetanus had a large effect on our observations, because of the >86% vaccination coverage in the Netherlands.³⁵ The last reported national measles outbreak was in 1999/2000, at least two years before the children in our study were born.³⁶ The vast moajority of infected children were not previously vaccinated, and a large-scale outbreak could be prevented due to the high overall vaccination coverage in the Netherlands.³⁵ Tetanus infection is rare in the Netherlands, with 5 or less reported cases of tetanus infection per year.³⁷ It will, therefore, be unlikely that natural exposure increased the antimeasles or anti-tetanus IgG titers and thereby interfered with our analyses.

Altogether, our data indicates that EBV is able to persist in all class-switched memory B-cell populations in blood, including CD27⁻IgA⁺ and CD27⁻IgG⁺ memory

B cells, and that although early EBV infection associated with a significantly smaller memory B-cell expansion in the first year of life, long-term effect on the composition of the blood memory B-cell compartment or its function in vaccination responses seem to be limited.

ACKNOWLEDGEMENTS

This work was supported by an Erasmus MC Fellowship to M.C.v.Z. The Generation R Study is conducted by the Erasmus MC, Erasmus University Rotterdam in close collaboration with the School of Law and Faculty of Social Sciences of the Erasmus University Rotterdam, the Municipal Health Service Rotterdam Metropolitan Area, the Rotterdam Homecare Foundation, and the Stichting Trombosedienst & Artsenlaboratorium Rijnmond. We gratefully acknowledge the contributions of children and parents, general practitioners, hospitals, midwives and pharmacies in Rotterdam. We thank H. Bouallouch-Charif, M.A.W. Smits-te Nijenhuis, M.J. Kolijn-Couwenberg, D. Zhao, K.A.M. van Kester, N.M.A. Nagtzaam, B.C.M. Dufourvan den Goorbergh and S.J.W Bartol for technical support, and K. Nasserinejad for assistance with biostatistics.

CONFLICT OF INTEREST

All authors declare that no competing interests exist.

D.v.d.H., A.B., A.B.R., M.C.v.Z., and J.J.M.v.D. designed the experiments; D.v.d.H., M.A.E.J. performed experiments and most of the data analyses; A.B. and A.B.R. contributed to determining EBV persistence in memory B-cell subsets. V.W.V.J. and H.A.M. designed the Generation R study and coordinated material collection. D.v.d.H. and M.C.v.Z wrote the manuscript; and all authors commented on the manuscript.

REFERENCES

- 1. Babcock GJ, Decker LL, Volk M, Thorley-Lawson DA. EBV persistence in memory B cells in vivo. *Immunity*. 1998; 9(3): 395-404.
- Niedobitek G, Agathanggelou A, Herbst H, Whitehead L, Wright DH, Young LS. Epstein-Barr virus (EBV) infection in infectious mononucleosis: virus latency, replication and phenotype of EBV-infected cells. J Pathol. 1997; 182(2): 151-159.
- 3. Fingeroth JD, Weis JJ, Tedder TF, Strominger JL, Biro PA, Fearon DT. Epstein-Barr virus receptor of human B lymphocytes is the C3d receptor CR2. *Proc Natl Acad Sci U S A*. 1984; 81(14): 4510-4514.
- 4. Li Q, Spriggs MK, Kovats S, Turk SM, Comeau MR, Nepom B, et al. Epstein-Barr virus uses HLA class II as a cofactor for infection of B lymphocytes. *J Virol*. 1997; 71(6): 4657-4662.
- 5. Dorner M, Zucol F, Alessi D, Haerle SK, Bossart W, Weber M, et al. beta1 integrin expression increases susceptibility of memory B cells to Epstein-Barr virus infection. *J Virol*. 2010; 84(13): 6667-6677.
- Chaganti S, Heath EM, Bergler W, Kuo M, Buettner M, Niedobitek G, et al. Epstein-Barr virus colonization of tonsillar and peripheral blood B-cell subsets in primary infection and persistence. *Blood*. 2009; 113(25): 6372-6381.
- Kurth J, Spieker T, Wustrow J, Strickler GJ, Hansmann LM, Rajewsky K, et al. EBV-infected B cells in infectious mononucleosis: viral strategies for spreading in the B cell compartment and establishing latency. *Immunity*. 2000; 13(4): 485-495.
- Kurth J, Hansmann ML, Rajewsky K, Kuppers R. Epstein-Barr virus-infected B cells expanding in germinal centers of infectious mononucleosis patients do not participate in the germinal center reaction. *Proc Natl Acad Sci U S A*. 2003; 100(8): 4730-4735.

- 9. Heath E, Begue-Pastor N, Chaganti S, Croom-Carter D, Shannon-Lowe C, Kube D, et al. Epstein-Barr virus infection of naive B cells in vitro frequently selects clones with mutated immunoglobulin genotypes: implications for virus biology. *PLoS Pathog.* 2012; 8(5): e1002697.
- Ehlin-Henriksson B, Gordon J, Klein G. B-lymphocyte subpopulations are equally susceptible to Epstein-Barr virus infection, irrespective of immunoglobulin isotype expression. *Immunology*. 2003; 108(4): 427-430.
- 11. Chaganti S, Ma CS, Bell AI, Croom-Carter D, Hislop AD, Tangye SG, et al. Epstein-Barr virus persistence in the absence of conventional memory B cells: IgM+IgD+CD27+ B cells harbor the virus in X-linked lymphoproliferative disease patients. *Blood*. 2008; 112(3): 672-679.
- 12. Souza TA, Stollar BD, Sullivan JL, Luzuriaga K, Thorley-Lawson DA. Influence of EBV on the peripheral blood memory B cell compartment. *J Immunol*. 2007; 179(5): 3153-3160.
- Heuvel Dvd, Jansen MAE, Dik WA, Bouallouch-Charif H, Zhao D, Kester KAMv, et al. CMV- and EBVinduced T-cell expansions in young children do not impair naive T-cell populations or vaccination responses. The Generation R Study *Journal of Infectious Diseases* 2015.
- 14. Dowd JB, Palermo T, Brite J, McDade TW, Aiello A. Seroprevalence of Epstein-Barr virus infection in U.S. children ages 6-19, 2003-2010. *PLoS One*. 2013; 8(5): e64921.
- 15. Driessen GJ, Dalm VA, van Hagen PM, Grashoff HA, Hartwig NG, van Rossum AM, et al. Common variable immunodeficiency and idiopathic primary hypogammaglobulinemia: two different conditions within the same disease spectrum. *Haematologica*. 2013; 98(10): 1617-1623.
- Berkowska MA, Driessen GJ, Bikos V, Grosserichter-Wagener C, Stamatopoulos K, Cerutti A, et al. Human memory B cells originate from three distinct germinal center-dependent and -independent maturation pathways. *Blood*. 2011; 118(8): 2150-2158.
- He B, Xu W, Santini PA, Polydorides AD, Chiu A, Estrella J, et al. Intestinal bacteria trigger T cellindependent immunoglobulin A(2) class switching by inducing epithelial-cell secretion of the cytokine APRIL. *Immunity*. 2007; 26(6): 812-826.
- Weller S, Braun MC, Tan BK, Rosenwald A, Cordier C, Conley ME, et al. Human blood IgM "memory" B cells are circulating splenic marginal zone B cells harboring a prediversified immunoglobulin repertoire. *Blood*. 2004; 104(12): 3647-3654.
- 19. Junying J, Herrmann K, Davies G, Lissauer D, Bell A, Timms J, et al. Absence of Epstein-Barr virus DNA in the tumor cells of European hepatocellular carcinoma. *Virology*. 2003; 306(2): 236-243.
- Kalina T, Flores-Montero J, van der Velden VH, Martin-Ayuso M, Bottcher S, Ritgen M, et al. EuroFlow standardization of flow cytometer instrument settings and immunophenotyping protocols. *Leukemia*. 2012; 26(9): 1986-2010.
- 21. Jaddoe VW, van Duijn CM, Franco OH, van der Heijden AJ, van Iizendoorn MH, de Jongste JC, et al. The Generation R Study: design and cohort update 2012. *Eur J Epidemiol*. 2012; 27(9): 739-756.
- 22. Kruithof CJ, Kooijman MN, van Duijn CM, Franco OH, de Jongste JC, Klaver CC, et al. The Generation R Study: Biobank update 2015. *Eur J Epidemiol*. 2014; 29(12): 911-927.
- 23. R Core Team; R Foundation for Statistical Computing V, Austria. R: A language and environment for statistical computing. 2015.
- 24. van Gent R, van Tilburg CM, Nibbelke EE, Otto SA, Gaiser JF, Janssens-Korpela PL, et al. Refined characterization and reference values of the pediatric T- and B-cell compartments. *Clin Immunol*. 2009; 133(1): 95-107.
- 25. Duchamp M, Sterlin D, Diabate A, Uring-Lambert B, Guerin-El Khourouj V, Le Mauff B, et al. B-cell subpopulations in children: National reference values. *Immun Inflamm Dis.* 2014; 2(3): 131-140.
- Huck K, Feyen O, Ghosh S, Beltz K, Bellert S, Niehues T. Memory B-cells in healthy and antibodydeficient children. *Clin Immunol.* 2009; 131(1): 50-59.
- 27. Morbach H, Eichhorn EM, Liese JG, Girschick HJ. Reference values for B cell subpopulations from infancy to adulthood. *Clin Exp Immunol*. 2010; 162(2): 271-279.
- F. Abbink, H.G.A.M. van de Avoort, W.A.M. Berbers, R.S. van Binnendijk, H.J. Boot, Y.T.H.P. van Duynhoven, et al. The National Immunisation Programme in the Netherlands_Developments in 2006 *Centre for Infectious Diseases Control, RIVM.* 2006.
- 29. Weller S, Faili A, Garcia C, Braun MC, Le Deist FF, de Saint Basile GG, et al. CD40-CD40L independent Ig gene hypermutation suggests a second B cell diversification pathway in humans. *Proc Natl Acad Sci U S A*. 2001; 98(3): 1166-1170.
- 30. Seifert M, Kuppers R. Molecular footprints of a germinal center derivation of human IgM+(IgD+) CD27+ B cells and the dynamics of memory B cell generation. *J Exp Med*. 2009; 206(12): 2659-2669.

- Schillinger M. KM, Hennigner K., Murray G., Hanselmann I., Bauer G. Variability of humoral immune response to acute Epstein-Barr virus (EBV) infection: evaluation of the significance of serological markers. *Med Microbiol Lett.* 1993; 2(296-303.
- 32. Appay V, van Lier RA, Sallusto F, Roederer M. Phenotype and function of human T lymphocyte subsets: consensus and issues. *Cytometry A*. 2008; 73(11): 975-983.
- 33. Kakoulidou M, Ingelman-Sundberg H, Johansson E, Cagigi A, Farouk SE, Nilsson A, et al. Kinetics of antibody and memory B cell responses after MMR immunization in children and young adults. *Vaccine*; 31(4): 711-717.
- 34. Buisman AM, de Rond CG, Ozturk K, Ten Hulscher HI, van Binnendijk RS. Long-term presence of memory B-cells specific for different vaccine components. *Vaccine*. 2009; 28(1): 179-186.
- 35. de Melker HE, van den Hof S, Berbers GA, Conyn-van Spaendonck MA. Evaluation of the national immunisation programme in the Netherlands: immunity to diphtheria, tetanus, poliomyelitis, measles, mumps, rubella and Haemophilus influenzae type b. *Vaccine*. 2003; 21(7-8): 716-720.
- van den Hof S, Meffre CM, Conyn-van Spaendonck MA, Woonink F, de Melker HE, van Binnendijk RS. Measles outbreak in a community with very low vaccine coverage, the Netherlands. *Emerg Infect Dis*. 2001; 7(3 Suppl): 593-597.
- 37. te Wierik MJ, Hahne SJ, van Ooik PC, van Lier AM, Swaan C. [Tetanus prophylaxis after an injury; check the need for vaccination and immunoglobulin]

Tetanusprofylaxe na verwonding: check de indicatie voor vaccinatie en immunoglobuline. *Ned Tijdschr Geneeskd*. 2013; 157(38): A5906.
CHAPTER 7

PERSISTENT SUBCLINICAL IMMUNE DEFECTS IN HIV-1 INFECTED CHILDREN TREATED WITH ANTIRETROVIRAL THERAPY

Running title: Immune defects in HIV+ children on ART

D. van den Heuvel,¹ G.J.A. Driessen,² M.A. Berkowska,^{1,*} M. van der Burg,¹ A.W. Langerak,¹ D. Zhao,¹ H. Charif,¹ N.G. Hartwig,^{2,#} A.M.C. van Rossum,² P.L.A. Fraaij,^{2,3} J.J.M. van Dongen,¹ M.C. van Zelm.¹

¹ Dept. Immunology, Erasmus MC, University Medical Center, Rotterdam, NL ² Dept. Pediatrics, division of Infectious Diseases, Immunology and Rheumatology, Erasmus MC, University Medical Center-Sophia, Rotterdam, NL ³ Dept. Viroscience, Erasmus MC, University Medical Center, Rotterdam, NL ^{*} Present address: Dept. Experimental Immunohematology, Sanquin Research, and Landsteiner Laboratory, Academic Medical Center, University of Amsterdam, Amsterdam, NL [#] Present address: Sint Franciscus Gasthuis, Rotterdam, NL

AIDS (2015), in press



ABSTRACT

OBJECTIVES. With the introduction of combined antiretroviral treatment (cART), HIV-infected children can reach adulthood with minimal clinical complications. However, long-term HIV and cART in adults is associated with immunosenescence and end-organ damage. Long-term consequences of HIV and cART in children are currently unknown.

DESIGN AND METHOD. We studied 69 HIV-infected children and adolescents under cART (0-23y) for the occurrence of subclinical immunological aberrations in blood B and T cells, using detailed flow cytometric immunophenotyping and molecular analyses.

Results. Children with undetectable plasma HIV viral loads for >1 year showed near-normal to normal CD4⁺ T-cell numbers and near-normal numbers of most class-switched memory B cells. Furthermore, expansions of aberrant CD21^{low} B cells contracted in patients with virus suppression. In contrast, CD8⁺ effector T cells were increased, and CD4⁺ memory T cells, V γ 9⁺V δ 2⁺ T cells and CD27⁻ IgA⁺ memory B cells were decreased and did not normalize under ART. Moreover, V γ 9⁺V δ 2⁺ T cells showed defects in their T-cell receptor repertoire selection.

CONCLUSION. Our results show the effectiveness of current cART to enable the buildup of phenotypically diverse B-cell and T-cell memory in HIV-infected children. However, several subclinical immune abnormalities were detected, which were partially caused by defective immune maturation. These persistent abnormalities were most severe in adolescents and therefore warrant long-term follow-up of HIV-infected children. Early identification of such immune defects might provide targets for monitoring future treatment optimization.

INTRODUCTION

Current combined antiretroviral therapy (cART) protocols have been successful in the treatment of human immunodefiency virus (HIV)-infected children, who can now reach adulthood with minimal clinical complications.¹ cART effectively induces viral suppression and in most patients plasma HIV RNA levels are below the detection limit within 1 year after start of treatment.² Furthermore, treatment restores total CD4⁺ T cell numbers to levels within the normal range,³ suggesting that the treatment is effectively inhibiting viral replication and inducing immune recovery.^{4,5} Signs of immune activation in HIV-infected adults and children, i.e. the CD8⁺ T-cell expansions and hypergammaglobulinemia, resolve after the initiation of cART.¹ Thus, cART changes HIV-infection from a lethal disease into a chronic, treatable disease.

However, long-term HIV infection has in adults been associated with clinical complications, such as cardiovascular complications, neurological diseases and malignancies.⁶⁻⁸ Both HIV-associated immune aberrations and immunosenescence, as well as long-term cART are important contributors to the development of these diseases.⁶⁻⁸ Immune aberrations in HIV-infected adults include an expansion of CD8⁺ T cells with an effector memory phenotype: CD45RA⁻CCR7⁻. These HIV-specific cells display reduced cytolytic potential and are more prone to apoptosis,⁹⁻¹¹ mechanisms that could contribute to impaired clearing of HIV.^{12,13} Furthermore, the V δ 2-expressing subset of TCRy δ ⁺ T cells is contracted both in mucosal tissues and in blood. This defect is associated with increased microbial translocation in the intestine.¹⁴⁻¹⁶ Finally, HIV-infection affects B-cell responses leading to hypergammaglobulinemia,^{17,18} an expansion of CD27⁺ class-switched memory B cells,^{18,23} of which especially the latter persists despite ART.^{18,24-26}

Long-term effects of HIV infection and cART in children currently remain unclear. The first few years of life are crucial for the proper human immune maturation, and thereby to prepare them for adulthood.²⁷ Even though cART successfully suppressed HIV, it is unclear whether the dynamic adaptive immune maturation is normal in HIV-infected children. Early detection of immune aberrations could be predictive of future complications. Some persistent aberrancies in T-cell and B-cell memory have been described in HIV-infected children,²⁸⁻³⁰ and their nature seems to depend on the child's age at the start of ART.³⁰⁻³⁷ Still, an in-depth study into immune aberrations that might have long-term pathogenic effects has, to our knowledge, not been performed.

We, therefore, studied the blood B- and T-cell compartments in a cohort of 69 perinatally HIV-infected children and adolescents (0-23 years). The vast majority of these individuals responded well to ART with HIV levels below the detection limit and total CD4⁺ T-cell numbers within the normal range. However, several persistent subclinical immune abnormalities could be identified in both B and T cells that warrant long-term follow-up of these perinatally HIV-infected children and might be predictive for future complications.

METHODS

STUDY SUBJECTS

Blood was obtained from 69 HIV-positive children between 0 and 23 years of age during routine outpatient clinic visits (Supplemental Table 1).^{1,5,38} HIV-infected individuals were stringently treated and monitored every 3-6 months at Erasmus MC-Sophia Children's hospital and part of The Dutch vertically HIV-infected pediatric population cohort as registered by the Dutch HIV Monitoring Foundation.^{1,5,39} Except for 1 child that did not give consent, all children of the cohort that were below 23 year of age and visited the outpatient clinic between December 2009 and December 2012 were included in the study. 148 age- and sex-matched HIVnegative healthy individuals were recruited in the same hospital.^{1,5,38} Included were children undergoing orthopedic, ophthalmic, urologic or other non-infectious surgical procedures in otherwise healthy condition. Excluded were children having fever, burns or using antibiotics at the moment of sampling, and children having a known immunodeficiency, cardiovascular disease, coagulopathy, renal or hepatic impairment, and children undergoing organ transplantation. Patients and controls were included following informed consent from the parents according to the quidelines of the Medical Ethics Committee of Erasmus MC.

Flow cytometric immunophenotyping and isolation of B- and T-cell subsets from peripheral blood

Absolute counts of blood CD4⁺ and CD8⁺ T cells were obtained with a diagnostic lyseno-wash protocol. Detailed flow cytometric immunophenotyping was performed on fresh blood samples after red blood cell lysis with ammonium chloride. CD4⁺ T cells, CD8⁺ T cells, TCR $\gamma\delta^+$ T cells and B cells were characterized using the antibodies described in Supplemental Table 2. Flowcytometric analyses were performed on an LSRII or CantoII (both from BD Biosciences) using standardized measurement settings.⁴⁰

Post-Ficoll PBMCs were stored in liquid nitrogen and later thawed for sorting of CD24^{dim}CD38^{dim}CD27⁻CD21^{low} (CD21^{low}) B cells and three CD24^{dim}CD38^{dim}CD21⁺ subsets: CD27⁻IgD⁺ naive mature, CD27⁺IgD⁺ natural effector and CD27⁺IgD⁻ memory B cells on a FACS Aria I (BD Biosciences).

SEQUENCE ANALYSIS OF REARRANGED TCR AND IGH TRANSCRIPTS

The presence of the invariant-T nucleotide within the V δ 2-J δ 1 junctional region ⁴¹ was determined in rearranged transcripts amplified from PBMC cDNA, using the BIOMED-2 V δ 2 forward and J δ 1 reverse primers.⁴² Rearranged V γ 9-J γ 1.2 transcripts were amplified with the BIOMED-2 V γ 9 forward primer and the fluorescently labeled BIOMED-1 J γ 1.2 reverse primer.^{42,43} Amplified products were subjected to GeneScan analyses using an ABI PRISM 3130XL fluorescent sequencer (Applied Biosystems) to determine the frequencies of canonical rearrangements.^{41,44}

IgM and IgG transcripts were amplified and cloned from cDNA of sorted B-cell subsets, using *IGHV3* and *IGHV4/6* leader forward primers and *IGHM* and *IGHG* reverse primers.^{45,46} Sequences were generated on an ABI PRISM 3130XL and analyzed with the IMGT (http://imgt.org/), and BASELINe (http://selection.med. yale.edu/baseline/) software.^{47,48}

B-CELL REPLICATION HISTORY USING THE **KREC** ASSAY

DNA was isolated from sorted B-cell subsets with the GenElute Mammalian Total DNA Miniprep Kit (Sigma-Aldrich) to determine the replication history with the Kappa-deleting Recombination Excision Circles (KREC) assay as described previously on a StepOnePlus Real-Time PCR system (Applied Biosystems).⁴⁹

STATISTICAL ANALYSES

Statistical analyses were performed using the Mann-Whitney U test, unpaired T-test, Fisher exact test or Chi-square test. P-values <0.05 were considered statistically significant.

RESULTS

PATIENT CHARACTERISTICS

In this study, we included 69 children and adolescents who were infected with HIV in early childhood (Supplemental Table 1). Of these, 4 patients had plasma HIV viral loads (VL) >15,000 copies/ml at the moment of sampling (HIVhigh). Of the remaining 65 patients (HIVlow), 26 had detectable VL <900 copies/ml at the time of first sampling or detectable VL at least once in the preceding year (suboptimal viral suppression; HIVdetect) and 39 patients had undetectable VL for >1y (stable viral suppression; HIVund) (Supplemental Table 1).

All HIVlow patients received ART treatment for an average of 7.3y (HIVund) and 5.1y (HIVdetect). Two of the four HIVhigh patients received ART for 9.9 and 12.3y; the other two patients were clinically well and treatment was postponed according to auidelines at that time. Disease severity scores at diagnosis, according to Center for Disease Control and Prevention (CDC) guidelines,⁵⁰ varied between N1 and C3. The nadir CD4⁺ T-cell counts were not significantly different between the three patient groups, as was the age at which the patients reached their nadir CD4. During the study follow-up, patients were clinically well; only 8 children needed a short-term hospitalization within the cumulative cohort follow-up. Total CD4⁺ T-cell numbers at the moment of inclusion were normal in 81.5% of HIVlow patients (87.2% HIVund; 73.1% HIVdetect) and in 50% of HIVhigh patients (1 treated, 1 untreated).²⁷ Serum IgA, IgG and/or IgM levels above the normal range were detected in only 17.9% of HIVund patients, but were significantly more frequent in HIVdetect (53.8%) and HIVhigh (75%) patients.⁵¹ Vaccinations against Hepatitis A and B induced a protective response (>100 International units/L (IU/L)) in 29/29 vaccinated HIVlow children and 1/2 vaccinated HIVhigh patients. Within the HIVlow group, increasing age of the patients correlated to significantly lower nadir CD4 counts, which were furthermore reached at significantly later age. Also treatment was initiated later in older patients and children of 16-23v more frequently showed reduced CD4⁺ T-cell counts at the moment of sampling, suggesting persistent CD4⁺ T-cell defects. Clinical severity score, high serum Ig and plasma HIV RNA levels were, however, independent of the age of the child at inclusion or treatment duration.

Altogether, cART appeared effective in most patients as evidenced by the limited clinical complications, the normal response to vaccinations and the increase in $CD4^+$ T-cell counts.

LOW CD4⁺ MEMORY T-CELL NUMBERS IN CART-RESPONSIVE ADOLESCENTS

CD4⁺ T-cell numbers were within the normal range in most children <16y, but were reduced in many older children and adolescents (Supplemental Table 1). When divided over 5 age-groups (<2y, 2-4y, 5-9y, 10-15y, 16-23y), CD4⁺ T-cell numbers in HIVlow patients were comparable to age-matched controls, except for a significant reduction in adolescents of 16-23y (Figure 1A).

To address whether CD4⁺ T-cell numbers were related to plasma HIV RNA levels, children of 5-20y were separated into HIVund, HIVdetect and HIVhigh (Supplemental Table 1). CD4⁺ T-cell numbers were only slightly, but not significantly, reduced in HIVund patients, and more severely reduced in HIVdetect and HIVhigh patients (Figure 1A).

To study the nature of the CD4⁺ T-cell reduction, we dissected total CD4⁺ T cells into CD45RO⁻CCR7⁺CD27⁺CD28⁺ naive, CD45RO⁺CCR7⁺CD27⁺CD28⁺ central memory (Tcm), CD45RO⁺CCR7⁻ effector memory (TemRO) and CD45RO⁻CCR7⁻ effector memory (TemRA) (Figure 1B). These four subsets were all significantly reduced in 16-23y HIVlow patients (Figure 1C). In younger children, these subsets were normally present, except for significantly increased TemRA in 2-4y and decreased TemRO in 5-9y. Within TemRO, the early differentiated CD27⁺CD28⁺ subset was not only significantly reduced in the 16-23y, but all age groups >5y (Supplemental Figure 1A). The reductions in Tcm and TemRO numbers were similar in HIVund or HIVdetect patients, but slightly stronger in patients with high viral load (Figure 1C and Supplemental Figure 1A). Thus, viral suppression seems to restore the CD4⁺ T-cell compartment to a large extent in HIVlow patients, but especially adolescents showed reduced Tcm and TemRO numbers that are typically seen in viremic patients.





Figure 2. The CD8⁺ T-cell compartment in HIV-infected children and adolescents.

A) CD8⁺ T-cell counts in controls and HIVlow patients under ART (left) and in uninfected controls (ctrl), HIVund, HIVdetect or HIVhigh patients of 5-23y old (right). The number of individuals per group are indicated in parentheses. Plots depict 25-75th percentiles (box) and 10-90th percentile (whiskers). **B)** Definition of naive (Tnaive), central memory (Tcm), CD45RO⁺ effector memory (TemRO) and CD45RO⁻ effector memory (TemRA) subsets within CD3⁺CD8⁺ T cells. **C)** Absolute numbers of CD8⁺ Tnaive, Tcm, TemRO and TemRA subsets presented similarly as in panel A. Statistical significance was determined using the Mann-Whitney U test; *, p<0.05; **, p<0.01; ***, p<0.001; ****, p<0.0001

CD8⁺ EFFECTOR MEMORY T-CELL EXPANSIONS IN YOUNG CHILDREN

Of the 65 HIVlow patients, only 4 showed the increase in total CD8⁺ T-cell numbers that is typically associated with HIV infection. Still, direct comparison of patients groups with age-matched controls showed significantly increased numbers of CD8⁺ T cells in children of 2-4 and 5-9y. (Figure 2A). Subset analysis revealed that this increase was mainly due to significantly high numbers of TemRA and to a lesser extent of TemRO (Figure 2B-C). Within these effector subsets, especially CD27⁻CD28⁻ late effectors were significantly increased in HIV-infected children (Supplemental Figure 1B-C). In line with this, more CD8⁺ T cells expressed CD57 and HLA-DR (not shown).⁵²⁻⁵⁴ The expansions of total and TemRO CD8⁺ effector T

Figure 1. The CD4⁺ T-cell compartment in HIV-infected children and adolescents.

A) CD4⁺ T-cell counts in controls and HIVlow patients under ART (left) and in uninfected controls (ctrl), HIVund, HIVdetect or HIVhigh patients of 5-23y old (right). The number of individuals per group are indicated in parentheses. Plots depict 25-75th percentiles (box) and 10-90th percentile (whiskers). **B)** Definition of naive (Tnaive), central memory (Tcm), CD45RO⁺ effector memory (TemRO) and CD45RO⁻ effector memory (TemRA) subsets within CD3⁺CD8⁻ T cells. **C)** Absolute numbers of CD4⁺ Tnaive, Tcm, TemRO and TemRA subsets presented similarly as in panel A. Statistical significance was determined using the Mann-Whitney U test; *, p<0.05; **, p<0.01; ***, p<0.001; ****, p<0.001.

cells significantly related to the presence of virus, whereas TemRA T-cell expansions were found in all three VL-groups (Figure 2C and Supplemental Figure 1B-C). Thus, good control of viremia seems to inhibit the CD8⁺ effector memory T-cell expansions, but abnormalities can still be found, especially in young children.

Cellular and molecular $TCR\gamma\delta^+$ T-cell repertoire changed in HIV patients

TCR $\gamma\delta^+$ T-cell numbers were low to normal in our patients with the typically increased ratio of V $\delta1^+$ over V $\delta2^+$ T cells (Figure 3A).^{15,16} In young children, this was mainly due to significantly increased numbers of V $\delta1^+$ T cells, whereas older children and adolescents had significantly reduced V $\delta2^+$ T-cell numbers compared to controls (Figure 3B-C). Moreover, V $\gamma9^+$ T-cell numbers were slightly reduced in children of 5-9y and 16-23y, suggestive of a reduced number of V $\gamma9^+$ V $\delta2^+$ T cells, the dominant population in healthy adults (Supplemental Figure 2A-B).⁴⁴

To study the nature of the V γ 9⁺V δ 2⁺ T-cell defects, we analyzed two typical selection determinants in these cells: 1) selection for the invariant-T nucleotide in the V δ 2-J δ 1 junctional regions, normally occurring within the first year of life and present in 90% of healthy adults, and 2) the canonical V γ 9-J γ 1.2 rearrangement, formed through homology-mediated repair and yielding a canonical Complementarity Determining Region 3 (CDR3) of 14 amino acids.^{41,44} In 15 patients, we did not find the invariant-T (Figure 3D). These mainly concerned young children <2y, but their frequency was also significantly reduced in the patients of 2-23y. Furthermore, patients showed a trend towards reduced usage of the canonical V γ 9-J γ 1.2 rearrangements, even despite effective virus suppression in HIVund patients (Supplementary figure 2C). Thus, in adolescents V γ 9⁺V δ 2⁺ T-cells were impaired in numbers and in their TCR maturation and repertoire selection, irrespective of ART.

REDUCTIONS IN IGA⁺ B-CELL MEMORY DESPITE ART

Total B-cell numbers, including the transitional and naive mature B-cell subsets, were normal in HIV patients of all age-groups (Figure 4A and data not shown). Within the memory B-cell compartment, all subsets were decreased in number, except for a slight increase in IgM-only B cells (Figure 4BC). Partial normalization of CD27⁻IgG⁺, CD27⁺IgG⁺ and CD27⁺IgA⁺ memory B-cell numbers was related to viral suppression, whereas the IgM⁺ populations were equally affected in HIVund, HIVdetect and HIVhigh patients. The strongest effect was observed in the CD27⁻IgA⁺ subset that is thought to originate from T-cell independent responses in the gut.⁴⁶ This population was consistently reduced in all age categories and to the same extent in HIVhigh, as well as HIVdetect or HIVund children (Figure 4C). Thus, even though ART seems to partially restore class-switched memory B-cell numbers persist during ART in HIV-infected children.

CD21^{LOW} B CELLS ARE ASSOCIATED WITH HIGH HIV LOADS

HIV-infected adults show expansions of atypical CD21^{low} B cells with molecular signs of antigen-maturation and reactivity to HIV.⁵⁵ They are functionally impaired due to reduced replication history and somatic hypermutation (SHM) levels,

з.

Cellular

rearrangements

significance

of the

and

for

was

invariant-T

(whiskers).

p<0.001;

determined,





Figure 4. The memory B-cell compartment in HIV-infected children and adolescents.

A) CD19⁺ B-cell counts in controls and HIVlow patients under ART (left) and in uninfected controls (ctrl), HIVund, HIVdetect or HIVhigh patients of 5-23y old (right). The number of individuals per group are indicated in parentheses. Plots depict 25-75th percentiles (box) and 10-90th percentile (whiskers). **B)** Definition of 6 memory B-cell subsets: CD27+IgD-IgM+ (IgM-only), CD27+IgD+IgM+ (natural effector), CD27-IgA+, CD27+IgA+, CD27-IgG+ and CD27+IgC+.⁴⁶ **C)** Absolute numbers of 6 memory B-cell subsets in HIV-infected children presented similarly as in panel A. Statistical significance was determined, using the Mann-Whitney U test; *, p<0.05; **, p<0.01; ****, p<0.001.

and reduced responsiveness to stimulation.⁵⁵ Our HIVhigh children also carried significant expansions of CD38^{dim}CD27⁻CD21^{low} B cells (Figure 5AB). Importantly, numbers of these CD21^{low} B cells were near-normal in HIVlow patients in all age groups (Figure 5A-C). To study whether CD21^{low} B-cell numbers were directly related to HIV viral loads, we performed longitudinal analysis (1.7-2.9 year) of 14 patients (Figure 5D and Supplemental Figure 3). In all four patients with high viral loads, we observed increased CD21^{low} B-cell numbers. In 2 patients, the CD21^{low} B-cell numbers increased within 0.5 year following the rise of plasma HIV RNA



Figure 5. legend on the next page

Figure 5. Kinetics and molecular diversity of CD21^{low} B cells of HIV-infected children. A) Definition of CD19⁺CD38^{dm}CD21^{low}CD27⁻ "CD21^{low} B cells". B) CD21^{low} B-cell counts in controls and HIVlow patients under ART (left) and in uninfected controls (ctrl), HIVund, HIVdetect or HIVhigh patients of 5-23y old (right). The number of individuals per group are indicated in parentheses. Plots depict 25-75th percentiles (box) and 10-90th percentile (whiskers). C) CD21tow B-cell frequencies within total CD19+ B cells presented similarly as panel B. D) Longitudinal follow up of HIV plasma loads and CD21^{low} B-cell frequencies in one representative patients with high-VL (left column), low-VL (middle) and decreasing VL (right). E) Sorting strategy and definition of CD21^{low}, naive mature, natural effector and CD27⁺IgD⁻ memory B cells. F) Replication history of purified B-cell subsets of 3 HIV-infected children (patients 28 29 and 34) as determined with the KREC assay.⁴⁹ G) Mutation frequencies in IGHV regions of IgM and IgG transcripts in 5 HIV-infected children (patients 52, 56, 59, 67 and 68). The number of transcripts analyzed in each category is indicated in parentheses. H) BASELINe analysis of IgM and IgG transcripts to determine the selection strength for replacement mutations.^{47,48} **I)** IgG subclass distributions of sequenced IgG transcripts from CD21^{low} and CD27⁺IgD⁻B cells. The number of sequences included is depicted in the center of the chart. Statistical significance was determined using the Mann-Whitney U test (B/C/G) or Chi-Square test (I); *, p<0.05; **, p<0.01; ***, p<0.001; ****, p<0.0001.

levels. Patients with consistently low plasma HIV RNA levels carried a persistently small CD21^{low} population of <20% of total B cells. Finally, patients who successfully started ART showed a concomitant decrease in CD21^{low} B-cell numbers within 0.5 year of decreased plasma HIV RNA levels (Figure 5D and Supplemental Figure 3). Thus, CD21^{low} B-cell numbers seem directly related to active HIV infection and can be normalized by ART in children.

CD21^{LOW} B-CELLS IN HIV PATIENTS ARE A MIXTURE OF NAIVE AND ANTIGEN-EXPERIENCED **B** CELLS

To study the nature of these CD21^{low} B cells, we purified these, as well as the CD21expressing CD27⁻IqD⁺ naive B cells, CD27⁺IqD⁺ natural effector and CD27⁺IqD⁻ memory B cells from 5 HIV-infected children over 13v for comparative analysis of replication history and SHM levels (Figure 5E). Similar to previous observations in adults,⁵⁵ CD21^{low} B cells had a replication history of ~6 cell divisions, which was comparable with natural effector B cells, but clearly less than CD27⁺IgD⁻ memory B cells (Figure 5F). The CD21^{low} population consisted mainly of IgM⁺ (with or without IqD) and IqG^+ B cells (data not shown). To distinguish between these subsets for SHM analysis, we sequenced IGHV genes from separately amplified IGHM and IGHG transcripts. IgG transcripts of CD21^{low} B cells were nearly all mutated, although the SHM loads were slightly lower than in IgG transcripts from classical CD27⁺IqD⁻ memory B cells (Figure 5G). Still, the transcripts of CD21^{low}IqG⁺ and CD27⁺IgD⁻ cells showed similar selection for replacement mutations in CDRs (Figure 5H). Furthermore, both subsets showed similar IgG subclass distributions with the IgM-proximal IgG1 and IgG3 mostly used (Figure 5I). In contrast, about half of the IgM transcripts in CD21^{low} B cells were unmutated. The range of SHM levels and selection for replacement mutations of the mutated clones were more similar to natural effector B cells, but clearly lower than CD27⁺IgD⁻ memory B cells (Figure 5G-H). Thus, our molecular analyses indicate that the CD21^{low} population in HIV patients is actually a mixture of naive and memory B cells, and that the molecular signs of antigen maturation in the IgG⁺ subset are more similar to classical CD27⁺IgG⁺ memory B cells than might have been previously appreciated.

DISCUSSION

We here performed an in-depth study on the effect of perinatal HIV infection and cART treatment on the immune compartment of children and adolescents. Our results show the effectiveness of current cART to enable the build-up of phenotypically diverse B- and T-cell memory in HIV-infected children. This included,

at least in part, the normalization of CD4⁺ T-cell numbers in the youngest children, normalization of class-switched memory B-cell numbers, and the reduction of CD21^{low} B-cell expansions. However, persistent expansions of CD8⁺ effector T-cells, and the reductions in V γ 9⁺V δ 2⁺ T-cells and CD27⁻IgA⁺ memory B cells were not restored by cART.

Nearly all children in our cohort responded well to ART; they hardly had clinical complaints, mounted protective vaccination responses, showed low to undetectable HIV counts in plasma, and carried near-normal blood CD4⁺ T-cell counts.^{4,5} Adolescents more often showed a persistent reduction in their CD4⁺ T-cell numbers. Various reasons might underlie this age-associated effect, including a lack of medication adherence, waning of the T-cell compartment over time due to prolonged HIV infection,⁵⁶ or the change in treatment protocols over the last 10 years (Supplemental Table 1).⁵⁷ The later start of ART treatment in children >10y of age was associated with significantly lower nadir CD4⁺ T-cell counts than in children <10y, suggesting a more disrupted immune compartment in the adolescents. The near-normal CD4⁺ T-cell compartment in the young children might therefore highlight a strong improvement in ART regimens in recent years and might further emphasize the importance of an early start of ART treatment. Still, future longitudinal studies will be needed to address whether the waning of CD4⁺ T cells over time is stably inhibited in these young children.

While the depletion of CD4⁺ T cells in adolescent patients concerned all naive, memory and effector subsets, Tcm and early TemRO cells were most severely affected. Even in adolescents with undetectable HIV for >1y CD4⁺ Tcm and TemRO cells were lower than in uninfected controls, indicating that, once lost, these populations might be impossible to fully restore by ART. Loss of Tcm cells was reported to correlate with rapid disease progression,⁵⁸ whereas a small HIV reservoir in Tcm cells and stable Tcm functionality were observed in long-term non-progressors and natural HIV-controllers.^{59,60} The higher numbers of Tcm cells in young HIVlow patients in our study might, therefore, again indicate an important immunological improvement due to ART, suggesting that early diagnosis and treatment might be important to prevent the initial loss of these populations.

All patients >2y had increased numbers of effector memory CD8⁺ T cells, which might reflect ongoing response caused by residual HIV replication or might potentially be an early sign of immunesenescence. It will be important to longitudinally follow-up these CD8⁺ TemRA expansions, to address whether they indeed correlate to chronic immune activation or possibly indicate early signs of immunosenescence.

The HIV-infected children in our cohort had increased numbers of V δ 1⁺ T cells and decreased V δ 2⁺ T cells, extending previous TCR γ δ ⁺ T-cell aberrations found in blood and the intestinal mucosa of HIV-infected adults.^{15,16,61} The increase in V δ 1⁺ T cells did not result from clonal expansion (data not shown), but could still be the result of chronic stimulation.¹⁵ The low V δ 2⁺ T cell numbers were mainly due to a reduction in the V γ 9⁺V δ 2⁺ subset. On top of the depletion that has previously been described in adults,⁶² we here showed that in children V γ 9⁺V δ 2⁺ T cells showed defective repertoire selection. Because the defects in V γ 9⁺V δ 2⁺ T cells were not restored by ART, their levels could be good biomarkers to monitor future treatment optimization. Still, longitudinal follow-up of our patients will be important to define potential clinical complications that are associated with defects in V γ 9⁺V δ 2⁺ T cells. In our cohort, we observed severely reduced numbers of blood CD27⁻IgA⁺ memory B cells, which are derived from T-cell independent responses in the intestinal lamina propria.⁴⁶ Because these responses do not critically depend on CD4⁺ T-cell help, this defect is most likely independent of any impaired CD4⁺ T-cell responses, but likely depend on other, possibly HIV-targeted, mechanisms.^{63,64} In line with the increased monocyte activation and intestinal microbial translocation observed by others,³⁷ the persistent loss of CD27⁻IgA⁺ B cells might result from persistent intestinal complications and ongoing HIV replication. However, we cannot exclude a possible effect of the oral intake of antivirals, which might affect B-cell responses in the intestine. Studies in HIV-negative individuals receiving post-exposure prophylaxis treatment might help to unravel the effect of treatment on the intestinal immune responses.

CD27⁻IgG⁺, CD27⁺IgG⁺ and CD27⁺IgA⁺ memory B-cell numbers were nearnormal in children under ART. Ig serum levels and memory responses to common vaccination antigens, such as measles, pneumococci, influenza and tetanus antigens are reported to be reduced in HIV-infected patients, and it is debated whether ART is able to restore these.⁶⁵⁻⁶⁸ The normalization of these T-cell dependent memory B-cell subsets in our patients, coinciding with a normalization of CD4⁺ memory T cells and protective hepatitis A/B vaccination responses suggests maintenance or recovery of T-cell dependent humoral responses in our cohort.

We found that CD21^{Iow} B-cell expansions correlated with plasma HIV RNA levels. We, moreover, showed that this subset was composed of both naive and memory IgM⁺ B cells and IgG⁺ B cells. This could explain the low replication history of the total subset. The CD21^{Iow}IgG⁺ subset contained SHM and IgG1 and IgG3 usage that was more similar to CD27⁻IgG⁺ than to CD27⁺IgG⁺ memory B cells.^{46,69} Considering that CD21^{Iow} B cells display poor response to antigen and contain high frequencies of HIV-reactive B cells,⁵⁵ downregulation of CD21 could be a mechanism for HIV to impair the host's protective Ig responses, which, we here showed for the first time, could affect both naive and memory B-cell responsiveness. The direct and stable contraction in CD21^{Iow} B-cell numbers following declines in plasma HIV RNA levels after cART treatment, therefore, indicates an important beneficial effect of cART treatment and makes it a good candidate marker for successful cART.

Altogether, our study showed the effectiveness of current ART to enable the build-up of phenotypically diverse B- and T-cell memory in HIV-infected children, especially in the younger children in our study, who receive the most recent treatment protocols and in whom treatment was started significantly earlier than in adolescents. However, even with undetectable viral loads, subclinical defects in CD4⁺ TemRO and Tcm, CD8⁺ TemRA, V δ 2⁺ T cells and CD27⁻IgA⁺ memory B cells persisted, which were partially caused by defective immune maturation. Careful prospective monitoring of these persistent defects will be important for the early detection of clinical complications, ongoing virus replication or immunosenescence that might arise from these defects when these children grow older.^{37,56}

ACKNOWLEDGEMENTS

The authors are indebted to E.F.E. de Haas and S.J.W. Bartol, M.A.W. Smits-te Nijenhuis, M.J. Kolijn-Couwenberg and N.M.A. Nagtzaam for technical support and I.P.E. Gondrie for patient database management. This work was supported by an Erasmus MC Fellowship to M.C. van Zelm.

The studies were performed in the Department of Immunology (headed by Prof. P. Katsikis) as part of the Molecular Medicine Postgraduate School of the Erasmus MC, Rotterdam, The Netherlands.

CONFLICT OF INTEREST

All authors declare that no competing interests exist.

D.v.d.H., M.A.B., J.J.M.v.D. and M.C.v.Z. designed the experiments; G.J.A.D., M.A.B., M.v.d.B., A.W.L., N.G.H., A.M.C.v.R. and P.L.A.F. provided conceptual advice; D.v.d.H., D.Z. and H.C. performed and analyzed most of the experiments and contributed to data analyses; G.J.A.D., N.G.H., A.M.C.v.R. and P.L.A.F. provided material necessary for performing experiments; D.v.d.H. and M.C.v.Z. wrote the manuscript; and all authors commented on the manuscript.

REFERENCES

- Cohen S, Smit C, van Rossum AM, Fraaij PL, Wolfs TF, Geelen SP, et al. Long-term response to combination antiretroviral therapy in HIV-infected children in the Netherlands registered from 1996 to 2012. *Aids*. 2013; 27(16): 2567-2575.
- 2. Duong T, Judd A, Collins IJ, Doerholt K, Lyall H, Foster C, et al. Long-term virological outcome in children on antiretroviral therapy in the UK and Ireland. *Aids*. 2014; 28(16): 2395-2405.
- 3. van Rossum AM, Scherpbier HJ, van Lochem EG, Pakker NG, Slieker WA, Wolthers KC, et al. Therapeutic immune reconstitution in HIV-1-infected children is independent of their age and pretreatment immune status. *Aids*. 2001; 15(17): 2267-2275.
- Pakker NG, Notermans DW, de Boer RJ, Roos MT, de Wolf F, Hill A, et al. Biphasic kinetics of peripheral blood T cells after triple combination therapy in HIV-1 infection: a composite of redistribution and proliferation. *Nat Med.* 1998; 4(2): 208-214.
- Cohen Stuart JW, Slieker WA, Rijkers GT, Noest A, Boucher CA, Suur MH, et al. Early recovery of CD4+ T lymphocytes in children on highly active antiretroviral therapy. Dutch study group for children with HIV infections. *Aids*. 1998; 12(16): 2155-2159.
- 6. Guaraldi G, Prakash M, Moecklinghoff C, Stellbrink HJ. Morbidity in older HIV-infected patients: impact of long-term antiretroviral use. *AIDS Rev.* 2014; 16(2): 75-89.
- 7. Kaplan RC, Sinclair E, Landay AL, Lurain N, Sharrett AR, Gange SJ, et al. T cell activation predicts carotid artery stiffness among HIV-infected women. *Atherosclerosis*. 2011; 217(1): 207-213.
- 8. Rajasuriar R, Wright E, Lewin SR. Impact of antiretroviral therapy (ART) timing on chronic immune activation/inflammation and end-organ damage. *Curr Opin HIV AIDS*. 2015; 10(1): 35-42.
- 9. Petrovas C, Mueller YM, Katsikis PD. Apoptosis of HIV-specific CD8+ T cells: an HIV evasion strategy. *Cell Death Differ*. 2005; 12 Suppl 1(859-870.
- Appay V, Dunbar PR, Callan M, Klenerman P, Gillespie GM, Papagno L, et al. Memory CD8+ T cells vary in differentiation phenotype in different persistent virus infections. *Nat Med.* 2002; 8(4): 379-385.
- Champagne P, Ogg GS, King AS, Knabenhans C, Ellefsen K, Nobile M, et al. Skewed maturation of memory HIV-specific CD8 T lymphocytes. *Nature*. 2001; 410(6824): 106-111.
- Migueles SA, Weeks KA, Nou E, Berkley AM, Rood JE, Osborne CM, et al. Defective human immunodeficiency virus-specific CD8+ T-cell polyfunctionality, proliferation, and cytotoxicity are not restored by antiretroviral therapy. *J Virol.* 2009; 83(22): 11876-11889.
- Papagno L, Spina CA, Marchant A, Salio M, Rufer N, Little S, et al. Immune activation and CD8+ T-cell differentiation towards senescence in HIV-1 infection. *PLoS Biol*. 2004; 2(2): E20.
- 14. Autran B, Triebel F, Katlama C, Rozenbaum W, Hercend T, Debre P. T cell receptor gamma/delta+ lymphocyte subsets during HIV infection. *Clin Exp Immunol*. 1989; 75(2): 206-210.
- Poles MA, Barsoum S, Yu W, Yu J, Sun P, Daly J, et al. Human immunodeficiency virus type 1 induces persistent changes in mucosal and blood gammadelta T cells despite suppressive therapy. J Virol. 2003; 77(19): 10456-10467.
- 16. Bordon J, Evans PS, Propp N, Davis CE, Jr., Redfield RR, Pauza CD. Association between longer

duration of HIV-suppressive therapy and partial recovery of the V gamma 2 T cell receptor repertoire. *J Infect Dis.* 2004; 189(8): 1482-1486.

- Lane HC, Masur H, Edgar LC, Whalen G, Rook AH, Fauci AS. Abnormalities of B-cell activation and immunoregulation in patients with the acquired immunodeficiency syndrome. *N Engl J Med.* 1983; 309(8): 453-458.
- De Milito A, Morch C, Sonnerborg A, Chiodi F. Loss of memory (CD27) B lymphocytes in HIV-1 infection. *Aids.* 2001; 15(8): 957-964.
- 19. Scott ME, Landay AL, Lint TF, Spear GT. In vivo decrease in the expression of complement receptor 2 on B-cells in HIV infection. *Aids*. 1993; 7(1): 37-41.
- Moir S, Malaspina A, Ogwaro KM, Donoghue ET, Hallahan CW, Ehler LA, et al. HIV-1 induces phenotypic and functional perturbations of B cells in chronically infected individuals. *Proc Natl Acad Sci U S A*. 2001; 98(18): 10362-10367.
- 21. Moir S, Fauci AS. B cells in HIV infection and disease. Nat Rev Immunol. 2009; 9(4): 235-245.
- 22. Amu S, Ruffin N, Rethi B, Chiodi F. Impairment of B-cell functions during HIV-1 infection. *Aids*. 2013; 27(15):2323–2334.
- 23. D'Orsogna LJ, Krueger RG, McKinnon EJ, French MA. Circulating memory B-cell subpopulations are affected differently by HIV infection and antiretroviral therapy. *Aids*. 2007; 21(13): 1747-1752.
- 24. Moir S, Malaspina A, Ho J, Wang W, Dipoto AC, O'Shea MA, et al. Normalization of B cell counts and subpopulations after antiretroviral therapy in chronic HIV disease. *J Infect Dis.* 2008; 197(4): 572-579.
- 25. Morris L, Binley JM, Clas BA, Bonhoeffer S, Astill TP, Kost R, et al. HIV-1 antigen-specific and -nonspecific B cell responses are sensitive to combination antiretroviral therapy. *J Exp Med.* 1998; 188(2): 233-245.
- Titanji K, Chiodi F, Bellocco R, Schepis D, Osorio L, Tassandin C, et al. Primary HIV-1 infection sets the stage for important B lymphocyte dysfunctions. *Aids.* 2005; 19(17): 1947-1955.
- 27. Comans-Bitter WM, de Groot R, van den Beemd R, Neijens HJ, Hop WC, Groeneveld K, et al. Immunophenotyping of blood lymphocytes in childhood. Reference values for lymphocyte subpopulations. *J Pediatr*. 1997; 130(3): 388-393.
- Sharp ER, Willberg CB, Kuebler PJ, Abadi J, Fennelly GJ, Dobroszycki J, et al. Association of differentiation state of CD4+ T cells and disease progression in HIV-1 perinatally infected children. *PLoS One.* 2012; 7(1): e29154.
- 29. Ghosh S, Feyen O, Jebran AF, Huck K, Jetzek-Zader M, Bas M, et al. Memory B cell function in HIVinfected children-decreased memory B cells despite ART. *Pediatr Res.* 2009; 66(2): 185-190.
- Bekker V, Scherpbier H, Pajkrt D, Jurriaans S, Zaaijer H, Kuijpers TW. Persistent humoral immune defect in highly active antiretroviral therapy-treated children with HIV-1 infection: loss of specific antibodies against attenuated vaccine strains and natural viral infection. *Pediatrics*. 2006; 118(2): e315-322.
- Romiti ML, Cancrini C, Castelli-Gattinara G, Di Cesare S, Ciaffi P, Bernardi S, et al. Kinetics of the T-cell receptor CD4 and CD8 V beta repertoire in HIV-1 vertically infected infants early treated with HAART. *Aids.* 2001; 15(16): 2075-2084.
- 32. Pensieroso S, Cagigi A, Palma P, Nilsson A, Capponi C, Freda E, et al. Timing of HAART defines the integrity of memory B cells and the longevity of humoral responses in HIV-1 vertically-infected children. *Proc Natl Acad Sci U S A*. 2009; 106(19): 7939-7944.
- 33. Palma P, Romiti ML, Cancrini C, Pensieroso S, Montesano C, Bernardi S, et al. Delayed early antiretroviral treatment is associated with an HIV-specific long-term cellular response in HIV-1 vertically infected infants. *Vaccine*. 2008; 26(40): 5196-5201.
- 34. Hainaut M, Ducarme M, Schandene L, Peltier CA, Marissens D, Zissis G, et al. Age-related immune reconstitution during highly active antiretroviral therapy in human immunodeficiency virus type 1-infected children. *Pediatr Infect Dis J.* 2003; 22(1): 62-69.
- 35. Cotugno N, Douagi I, Rossi P, Palma P. Suboptimal immune reconstitution in vertically HIV infected children: a view on how HIV replication and timing of HAART initiation can impact on T and B-cell compartment. *Clin Dev Immunol.* 2012; 2012(805151):1-11.
- 36. Carcelain G, Debre P, Autran B. Reconstitution of CD4+ T lymphocytes in HIV-infected individuals following antiretroviral therapy. *Curr Opin Immunol*. 2001; 13(4): 483-488.
- 37. Persaud D, Patel K, Karalius B, Rainwater-Lovett K, Ziemniak C, Ellis A, et al. Influence of Age at Virologic Control on Peripheral Blood Human Immunodeficiency Virus Reservoir Size and Serostatus

in Perinatally Infected Adolescents. JAMA Pediatr. 2014; 168(12):1138-1146.

- 38. Driessen GJ, Dalm VA, van Hagen PM, Grashoff HA, Hartwig NG, van Rossum AM, et al. Common variable immunodeficiency and idiopathic primary hypogammaglobulinemia: two different conditions within the same disease spectrum. *Haematologica*. 2013; 98(10): 1617-1623.
- 39. Fraaij PL, Verweel G, van Rossum AM, van Lochem EG, Schutten M, Weemaes CM, et al. Sustained viral suppression and immune recovery in HIV type 1-infected children after 4 years of highly active antiretroviral therapy. *Clin Infect Dis.* 2005; 40(4): 604-608.
- Kalina T, Flores-Montero J, van der Velden VH, Martin-Ayuso M, Bottcher S, Ritgen M, et al. EuroFlow standardization of flow cytometer instrument settings and immunophenotyping protocols. *Leukemia*. 2012; 26(9): 1986-2010.
- Breit TM, Wolvers-Tettero IL, van Dongen JJ. Unique selection determinant in polyclonal V delta 2-J delta 1 junctional regions of human peripheral gamma delta T lymphocytes. J Immunol. 1994; 152(6): 2860-2864.
- 42. van Dongen JJ, Langerak AW, Bruggemann M, Evans PA, Hummel M, Lavender FL, et al. Design and standardization of PCR primers and protocols for detection of clonal immunoglobulin and T-cell receptor gene recombinations in suspect lymphoproliferations: report of the BIOMED-2 Concerted Action BMH4-CT98-3936. *Leukemia*. 2003; 17(12): 2257-2317.
- 43. Pongers-Willemse MJ, Seriu T, Stolz F, d'Aniello E, Gameiro P, Pisa P, et al. Primers and protocols for standardized detection of minimal residual disease in acute lymphoblastic leukemia using immunoglobulin and T cell receptor gene rearrangements and TAL1 deletions as PCR targets: report of the BIOMED-1 CONCERTED ACTION: investigation of minimal residual disease in acute leukemia. *Leukemia*. 1999; 13(1): 110-118.
- 44. Sandberg Y, Almeida J, Gonzalez M, Lima M, Barcena P, Szczepanski T, et al. TCRgammadelta+ large granular lymphocyte leukemias reflect the spectrum of normal antigen-selected TCRgammadelta+ T-cells. *Leukemia*. 2006; 20(3): 505-513.
- 45. Tiller T, Meffre E, Yurasov S, Tsuiji M, Nussenzweig MC, Wardemann H. Efficient generation of monoclonal antibodies from single human B cells by single cell RT-PCR and expression vector cloning. *J Immunol Methods*. 2008; 329(1-2): 112-124.
- 46. Berkowska MA, Driessen GJ, Bikos V, Grosserichter-Wagener C, Stamatopoulos K, Cerutti A, et al. Human memory B cells originate from three distinct germinal center-dependent and -independent maturation pathways. *Blood*. 2011; 118(8): 2150-2158.
- 47. Uduman M, Yaari G, Hershberg U, Stern JA, Shlomchik MJ, Kleinstein SH. Detecting selection in immunoglobulin sequences. *Nucleic Acids Res.* 2011; 39(Web Server issue): W499-504.
- 48. Yaari G, Uduman M, Kleinstein SH. Quantifying selection in high-throughput Immunoglobulin sequencing data sets. *Nucleic Acids Res.* 2012; 40(17): e134.
- van Zelm MC, Szczepanski T, van der Burg M, van Dongen JJ. Replication history of B lymphocytes reveals homeostatic proliferation and extensive antigen-induced B cell expansion. J Exp Med. 2007; 204(3): 645-655.
- 50. M. Blake Caldwell, Margaret J. Oxtoby, Robert J. Simonds, Mary Lou Lindegren, Martha F. Rogers. Centers for Disease Control and Prevention. 1994 Revised classification system for human immunodeficiency virus infection in children less than 13 years of age. Official authorized addenda: human immunodeficiency virus infection codes and official guidelines for coding and reporting ICD-9-CM MMWR. 1994; 43(No. RR-12).
- sanquin. Normale waarden Ig's voor verschillende leeftijdsgroepen. http://wwwsanquinnl/productendiensten/diagnostiek/vademecum/theoriehoofdstukken/afweerstoornissen. data obtained at 03-07-2013.
- 52. Appay V, van Lier RA, Sallusto F, Roederer M. Phenotype and function of human T lymphocyte subsets: consensus and issues. *Cytometry A*. 2008; 73(11): 975-983.
- 53. Effros RB, Allsopp R, Chiu CP, Hausner MA, Hirji K, Wang L, et al. Shortened telomeres in the expanded CD28-CD8+ cell subset in HIV disease implicate replicative senescence in HIV pathogenesis. *Aids*. 1996; 10(8): F17-22.
- 54. Saukkonen JJ, Kornfeld H, Berman JS. Expansion of a CD8+CD28- cell population in the blood and lung of HIV-positive patients. *J Acquir Immune Defic Syndr*. 1993; 6(11): 1194-1204.
- Moir S, Ho J, Malaspina A, Wang W, DiPoto AC, O'Shea MA, et al. Evidence for HIV-associated B cell exhaustion in a dysfunctional memory B cell compartment in HIV-infected viremic individuals. J Exp Med. 2008; 205(8): 1797-1805.
- 56. Corbeau P, Reynes J. Immune reconstitution under antiretroviral therapy: the new challenge in HIV-1

infection. Blood. 2011; 117(21): 5582-5590.

- 57. WHO. Antiretroviral Therapy for HIV Infection in Infants and Children: Towards Universal Access: Recommendations for a Public Health Approach: 2010 Revision. *WHO Guidelines Approved by the Guidelines Review Committee*. 2010; ISBN-13: 978-92-4-159980-1
- 58. Yang X, Jiao YM, Wang R, Ji YX, Zhang HW, Zhang YH, et al. High CCR5 density on central memory CD4+ T cells in acute HIV-1 infection is mostly associated with rapid disease progression. *PLoS One*. 2012; 7(11): e49526.
- 59. Descours B, Avettand-Fenoel V, Blanc C, Samri A, Melard A, Supervie V, et al. Immune responses driven by protective human leukocyte antigen alleles from long-term nonprogressors are associated with low HIV reservoir in central memory CD4 T cells. *Clin Infect Dis.* 2012; 54(10): 1495-1503.
- Potter SJ, Lacabaratz C, Lambotte O, Perez-Patrigeon S, Vingert B, Sinet M, et al. Preserved central memory and activated effector memory CD4+ T-cell subsets in human immunodeficiency virus controllers: an ANRS EP36 study. *J Virol*. 2007; 81(24): 13904-13915.
- 61. Fenoglio D, Poggi A, Catellani S, Battaglia F, Ferrera A, Setti M, et al. Vdelta1 T lymphocytes producing IFN-gamma and IL-17 are expanded in HIV-1-infected patients and respond to Candida albicans. *Blood*. 2009; 113(26): 6611-6618.
- 62. Li H, Pauza CD. HIV envelope-mediated, CCR5/alpha4beta7-dependent killing of CD4-negative gammadelta T cells which are lost during progression to AIDS. *Blood*. 2011; 118(22): 5824-5831.
- 63. Moir S, Malaspina A, Li Y, Chun TW, Lowe T, Adelsberger J, et al. B cells of HIV-1-infected patients bind virions through CD21-complement interactions and transmit infectious virus to activated T cells. *J Exp Med*. 2000; 192(5): 637-646.
- 64. Xu W, Santini PA, Sullivan JS, He B, Shan M, Ball SC, et al. HIV-1 evades virus-specific IgG2 and IgA responses by targeting systemic and intestinal B cells via long-range intercellular conduits. *Nat Immunol.* 2009; 10(9): 1008-1017.
- Cagigi A, Nilsson A, Pensieroso S, Chiodi F. Dysfunctional B-cell responses during HIV-1 infection: implication for influenza vaccination and highly active antiretroviral therapy. *Lancet Infect Dis.* 2010; 10(7): 499-503.
- Bussmann BM, Reiche S, Bieniek B, Krznaric I, Ackermann F, Jassoy C. Loss of HIV-specific memory B-cells as a potential mechanism for the dysfunction of the humoral immune response against HIV. *Virology*. 2010; 397(1): 7-13.
- 67. Moir S, Buckner CM, Ho J, Wang W, Chen J, Waldner AJ, et al. B cells in early and chronic HIV infection: evidence for preservation of immune function associated with early initiation of antiretroviral therapy. *Blood*. 2010; 116(25): 5571-5579.
- Titanji K, De Milito A, Cagigi A, Thorstensson R, Grutzmeier S, Atlas A, et al. Loss of memory B cells impairs maintenance of long-term serologic memory during HIV-1 infection. *Blood*. 2006; 108(5): 1580-1587.
- 69. Fecteau JF, Cote G, Neron S. A new memory CD27-IgG+ B cell population in peripheral blood expressing VH genes with low frequency of somatic mutation. *J Immunol*. 2006; 177(6): 3728-3736.

Supplemental Table 1A. Characteristic	s of HIV patient g	roups				
				HIVIow		
	HIVIow	0-1y	2-4y	5-9y	10-15y	16-23y
E	65	9	6	15	20	15
Plasma HIV RNA copies/ml at inclusion (average (range)) ^a	25.3 (0-900)	192 (0-900)	27.0 (0-123)	6.8 (0-102)	0-0) 0	9.7 (0-145)
Age at inclusion in years (average (range))	10.6 (0.4-20.6)	1.2 (0.4-1.9)	3.5 (2.3-4.9)	7.6 (5.1-9.9)	13.0 (10.2-15.9)	18.4 (16.2-20.6)
ART treatment (n (%))	65 (100)	6 (100)	9 (100)	15 (100)	20 (100)	15 (100)
CDC score (at diagnosis) ^b	N1-C3	A1-C3	A1-C1	N1-C3	N1-C3	N1-C3
Age at start ART in years (average (range))	4 (0-17)	0.3 (0-0.6)	1.2 (0.1-3.2)	2.5 (0-7.3)	3.7 (0.3-14.7)	9.1 (5.4-17)
ART duration in years (average (range))	6.4 (0.2-13.3)	0.7 (0.2-1.6)	2.3 (1-4.7)	5.1 (0.5-8.8)	9.1 (1.2-13.3)	9.3 (0.9-12.7)
Nadir CD4 in % of age-matched reference value (average (range)) °	61.6 (0-301.5)	86 (29.6-112.7)	110.5 (36.8-301.5)	60.6 (1.4-215.0)	51.1 (0-118.8)	37.5 (1.0-91.4)
Age at nadir CD4 in years (average (range))	6.4 (0.2-19.4)	1.0 (0.2-1.9)	3.1 (1.6-5.8)	3.8 (0.3-8.4)	6.8 (0.3-15.0)	12.6 (5.4-19.4)
Reduced CD4 ⁺ T-cell numbers at inclusion (n (%)) ^c	12 (18.5)	0 (0)	1 (11.1)	0 (0)	2 (10)	6 (60)
Hyperglobulinemia (n (%)) d	21 (32.3)	4 (66.7)	2 (22.2)	5 (33.3)	3 (15)	7 (46.7)
Protective Hepatitis A/ B response (n[protected]/n[tested] (%))	29/29 (100)	3/3 (100)	3/3 (100)	11/11 (100)	11/11 (100)	1/1 (100)
^a Undetectable virus load was depicted as C ^b CDC= Center for Disease Control and Pre-) vention; N/A/B/C, i	ncreasing clinical s	everity; 1/2/3, increa	sing CD4 depletion	20	

CD4+ T-cell numbers below age-matched reference values ^{27,50}
Ig serum levels above age-matched reference values ³¹

	HIVlow			
	HIVund	HIVdetect	HIVhigh	
n	39	26	4	
Plasma HIV RNA copies/ml at inclusion (average (range)) ^a	0 (0-0)	63.2 (0-900)	111,275 (16,600-328,000)	
Age at inclusion in years	10.4	10.9	15.6	
(average (range))	(1.0-18.9)	(0.4-20.6)	(12.2-19.0)	
ART treatment	39	26	2	
(n (%))	(100)	(100)	(50)	
CDC score (at diagnosis) ^b	N1-C3	N1-C3	N1-A1	
Age at start ART in years	2.6	6.2	4.4	
(average (range))	(0-11.1)	(0-17.0)	(1.1-7.8)	
ART duration in years	7.3	5.1	11.1	
(average (range)) Nadir CD4 in % of age-matched reference value	(0.8-13.3) 67.4	(0.2-12.8) 53.1	(9.9-12.3) 43.5	
(average (range)) ^c	(6.9-301.5)	(0-160)	(21-64.3)	
Age at nadir CD4 in years	6.1	6.9	9.9	
(average (range)) Reduced CD4+ T-cell numbers at inclusion	(0.3-19.4) 5	(0.2-17.0) 7	(1.6-16.8)	
(n (%)) ^c	(12.8)	(26,9)	(50)	
Hyperglobulinemia	7	14	3	
(n (%)) ^d	(17.9)	(53.8)	(75)	
Protective Hepatitis A/ B response	16/16	13/13	1/2	
(n _{protected} /n _{tested} (%))	(100)	(100)	(50)	

Supplemental Table 1B. Characteristics of HIV patient groups

Undetectable virus load was depicted as 0

^b CDC= Center for Disease Control and Prevention; N/A/B/C, increasing clinical severity; 1/2/3,

increasing CD4 depletion 50

^c CD4+ T-cell numbers below age-matched reference values ^{27,50}

^d Ig serum levels above age-matched reference values ⁵¹

	Supplemental Table 2. Antibodies used for flow cytometry				
	Antibody	Fluorochrome	Clone	Manufacturer	
	CD3	PerCP-Cy5.5	SK7	BD Biosciences	
	CD4	PB	RPA-T4	BD Biosciences	
	CD8	APC-H7	SK1	BD Biosciences	
	CCR7	PE	3D13	eBiosciences	
	CD45RO	FITC	UCHL1	Zebra/Dako	
	CD27	APC	L128	BD Biosciences	
	CD28	PE-Cy7	CD28.2	eBiosciences	
	TCRaβ	APC	IP26	eBiosciences	
I Cells	TCRγδ	PE-Cy7	11F2	BD Biosciences	
	Vδ1	FITC	TS8.2	Thermo Scientific	
	νδ2	FITC	B6.1	BD Biosciences	
	νδ2	PE	B6	BD Biosciences	
	Vy9	PE	B3.1	BD Biosciences	
	CD57	FITC	HNK-1	BD Biosciences	
	HLADR	PE-Cy7	L243	BD Biosciences	
	CD45	PO	HI30	Invitogen	
	CD19	PerCP-Cy5.5	SJ25C1	BD Biosciences	
	CD21	PE-Cy7	B-ly4	BD Biosciences	
	CD24	PB	SN3	Exbio	
	CD27	APC	L128	BD Biosciences	
	CD38	APC-H7	HB7	BD Biosciences	
	IgA	FITC	IS11-8E10	Miltenyi Biotech	
R colle	IgA	PE	IS11-8E10	Miltenyi Biotech	
D Cells	IgA	FITC	polyclonal	Kallestad	
	IgD	bio	IA6-2	BD Biosciences	
	IgD	FITC	polyclonal	SBA	
	IgG	PE	G18-145	BD Biosciences	
	IgG	PE	polyclonal	SBA	
	IgM	PB	G20-127	BD Biosciences	
	IgM	PE	polyclonal	SBA	





of CD27 and CD28.³⁰ Left plot depicts the absolute numbers of each cell type in controls and HIVlow patients under ART. Right plot depicts the subset of 5-23y-old children in either uninfected controls (ctrl), HIVund, HIVdetect or HIVhigh patients. The number of individuals in each category is represented in parentheses underneath the plots. Plots depict 25-75th percentile (box) and 10-90th percentile (whiskers) of data. Statistical significance was determined, using the Mann-Whitney U test; *, p<0.05; **, p<0.01; ***, p<0.001; ****, p<0.0001.



Supplemental Figure 2. Cellular and molecular defects in Vy9+Vo2+ T cells of HIV-infected children and adolescents

A) Gating strategy for $V\gamma 9^+V\delta 2^+$ T cells. **B)** Left plot depicts the absolute numbers of $V\gamma 9^+V\delta 2^+$ T cells in controls and HIVlow patients under ART. Right plot depicts the subset of 5-23y-old children in either uninfected controls (ctrl), HIVund, HIVdetect or HIVhigh patients. The number of individuals in each category is represented in parentheses underneath the plots. Plots depict 25-75th percentile (box) and 10-90th , percentile Statistical (whiskers) of data. significance was determined. using the Mann-Whitney U test; *, p < 0.05. C) Frequency of Vy9-Jy1.2 rearrangement of 14 amino acids in size, representing the canonical Patients rearrangement. are represented as in panel B. Statistical significance was determined, using the unpaired T-test; *, p<0.05.



20

0

<2v





Supplemental Figure 3: Longitudinal follow-up of CD21^{low} B-cell frequencies and HIV VL in HIV+ children.

Longitudinal follow up of plasma HIV RNA levels and CD21^{low} B-cell frequencies in 4 patients with high-VL (left column), 5 patients with low-VL (middle) and 5 patients with decreasing VL (right). The 3 top panels are the same as shown in Figure 5D in the main paper.

CHAPTER 8

GENERAL DISCUSSION

During the first few years of life, newborns mount immune responses to diverse pathogenic threats and generate immunological memory post infection. This will prevent clinical illness during subsequent encounters with the same pathogen and thereby provide functional immunity. Maturation of the largely immature neonatal immune system is necessary to provide a young child with an immune system that supports long-term survival without the development of clinical complications due to either insufficient responses (immunodeficiency or immunosenescence) or unwanted responses (autoimmunity or allergies). In adults and elderly, persistent viral infections have been shown to skew T-cell memory. This mostly concerns the cytomegalovirus (CMV), Epstein Barr virus (EBV) and human immunodeficiency virus (HIV).¹⁻³ Little is known about the factors that affect leukocyte subset numbers and formation of immunological memory in young children.

In this thesis, we aimed to identify the effects of various external determinants during pregnancy, around child birth and during the first years of life on the leukocyte populations in young children. In **Chapter 2**, we studied leukocyte population dynamics between birth and 6 years of age in relation to a large number of external determinants. The association between breastfeeding and reduced B-cell memory was further studied in more detail in **Chapter 3**. Various aspects of persistent viral infections were studied: i.e. factors that affect the prevalence of herpesvirus infections (**Chapter 4**), the CMV- and EBV-associated changes in T-cell memory (**Chapter 5**), the EBV-associated reduction of memory B cells (**Chapter 6**), and the effect of perinatal HIV infection on the B-cell and T-cell compartments of HIV-infected children receiving combined antiretroviral (cART) treatment (**Chapter 7**). Together, the studies in this thesis illustrate the dynamics of childhood immune maturation and the plasticity of the childhood immune system to cope with persistent viral infections.

BREASTFEEDING PROVIDES STEPWISE MATURATION OF CELLULAR AND HUMORAL IMMUNITY

In **Chapter 2** we observed a selective negative effect of breastfeeding on the pattern of CD27 IgA+ memory B-cell dynamics, but not the germinal center-dependent memory B-cell populations, between birth and the age of 6 years. Breastfeeding is known to provide passive protection to the newborn during the first few months of life, in the form of maternal immunoglobuling or anti-pathogenic molecules, growth factors, lactoferrin and exosomes.⁴⁻⁶ Maternal IgA is important to reduce bacterial adhesion to epithelial cells and intestinal translocation in the neonate and thereby to prevent bacterial exposure to the newborn's humoral immune system.⁴ The CD27 IgA⁺ memory B-cell population is generated from local intestinal IgA responses,⁷ which is the direct target organ of breastfeeding. This would suggest that breastfeeding reduces especially the formation of local humoral responses, whereas systemic germinal center-dependent responses are less or not affected. However, in **Chapter 3**, we observed that breastfeeding duration was associated with smaller T-cell dependent memory B-cell populations, an effect that resolved after discontinuation of breastfeeding. Growth factor proteins (e.g. epidermal growth factor and insulin-like growth factor) and hormones (e.g. erythropoietin) in breast milk were observed to reduce the intestinal permeability and subsequent microbial translocation.^{6,8-10} This would be in line with our observation of a reducing effect of breastfeeding on systemic memory B-cell populations during breastfeeding duration. The longitudinal approach in Chapter 2 focused on

General Discussion

the impact of breastfeeding on the overall pattern between birth and 6 years of age without including information on breastfeeding duration, which might in part explain why we did not observe the reduced numbers of systemic T-cell dependent memory B-cell populations in breastfed children in this study. Though speculative, breastfeeding might transiently delay the induction of systemic B-cell responses during the period of breastfeeding, whereas the effect of breastfeeding on the intestinal microbiota and subsequent intestinal CD27⁻IgA⁺ memory B-cell responses might be more long-lasting and therefore detectable on the overall pattern of CD27⁻IgA⁺ memory B-cell development. Detailed follow-up analyses of CD27⁻IgA⁺ memory B cells in association with breastfeeding or formula feeding will be important to understand the long-term consequences of breastfeeding on this population.

In contrast to inhibiting effects on B-cell memory, we observed expansions of central memory T (Tcm) cells in the presence of breastfeeding on. Expansion of the Tcm subset depended on breastfeeding exposure, but not on breastfeeding duration. This might suggest that it primes the newborn's T-cell compartment, and would be in line with the more effective vaccination responses observed in breastfed children.¹¹ Currently no direct CD8⁺ T-cell stimulating component has been identified in breastfeeding. Potential factors in breast milk that might stimulate CD8⁺ T-cell maturation include vitamins, cytokines, exosomes, and potentially viral or bacterial structures.⁴⁻⁶ Cytokines, such as interleukin (IL)-15, can have CD8⁺ T-cell stimulatory potential,¹² and IL-15 in breast milk has been associated to protection from mother-to-child HIV-transmission.¹³ Exosomes in breast milk contain major histocompatibility complex (MHC)-class I and MHC-class II proteins, as well as various co-stimulatory molecules, and functional micro-RNAs, all of which might provide them with immunomodulatory potential.¹⁴⁻¹⁶ Exosomes might be involved in presentation of maternal viral or bacterial antigens, and thereby stimulate T-cell maturation, either directly, or by bringing together CD4+ and CD8⁺ T cells.¹⁵ This is in line with the described increase in CD4⁺ regulatory T-cell differentiation under the control of exosomes in breast milk,¹⁴ Based on the complex membrane protein composition of exosome pools and the extensive content of these vesicles, I am convinced that more regulatory properties are likely to be identified in the future, which might provide an explanation for the CD8⁺ T-cell stimulation that we observed in breastfed children.

One might speculate that breastfeeding reduces the need for humoral immune responses, while T-cell responses are primed, thereby boosting cellular immunity. It will be important to study the receptor diversity and reactivity of this expanded CD8⁺ Tcm pool, to define whether they are the result of antigen-independent or antigen-driven differentiation and proliferation, the first of which might provide a broadly reactive Tcm pool, whereas the latter would likely skew the repertoire of these cells. Future research into the immune stimulatory properties of exosomes, cytokines, and other components of breast milk, will be important to provide insights into the broad or targeted effect of breastfeeding-associated CD8⁺ T-cell expansions.

PERSISTENT VIRAL INFECTIONS INDUCE DISTINCT CD8⁺ EFFECTOR MEMORY T-CELL EXPANSIONS IN YOUNG CHILDREN

We observed that in 6-year-old children, CMV and EBV (**Chapter 5**), and HIV (**Chapter 7**), induces distinct CD8⁺ effector memory T-cell expansions: EBV was

predominantly associated with expansions of early CD45RO⁺ effector memory; HIV predominantly with intermediate to late CD45RO⁺ or CD45RA⁺ effector memory; and CMV predominantly with late CD45RA⁺ effector memory (Figure 1). Why infections with these viruses result in CD8⁺ T-cell expansions with distinct immunophenotypes is currently still under debate.

An important aspect will be the distinct functional properties of early differentiated T cells (high proliferative potential, but low cytotoxicity) and terminally differentiated T cells (low proliferative potential, but high cytotoxicity) in relation to the optimal anti-viral response required to clear each distinct virus.^{12,17-19} Poly-functionality, measured as simultaneous production of IL-2, TNFa and IFNY, is assumed to be essential for long term viral control.²⁰⁻²² However, these poly-functional T cells might have a more intermediate differentiation phenotype, in both CMV and HIV infection,^{23,24} whereas the predominantly late-differentiated CMV-specific cells might be enriched for low avidity cells that are less responsive to antigenic stimulation.²⁵ Apparently, these functional properties alone do not explain the phenotypic differences in anti-viral responses.

CMV-specific T cells in children (**Chapter 5**) were mostly CD45RA⁻ rather than CD45RA⁺. This was in line with previous data on children,²⁵ but contrasted observation in adults of expansions of CMV-specific CD45RA⁺ effector memory



Figure 1. CD8⁺ **effector memory T-cell expansions in children infected with CMV, EBV or HIV.** Absolute numbers of early, intermediate are late differentiated CD45RO⁺ (TemRO; left) or CD45RO⁻ (CD45RA⁺ TemRA; right) CD8⁺ effector memory T-cell populations in 5-9 year old children of the Generation R birth cohort (Uninfected, CMV+ and/or EBV+) and the HIV-cohort (HIV+). Included HIV-infected children showed undetectable viral load for more than 1 year, or detectable viral load in the preceding year, but below 900 copies/ml at the time of inclusion. Data in HIV infected children was not corrected for co-infection with CMV or EBV. Bars depict stacked median values per T-cell population. The number of individuals per category is indicated underneath each plot. Significance in CMV- and EBV-infected children was tested first by a Kruskal-Wallis test per lymphocyte population, and in case of significance (p<0.05) followed by a Dunn's test of individual patient groups relative to the uninfected controls. Significance for the Dunn's test or Mann Whitney U test. Significance for the Dunn's test or Mann Whitney U test. Significance for the Dunn's test or Mann Whitney U test. significance in the plots: *, p<0.001; ****, p<0.001; ****, p<0.0001.

T cells. This might mean that terminally differentiated CMV-specific CD45RA+ effector memory T cells accumulate with increasing age as a result from repetitive antigenic stimulation.^{18,21,26} As these terminally-differentiated CD45RA⁺ effector memory T cells show signs of replicative senescence,^{27,28} this might result in the eventual loss of this response. In line with this hypothesis, especially T-cell clones with high HIV-antigen binding sensitivity that was associated with high initial anti-viral functionality and anti-viral protection, showed CD57 expression and frequent loss of dominant clones.²¹ This might also explain the need for large accumulations of terminally differentiated T cells to maintain viral control in CMVinfected elderly.²⁵ It will be of interest to longitudinally follow-up the phenotype of CMV-specific cells in the children studied in **Chapter 5**, as this will show whether the phenotype is stable or changes over time into a terminally-differentiated phenotype. Furthermore, the effect of high antigenic stimulation on the phenotype of these T cells might be tested by comparing early treated HIV-infected children that show virus-control prior to the onset of high viremia, with children treated after the development of high HIV viremia.

Multiple findings in literature contrast the hypothesis of replicative senescence upon repetitive stimulation. The terminally differentiated CMV-specific T cells that predominate in elderly retain the potential to function upon re-stimulation.²⁹ Furthermore, poly-functional T cells accumulate in CMV-infected elderly,²² and a low HIV viral setpoint and high CD4⁺ T-cell count correlated with the presence of T cells with a terminally-differentiated phenotype, instead of the intermediate phenotype related to poly-functionality.^{23,24,30,31} Furthermore, initiation of cART treatment, and subsequent reduction in HIV loads in serum, in HIV infected patients, might not change the phenotype of the initially developed CD8⁺ T-cell populations. This latter observation might suggest that the initial anti-viral response defines the long-term phenotypic (and potentially functional) characteristics of anti-viral CD8⁺ T-cell populations.

In Chapter 5, we observed higher CMV than EBV prevalence in 2-year-olds, suggesting younger infection with CMV than EBV. Furthermore, we observed in **Chapter 4** that CMV infection might depend more on vertical transmission, such as via breastfeeding, whereas EBV and HSV-1 depended more on horizontal transmission as they were associated with having siblings or attending daycare. As childhood immune maturation is associated with strong changes in the inflammatory milieu,³²⁻³⁴ the inflammatory milieu at the time of infection might be an additional factor adding to the complexity of determining the T-cell phenotype. Antigen-independent cytokine stimulation (e.g. by IL-15) has been described to induce a terminally-differentiated phenotype in otherwise functional CMV- and EBV-specific T cells in elderly.^{12,17,24,25,35,36} Their increased resistance to apoptosis might subsequently contribute to their accumulation. A similar increase in survival and effector functions has also been described upon IL-15 treatment of HIVspecific effector memory T cells, although this did not directly induce their terminal differentiation.³⁷ In the studies presented in this thesis, we were, unfortunately, unable to correlate the age of infection with the phenotype of the virus-specific CD8⁺ T cells and the inflammatory state of the immune system. Correlating this information will be essential to further test the importance of the inflammatory milieu on the priming of anti-viral responses.

The potentially distinct routes of infection for CMV (vertical transmission, such as via breastfeeding), compared to EBV and HSV-1 (more horizontal transmission, via contact between children) (**Chapter 4**), might be another important factor

influencing the primary anti-viral response as it can affect virus load or virusstability. Cytokines and growth factors in breast milk have been suggested to perform protective functions either by enhancing activation of CD8⁺ T cells in breast milk, via the reduction of intestinal permeability, or via the activation of the infant's intestinal epithelial intestinal tissue to prime the infant against intestinal infection.^{13,38} Although this was only shown for HIV infection, a similar situation might apply for infectious CMV in breast milk.¹² While inducing a more terminally differentiated phenotype of CD8⁺ effector memory T cells, ^{12,24} it might induce early and strong control of infection, and could explain the relatively small CD8⁺ T-cell expansions that we observed in **Chapter 5**. This hypothesis could be tested by comparing the phenotype and size of virus-specific or virus-associate CD8⁺ T-cell populations in children fed with breast milk versus formula milk. Additional insights into the effect of maternal cytokines on direct immune priming in the breastfed infant, instead of on the virus within the breast milk, could be obtained by studying EBV- and HSV-1-infected children. These two viruses are less frequently acquired via breastfeeding (Chapter 4). Differences between EBVor HSV-1-infected children fed with breast milk or formula milk would therefore indicate a direct effect of breastfeeding on the infant's immune system.

INHIBITION OF IMMUNE RESPONSES BY PERSISTENT VIRAL INFECTIONS IN YOUNG CHILDREN

Besides the strong induction of anti-viral CD8⁺ effector memory T-cell expansions, we also observed two negative associations of these viruses with B-cell and T-cell numbers. First, EBV infection was found to associate with a transient reduction in memory B-cell numbers, likely resulting from direct infection and persistence of the virus in these cells (**Chapters 2** and **6**). Second, HSV-1 infection did not induce CD8⁺ T-cell expansions, but instead was associated with reduced naive B-cell numbers (Chapter 2). Naive B cells themselves are not permissive for HSV-1 infection, suggesting indirect effects. The reduction in naive B cells might involve increased differentiation into memory or effector cells, which would suggest the involvement of more humoral responses against HSV-1 instead of cytotoxic T-cell responses. However, we did not observe an increase in memory B cells in our analyses and were unable to study plasma cell responses. It will be important to define whether the effects are persistent or transient in nature. The effect of EBV seemed mainly due to a reduced expansion of memory B cells within the first year of life. Although these cell numbers have normalized after their initial decline at 14 months (Chapter 6), it remains unclear whether these are as diverse as in uninfected children, or children infected after 14 months of age. Specifically the clonal diversity and replication history could reveal insights into whether memory B-cell numbers normalized through replenishment from naive cells or through enhanced compensatory proliferation of the few remaining memory cells. When the normalization of cell numbers would be due to increased bone marrow output, it could potentially restore this diversity, whereas we might speculate that normalization by increased homeostatic proliferation of circulating memory B cells could induce permanent reduced diversity of the memory B-cell compartment, and potentially reduced humoral immunity, in these EBV-infected children.

PERSISTENT SUBCLINICAL IMMUNE DEFECTS IN HIV-INFECTED CHILDREN

In **Chapter 7**, we found that current cART protocols restored (or prevented the loss of) many aspects of memory B and T cells in perinatally HIV-infected children and adolescents. However, we also identified a selected set of persistent defects that were not prevented nor restored under cART treatment. These especially concerned: expansions of CD8⁺ effector memory T cells, and reductions in CD4⁺ central memory and CD45RO⁺ effector memory T cells, $V\delta^{2+}Vy^{9+}$ T cells and CD27⁻ IqA⁺ memory B cells. Even though these defects were currently still subclinical, complete normalization of these populations might be essential for an efficient immune system and for long-term protection of HIV-infected individuals.³⁹⁻⁴² Current cART protocols provide HIV-infected children with a relatively good prognosis, but the challenge will be to have these children to age with minimal clinical complications. A detailed understanding of the immunological conditions of these children and the identification of potential early signs of clinical complications in the future, will be highly important for future health care and potential further treatment optimization. Whereas follow-up of the normalized memory B-cell and T-cell populations will be important to define the stability of the beneficial effects of cART, the identified persistent defects might indicate potential pitfalls in the current treatment that could in the long term lead to clinical complications. Therefore, follow-up of these subclinical defective lymphocyte populations is highly important.

Combined, the persistent, but still subclinical, defects identified in Chapter 7, might be signs of incomplete HIV control, CD4⁺ central memory and CD45RO⁺ effector memory T-cell populations might be preferentially infected and provide important HIV reservoirs that can be established in a little as 10 days after the onset of clinical symptoms.⁴³⁻⁴⁷ Though clearly speculative, the observed subclinical defects might suggest that CD4⁺ memory T cells are persistently depleted and that latent reservoirs in CD4⁺ memory T cells might not be fully eliminated by cART.⁴⁸ Such a persistent reservoir and low-level ongoing virus replication, might subsequently trigger persistent CD8⁺ T-cell expansions, including the highly HIVrelated CD8⁺ intermediate differentiated effector memory T cells (Figure 1).^{1,2} The intestine has been described to be an important reservoir for HIV infection.^{43,49-51} In line with this, we observed a severe decrease of intestinal-derived CD27-IqA⁺ memory B-cell numbers in blood. Whether this population is reduced due to increased retention, lack of generation, or depletion of these cells in the intestine will require additional studies. Moreover, it will be highly interesting to study whether this population is affected as a direct result of ongoing local HIV-replication, of increased intestinal bacterial translocation after HIV-induced intestinal damage,^{49,52} or of the effect of the predominantly oral treatment on the integrity of the intestinal tissue. The latter might be tested in HIV-exposed, but HIV-uninfected individuals receiving post-exposure prophylaxis. Finally, the reduction of $V\delta^{2+}Vy^{9+}$ T-cell numbers in HIV-infected children, which was partially due to impaired maturation of these cells in response to phospho-antigens,⁵³⁻⁵⁶ might be an early predictor for potential mycobacterial complications and impaired anti-HIV responses.^{42,57} Altogether, detailed monitoring of these children will be needed to address the clinical relevance of these, still subclinical, defects at the long-term, but the populations identified here might provide important targets for longitudinal follow-up.

PLASTICITY IN THE CHILDHOOD IMMUNE SYSTEM UPON VIRUS ENCOUNTER

In Chapters 5 and 7, we observed that the CMV-, EBV- and HIV-associated CD8+ effector memory T-cell expansions in 6-year-old children did not affect the naive and central memory T-cell compartments, Furthermore, in CMV and EBV infection it, at the same age, did not result in the loss of vaccination responses against nonrelated measles or tetanus infection. Moreover, infection before the age of 2 years induced smaller T-cell expansions than infection between 2-6y. In **Chapter 6**, we observed that EBV infection within the first two years of life resulted in an only transient decline of memory B-cell numbers around 24m that normalized before the age of 6y. Combined with the mostly asymptomatic nature of CMV and EBV infection in early childhood, and the potential protective effect of these viruses on the development of celiac disease (Jansen et al submitted) and allergies, 58-60 this might suggest only mild changes in immune homeostasis after childhood infection. However, it is important to realize that the studies in this thesis focused predominantly on quantitative measures, and less on functional responsiveness or the development of a broad repertoire of lymphocyte receptors. Even though our data in Chapter 5 suggests that effector memory T-cell populations did not increase in early CMV- and EBV-infected children between 2 and 6 years of age, long-term follow-up of these children will be essential to confirm the stability of this population and the associated mild changes in the composition of the immune system, without excessive accumulation of CD8⁺ effector memory T cells, the loss of naive cells or immunity against non-related pathogens. Furthermore, the reduction of memory B-cell numbers upon EBV-infection normalized before the age of 6 years and vaccination responses against measles and tetanus seemed normal. However, it needs to be investigated whether also the repertoire and responsiveness of this memory B-cell population fully normalized compared to EBV-uninfected children. When cell numbers normalize due to homeostatic proliferation, this might reduce the diversity and impair subsequent responsiveness of the memory B-cell population in EBV-infected children.

The importance of a proper immune maturation might be even more clearly visible in perinatally HIV-infected children. In Chapter 7, we observed that the HIVinfected children below 10y were treated at significantly younger age and showed less immune dysbalance than children >10y. Though current cART treatment is able to reduce the virus levels in serum to below the detection limit, and to recover a large part of the immune system, the extent of immune recovery was described to be inversely related to the patient's age at the start of cART treatment.^{61,62} (Reviewed in ⁶³) Whereas the start of treatment early during immune maturation has been described to prevent the loss of humoral vaccination responsiveness, this responsiveness could not be restored by treatment started later during childhood.⁶⁴ Based on these beneficial effects of early treatment, current cART protocols changed in the last decade; whereas according to the 2006 guidelines cART had to be initiated in <11-months-old HIV-infected infants with clinical complications, CD4⁺ T-cell frequencies below 25% of the age-related normal frequency, CD4⁺ T-cell count <1,500 cells/ul, or a total lymphocyte counts below 4,000 cells/µl,65 these recommendations have been updated, suggesting to initiate treatment in all HIV-infected children below 24 months of age (2010),⁶⁶ or even in all HIV-infected children below 5 years of age (2013).⁶⁷ Long-term follow-up of these early treated HIV-infected children will be essential to confirm whether the beneficial effects of early treatment might wane over time or persist into adulthood.

THE CLINICAL RELEVANCE OF UNDERSTANDING HERPESVIRUS IMMUNITY

In **Chapter 4**, we describe the large variation in herpesvirus seropositivity within populations with different ethnicities; the incidence of herpesvirus infection was significantly lower in the Western world compared to non-Western populations. Infection with CMV, EBV, HSV-1 and VZV are common, and might reach >50 to even 100% seropositivity in the adult population in the Netherlands.⁶⁸⁻⁷³ Genetic variation between different ethnic groups likely contributes to infection susceptibility. However, we described in **Chapter 4**, that a significant part of the difference in infection prevalence between ethnic groups could also be explained by differences in external determinants associated with (likely cultural) behavior. Lower socioeconomic situation and larger family size (more siblings) in certain ethnic population were associated to the increased prevalence of EBV and HSV-1 infection, whereas the prevalence of CMV infection might be increased by more frequent breastfeeding over bottle feeding of infants. As a consequence, the smaller family size and use of bottle feeding in Western populations might delay or fully prevent infection. As especially infections with CMV and EBV have been associated with severe clinical complication in neonates, immunosuppressed individuals and elderly,⁷⁴⁻⁸³ the reduced prevalence of herpesvirus infection in the Western world might seem beneficial.

However, we observed that a lack of CMV infection or seropositivity for less than 2 herpesviruses was associate with an increased risk for the development of celiac disease autoimmunity (Jansen et al submitted), suggesting a protective effect of herpesvirus infection. A similar effect of herpesviruses has previously also been described for allergic diseases.⁵⁸⁻⁶⁰ Importantly, this protective effect might be limited to infection before the second year of life,^{60,84} and might require wild type infection and not vaccination.⁵⁹ These observations are in line with the relatively limited immune disturbance that we observed upon early CMV and EBV infection in both the T-cell compartment (**Chapter 5**) and the B-cell compartment (**Chapter 6**), and are in line with the hygiene hypothesis. The increasing hygiene, smaller family size and increasing use of bottle feeding in the Western world, and the concomitant change in our microbiome and infection pressure in the Western population, is negatively correlated with the prevalence of allergic or autoimmune diseases.^{33,34}

As a result of the diverse range of above-described clinical aspects to be considered, opinions on whether or not herpesvirus infection should be prevented by universal immunization are varying. Our data might argue against herpesvirus vaccination in young childhood (e.g. below 2 years of age) as herpesvirus infection at young age might be beneficial in priming immune maturation. However, as delayed infection poses important clinical risks, especially upon primary CMV infection during transplantation or pregnancy,⁷⁴ delayed herpesvirus infection might be disadvantageous and should preferably be prevented. Therefore, immunization against especially CMV and EBV might prove beneficial in children that were not naturally infected during the first few years of life.

THE UNIQUE OPPORTUNITIES PROVIDED BY LARGE COHORT STUDIES

The studies described in this thesis were performed in two unique population cohorts: the Generation R cohort and the Dutch vertically HIV-infected pediatric population cohort (HIV-cohort).⁸⁵⁻⁸⁷ Both cohorts are prospective population-based cohort studies involving long-term follow-up of young children and include a detailed clinical or environmental characterization of the included individuals.

The large size of the Generation R cohort enabled studies on clinically minor patient groups, which would otherwise be too small to be studied in detail. One of those opportunities was the inclusion of children with slightly delayed or lack of VZV infection. As VZV infection reaches nearly 100% in adults, and already >90% in 6-year-old children in the Netherlands,⁷³ effects of VZV infection on the composition of the immune compartment could never be studied in detail before.⁸⁸ Moreover, in such studies, the effects of other persistent viruses, such as CMV and EBV, would need to be excluded. By studying this large Generation R population cohort, and focusing on 6-year-old children, we could identify 69 children being negative for HSV-1 and VZV. This allowed us, for the first time, to study the effect of VZV on the immune system (Chapters 5), while correcting our analyses for confounding effects that would otherwise interfere with proper interpretation of the data, including potential interference by CMV- and or EBV-associated effects. Thus, we showed that VZV did not affect memory T cells, in contrast to CMV or EBV. Consequently, VZV infection might be either relatively mild in terms of T-cell responses, or persistent T-cell expansions are only present locally and are not detectable in blood.

Our studies on the HIV-cohort, including 69 children, enabled a detailed analysis of blood B and T cells (**Chapter 7**). This number of children, with different age between 0-23 years, made it possible to compare the effect of treatment in different age-groups. Furthermore, the detailed and longitudinal characterization of these children further enabled the discrimination between children with stable HIV suppression for >1y, children with suboptimal HIV suppression presenting low virus loads at the moment of sampling or detectable virus loads within the preceding year, and children that lacked virus suppression. Thereby we were able to correlate our findings to the presence of HIV in serum, and to define the relevance of optimal virus control. The detailed 8-color flow cytometry analyses of this large HIV-cohort enabled the identification of persistent immune defects, despite current cART protocols, that might provide important new targets for future studies (see Section "Persistent subclinical immune defects in HIV-infected children")

CONCLUDING REMARKS

A properly build immune system during the first few years of life is essential for long-term survival without the development of clinical complications. Persistent viral infections are known to have a strong impact on this immune maturation. The studies presented in this thesis combine both large-scale analyses on the complexity of immune maturation, and detailed immunological analyses on the origin, maturation or responsiveness of selected lymphocyte populations. Where large-scale cohort studies provide the unique possibility to analyze isolated determinants while excluding interference by the large variety of confounding
factors, only the addition of detailed cellular and molecular analyses will allow the study of immune maturation in its complete depth and complexity.

Combined, the studies presented in this thesis stress the plasticity of the childhood immune system and its capability to control viral infection without large-scale immune dysfunction. This information will be important, not only for our basic understanding of healthy immune maturation, but might also contribute to our understanding of immune dysfunction during chronic immune stimulation. Furthermore, it might add to the ongoing debate of herpesvirus vaccination. In addition, the identification of persistent subclinical defects in selected lymphocyte populations in blood of cART treated HIV-infected children, might directly contribute to targeted long-term clinical follow-up of these children.

It will be essential to continue combined epidemiological and detailed molecular immunological studies on the composition of the immune system in children under the influence of different inflammatory conditions or microbial colonization, as well as in children carrying different viral infections and being infected at different age. These studies will even further improve our understanding of what drives the maturation of the immune compartment during childhood and what defines its homeostasis.

REFERENCES

- 1. Appay V, Dunbar PR, Callan M, Klenerman P, Gillespie GM, Papagno L, et al. Memory CD8+ T cells vary in differentiation phenotype in different persistent virus infections. *Nat Med*. 2002; 8(4): 379-385.
- 2. Appay V, van Lier RA, Sallusto F, Roederer M. Phenotype and function of human T lymphocyte subsets: consensus and issues. *Cytometry A*. 2008; 73(11): 975-983.
- Champagne P, Ogg GS, King AS, Knabenhans C, Ellefsen K, Nobile M, et al. Skewed maturation of memory HIV-specific CD8 T lymphocytes. *Nature*. 2001; 410(6824): 106-111.
- 4. M'Rabet L, Vos AP, Boehm G, Garssen J. Breast-feeding and its role in early development of the immune system in infants: consequences for health later in life. *J Nutr*. 2008; 138(9): 1782S-1790S.
- 5. Mincheva-Nilsson L, Hammarstrom ML, Juto P, Hammarstrom S. Human milk contains proteins that stimulate and suppress T lymphocyte proliferation. *Clin Exp Immunol*. 1990; 79(3): 463-469.
- 6. Brugman S, Perdijk O, van Neerven RJ, Savelkoul HF. Mucosal Immune Development in Early Life: Setting the Stage. *Arch Immunol Ther Exp (Warsz)*. 2015; 63(4): 251-268.
- Berkowska MA, Driessen GJ, Bikos V, Grosserichter-Wagener C, Stamatopoulos K, Cerutti A, et al. Human memory B cells originate from three distinct germinal center-dependent and -independent maturation pathways. *Blood*. 2011; 118(8): 2150-2158.
- 8. Weaver LT, Laker MF, Nelson R, Lucas A. Milk feeding and changes in intestinal permeability and morphology in the newborn. *J Pediatr Gastroenterol Nutr*. 1987; 6(3): 351-358.
- 9. Steinwender G, Schimpl G, Sixl B, Kerbler S, Ratschek M, Kilzer S, et al. Effect of early nutritional deprivation and diet on translocation of bacteria from the gastrointestinal tract in the newborn rat. *Pediatr Res.* 1996; 39(3): 415-420.
- Ballard O, Morrow AL. Human milk composition: nutrients and bioactive factors. *Pediatr Clin North* Am. 2013; 60(1): 49-74.
- Pabst HF, Godel J, Grace M, Cho H, Spady DW. Effect of breast-feeding on immune response to BCG vaccination. *Lancet*. 1989; 1(8633): 295-297.
- Geginat J, Lanzavecchia A, Sallusto F. Proliferation and differentiation potential of human CD8+ memory T-cell subsets in response to antigen or homeostatic cytokines. *Blood*. 2003; 101(11): 4260-4266.
- Walter J, Ghosh MK, Kuhn L, Semrau K, Sinkala M, Kankasa C, et al. High concentrations of interleukin 15 in breast milk are associated with protection against postnatal HIV transmission. *J Infect Dis*. 2009; 200(10): 1498-1502.
- 14. Admyre C, Johansson SM, Qazi KR, Filen JJ, Lahesmaa R, Norman M, et al. Exosomes with immune

modulatory features are present in human breast milk. J Immunol. 2007; 179(3): 1969-1978.

- 15. Yanez-Mo M, Siljander PR, Andreu Z, Zavec AB, Borras FE, Buzas EI, et al. Biological properties of extracellular vesicles and their physiological functions. *J Extracell Vesicles*. 2015; 4(27066.
- Zhou Q, Li M, Wang X, Li Q, Wang T, Zhu Q, et al. Immune-related microRNAs are abundant in breast milk exosomes. *Int J Biol Sci.* 2012; 8(1): 118-123.
- 17. Geginat J, Sallusto F, Lanzavecchia A. Cytokine-driven proliferation and differentiation of human naive, central memory, and effector memory CD4(+) T cells. *J Exp Med.* 2001; 194(12): 1711-1719.
- Gamadia LE, van Leeuwen EM, Remmerswaal EB, Yong SL, Surachno S, Wertheim-van Dillen PM, et al. The size and phenotype of virus-specific T cell populations is determined by repetitive antigenic stimulation and environmental cytokines. *J Immunol.* 2004; 172(10): 6107-6114.
- Cellerai C, Perreau M, Rozot V, Bellutti Enders F, Pantaleo G, Harari A. Proliferation capacity and cytotoxic activity are mediated by functionally and phenotypically distinct virus-specific CD8 T cells defined by interleukin-7R{alpha} (CD127) and perforin expression. J Virol. 2010; 84(8): 3868-3878.
- Betts MR, Nason MC, West SM, De Rosa SC, Migueles SA, Abraham J, et al. HIV nonprogressors preferentially maintain highly functional HIV-specific CD8+ T cells. *Blood*. 2006; 107(12): 4781-4789.
- Almeida JR, Price DA, Papagno L, Arkoub ZA, Sauce D, Bornstein E, et al. Superior control of HIV-1 replication by CD8+ T cells is reflected by their avidity, polyfunctionality, and clonal turnover. J Exp Med. 2007; 204(10): 2473-2485.
- 22. Lachmann R, Bajwa M, Vita S, Smith H, Cheek E, Akbar A, et al. Polyfunctional T cells accumulate in large human cytomegalovirus-specific T cell responses. *J Virol*. 2012; 86(2): 1001-1009.
- Riddell NE, Griffiths SJ, Rivino L, King DC, Teo GH, Henson SM, et al. Multifunctional cytomegalovirus (CMV)-specific CD8(+) T cells are not restricted by telomere-related senescence in young or old adults. *Immunology*. 2015; 144(4): 549-560.
- Riou C, Treurnicht F, Abrahams MR, Mlisana K, Liu MK, Goonetilleke N, et al. Increased memory differentiation is associated with decreased polyfunctionality for HIV but not for cytomegalovirusspecific CD8+ T cells. J Immunol. 2012; 189(8): 3838-3847.
- Griffiths SJ, Riddell NE, Masters J, Libri V, Henson SM, Wertheimer A, et al. Age-associated increase of low-avidity cytomegalovirus-specific CD8+ T cells that re-express CD45RA. J Immunol. 2013; 190(11): 5363-5372.
- Almeida JR, Sauce D, Price DA, Papagno L, Shin SY, Moris A, et al. Antigen sensitivity is a major determinant of CD8+ T-cell polyfunctionality and HIV-suppressive activity. *Blood*. 2009; 113(25): 6351-6360.
- 27. Appay V, Rowland-Jones SL. Lessons from the study of T-cell differentiation in persistent human virus infection. *Semin Immunol.* 2004; 16(3): 205-212.
- Brenchley JM, Karandikar NJ, Betts MR, Ambrozak DR, Hill BJ, Crotty LE, et al. Expression of CD57 defines replicative senescence and antigen-induced apoptotic death of CD8+ T cells. *Blood*. 2003; 101(7): 2711-2720.
- 29. Wallace DL, Masters JE, De Lara CM, Henson SM, Worth A, Zhang Y, et al. Human cytomegalovirusspecific CD8(+) T-cell expansions contain long-lived cells that retain functional capacity in both young and elderly subjects. *Immunology*. 2011; 132(1): 27-38.
- Northfield JW, Loo CP, Barbour JD, Spotts G, Hecht FM, Klenerman P, et al. Human immunodeficiency virus type 1 (HIV-1)-specific CD8+ T(EMRA) cells in early infection are linked to control of HIV-1 viremia and predict the subsequent viral load set point. *J Virol*. 2007; 81(11): 5759-5765.
- Barbour JD, Ndhlovu LC, Xuan Tan Q, Ho T, Epling L, Bredt BM, et al. High CD8+ T cell activation marks a less differentiated HIV-1 specific CD8+ T cell response that is not altered by suppression of viral replication. *PLoS One*. 2009; 4(2): e4408.
- Prescott SL, Macaubas C, Holt BJ, Smallacombe TB, Loh R, Sly PD, et al. Transplacental priming of the human immune system to environmental allergens: universal skewing of initial T cell responses toward the Th2 cytokine profile. *J Immunol*. 1998; 160(10): 4730-4737.
- 33. Strachan DP. Hay fever, hygiene, and household size. BMJ. 1989; 299(6710): 1259-1260.
- Daley D. The evolution of the hygiene hypothesis: the role of early-life exposures to viruses and microbes and their relationship to asthma and allergic diseases. *Curr Opin Allergy Clin Immunol*. 2014; 14(5): 390-396.
- 35. Dunne PJ, Belaramani L, Fletcher JM, Fernandez de Mattos S, Lawrenz M, Soares MV, et al. Quiescence and functional reprogramming of Epstein-Barr virus (EBV)-specific CD8+ T cells during persistent

infection. Blood. 2005; 106(2): 558-565.

- Dunne PJ, Faint JM, Gudgeon NH, Fletcher JM, Plunkett FJ, Soares MV, et al. Epstein-Barr virusspecific CD8(+) T cells that re-express CD45RA are apoptosis-resistant memory cells that retain replicative potential. *Blood*. 2002; 100(3): 933-940.
- 37. Mueller YM, Bojczuk PM, Halstead ES, Kim AH, Witek J, Altman JD, et al. IL-15 enhances survival and function of HIV-specific CD8+ T cells. *Blood*. 2003; 101(3): 1024-1029.
- Arsenault JE, Webb AL, Koulinska IN, Aboud S, Fawzi WW, Villamor E. Association between breast milk erythropoietin and reduced risk of mother-to-child transmission of HIV. J Infect Dis. 2010; 202(3): 370-373.
- Yang X, Jiao YM, Wang R, Ji YX, Zhang HW, Zhang YH, et al. High CCR5 density on central memory CD4+ T cells in acute HIV-1 infection is mostly associated with rapid disease progression. *PLoS One*. 2012; 7(11): e49526.
- 40. Descours B, Avettand-Fenoel V, Blanc C, Samri A, Melard A, Supervie V, et al. Immune responses driven by protective human leukocyte antigen alleles from long-term nonprogressors are associated with low HIV reservoir in central memory CD4 T cells. *Clin Infect Dis.* 2012; 54(10): 1495-1503.
- Potter SJ, Lacabaratz C, Lambotte O, Perez-Patrigeon S, Vingert B, Sinet M, et al. Preserved central memory and activated effector memory CD4+ T-cell subsets in human immunodeficiency virus controllers: an ANRS EP36 study. J Virol. 2007; 81(24): 13904-13915.
- 42. Poccia F, Boullier S, Lecoeur H, Cochet M, Poquet Y, Colizzi V, et al. Peripheral V gamma 9/V delta 2 T cell deletion and anergy to nonpeptidic mycobacterial antigens in asymptomatic HIV-1-infected persons. *J Immunol*. 1996; 157(1): 449-461.
- Brenchley JM, Schacker TW, Ruff LE, Price DA, Taylor JH, Beilman GJ, et al. CD4+ T cell depletion during all stages of HIV disease occurs predominantly in the gastrointestinal tract. J Exp Med. 2004; 200(6): 749-759.
- 44. Krzysiek R, Rudent A, Bouchet-Delbos L, Foussat A, Boutillon C, Portier A, et al. Preferential and persistent depletion of CCR5+ T-helper lymphocytes with nonlymphoid homing potential despite early treatment of primary HIV infection. *Blood*. 2001; 98(10): 3169-3171.
- Chun TW, Engel D, Berrey MM, Shea T, Corey L, Fauci AS. Early establishment of a pool of latently infected, resting CD4(+) T cells during primary HIV-1 infection. *Proc Natl Acad Sci U S A*. 1998; 95(15): 8869-8873.
- 46. Chun TW, Stuyver L, Mizell SB, Ehler LA, Mican JA, Baseler M, et al. Presence of an inducible HIV-1 latent reservoir during highly active antiretroviral therapy. *Proc Natl Acad Sci U S A*. 1997; 94(24): 13193-13197.
- Finzi D, Hermankova M, Pierson T, Carruth LM, Buck C, Chaisson RE, et al. Identification of a reservoir for HIV-1 in patients on highly active antiretroviral therapy. *Science*. 1997; 278(5341): 1295-1300.
- Chun TW, Nickle DC, Justement JS, Large D, Semerjian A, Curlin ME, et al. HIV-infected individuals receiving effective antiviral therapy for extended periods of time continually replenish their viral reservoir. J Clin Invest. 2005; 115(11): 3250-3255.
- 49. Costiniuk CT, Angel JB. Human immunodeficiency virus and the gastrointestinal immune system: does highly active antiretroviral therapy restore gut immunity? *Mucosal Immunol*. 2012; 5(6): 596-604.
- Arthos J, Cicala C, Martinelli E, Macleod K, Van Ryk D, Wei D, et al. HIV-1 envelope protein binds to and signals through integrin alpha4beta7, the gut mucosal homing receptor for peripheral T cells. *Nat Immunol.* 2008; 9(3): 301-309.
- 51. Cicala C, Martinelli E, McNally JP, Goode DJ, Gopaul R, Hiatt J, et al. The integrin alpha4beta7 forms a complex with cell-surface CD4 and defines a T-cell subset that is highly susceptible to infection by HIV-1. *Proc Natl Acad Sci U S A*. 2009; 106(49): 20877-20882.
- 52. Brenchley JM, Price DA, Schacker TW, Asher TE, Silvestri G, Rao S, et al. Microbial translocation is a cause of systemic immune activation in chronic HIV infection. *Nat Med*. 2006; 12(12): 1365-1371.
- Sandberg Y, Almeida J, Gonzalez M, Lima M, Barcena P, Szczepanski T, et al. TCRgammadelta+ large granular lymphocyte leukemias reflect the spectrum of normal antigen-selected TCRgammadelta+ T-cells. *Leukemia*. 2006; 20(3): 505-513.
- 54. Parker CM, Groh V, Band H, Porcelli SA, Morita C, Fabbi M, et al. Evidence for extrathymic changes in the T cell receptor gamma/delta repertoire. *J Exp Med*. 1990; 171(5): 1597-1612.
- 55. Breit TM, Wolvers-Tettero IL, van Dongen JJ. Unique selection determinant in polyclonal V delta 2-J delta 1 junctional regions of human peripheral gamma delta T lymphocytes. J Immunol. 1994; 152(6): 2860-2864.

- 56. Pauza CD, Poonia B, Li H, Cairo C, Chaudhry S. gammadelta T Cells in HIV Disease: Past, Present, and Future. *Front Immunol.* 2014; 5(687.
- 57. Poccia F, Battistini L, Cipriani B, Mancino G, Martini F, Gougeon ML, et al. Phosphoantigen-reactive Vgamma9Vdelta2 T lymphocytes suppress in vitro human immunodeficiency virus type 1 replication by cell-released antiviral factors including CC chemokines. *J Infect Dis.* 1999; 180(3): 858-861.
- Veiga RV, Cunha SS, Dattoli VC, Cruz AC, Cooper PJ, Rodrigues LC, et al. Chronic virus infections supress atopy but not asthma in a set of children from a large Latin American city: a cross-section study. *BMC Pulm Med.* 2011; 11(24.
- 59. Silverberg JI, Norowitz KB, Kleiman E, Silverberg NB, Durkin HG, Joks R, et al. Association between varicella zoster virus infection and atopic dermatitis in early and late childhood: a case-control study. J Allergy Clin Immunol. 2010; 126(2): 300-305.
- 60. Saghafian-Hedengren S, Sverremark-Ekstrom E, Linde A, Lilja G, Nilsson C. Early-life EBV infection protects against persistent IgE sensitization. *J Allergy Clin Immunol*. 2010; 125(2): 433-438.
- Walker AS, Doerholt K, Sharland M, Gibb DM, Collaborative HIVPSSC. Response to highly active antiretroviral therapy varies with age: the UK and Ireland Collaborative HIV Paediatric Study. *Aids*. 2004; 18(14): 1915-1924.
- 62. Hainaut M, Ducarme M, Schandene L, Peltier CA, Marissens D, Zissis G, et al. Age-related immune reconstitution during highly active antiretroviral therapy in human immunodeficiency virus type 1-infected children. *Pediatr Infect Dis J.* 2003; 22(1): 62-69.
- 63. Cotugno N, Douagi I, Rossi P, Palma P. Suboptimal immune reconstitution in vertically HIV infected children: a view on how HIV replication and timing of HAART initiation can impact on T and B-cell compartment. *Clin Dev Immunol.* 2012; 2012(805151.
- 64. Pensieroso S, Cagigi A, Palma P, Nilsson A, Capponi C, Freda E, et al. Timing of HAART defines the integrity of memory B cells and the longevity of humoral responses in HIV-1 vertically-infected children. *Proc Natl Acad Sci U S A*. 2009; 106(19): 7939-7944.
- 65. WHO. Antiretroviral therapy of HIV infection in infants and children: towards universal access. Geneva, Switzerland: WHO Press; 2006.
- 66. WHO. Antiretroviral Therapy for HIV Infection in Infants and Children: Towards Universal Access: Recommendations for a Public Health Approach: 2010 Revision. (ed 2013/06/07); 2010.
- 67. WHO. Global update on HIV treatment 2013: results, impact and opportunities: WHO report in partnership with UNICEF and UNAIDS. Geneva, Switzerland WHO Library Cataloguing-in-Publication Data; 2013.
- 68. Korndewal MJ, Mollema L, Tcherniaeva I, van der Klis F, Kroes AC, Oudesluys-Murphy AM, et al. Cytomegalovirus infection in the Netherlands: seroprevalence, risk factors, and implications. *J Clin Virol*. 2015; 63(53-58.
- Gaytant MA, Galama JM, Semmekrot BA, Melchers WJ, Sporken JM, Oosterbaan HP, et al. The incidence of congenital cytomegalovirus infections in The Netherlands. J Med Virol. 2005; 76(1): 71-75.
- 70. Dowd JB, Palermo T, Brite J, McDade TW, Aiello A. Seroprevalence of Epstein-Barr virus infection in U.S. children ages 6-19, 2003-2010. *PLoS One*. 2013; 8(5): e64921.
- 71. Smith JS, Robinson NJ. Age-specific prevalence of infection with herpes simplex virus types 2 and 1: a global review. *J Infect Dis*. 2002; 186 Suppl 1(S3-28.
- Pembrey L, Raynor P, Griffiths P, Chaytor S, Wright J, Hall AJ. Seroprevalence of cytomegalovirus, Epstein Barr virus and varicella zoster virus among pregnant women in Bradford: a cohort study. *PLoS One*. 2013; 8(11): e81881.
- 73. van Lier A, Smits G, Mollema L, Waaijenborg S, Berbers G, van der Klis F, et al. Varicella zoster virus infection occurs at a relatively young age in The Netherlands. *Vaccine*. 2013; 31(44): 5127-5133.
- 74. Crough T, Khanna R. Immunobiology of human cytomegalovirus: from bench to bedside. *Clin Microbiol Rev.* 2009; 22(1): 76-98, Table of Contents.
- 75. Derhovanessian E, Maier AB, Hahnel K, McElhaney JE, Slagboom EP, Pawelec G. Latent infection with cytomegalovirus is associated with poor memory CD4 responses to influenza A core proteins in the elderly. *J Immunol.* 2014; 193(7): 3624-3631.
- 76. Frasca D, Diaz A, Romero M, Landin AM, Blomberg BB. Cytomegalovirus (CMV) seropositivity decreases B cell responses to the influenza vaccine. *Vaccine*; 33(12): 1433-1439.
- 77. Jonasson L, Tompa A, Wikby A. Expansion of peripheral CD8+ T cells in patients with coronary artery disease: relation to cytomegalovirus infection. *J Intern Med.* 2003; 254(5): 472-478.

- 78. van de Berg PJ, Yong SL, Remmerswaal EB, van Lier RA, ten Berge IJ. Cytomegalovirus-induced effector T cells cause endothelial cell damage. *Clin Vaccine Immunol*. 2012; 19(5): 772-779.
- 79. Bolovan-Fritts CA, Trout RN, Spector SA. High T-cell response to human cytomegalovirus induces chemokine-mediated endothelial cell damage. *Blood*. 2007; 110(6): 1857-1863.
- 80. Savva GM, Pachnio A, Kaul B, Morgan K, Huppert FA, Brayne C, et al. Cytomegalovirus infection is associated with increased mortality in the older population. *Aging Cell*; 12(3): 381-387.
- 81. de-The G. Is Burkitt's lymphoma related to perinatal infection by Epstein-Barr virus? *Lancet*. 1977; 1(8007): 335-338.
- 82. Piriou E, Asito AS, Sumba PO, Fiore N, Middeldorp JM, Moormann AM, et al. Early age at time of primary Epstein-Barr virus infection results in poorly controlled viral infection in infants from Western Kenya: clues to the etiology of endemic Burkitt lymphoma. J Infect Dis. 2012; 205(6): 906-913.
- Draborg AH, Duus K, Houen G. Epstein-Barr virus in systemic autoimmune diseases. Clin Dev Immunol. 2013; 2013(535738.
- 84. Calvani M, Alessandri C, Paolone G, Rosengard L, Di Caro A, De Franco D. Correlation between Epstein Barr virus antibodies, serum IgE and atopic disease. *Pediatr Allergy Immunol*. 1997; 8(2): 91-96.
- 85. Jaddoe VW, van Duijn CM, Franco OH, van der Heijden AJ, van Iizendoorn MH, de Jongste JC, et al. The Generation R Study: design and cohort update 2012. *Eur J Epidemiol*. 2012; 27(9): 739-756.
- 86. Kruithof CJ, Kooijman MN, van Duijn CM, Franco OH, de Jongste JC, Klaver CC, et al. The Generation R Study: Biobank update 2015. *Eur J Epidemiol*. 2014; 29(12): 911-927.
- Cohen S, Smit C, van Rossum AM, Fraaij PL, Wolfs TF, Geelen SP, et al. Long-term response to combination antiretroviral therapy in HIV-infected children in the Netherlands registered from 1996 to 2012. *Aids*. 2013; 27(16): 2567-2575.
- Derhovanessian E, Maier AB, Hahnel K, Beck R, de Craen AJ, Slagboom EP, et al. Infection with cytomegalovirus but not herpes simplex virus induces the accumulation of late-differentiated CD4+ and CD8+ T-cells in humans. J Gen Virol. 2011; 92(Pt 12): 2746-2756.





Addendum

SUMMARY SAMENVATTING



SUMMARY

During pregnancy, a fetus is protected from a large part of the pathogens of the environment. As a result, a newborn's immune system is immature and unexperienced, and mainly composed of innate leukocytes and naive lymphocytes. Memory lymphocytes, and concomitant functional immunity, needs to be formed in response to various pathogen encounters. Most of the immune maturation occurs during the first few years of childhood, during which the child comes into contact with a large variety of pathogens. Though most of the pathogens are cleared, after which life-long immunity is generated, some viruses evade virus clearance and induce a state of latency and viral persistence. Due to their continuous presence and continuous pressure on the immune system, persistent viral infections are known to have a strong impact on the immune system of especially elderly. The research described in this thesis was set out to characterize immune maturation during early childhood, in response to various environmental determinants with a specific focus on persistent viral infections.

In **Chapter 2**, we studied the dynamics of leukocyte subsets within the innate leukocyte, naive and memory B-cell and naive and memory T-cell lineages, and investigated which external determinants were associated with childhood leukocyte dynamics. We demonstrated that the dynamics of leukocyte subsets could be divided in four main age-related patterns, with functionally similar leukocyte populations following similar kinetics. Furthermore, the effect on immune dynamics by various external determinants related to prenatal maternal life style, maternal immune-mediated diseases, birth characteristics or bacterial/ viral exposure, were described. Our data confirms previously described effects of persistent Cytomegalovirus (CMV) and Epstein Barr virus (EBV) infections on memory B-cell and T-cell dynamics, and identified new effects, including the association between Herpes Simplex virus 1 (HSV-1) seropositivity and reduced naive B-cell numbers. Our large-scale analysis of all leukocyte subsets and various external determinants allowed for the identification of new determinants that can provide targets for studies on the molecular processes that regulate leukocyte development and immune responses.

In **Chapter 3**, the effect of breastfeeding was studied in more detail in a crosssectional study on breastfed versus bottle-fed children at the age of 6, 14, 25 and 72 months. We demonstrated a negative association between longer breastfeeding duration and the numbers of T-cell dependent CD27⁻IgG⁺, CD27⁺IgM⁺ and CD27⁺IgA⁺ memory B-cell populations at the age of 6 months. These effects disappeared after breastfeeding discontinuation. Furthermore, we demonstrated that breastfeeding exposure, independent of breastfeeding duration, had a positive effect on the size of the cytotoxic CD8⁺ T-cell compartment, potentially suggesting a lasting priming of CD8⁺ T-cell immunity via breastfeeding. Combined these observations might suggest that breastfeeding reduces the need for humoral immune responses, while priming cellular responses, thereby enabling step-wise immune maturation.

In **Chapter 4**, we demonstrated the differences in infection prevalence for CMV, EBV and HSV-1 in populations with different ethnicities. The ethnic differences in EBV and HSV-1 prevalence could in part (up to 39%) be ascribed to socioeconomic position and factors related to life style. This might suggest an important involvement of horizontal transmission for EBV and HSV-1 infection, which would be related to factors such as family size or daycare attendance. Ethnic differences

in the prevalence of CMV infection were independent of socioeconomic position and factors related to life style. In contrast CMV infection seemed to be positively associated to breastfeeding, suggesting the involvement of vertical transmission from mother to child.

CMV and EBV are associated with changes in immunological memory. Infected adults display persistent expansions of virus-specific effector memory T cells in both the CD8⁺ and the CD4⁺ T-cell lineages. In **Chapter 5**, we studied the memory T-cell compartment in detail in 6-year-old children with or without infection with CMV and/or EBV. We demonstrated similar virus-associated T-cell expansions as those described in adults. Furthermore, we demonstrated that these effects were independent of HSV-1 and Varicella Zoster virus (VZV) infection. However, in contrast to suggestions in elderly, CMV- and/or EBV-associated effector memory T-cell expansions in 6-year-old children did not result in a loss of the naive or central memory T cells, nor in a loss of vaccination responses against nonrelated Measles or Tetanus vaccination. Moreover, in 6-year-old children infected before age 2 years, stable control of these persistent viruses was maintained with only limited effector memory T-cell expansions. This might suggest an only mild change in the CD8⁺ T-cell compartment upon childhood infection with CMV or EBV. These new insights into the immunomodulatory effects of herpesviruses in young children are important for our understanding of herpesvirus-associated immunosenescence in the elderly.

EBV has been shown to persist in memory B cells. Still, it is unclear whether persistent EBV infection results in altered B-cell memory. Therefore, in **Chapter 6**, we studied persistence of EBV in the various memory B-cell subsets and the memory-B-cell compartments in children persistently infected with EBV. We demonstrated that, in adults, EBV was able to persist in all class-switched memory B-cell populations, independent of their germinal center dependent or independent origin. In addition, through extensive flowcytometric immunophenotyping in EBV-infected children, we were able to demonstrate that EBV infection in the first year of life resulted in a reduction in the memory B-cell expansion that normally occurs at the age of 14 months. However, this reduction seemed transient as it normalized before the age of 6 years. Moreover, it did not seem to have long-term consequences on vaccination responses against tetanus and measles vaccination. Combined, this study demonstrates the impact of EBV infection on immunological memory in young children, as well as the plasticity of the immune system to overcome this.

With the introduction of combined antiretroviral therapy (cART), Human Immunodeficiency virus (HIV) infection is becoming a chronic instead of lethal disease, and perinatally HIV-infected children can reach adulthood with minimal clinical complications. However, long-term effects of persistent HIV infection and cART in children currently remain unclear. In **Chapter 7** we studied a cohort HIV-infected children and adolescents receiving cART. We demonstrated that cART is able to effectively reduce virus load in serum and restore total CD4+ T-cell numbers and many of the other HIV-related immune defects. Still, a selection of subclinical immunological defects was identified in blood of cART treated HIV-infected children. These involved the CD4+ T-cell, CD8+ T-cell, TCR $\gamma\delta^+$ T-cell and memory B-cell lineages, and were shown to be in part caused by defective immune maturation. A detailed understanding of the immunological conditions of HIV-infected children and the identification of potential early signs of clinical complications in the future, will be highly important for future health care and

potential further treatment optimization in these HIV-infected children. Though still subclinical, the identified defects might provide important targets for targeted clinical follow-up of the HIV-infected children.

Combined, the studies presented in this thesis underline the complexity of the childhood immune dynamics and identify various external factors associated with immune maturation, with a focus on breastfeeding and persistent viral infections. Moreover, these studies stress the plasticity of the childhood immune system upon viral infection with herpesviruses and the HIV virus. Children seem to control persistent herpesvirus infections without the negative effects that have been observed in elderly. This information will be important, not only for our basic understanding of healthy immune maturation, but might also contribute to our understanding of immune dysfunction during chronic immune stimulation. Knowledge on immune dynamics and the external determinants that shape the childhood immune system will provide more understanding of the processes that underlie the formation of long-lasting immunity without inducing destructive, excessive or insufficient immune responses.

SAMENVATTING

Het afweersysteem van de mens bestaat uit een complex geheel van cellen en moleculen. Het kan worden onderverdeeld in aangeboren afweer en verworven afweer. Aangeboren afweercellen functioneren direct na hun aanmaak en deze eigenschappen veranderen niet tijdens of na een afweerreactie. In tegenstelling, wordt de verworven afweer grotendeels gevormd als gevolg van contact met ziekteverwekkers. De verworven afweer bestaat uit B en T cellen. B en T cellen starten hun ontwikkeling als naïeve cellen die nog niet in contact zijn gekomen met een ziekteverwekker. Pas na contact met een ziekteverwekker zal een naïeve B of T cel gaan delen en uitrijpen tot effectorcellen die de ziekteverwekker bestrijden, of tot geheugencellen. Deze geheugencellen blijven lange tijd in het lichaam aanwezig en kunnen zich bij een volgend contact met dezelfde ziekteverwekker snel ontwikkelen tot een effectorcel. Daardoor kan de ziekteverwekker bij een volgend contact meestal snel uit het lichaam worden verwijderd voordat er ziekteverschijnselen ontstaan. In dit geval spreken we van immuniteit tegen deze ziekteverwekker. Het verworven afweersysteem past zich op deze manier als het ware aan aan de omgeving.

Tijdens de zwangerschap wordt een foetus in de baarmoeder beschermd tegen een groot deel van de ziekteverwekkers uit de omgeving. Daarom bestaat het afweersysteem in pasgeborenen voornamelijk uit aangeboren afweercellen en naïeve B en T cellen. Geheugencellen en bijbehorende immuniteit, zullen grotendeels gevormd worden na de geboorte als gevolg van contact met de vele verschillende ziekteverwekkers in de omgeving. De veelheid aan ziekteverwekkers die een kind vooral in de eerste levensjaren tegenkomt bepalen dus grotendeels de vorming van zijn afweersysteem.

Hoewel de meeste virussen door middel van een afweerreactie worden verwijderd uit het lichaam, zijn sommige virussen in staat het afweersysteem te ontwijken en zich blijvend in het lichaam te 'verschuilen'. Dit zijn zogeheten persistente virussen. Een aantal bekende persistente virussen zijn het humaan immunodeficiëntie virus (HIV), en leden van de herpesvirusfamilie, waaronder het cytomegalovirus (CMV), Epstein Barr virus (EBV), herpes simplex virus-1 (HSV-1) en varicella zoster virus (VZV). Doordat deze virussen continu in het lichaam aanwezig zijn, oefenen ze een sterke druk uit op het afweersysteem. Bij een deel van de ouderen beïnvloedt dragerschap van deze virussen het afweersysteem sterk, waardoor ze minder goed kunnen reageren op andere ziekteverwekkers, waaronder het influenza (griep) virus. Het is echter niet duidelijk of deze effecten al op jonge leeftijd ontstaan, of dat het afweersysteem bij jongeren anders reageert op deze virussen dan bij ouderen. Daarom was het doel van het onderzoek beschreven in dit proefschrift het bestuderen van de vorming van het afweersysteem in jonge kinderen en hoe infecties met persistente virussen deze vorming beïnvloeden.

In **Hoofdstuk 2** hebben we bestudeerd hoe de aantallen van de diverse typen afweercellen in bloed van kinderen veranderen tijdens de eerste 6 jaren van hun leven. Deze metingen zijn verricht binnen de Generation R Study waarin 10.000 kinderen vanaf het eerste trimester van de zwangerschap worden gevolgd. Hierdoor zijn van veel kinderen gegevens van meerdere tijdspunten beschikbaar evenals vele sociale en biologische gegevens. Bij deze kinderen hebben we aangetoond dat de afweercellen in het bloed kunnen worden onderverdeeld in 4 groepen op basis van de dynamiek in de cel aantallen. Hierbij volgden functioneel vergelijkbare afweercelpopulaties hetzelfde patroon. Deze grote Generation R Study maakte het daarnaast mogelijk om te bestuderen welke omgevingsfactoren geassocieerd zijn met de dynamiek van afweercellen. We hebben hierbij gefocust op omgevingsfactoren die gerelateerd zijn aan 1) de levensstijl van de moeder tijdens de zwangerschap (bijvoorbeeld roken en alcoholgebruik), 2) de aanwezigheid van auto-immuunziektes of allergieën bij de moeder, 3) karakteristieken van de geboorte (bijvoorbeeld of een kind is geboren via een keizersnede of vaginale bevalling, of informatie over het geboortegewicht), of 4) het contact met bacteriën/virussen. Allereerst bevestigen onze resultaten eerdere bevindingen uit de literatuur over het effect van CMV en EBV op de verandering in de aantallen B en T geheugencellen. Daarnaast hebben we nieuwe effecten geïdentificeerd, zoals een verminderd aantal naïeve B cellen in bloed van HSV-1-geïnfecteerde kinderen. Onze grootschalige analyses van dit grote aantal leukocytenpopulaties en de vele omgevingsfactoren hebben het mogelijk gemaakt om nieuwe factoren te identificeren die kunnen worden gebruikt in vervolgstudies naar de processen die de vorming van het afweersysteem reguleren.

In **Hoofstuk 3** hebben we op de leeftijden van 6, 14, 25 of 72 maanden de effecten van borstvoeding en flesvoeding op het afweersysteem bestudeerd. Op de leeftijd van 6 maanden vonden we dat naar mate een kind langer borstvoeding kreeg, verschillende populaties binnen de B geheugencellen kleiner werden. Deze effecten waren niet meer detecteerbaar nadat de borstvoeding werd gestopt. Daarnaast hebben we gevonden dat kinderen die borstvoeding kregen, een vergrootte populatie van T geheugencellen hadden, namelijk de CD8⁺ centrale T geheugencellen. Dit effect was onafhankelijk van de hoe lang een kind borstvoeding kreeg. Samen, suggereren deze bevindingen dat een kind dankzij de beschermende werking van borstvoeding tijdelijk minder B geheugencellen hoeft te vormen, terwijl borstvoeding de vorming van T geheugencellen juist stimuleert. Mogelijk faseert het geven van borstvoeding de vorming van het afweersysteem door eerst de vorming van T geheugencellen te laten plaatsvinden en pas later de B geheugencellen.

In **Hoofdstuk 4** hebben we laten zien dat de frequentie van CMV, EBV en HSV-1 infectie binnen een bevolking sterk afhangt van de etniciteit van de geteste groep mensen. Deze variatie tussen verschillende etnische groepen kon in het geval van EBV en HSV-1 infectie voor ≤39% worden verklaard door verschillen tussen de etnische groepen in de levensstijl en socio-economische status van de moeder. Dit hangt mogelijk samen met factoren zoals familiegrootte of het gebruik van kinderopvang en sluit aan bij eerdere bevindingen dat HSV-1 en EBV veelal horizontaal worden overgedragen door contact tussen familieleden of tussen kinderen op kinderdagverblijven. De verschillen in de frequentie van CMV infectie tussen verschillende etnische groepen waren niet afhankelijk van de socio-economische status of levensstijl van de moeder; CMV infectie kwam juist frequenter voor in bevolkingsgroepen die vaker borstvoeding gaven dan flesvoeding. Dit suggereert een rol voor de zogeheten 'verticale overdracht' van CMV van moeder naar kind.

Een selectie van CMV- en EBV-geïnfecteerde ouderen hebben sterk toegenomen aantallen CMV- en EBV-specifieke T geheugencellen en een verminderde afweerrespons op vaccinaties tegen bijvoorbeeld influenza. Om te bepalen of CMV en EBV ook op jonge leeftijd zulke effecten geven, hebben we, in **Hoofdstuk 5**, T geheugencellen in detail bestudeerd bij 6-jaar-oude kinderen met of zonder CMV en/of EBV infectie. We hebben aangetoond dat CMV en EBV infectie elk resulteerden in een toename van een ander type T geheugencel, en dat beide toenames aanwezig waren in dubbel-geïnfecteerde kinderen. Deze toenames waren onafhankelijk van infectie met twee andere herpesvirussen: HSV-1 en VZV. In tegenstelling tot ouderen, leidden deze vergrote populaties van T geheugencellen bij kinderen niet tot verminderde aantallen naïeve T cellen of andere T geheugencellen, of tot het verlies van vaccinatieresponsen tegen de niet-gerelateerde ziekteverwekkers mazelen en tetanus. Het toegenomen aantal T geheugencellen was op 6 jarige leeftijd kleiner bij kinderen die al voor het 2^e levensjaar waren geïnfecteerd met CMV of EBV, dan bij kinderen die op latere leeftijd waren geïnfecteerd. Onze resultaten suggereren dat er slechts een milde verschuiving ontstaat in de T-celpopulaties na infectie met CMV of EBV in de vroege kindertijd. Deze bevindingen zijn in tegenstelling met eerder beschreven sterke effecten van deze virussen op de T cellen in ouderen. Deze nieuwe inzichten in de effecten van herpesvirussen op het afweersysteem van jonge kinderen kunnen een belangrijk bijdrage leveren aan onze kennis over de herpesvirusgeassocieerde veroudering van ons afweersysteem bij ouderen.

In de literatuur is beschreven dat het EBV persisteert in B geheugencellen, maar het is niet bekend wat het effect daarvan is op de grootte en de verdeling van de verschillende B geheugencelpopulaties. Daarom hebben wij in **Hoofdstuk 6** bestudeerd in welke B geheugencelpopulaties EBV persisteert in het bloed van volwassenen. Daarnaast hebben we bestudeerd wat het effect is van persistente EBV infectie op de aantallen B geheugencellen in bloed van kinderen. We hebben aangetoond dat EBV in alle 6 typen B geheugencellen kan persisteren, maar dat de frequentie het hoogst was in de IgA⁺ en IgG⁺ subsets. Op de leeftijd van 14 maanden waren de aantallen van deze B geheugencellen significant lager in het bloed van EBV-geïnfecteerde kinderen, in vergelijking met EBV-negatieve kinderen. Dit effect was kortstondig en op 6-jarige leeftijd waren de aantallen B geheugencellen in deze EBV-geïnfecteerde kinderen vergelijkbaar met die in EBV-negatieve kinderen en kinderen die pas tussen 14 maanden en 6 jarige leeftijd waren geïnfecteerd met EBV. Bovendien hadden de EBV infectie en de bijbehorende tijdelijke verlaging van B geheugencellen geen langdurig effect op vaccinatieresponsen tegen tetanus en mazelen. Hiermee hebben we voor het eerst een negatief effect gezien van EBV persistentie op B geheugencellen en hebben we vooral het vermogen aangetoond van het afweersysteem om deze gevolgen te herstellen.

Infectie met HIV heeft een enorme invloed op het afweersysteem, maar dankzij de invoering van gecombineerde antiretrovirale therapie (cART) zijn de effecten veel kleiner dan voorheen. Veelal kunnen HIV-geïnfecteerde kinderen tegenwoordig de volwassen leeftijd bereiken zonder veel klinische problemen. Toch is het lange-termijn effect van HIV infectie en de bijbehorende cART behandeling nog onvoldoende bekend. In **Hoofdstuk 7** hebben we het verworven afweersysteem bestudeerd in een groep met HIV-geïnfecteerde kinderen en jongvolwassenen die onder behandeling waren van cART. We hebben gevonden dat cART in staat was de hoeveelheid virus in het bloed te verlagen tot beneden de detectiegrens van de meettechniek. Als gevolg werden CD4⁺ T cellen niet massaal geïnfecteerd en waren deze aantallen vrijwel normaal. Daarnaast zorgde cART behandeling ervoor dat de aantallen B en T geheugencellen, en de meeste andere effecten van het virus op het afweersysteem, bijna volledig normaliseerden. Desalniettemin konden we een aantal subklinische defecten identificeren die niet verdwenen door de behandeling. Dit betrof onder andere een blijvend verlies van CD4⁺ T cellen, TCRy δ^+ T cellen en B geheugencellen, en een blijvend toegenomen CD8⁺ T celpopulatie. Een gedetailleerd beeld van de conditie van het afweersysteem van HIV-geïnfecteerde kinderen, en het herkennen van signalen die mogelijk wijzen op toekomstige klinische problemen, is uiterst belangrijk voor de gezondheidszorg en eventuele verdere optimalisatie van behandeling. Hoewel de geïdentificeerde defecten (nog) subklinisch zijn, kunnen de geïdentificeerde populaties mogelijke belangrijke targetpopulaties zijn voor immunologische monitoring van HIVgeïnfecteerde kinderen.

Gezamenlijk illustreren de studies in dit proefschrift de complexe dynamiek in het afweersysteem van jonge kinderen en het effect van persistente virussen hierop. Onze bevindingen benadrukken het vermogen van het afweersysteem van kinderen om een infectie met een persistent virus zoals de herpesvirussen of HIV onder controle te houden zonder de negatieve effecten die beschreven zijn bij ouderen. Deze informatie kan niet alleen bijdragen aan onze basale kennis over de vorming van het gezonde afweersysteem, maar kan mogelijk ook een belangrijke bijdrage leveren aan onze kennis over de disfunctie van het afweersysteem tijdens chronische (virus-gerelateerde) stimulatie. Bovenal zal deze informatie bruikbaar zijn bij het verder ontrafelen van de processen die de vorming van levenslange immuniteit reguleren zonder de vorming van ontoereikende, excessieve of zelfs destructieve responsen.



LIST OF ABBREVIATIONS

AID	Activation-induced cytidine deaminase
AIDS	Acquired immunodeficiency syndrome
anti-TPO	anti-thyroid peroxidase IgG
anti-tTG	Anti-tissue transglutaminase IgA antibody
APC	Antigen presenting cell
BCR	B-cell receptor
BMI	Body mass index
cART	Combined antiretroviral therapy
CCR	Chemokine receptor
CD	Cluster of differentiation
CI	Confidence interval
CMV	Cytomegalovirus
CR	Complement receptor
CSR	Class switch recombination
D	Diversity
DC	Dendritic cell
EBV	Epstein Barr virus
FSC	Forward scatter
HIV	Human immunodeficiency virus
HLA	Human leukocyte antigen
HSC	Hematopoietic stem cell
HSV-1	Herpes simplex virus-1
Ig	Immunoglobulin
IgH	Immunoglobulin heavy chain
IgL	Immunoglobulin light chain
Ідк	Immunoglobulin kappa
Igλ	Immunoglobulin lambda
IL	Interleukin
INF	Interferron
J	Joining
KLRG1	Killer cell lectin like receptor G1
MHC	Major histocompatibility complex
NK cell	Natural killer cell
OR	Odds ratio
PD-1	Programmed death
SD	Standard deviation
SES	Socio-economic status
SHM	Somatic hypermutation
SSC	Side scatter
ТІ	T cell-independent
TD	T cell-dependent

Тс	Cytotoxic T cell; CD8+ T cell
Tcm	Central memory T cell
TCR	T-cell receptor
Tem	Effector memory T cell
TemRA	CD45RO- effector memory T cell
TemRO	CD45RO+ effector memory T cell
TG2A	Anti-tissue transglutaminase IgA antibody
Th	Helper T cell; CD4+ T cell
TNF	Tumor necrosis factor
Treg	Regulatory T cell
V	Variable
VCA	Viral capsid antigen
VZV	Varicella zoster virus

Addendum

CURRICULUM VITAE PHD PORTFOLIO LIST OF PUBLICATIONS

CURRICULUM VITAE

Personal details:				
Name	Diana van den Heuvel			
Date of birth	12 July 1987			
Place of birth	Rotterdam; The Netherlands			
Nationality	Dutch			
EDUCATION:				
2010-2015	PhD study: Department of Immunology of the Erasmus MC <i>The Immune System out of Shape? Shaping of adaptive immunity</i> <i>by persistent viral infections in young children</i> Promotor: Prof.dr. J.J.M. van Dongen Co-promotor: Dr. M.C. van Zelm			
2008 - 2010 (cum laude)	Master of Science: Biomolecular Sciences Vrije Universiteit Amsterdam; Specialization Cell Biology			
2005 - 2008	Bachelor of Applied Science: Biology and Medical Laboratory research:			
	Hogeschool Rotterdam; Differentiation: research/ biochemistry; extra subjects: histology, gene cloning and signal transduction			
1999 - 2005	VWO (pre-university education) Comenius College, Capelle aan den IJssel.			
RESEARCH EXP	ERIENCE:			
2010	Literature study: An overprotective immune system: The origin, development and reactivity of IgE responses in immune protection and allergy Erasmus MC, Department of Immunology, Rotterdam Supervisors Dr. M.C. van Zelm and Drs. M.A. Berkowska			
2010	2 nd Master internship:			
	Characterization of B-cell defects in HIV-infected children Erasmus MC, Department of Immunology, Rotterdam Supervisors Dr. M.C. van Zelm and Drs. M.A. Berkowska			
2009	1st Master internship: <i>DC-exosome binding to T-cells: Retention at the plasma</i>			
	membrane and functional consequences University Utrecht, Department of Biochemistry and Cell biology, Utrecht			
	Supervisors Dr. M. Wauben Dr. E. Nolte-'t Hoen			
2008	Graduation internship bachelor: Determining the mutation rate and mutation characteristics of 18 Y-chromosomal Short Tandem Repeat Markers (Y-STRs) Erasmus MC, Department of Forensic Molecular Biology, Rotterdam			
	Supervisors Prof.dr. M. Kayser and M. Goedbloed			

2007 **1st Bachelor internship:** *Tuberous Sclerosis complex: Characterization of TSC2 and TSC1 variants* Erasmus MC, Department of Clinical Genetics, Rotterdam Supervisor Dr. M. Nellist

COURSES (ADDITIONAL TO THE PHD-PORTFOLIO):			
2005 (certificate)	Coaching of first year bachelor students		
Work history:			
2008	Research Technician Erasmus MC, Department of Forensic Molecular Biology Head of the group: Prof.dr. M. Kayser		

2006 - 2008 Student coach at the Hogeschool Rotterdam

PHD PORTFOLIO

Name PhD student	Diana van den Heuvel
Erasmus MC Department	Immunology
PhD period	September 2010 – February 2015
Promotor:	Prof.dr. J.J.M. van Dongen MD
Supervisor:	Dr. M.C. van Zelm
Research School:	Molmed

1.	PHD COURSES	YEAR			
G	General courses				
-	Teach the teacher I	2014			
-	Management voor promovendi en postdocs (NIBI)	2013			
-	Biomedical English Writing and Communication (15 weeks)	2012			
-	The Basic Introduction Course on SPSS	2011			
S	pecific courses				
-	Basic flow training (advanced)	2013			
-	Regulatory B-cell symposium	2012			
-	The 4th Symposium & master classes on Mucosal Immunology 'Adaptive immune response in the mucosa: B Cells and Beyond'	2011			
-	The Advanced Course 'Molecular Immunology' at the Molecular Medicine postgraduate school within the Erasmus MC	2011			
-	Molmed course (introduction to the topics of the 15th Molecular Medicine day 2011)	2011			
-	Mucosal Immunology symposium + masterclass; Molecular Medicine postgraduate school of the Erasmus MC; Rotterdam	2011			
-	Biomedical research techniques at the Molecular Medicine postgraduate school within the Erasmus MC	2010			
-	Lectures and refreshment classes of The Advanced Course 'Molecular Immunology' at the Molecular Medicine postgraduate school within the Erasmus MC	2009			
2.	Congresses, seminars, presentations	YEAR			
S	eminars and workshops				
-	Annual Molmed Day (4x)	2010-2014			
-	Medische Immunologie, Rotterdam	2010			
Pı	Presentations				
-	Mucosal Immunology symposium + masterclass; Molecular Medicine postgraduate school of the Erasmus MC; Rotterdam	2011			

(presentation in the masterclass)
29th annual meeting of the European Society for Paediatric 2011 Infectious Diseases (ESPID) (*ePoster presentation*)

(Inter)national conferences			
-	Keystone: 'HIV Vaccines' combined with 'B cell development and function'	2013	
-	The 5th Netherlands Conference on HIV Pathogenesis, Prevention and Treatment (NCHIV 2011)	2011	
-	29th annual meeting of the European Society for Paediatric Infectious Diseases (ESPID)	2011	
-	Nederlandse Vereniging Voor Immunologie (NVVI) Lunteren symposium 2011; Time for high T: Features and functions of T-cells in health and disease	2011	
-	Nederlandse Vereniging Voor Immunologie (NVVI) annual meeting; Noordwijkerhout (5x)	2010-2014	
Other			
-	Journal club department of Immunology	2009-2013	
-	Seminars and minisymposia	2009-2014	
3.	TEACHING	YEAR	
-	Vaardigheidsonderwijs immunologie 2 nd year Medical curriculum	2011-2014	
-	Interviews with students, $2^{\mbox{\scriptsize nd}}$ year Keuzeonderwijs "Paradise or Evil Empire"	2012-2013	
-	Supervising 3 Bachelor theses / 1 Master thesis	2011-2014	

PUBLICATIONS

ACCEPTED PUBLICATIONS

<u>D. van den Heuvel</u>, G.J.A. Driessen, M.A. Berkowska, M. van der Burg, A.W. Langerak, D. Zhao, H. Charif, N.G. Hartwig, A.M.C. van Rossum, P.L.A. Fraaij, J.J.M. van Dongen and M.C. van Zelm; **Persistent subclinical immune defects in HIV-1 infected children treated with antiretroviral therapy.**

AIDS (2015), in press

<u>D. van den Heuvel</u>, M.A.E. Jansen, W.A. Dik, H. Bouallouch-Charif, D. Zhao, K.A.M. van Kester, M.A.W. Smits-te Nijenhuis, M.J. Kolijn-Couwenberg, V.W.V. Jaddoe, R. Arens, J.J.M. van Dongen, H.A. Moll and M.C. van Zelm; **CMV- and EBV-induced T-cell expansions in young children do not impair naive T-cell populations or vaccination responses: The Generation R Study.**

Journal of Infectious Diseases (2015), in press

M.A.E. Jansen, <u>D. van den Heuvel</u>, M.C. van Zelm, V.W.V. Jaddoe, A. Hofman, J.C. de Jongste, H. Hooijkaas and H.A. Moll; **Decreased memory B cells and increased CD8 memory T cells in blood of breastfed children: The Generation R study.** *PLoS One. 2015; 10(5): e0126019.*

SUBMITTED PUBLICATIONS

M.A.E. Jansen, <u>D. van den Heuvel</u>, S.H. Bouthoorn, V.W.V. Jaddoe, H. Hooijkaas, H. Raat, P.L.A. Fraaij, M.C. van Zelm and H.A. Moll; **Can we explain ethnic differences in CMV, EBV and HSV-1 virus seroprevalences in childhood? The Generation R Study.** *Submitted*

M.A.E. Jansen, <u>D. van den Heuvel</u>, K.V.M. van der Zwet, V.W.V. Jaddoe, A. Hofman, J.C. Escher, P.L.A. Fraaij, H. Hooijkaas, M.C. van Zelm and H.A. Moll; **Herpesvirus infections and transglutaminase type 2 antibody positivity in childhood: The Generation R Study.**

Submitted

MANUSCRIPTS IN PREPARATION

D. van den Heuvel, M.A.E. Jansen, A. Bell, A.B. Rickinson, V.W.V. Jaddoe, J.J.M. van Dongen, H.A. Moll and M.C. van Zelm; **Transient reduction in IgA**⁺ **and IgG**⁺ **memory B-cell numbers in young children persistently infected with EBV: The Generation R study.** *Manuscript in preparation*

D. van den Heuvel, M.A.E. Jansen, K. Nasserinejad, W.A. Dik, E.G. van Lochem, L.E. Bakker-Jonges, H. Bouallouch-Charif, V.W.V. Jaddoe, H. Hooijkaas, J.J.M. van Dongen, H.A. Moll and M.C. van Zelm; Effects of external determinants on age-related patterns of innate leukocyte and naive and memory B- and T-lymphocyte numbers in early childhood: The Generation R Study.

Manuscript in preparation

Addendum

ACKNOWLEDGEMENTS/DANKWOORD

Hier zit ik dan, relaxed en languit op de bank mijn dankwoord te schrijven. Ik had een jaar, of zelfs een maand geleden niet gedacht dat dit moment ooit nog ging komen. Dit proefschrift was nooit tot stand gekomen zonder de hulp van heel veel mensen. Ik ga mijn best doen jullie allemaal te bedanken, dus ga er maar even goed voor zitten...

Allereerst wil ik mijn promotor Jacques van Dongen en copromotor Menno van Zelm bedanken voor alle hulp en begeleiding tijdens mijn uiteindelijk bijna 5-jarige promotieonderzoek. Ik heb veel van jullie geleerd. Jullie beiden zijn zeer enthousiast over het doen van onderzoek en stonden altijd klaar met nieuwe plannen en ideeën. Menno, ik begon al eerder in jouw groep als een master student. Ook al had ik officieel gekozen voor een project naar de reactiviteit van IgA⁺ B cellen, was mijn project na twee dagen al compleet veranderd en bestudeerde ik opeens de B en T cellen in HIV-geïnfecteerde kinderen. Deze switch was even wennen, maar uiteindelijk heb ik er toch maar mooi een hele interessante promotiestudie aan overgehouden. Niet alleen het onderwerp, maar ook zeker jouw tempo en niveau van denken en werken was even wennen. Zeker in het begin was het niet altijd even gemakkelijk om je bij te benen, maar ik heb hier wel heel veel van geleerd. Nu ben je zelf aan een nieuw avontuur begonnen in Australië. Heel veel succes daar. Je gaat het er vast fantastisch doen!

Vervolgens wil ik graag mijn commissieleden bedanken. Prof.dr. H.A. Moll, Prof.dr R.A.W. van Lier en Dr. A.M.C. van Rossum van de kleine commissie, en ook Prof. dr. R. de Groot en Prof.dr. G.M. Verjans: bedankt dat jullie plaats wilden nemen in mijn promotiecommissie en dat jullie de tijd wilden nemen om mijn stukken te beoordelen. Speciale dank gaat uit naar Henriette. Bedankt dat je niet alleen secretaris van de commissie wilde zijn, maar me ook altijd nuttige feedback hebt gegeven en me hebt geholpen bij alle epidemiologische analyses die we samen hebben gedaan. Hoewel we in het begin soms twee verschillende talen leken te spreken, heb je me geholpen om de 'epidemiologische taal' genoeg onder de knie te krijgen om mijn promotieonderzoek mooi te kunnen afronden. Daarnaast ben je een belangrijke buffer geweest tussen de verschillende expertises die allemaal meespeelden tijdens het doen van onderzoek op het grensgebied van de epidemiologie en immunologie, bedankt daarvoor.

Een tweede cruciale schakel naar de epidemiologie, zonder wie dit boekje nooit tot stand was gekomen, is Michelle. Michelle, wij zijn vrijwel tegelijk aan ons promotieonderzoek begonnen en hebben ons samen door de grote berg aan data gewerkt. Dankzij jou heb ik niet alleen twee mooie artikelen als tweede auteur aan dit proefschrift kunnen toevoegen, maar heb ik ook geleerd hoe de epidemiologie 'denkt' en 'werkt'. Jij hebt mij wegwijs gemaakt in de Generation R data. We zijn samen tegen wat muren op gelopen, maar er toch ook telkens samen overheen geklommen. Jij bent nu net als ik bijna klaar met je promotieonderzoek, maar ondertussen ook al druk in de kliniek en bent, op het moment dat ik dit schrijf, zelfs net getrouwd. Heel veel succes met al die nieuwe uitdagingen die je aan gaat. Maar na wat ik gezien heb tijdens onze promotieonderzoeken gaat dat zeker goed komen.

De volgende die ik wil bedanken zijn mijn paranimfen Britt en Christina. Meiden, bedankt dat jullie mijn paranimfen wilden zijn! Ik ben heel blij dat ik jullie heb leren kennen en met jullie heb mogen werken. Jullie hebben me niet alleen altijd geholpen bij werk-gerelateerde dingen en me aangehoord als ik weer eens wilde klagen omdat het niet lukte met mijn datasets of experimenten (pessimist dat ik er ben...), maar jullie kwamen zelfs langs mijn huis met muffins en cadeautjes voor mijn kat om me door de afronding van mijn promotieonderzoek te trekken op het moment dat ik er doorheen zat. En dan zijn jullie nóg verrast als ik jullie als mijn paranimfen vraag... meiden niet meer dan terecht!

Next, I would also like to thank the people that introduced me into the immunology department and that guided me through my master internship and into my PhD-project. Magda (Berkowska), you taught me very patiently everything I had to know during my internship, and after that we have been roommates for a while during my PhD-project. You have set the bar for us PhD-students in the group of Menno, which I am trying to reach. Thanks a lot for your help and guidance and I hope that we can meet once in a while to catch up. Maar ook Edwin, je bent ondertussen niet meer op het Erasmus, maar je hebt me tijdens mijn stage onder je hoede genomen wat betreft de flowcytometrie van HIV-geïnfecteerde kinderen. Ik kon altijd op je rekenen. Bedankt voor je hulp.

Dan zijn er natuurlijk ook nog de vele handen die mij door de grote aantallen metingen en digitale analyses hebben geholpen. Een groot aantal van de mensen die hebben gewerkt aan de (nu 10-jaar-durende) Generation R studie ken ik niet eens persoonlijk, maar zonder al deze mensen was deze studie nooit mogelijk geweest. Iedereen hiervoor bedankt.

In het bijzonder wil ik graag Marja, Marion en Nicole bedanken. Jullie hebben mij onwijs geholpen in zowel de HIV-studie als de Generation R studie. Al als student was het heel gemakkelijk werken met jullie binnen de HIV-studie. Ook al begon ik regelmatig te stuiteren als ik jullie zag aankomen met zo'n printje van de HIVdiagnostiek, omdat het weer eens niet uit kwam in mijn planning, hebben jullie me toch telkens weer vriendelijk geholpen met alle info die ik jullie vroeg. En wat voelde ik me ongemakkelijk toen ik het slechte nieuws moest komen brengen dat we binnen de Generation R studie opnieuw moesten beginnen met de analyses van alle ~2,400 FACS files met 17 buizen. Deze studie was, zoals jullie het zelf ook zeiden, jullie kindje, waar jullie heel hard aan hadden gewerkt (en wat bijna klaar was). Nu kwam ik even om de hoek kijken en alles op zijn kop zetten. Maar hoewel dit wel wat zenuwen heeft gekost voordat ik dat nieuws aan jullie durfde te brengen, was jullie reactie eigenlijk alleen: "poehhh,... vooruit dan maar ... wat is het plan?". Ik heb dat altijd onwijs gewaardeerd!

Andere onmisbare hulp heb ik gekregen van Halima en Kevin. Kevin, vers van het HLO na een stage bij Magda (Rother), was je bereid me te helpen met de Generation R analyses. Dit betekende dagenlang (en maandenlang) non-stop flowcytometrie analyses. Halima, jij bent iemand die altijd anderen wilt helpen. Ook al waren de dagenlange flowcytometrie analyses niet altijd even motiverend, heb je wel altijd doorgezet om het af te krijgen en mij daarmee te helpen. Dit deed je ook nog met positieve energie die je dan doorstraalde op mij. Ik heb veel waardering voor hoe jullie tweeën dag in dag uit weer aan de gang gingen met die analyses en hoe jullie probeerden om mij zo ver mogelijk vooruit te helpen.

Natuurlijk wil ik ook mijn studenten bedanken, Shirley, Dan en Rachid. Shirley, mocht je dit ooit lezen. Ik vind het nog steeds heel jammer dat het je niet is gelukt om je stage af te ronden. Je hebt een goeie stage gelopen, die je naar mijn mening had kunnen afronden met een mooi cijfer. Ik hoop dat het nu allemaal goed met je gaat. Dan, just starting your first bachelor internship you were still a bit insecure,

but you were always very good in everything you did. Plates full of bacteria, plates full of ELISA tests, or loads of flowcytometry analyses, nothing was too much. You decided (unfortunately) to continue a master outside the field of science, but you are of course also doing very well at that. Good luck with all your future plans. Rachid, je kreeg een technische uitdaging om de tetrameerkleuringen te starten en dat heeft zo zijn hobbels gekend, maar het is gelukt. Je bent inmiddels bezig met de I&I master. Heel veel succes! Ingrid (Snijdewind), ook al was je niet mijn student, je hebt me wel geholpen met de flowcytometrie van de HIV-studie. Jij bent zo'n bezige bij die alles tegelijk doet en de ballen in de lucht houdt. Je master, je coschappen, je promotieonderzoek en dan ook nog zwanger. Je bent altijd druk, maar nooit te beroerd om iemand anders te helpen. Succes met alles afronden (gaat zeker goed komen) en geniet van je gezinnetje!

Dan zijn er nog mijn kamergenootjes. Hanna, in de oudbouw hebben wij lange tijd een kamer gedeeld. Je was altijd de vraagbaak voor mij als startende PhDstudent. Ondertussen zie ik je eigenlijk als 'één van ons BCDers' en zit je ook op de BCD-kamer (al mogen we hem niet zo noemen ;-P). Nu nog loop ik bij je langs zodra ik ben vastgelopen in mijn Excel analyses en lossen we het op. Maar ook naast de praktische dingen weet ik je te vinden voor hulp en mental support. Ondertussen als postdoc aan de gang en net terug van je zwangerschapsverlof ben je al weer druk met alle studies. Maar hoe druk ook, jij lijkt altijd alles onder controle te hebben, daar ben ik stiekem jaloers op. Anna, wij waren in onze tijd samen op de kamer allebei aan het stoeien met flowcytometrie analyses en daardoor elkaars mental support. Ondertussen ben je gepromoveerd en druk in de kliniek. Succes met alles. Martine ("lekker ding") en Marjolein, het is altijd een drukke boel op de kamer. Met veel gekakel en koffietjes of theetjes. Altijd gezellig. Jullie zijn allebei ook al lekker op weg in jullie promotieonderzoek. Succes met afronden. Jullie kunnen het!

En dan is er natuurlijk nog de hele (ex-)BCD groep. Magda B., Magda R., Christina, Britt, Jorn, Marieke, Edwin, Benjamin, Halima en Katharina. Sommigen heb ik hierboven al bedankt. Magda R., we started together on the FRET project, are now defending on two consecutive days, and even party together. Both of us struggled through the process of writing and finishing our thesis together. When you were stressed out, I would try to listen, but what happened most of the time is that I would freak out and you would listen and tell me that the world is not going to fall apart. With your complex project and your ability to focus and organize your projects, I always considered you as the most 'true scientist' in our group. It was a tough start being alone in The Netherlands and running a project in which almost no-one could help. But you nailed it! Jorn, jij bent altijd de rust zelve en hebt alles onder controle. Je gaat bijna naar Australië en daarna je promotieonderzoek afronden. Veel stukken op de planning, maar ik heb er vertrouwen in dat je ook dat allemaal onder controle hebt. Marieke, hoewel de start van je aanstelling niet even soepel liep, heb je je helemaal in je project gegooid en ben je ook nu hard op weg. Ik hoop dat je stukken snel geaccepteerd worden. Je kunt wel een positieve review gebruiken. Benjamin, wij hebben heel wat gesprekken gehad over onderzoek en daarbuiten. Nu je op de 11^e zit spreek ik je helaas minder. Ik hoop dat we contact houden. Succes met je nieuwe vaste contract! Katharina, it was only for a relatively short time, but thanks for working together. You just started your new job. Good luck back in Germany!

Uiteraard wil ik ook alle andere collega's en coauteurs bedanken voor alle hulp en gezelligheid tijdens deze PhD-studie. Ik kan jullie in dit dankwoord helaas niet allemaal bij naam noemen, en zou ook alleen maar mensen vergeten als ik het toch ga proberen. Daarom aan jullie allemaal: dankzij jullie heb ik een hele fijne tijd gehad binnen de afdeling Immunologie.

En dan komen we uiteraard bij de mensen die misschien nog wel het allerbelangrijkst zijn geweest in de afgelopen jaren. Pa, Ma, Linda en Anita, wie had gedacht dat ik nog eens ontspannen en languit op mijn eigen bank mijn dankwoord zou schrijven. Bedankt voor jullie hulp. Vooral de laatste twee jaar hebben ook van jullie veel gevraagd, en dan heb ik het niet alleen over de promotietijd... Deze periode is zwaarder geweest dan ik me van tevoren kon bedenken, maar dankzij jullie ben ik toch op dit punt terecht gekomen en heb ik alles weer op de rit. Jullie hebben me er doorheen getrokken ook al was het voor jullie ook onwijs zwaar, daar kan ik jullie niet genoeg voor bedanken. Ik heb meer geleerd in deze periode dan tijdens de volledige rest van mijn promotieonderzoek. Het belangrijkste wat ik heb geleerd is om te genieten van wat je doet. Laat niemand oordelen over wie je bent. Geniet van de weg naar je doel, want die is soms belangrijker dan de uitkomst zelf. En pieker niet, want piekeren is als een schommelstoel; het houdt je bezig maar brengt je nergens.

Maar boven alles: Het leven is als een vlinder; ook al zou de wereld zonder ze misschien precies hetzelfde functioneren, hij zou minder mooi zijn.

Iedereen bedankt; Thank you all!

